CITY OF JOLIET, ILLINOIS RIDGEWOOD PHASE 3 WATER MAIN IMPROVEMENTS CITY OF JOLIET CONTRACT NO. 2609-0122

NOTICE

Note to Specifier: This Notice should be printed on blue paper and inserted immediately following the title page of the project manual.

ALL BIDDERS MUST COMPLY WITH THE CITY OF JOLIET'S POLICY REGARDING THE INCREASED USE OF DISADVANTAGED BUSINESSES. THIS POLICY REQUIRES ALL BIDDERS TO ADVERTISE IN A DAILY, REGIONAL NEWSPAPER OR IN AN ESTABLISHED. ONLINE BIDDER'S CLEARINGHOUSE SUCH AS THE "DODGE **REPORT** (HTTP://CONSTRUCTION.COM/DODGE/)", "iSqFt (https://www.isqft.com/start/)", OR EQUIVALENT, UNLESS THE BIDDER WILL AWARD NO SUB-AGREEMENTS. THE ADVERTISEMENT MUST RUN ONE (1) DAY AT LEAST SIXTEEN (16) DAYS PRIOR TO BID OPENING. SEE PROJECT MANUAL FOR DETAILS.

NON-COMPLIANCE WILL DISQUALIFY YOUR BID.

CITY OF JOLIET, ILLINOIS RIDGEWOOD PHASE 3 WATER MAIN IMPROVEMENTS

CITY OF JOLIET CONTRACT NO. 2609-0122

PROJECT MANUAL

Baxter & Woodman, Inc. Illinois 815.459.1260

Wisconsin 262.763.7834

Florida 561.655.6175

Texas 281.350.7027

PROJECT TITLE PAGE 00 01 01 (210039.40)

00 01 07

SEALS PAGE

November 10, 2021

- 1. Specifications of materials and labor required for the construction work shown on the Drawings are prepared by Baxter & Woodman, Inc., Consulting Engineers.
- 2. The Drawings which accompany these Specifications are titled:

"City of Joliet, Illinois, RIDGEWOOD PHASE 3 WATER MAIN IMPROVEMENTS City of Joliet Contract No. 2609-0122".

3. Copyright 2021 by Baxter & Woodman, Inc. All Rights Reserved. No part of these Specifications or the accompanying Drawings may be reproduced by any means, or otherwise reused without the prior written permission of Baxter & Woodman, Inc.

OR LAUREN MARIE SCHULD IN THE OR LLINGS

Project Manager License Expires 2/28/2022 62-70246
LICENSED
PROFESSIONAL
ENGINEER
OF
OF
ILL
INTERPROPERTIES
OF

Project Engineer License Expires 2/28/2022

BAXTER & WOODMAN, INC. STATE OF ILLINOIS – PROFESSIONAL DESIGN FIRM LICENSE NO. 184.001121-0014 - EXPIRES 4/30/2023

> SEALS PAGE 00 01 07-1 (210039.40)

CITY OF JOLIET, ILLINOIS RIDGEWOOD PHASE 3 WATER MAIN IMPROVEMENTS PROJECT MANUAL

TABLE OF CONTENTS

Section Number	Section
DIVISION 00	- PROCUREMENT AND CONTRACTING REQUIREMENTS
00 01 01 00 01 07 00 01 10 00 11 13 00 21 00.41	PROJECT TITLE PAGE SEALS PAGE TABLE OF CONTENTS ADVERTISEMENT FOR BIDS BIDDER INSTRUCTIONS Pages 10 WATER SUPPLY LOAN PROGRAM Pages 11-24 AFFIRMATIVE ACTION FOR EQUAL EMPLOYMENT OPPORTUNITY (E.O. 11246) Pages 25-28 CONSTRUCTION CONTRACTS OF LOAN RECIPIENT AND OTHER SECTIONS FROM "PROCEDURES FOR ISSUING LOANS FROM THE WATER SUPPLY LOAN PROGRAM" Pages 29-32 DISADVANTAGED BUSINESS ENTERPRISE
	PARTICIPATION SPECIFICATIONS Pages 44-45 ILLINOIS WORKS JOBS PROGRAM ACT – APPRENTICESHIP INITIATIVE INFORMATION FOR CONTRACTORS
00 31 24.51 00 31 32.11 00 41 00.41 00 43 13 00 43 36 00 43 93.41 00 51 05 00 51 10 00 52 00 00 55 00 00 60 01.41 00 61 13.13 00 61 13.16 00 61 19 00 62 01 00 62 03 00 62 04 00 62 05 00 62 06 00 62 07 00 62 08 00 62 16 00 62 16.13	CLEAN CONSTRUCTION OR DEMOLITION DEBRIS (CCDD) - LPC 662 SUBSURFACE DRILLING INFORMATION BID FORM BID SECURITY FORM PROPOSED SUBCONTRACTORS FORM BID SUBMITTAL CHECKLIST NOTICE OF INTENT TO AWARD NOTICE OF AWARD AGREEMENT NOTICE TO PROCEED REPRESENTATIONS, CERTIFICATIONS AND FORMS PERFORMANCE BOND FORM PAYMENT BOND FORM MAINTENANCE BOND FORM CITY OF JOLIET AFFIDAVITS ILLINOIS DRUG FREE WORKPLACE ACT ILLINOIS CONTRACTOR CERTIFICATION ILLINOIS COMPLIANCE WITH SAFETY REQUIREMENTS ILLINOIS TAX DELINQUENCY/DEFAULT SUBSTANCE ABUSE PREVENTION PROGRAM EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORKS ACT CERTIFICATE OF INSURANCE FORM ADDITIONAL INSUREDS

Table of Contents - 1 -(210039.40) IEPA-PWSLP

Section Number	Section
00 62 16.16 00 62 16.19 00 62 16.21 00 62 33.13	ADDITIONAL INSUREDS ADDITIONAL INSUREDS ADDITIONAL INSUREDS BIDDER CERTIFICATION REGARDING THE USE OF AMERICAN IRON AND STEEL PRODUCTS
00 62 76.01 00 62 76.02 00 62 76.03	SWORN STATEMENT PARTIAL WAIVER OF LIEN CONTRACTOR'S AFFIDAVIT
00 62 76.04 00 62 76.14 00 62 79 00 63 49	FINAL WAIVER OF LIEN TAX EXEMPTION CERTIFICATE STORED MATERIAL FORM WORK CHANGE DIRECTIVE FORM
00 63 63 00 65 16 00 65 19 00 65 36	CHANGE ORDER FORM CERTIFICATE OF SUBSTANTIAL COMPLETION FORM CERTIFICATE OF COMPLETION FORM WARRANTY FORM
00 65 36 00 65 36.03 00 72 00 00 73 00.41	CONTRACTOR'S AFFIDAVIT LETTER GENERAL CONDITIONS SUPPLEMENTARY CONDITIONS
00 73 00.42 00 74 23.01 00 74 23.02 00 74 23.03	DAVIS-BACON ACT WAGE DETERMINATION SCHEDULE DBE FORM #1 (CONTRACTOR CERTIFICATION) DBE FORM #3 (SUBCONTRACTOR UTILIZATION) DBE FORM #4 (BIDDER'S LIST)
00 74 23.04 00 74 23.05	BIDDER CERTIFICATION IN COMPLIANCE WITH ARTICLE 33E-11 TO THE "CRIMINAL CODE OF 2012" USEPA CERTIFICATION OF NONSEGREGATED FACILITIES
00 74 23.06 00 74 23.07	NOTICE TO LABOR UNIONS OR OTHER ORGANIZATIONS OF WORKERS NONDISCRIMINATION IN EMPLOYMENT USEPA CERTIFICATION REGARDING DEBARMENT, SUSPENSION AND OTHER RESPONSIBILITY MATTERS
00 74 23.52 00 74 23.55 00 74 23.57 00 74 23.58	WHISTLEBLOWERS RIGHTS POSTER (APPENDIX B) EMPLOYEE RIGHTS UNDER THE DAVIS-BACON ACT POSTER (APPENDIX E)
DIVISION 01	- GENERAL REQUIREMENTS
01 22 29 01 26 13 01 26 13.13 01 31 19 01 32 16 01 32 36 01 33 01 01 41 26	MEASUREMENT AND PAYMENT REQUESTS FOR INTERPRETATION RFI FORM PROJECT MEETINGS CONSTRUCTION PROGRESS SCHEDULES PRECONSTRUCTION VIDEO RECORDING SUBMITTALS PERMITS

Table of Contents
- 2 (210039.40) IEPA-PWSLP

Section Number	<u>Se</u>	<u>ection</u>
01 42 13 01 45 29 01 50 00 01 62 13 01 66 11 01 74 26 01 77 01 01 78 39		ABBREVIATIONS AND ACRONYMS TESTING LABORATORY SERVICES TEMPORARY FACILITIES AND CONTROLS AMERICAN IRON AND STEEL DOCUMENTATION STORAGE AND PROTECTION OF MATERIAL AND EQUIPMENT PIPELINE CONSTRUCTION SITE CLEANING CONTRACT CLOSEOUT PROJECT RECORD DOCUMENTS
DIVISION 31	-	EARTHWORK
31 23 79 31 25 00		TRENCHING, BACKFILLING, AND COMPACTING EROSION AND SEDIMENTATION CONTROLS
DIVISION 32	-	EXTERIOR IMPROVEMENTS
32 10 00.19 32 92 00.13		STREET, DRIVEWAY, AND SIDEWALK REPLACEMENT LAWNS AND GRASSES
DIVISION 33	-	UTILITIES
33 05 23.16 33 11 43		WATER SERVICES DRILLED UNDER PAVEMENTS WATER DISTRIBUTION SYSTEM

ADVERTISEMENT FOR BIDS

CITY OF JOLIET, ILLINOIS

1. <u>Time and Place of Opening Bids</u>. Sealed proposals for the construction of the Ridgewood Phase 3 Water Main Improvements for the City of Joliet, Will County, Illinois, will be received at the Council Chambers, City of Joliet Municipal Building, 150 West Jefferson Street, Joliet, Illinois 60432 until 2:00 P.M., December 20, 2021, and at that time will be publicly opened and unit price total read aloud. It is highly recommended that bids be tabbed to mark the Bid Bond and unit price total pages and that an electronic copy of the entire submission be included on a USB flash drive. Those in attendance must adhere to the current City of Joliet City Hall COVID-19 policy. The City of Joliet follows current CDC guidelines regarding COVID-19. Bid results will also be posted on the City's website, and emails sent out to individuals who have subscribed to the related RSS feed. It is preferred that you mail your bids/proposals. They should be addressed as follows:

OFFICE OF THE CITY CLERK 150 W. JEFFERSON ST. JOLIET, IL 60432

If you do choose to hand deliver your bid/proposal, they are to be hand delivered to the East or West side of City Hall, 150 W. Jefferson St., Joliet, IL 60432 and marked clearly on the outside of the SEALED package with the BID NUMBER AND NAME OF THE PROJECT, DATE AND TIME OF THE BID OPENING, NAME AND ADDRESS AND PHONE NUMBER OF YOUR COMPANY, and RECEIPT OF ALL ADDENDA (if applicable). All other doors will be locked. Please make sure to mention you are delivering a sealed bid/proposal, so the receiver knows to time stamp the envelope upon receipt. If dropping off a bid in person, bids must be dropped off during business hours only between 8:00 A.M. and 4:30 P.M.

- 2. <u>Description of Work.</u> The proposed construction consists of the installation of approximately 130 lineal feet of 6-inch ductile iron main, 4,145 lineal feet of 8-inch ductile iron main, 20 lineal feet of 12-inch ductile iron main; water services, lawn restoration; and other miscellaneous items of work. The contract will include all work necessary to install the water main, install water service lines via directional drilling or open cut methods, reconnect existing water service lines, restore the roadway, and perform all restoration to return the area to its original condition.
- 3. <u>Information for Bidders.</u> Any contract or contracts awarded under this invitation for bids are expected to be funded in part by a loan from the Illinois Environmental Protection Agency (IEPA). Neither the State of Illinois nor any of its departments, agencies, or employees is or will be a party to this invitation for bids or any resulting contract. The procurement will be subject to regulations contained in the procedures for issuing loans from the Public Water Supply Loan Program (35 IAC Part 662), the Davis-Bacon Act (40 USC 276a through 276a-5) as defined by the United States Department of Labor, the Employment of Illinois Workers on Public Works Act (30 ILCS 570), the Illinois Works Jobs Program Act Apprenticeship Initiative, the Disadvantaged Business Enterprise policy per 40 CFR Part 33, as amended, and the "Use of American Iron and

Steel" requirements as contained in Section 436 of H.R. 3547, the Consolidated Appropriations Act, 2014. This procurement is also subject to loan recipient's policy regarding the increased use of disadvantaged business enterprises. The loan recipient's policy requires all bidders to undertake specified affirmative efforts at least sixteen (16) days prior to bid opening. The policy is contained in the specifications.

Bidders are also required to comply with the President's Executive Order No. 11246, as amended. The requirements for Bidders and Contractors under this order are explained in 41 CFR 60-4. The City of Joliet Local Bidder Ordinance does not apply to this contract.

All pertinent documents may be examined at the Office of the City Clerk, City of Joliet Municipal Building, 150 West Jefferson Street, Joliet, Illinois 60432 between the hours of 8:00 A.M. and 4:30 P.M., Monday through Friday. Hard copies will not be available for purchase. Electronic copies of the Bidding Documents can be downloaded free of charge at https://www.joliet.gov/departments/finance/purchasing/bids-proposals.

Contractor(s) shall pay prevailing wages at rates not less than those under Davis-Bacon Wage Act Provisions as determined by U.S. Department of Labor to all laborers, workmen and mechanics performing work under this contract.

Bidding Document holder should sign up for feeds at https://www.joliet.gov/departments/finance/purchasing/bids-proposals/construction-publicutilities and provide first and last name and email address to automatically receive addendums. Addendums Joliet's also be posted on the City of https://www.joliet.gov/departments/finance/purchasing/bids-proposals. The potential Contractor remains responsible for obtaining all addenda to the original specifications and should check the specific bid page before submitting a bid to ensure they have received all addenda.

Prequalification pursuant to Ordinance No. 7345 is necessary. Bidders are required to be pre-qualified through the Illinois Department of Transportation, the Capital Development Board or the City of Joliet. It is the responsibility of the bidder to ensure that their pre-qualification information is provided to the City of Joliet Purchasing Division prior to the bid opening. If bidders are not prequalified through IDOT or Capital Development Board, then they must be prequalified with the City of Joliet. Financial prequalification forms can be obtained from the City of Joliet website at https://www.joliet.gov/departments/finance/purchasing/prequalification-process. This prequalification MUST be renewed yearly. To check on your current prequalification status, you can contact purchasing@joliet.gov. The current price for City of Joliet prequalification is \$175, which offsets the costs for independent auditor review of the documents. Those documents are to be submitted to the Purchasing Division, City of Joliet, 150 West Jefferson Street, Joliet, IL 60432 at least 6 days prior to the bid opening.

If at the time the Agreement for this Project is executed, or if during the term of the Agreement, there is excessive unemployment in Illinois as defined in the Employment of Illinois Workers on Public Works Act, 30 ILCS 570/0.01 et seq., as two consecutive months of unemployment exceeding 5%, the Bidder agrees to employ Illinois laborers in accordance with the "Employment of Illinois Workers on Public Works Act". An "Illinois laborer" is defined as any person who has resided in Illinois for at least thirty (30) days and intends to become or remain an Illinois resident.

All proposals are subject to the requirements of the City of Joliet Procurement Code (Section 2-430 - 2-453) of the Bidder Instructions.

All Bids must be accompanied by a Bidder's bond, certified check, bank cashier's check or bank draft payable to the City of Joliet for ten percent (10%) of the total amount of the Bid as provided in the Bidder Instructions.

A **mandatory** Pre-Bid Conference of all prospective Bidders and/or their representatives will be held virtually on November 30, 2021 at 10:00 A.M. Bidders are **required** to attend and participate in the conference. Email Lauren Schuld (lschuld@baxterwoodman.com) for conference call information.

Direct all questions about the meaning or intent of the Bidding Documents to the Engineer Lauren Schuld (<u>Ischuld@baxterwoodman.com</u>).

4. <u>Rejection of Bids</u>. The Council reserves the right to reject any or all Bids and to waive technicalities. Unless the Bids are rejected for good cause, award of contract shall be made to the lowest responsible and responsive Bidder.

Advertisement Date(s):	Herald News, November 18, 2021		
	James V. Capparelli, City Manager		
	Dawn A. Kochan, Purchasing/Contracts Administrator		

END OF ADVERTISEMENT FOR BIDS

00 21 00.41

BIDDER INSTRUCTIONS

1. DEFINED TERMS

1.01 Terms used in these Bidder Instructions which are defined in the "Standard General Conditions of the Construction Contract" (EJCDC No. C-700, 2007 edition) have the meanings assigned to them in the General Conditions. The term "Bidder" means one who submits a Bid directly to Owner, as distinct from a sub-bidder, who submits a bid to a Bidder. The term "Successful Bidder" means the lowest, qualified, responsible and responsive Bidder to whom Owner (on the basis of Owner's evaluation as hereinafter provided) makes an award. The term "Bidding Documents" includes the Advertisement for Bids, Bidder Instructions, the Bid Form, all submittals listed in 00 43 93.41 Bid Submittal Checklist, and the proposed Contract Documents, including all Addenda issued prior to receipt of Bids.

COPIES OF BIDDING DOCUMENTS

- 2.01 Complete sets of the Bidding Documents can be obtained from the City of Joliet, 150 West Jefferson Street, Joliet, Illinois 60432 in accordance with the Advertisement for Bids.
- 2.02 Complete sets of Bidding Documents must be used in preparing Bids; neither Owner nor Engineer assume any responsibility for errors or misinterpretations resulting from the use of incomplete sets of Bidding Documents.
- 2.03 Owner and Engineer in making copies of Bidding Documents available do so only for the purpose of obtaining Bids on the Work and do not confer a license or grant for any other use.

3. QUALIFICATIONS OF BIDDERS

- 3.01 To demonstrate qualifications to perform the Work, each Bidder must be prequalified to complete work for the City of Joliet. The Bidder/Contractor will be considered prequalified in one or more three ways as stated in 00 11 13 Advertisement for Bids:
 - 1) Being prequalified with the Illinois Department of Transportation;
 - 2) Being prequalified with the State of Illinois Capital Development Board;
 - 3) Being previously prequalified with the City of Joliet and included on their list of prequalified Contractors, or submitting the necessary from and financial statement to the City of Joliet at least six (6) days prior to the bid opening.

Financial prequalification forms can be obtained from the City of Joliet website at https://www.joliet.gov/departments/finance/purchasing/prequalification-process. This prequalification must be renewed yearly. The current price for City of Joliet prequalification is \$175, which offsets the costs for independent auditor review of the documents. Those documents are to be submitted to the Office of the City Clerk, City of Joliet, 150 West Jefferson Street, Joliet, Illinois 60432 at least six (6) days prior to the bid opening and must be accompanied by \$175. To check prequalification status, contact purchasing@joliet.gov.

4. EXAMINATION OF CONTRACT DOCUMENTS AND SITE

- 4.01 It is the responsibility of each Bidder before submitting a Bid to: (a) examine the Contract Documents thoroughly; (b) visit the site to become familiar with local conditions that may affect cost, progress, performance, or furnishing of the Work; (c) consider federal, state and local Laws and Regulations that may affect cost, progress, performance of furnishing of the Work; (d) study and carefully correlate Bidder's observations with the Contract Documents, and (e) notify Engineer of all conflicts, errors or discrepancies in the Contract Documents.
- 4.02 Provisions concerning responsibilities for the adequacy of data furnished to prospective Bidders on subsurface conditions, Underground Facilities, and other physical conditions, and possible changes in the Contract Documents due to differing conditions appear in Paragraphs 4.02, 4.03 and 4.04 of the General Conditions.
- 4.03 Before submitting a Bid, each Bidder will, at Bidder's own expense, make or obtain any additional examinations, investigations, explorations, tests and studies, and obtain any additional information and data which pertain to the physical conditions (surface, subsurface, and Underground Facilities) at or contiguous to the site or otherwise which may affect cost, progress, performance or furnishing of the Work, and which Bidder deems necessary to determine his Bid for performing and furnishing the Work in accordance with the time, price, and other terms and conditions of the Contract Documents.
- 4.04 On request in advance, Owner will provide each Bidder access to the site to conduct such explorations and tests as each Bidder deems necessary for submission of a Bid. Bidder shall fill all holes, clean up and restore the site to its former condition upon completion of such explorations.
- 4.05 The lands upon which the Work is to be performed, rights-of-way and easements for access thereto, and other lands designated for use in performing the Work are identified in the Contract Documents. All additional lands and access thereto required for temporary construction facilities or storage of materials and equipment are to be provided by the Successful Bidder. Easements for structures or changes in existing structures are to be obtained and paid for by Owner unless otherwise provided in the Contract Documents.
- 4.06 The submission of a Bid will constitute an incontrovertible representation by Bidder that Bidder has complied with every requirement of this Article 4, that without exception, the Bid is premised upon performing and furnishing the Work required by the Contract Documents and such means, methods, techniques, sequences or procedures of construction as may be indicated in or required by the Contract Documents, and that the Contract Documents are sufficient in scope and detail to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work.
- 4.07 The Contract Documents contain the provisions required for the construction of the Project. Information obtained from an officer, agent, or employee of the Owner or any person shall not affect the risks or obligations assumed by the Contractor or relieve him from fulfilling any of the conditions of the contract.

5. PRE-BID CONFERENCE

5.01 A **mandatory** Pre-Bid Conference will be held at the time and place given in the ADVERTISEMENT FOR BIDS. Representatives of Owner and Engineer will be present to discuss the Project. Bidders are **required** to attend and participate in the conference. Engineer will transmit to all prospective Bidders of record such Addenda as Engineer considered necessary in response to questions arising at the conference. Oral statements may not be relied upon and will not be binding or legally effective.

6. INTERPRETATIONS AND ADDENDA

- 6.01 Direct all questions about the meaning or intent of the Contract Documents to Engineer (Telephone 815.444.3306 or email lschuld@baxterwoodman.com). Interpretations or clarifications considered necessary by Engineer in response to such questions will be issued by Addendum emailed by the Engineer or the Engineer's designated independent distributor to all parties recorded by Engineer as having received the Bidding Documents. Questions received less than 72 hours prior to the Bid Opening might not be answered. Only questions answered by formal written Addendum will be binding. Oral and other interpretations or clarifications will be without legal effect.
- 6.02 Addenda may also be issued to modify the Bidding Documents as deemed advisable by Owner or Engineer.
- 6.03 Addenda will be provided in an electronic format (.pdf format) via email. All Bidding holders should Document sign up for the RSS feeds https://www.joliet.gov/departments/finance/purchasing/bids-proposals/construction-publicutilities and provide first and last name and email address to automatically receive addendums. Addendums will also be posted on the Citv of Joliet's https://www.joliet.gov/departments/finance/purchasing/bids-proposals/construction-publicutilities. All Bidding Document holders must provide a name, address, and e-mail address to receive electronic addenda and must reply to the e-mail to indicate receipt of the addenda.

7. BID SECURITY

- 7.01 Each Bid must be accompanied by Bid security made payable to Owner in an amount and in the form stated in the Advertisement for Bids.
- 7.02 The Bid securities of all except the three lowest responsive, responsible Bidders will be returned promptly after the Bids have been tabulated and checked. The Bid securities of the three lowest responsive, responsible Bidders will be returned as soon as the contract and bonds of the successful Bidder have been properly executed and approved.

8. CONTRACT TIMES

8.01 The number of days within which, or the dates by which, the Work is to be (a) Substantially Completed and (b) also completed and ready for final payment are set forth on the Bid Form.

9. LIQUIDATED DAMAGES

9.01 Provisions for liquidated damages, if any, are set forth in the Supplementary Conditions, Article 12.

10. SUBSTITUTE OR "OR EQUAL" ITEMS

10.01 The Contract, if awarded, will be based solely on the materials and equipment described in the Drawings and Specifications without consideration of possible substitute or "or equal" items.

11. SUBCONTRACTORS, SUPPLIERS, AND OTHERS

- 11.01 The apparent low Bidder, or any other Bidder so requested, shall submit to the office of Engineer within five (5) business days after the receipt of Bids, the following information on 00 43 36 Proposed Subcontractor Form:
 - A. A list of the names of Bidder's proposed Subcontractors whose portion of the Work exceeds \$10,000, along with a description of the Work to be performed by each.
- 11.02 Bidder shall be required to perform with his own Work forces at least 30% of the Work, unless written consent to subcontract a greater percentage of the Work is obtained from Owner.

12. BID FORM

- 12.01 Only the Bid Forms attached hereto shall be used. Where required, the Bid price of each item shall be in writing and in figures; in case of conflict, the former shall apply.
- 12.02 Bidder, when signing the Bid, shall meet the following requirements:
 - A. The full name, telephone number, and business address of each Bidder must be entered on the Bid submitted. The Bid shall be signed in the space provided therefore by written signature of the person or persons properly authorized to sign it. All names must be typed or printed below the signature.
 - B. A Bid submitted by an individual shall be signed by Bidder or by an authorized agent.
 - C. Bids by partnerships must be executed in the partnership names and signed by a partner, whose title must appear under the signature and the official address of the partnership must be shown below the signature.

- D. Bids by corporations must be executed in the corporate name by the president or a vice-president (or other corporate officer accompanied by evidence of authority to sign) and the corporate seal must be affixed and attested by the secretary or an assistant secretary. The state of incorporation shall be shown below the signature. Such corporation must be licensed to do business in the state in which the Project is located before a Contract to do the Work embraced in the Bid can be signed. If a foreign corporation, the state under which it is incorporated must be named.
- E. Bids which are signed by an attorney-in-fact for individuals, firms, partnerships or joint ventures shall have attached thereto a power-of-attorney evidencing authority to sign the Bid.
- 12.03 The Bid shall contain an acknowledgment of receipt of all Addenda, the numbers and dates of which shall be filled in on the Bid Form.
- 12.04 It is the responsibility of the Bidder to submit a neat, accurate, and complete Bid.

13. BASIS OF BID; EVALUATION OF BIDS

13.01 Unit price:

- A. Bidders shall submit a Bid on a unit price basis for each item of Work listed in the Bid schedule.
- B. The total of all estimated prices will be determined as the sum of the products of the estimated quantity of each item and the unit price Bid for the item. The final quantities and Contract Price will be determined in accordance with paragraph 11.03 of the General Conditions.
- C. Discrepancies between the multiplication of units of Work and unit prices will be resolved in favor of the unit prices. Discrepancies between the indicated sum of any column of figures and the correct sum thereof will be resolved in favor of the correct sum. Discrepancies between words and figures will be resolved in favor of the words.
- 13.02 The Bid price shall include such amounts as the Bidder deems proper for overhead and profit on account of cash allowances, if any, named in the Contract Documents as provided in Paragraph 11.02 of the General Conditions.

14 SUBMISSION OF BIDS

14.01 Bids shall be submitted at the time and place indicated in the Advertisement for Bids and shall be sealed in an opaque envelope, marked with the Project title (and if applicable, the designated portion of the Project for which the Bid is submitted and acknowledgement of any addenda), name and address of the Bidder, and accompanied by the Bid Security and other required documents. If the Bid is sent through the mail or other delivery system, the sealed envelope shall be enclosed in a separate envelope addressed to the Owner with the notation "BID

ENCLOSED" on the face thereof. Any Bids received after the official Bid closure time will be returned to the Bidder unopened.

15 MODIFICATION AND WITHDRAWAL OF BID

- 15.01 Bids may be modified or withdrawn by notice to Owner at the place designated for receipt of Bids any time prior to opening of Bids. Such notice shall be in writing and signed by Bidder.
- 15.02 Withdrawn Bids may be resubmitted up to the time designated for the receipt of Bids provided that they are then fully in conformance with these Bidder Instructions.
- 15.03 Bid security shall be in an amount sufficient for the Bid as modified or resubmitted.
- 15.04 After the Bid opening, no Bid may be modified, withdrawn or canceled by Bidder without consent of Owner.

16 OPENING OF BIDS

16.01 Bids will be publicly opened and read aloud. An abstract of the amounts of the base Bids and major alternates (if any) will be made available by Owner to Bidders after opening of Bids.

17 BIDS TO REMAIN SUBJECT TO ACCEPTANCE

17.01 All Bids will remain subject to acceptance for the period of time stated in the Bid Form, but Owner may, in its sole discretion, release any Bid and return the Bid security prior to the end of this period.

18 AWARD OF CONTRACT

- 18.01 Owner reserves the right to reject Bids, to waive informalities not involving price, time or changes in the Work, and to disregard all nonconforming, nonresponsive, unbalanced or conditional Bids. Also, Owner reserves the right to reject the Bid of any Bidder if Owner believes that it would not be in Owner's best interest to make an award to that Bidder, whether because the Bid is not responsive or the Bidder is unqualified or of doubtful financial ability or fails to meet any other pertinent criteria established by Owner.
- 18.02 In evaluating Bids, Owner shall consider the qualifications of the Bidders, whether or not the Bids comply with the prescribed requirements, and such alternates, unit prices, supplemental prices, and other data as may be requested in the Bid Form or prior to the Notice of Award.
- 18.03 Owner may conduct such investigations as Owner deems necessary to assist in the evaluation of any Bid and to establish the responsibility, qualifications, and financial ability of the Bidders, proposed Subcontractors, Suppliers, and other persons and organizations to perform

and furnish the Work in accordance with the Contract Documents to Owner's satisfaction within the prescribed time.

- 18.04 Owner shall have the right to accept alternates in any order or combination, unless specifically otherwise provided.
- 18.05 A Bid which has not been prepared according to the instructions contained herein or which does not contain a unit price which is both adequate and reasonable for each and every item named in the Bid may be considered irregular and subject to rejection. Unbalanced Bids which cannot be resolved by mathematical correction without resorting to information not contained in the Bid, are subject to rejection.
- 18.06 If the Contract is to be awarded, it will be awarded to the lowest responsive, responsible Bidder.
- 18.07 If the Contract is awarded, Owner will give Successful Bidder notification in writing at the earliest possible date.

19 CONTRACT SECURITY AND INSURANCE

19.01 Article 5 of the General Conditions and the Supplementary Conditions set forth Owner's requirements as to Performance, Payment and other bonds, and insurance requirements. When the Successful Bidder delivers the executed Agreement to Owner, it shall be accompanied by the required Performance Bond, Payment Bond, and Certificate of Insurance.

20 SIGNING OF AGREEMENT

- 20.01 When Owner gives a Notice of Award to the Successful Bidder, it will be accompanied by the appropriate number of unsigned counterparts of the Agreement with all other Contract Documents attached. Within ten (10) days after the date of the Notice of Award, the Successful Bidder shall sign and deliver the required number of counterparts of the Agreement and attached documents to Owner or Engineer, as so directed, with the required bonds and insurance certificate(s). Within ten (10) days thereafter, Owner will deliver fully signed counterparts to Engineer for further distribution to Contractor.
- 20.02 Failure on the part of the Successful Bidder to execute the Agreement and provide acceptable bonds and insurance certificate(s) within ten (10) days after the date the contract was mailed or otherwise delivered to him, will be just cause for the annulment of the award and the forfeiture of the Bid security to Owner, not as a penalty but in payment of liquidated damages sustained as a result of such failure. If Owner fails to execute the Agreement within a reasonable time, not to exceed ninety (90) days after receiving the executed Agreement and acceptable bonds from the Successful Bidder, the Successful Bidder shall have the right to withdraw his proposal without forfeiture of his Bid security.

21 SALES AND USE TAXES

21.01 Owner is exempt from Illinois State sales and use taxes on materials and equipment to be incorporated in the Work. Said taxes shall not be included in the Bid.

22 WAGE RATE REQUIREMENTS

- 22.01 No less than the prevailing rate of wages under the Davis-Bacon Wage Act (40 USC 276a through 276a-5) as determined by the U.S. Department of Labor shall be paid to all laborers, workmen and mechanics performing work under this contract. Contractor and all Subcontractors engaged in any part of the Project will be required to comply with the Davis-Bacon Wage Act requirements and the cost of doing so must be included in the Bid as no extra payment on account of this provision of law will be allowed.
- 22.02 If the Department of Labor revises prevailing wage rates, the revised prevailing wage rates on the Department of Labor's website (https://beta.sam.gov) shall apply to this contract and the Contractor will not be allowed additional compensation on account of said revisions. The Contractor shall review the wage rates applicable to the work of the contract at regular intervals to ensure timely payment of current wage rates. The Contractor agrees no additional notice is required. The Contractor shall be responsible for notifying each subcontractor of the wage rates set forth in this contract and any revisions thereto.
- 22.03 Provide certified weekly payrolls to the Owner on a weekly basis to demonstrate compliance.

23 STATE OF ILLINOIS DRUG FREE WORKPLACE CERTIFICATION

- 23.01 The Drug Free Workplace Act (Illinois Compiled Statutes Chapter 30-580) requires that no Contractor be considered for the purposes of being awarded a contract for the procurement of any property or services from the State unless that Contractor has certified to the State that the Contractor will provide a drug free workplace.
- 23.02 For the purpose of this certification, "Contractor" means a corporation, partnership, or other entity with twenty-five (25) or more employees at the time of issuing the grant, or a department, division, or other unit thereof, directly responsible for the specific performance under a contract of \$5,000 or more from the State.
- 23.03 Provide a signed certification of Drug Free Workplace Act compliance (Certificate 00 62 03) with the Bid.

24 STEEL PRODUCTS PROCUREMENT

24.01 Comply with the provisions of the Illinois Steel Products Procurement Act (30 ILCS 565/).

25 SUBSTANCE ABUSE PREVENTION PROGRAM.

25.01 Illinois P. A. 095-0635, Substance Abuse Prevention on Public Works Projects Act requires that all Contractors and Subcontractors have a written Substance Abuse Prevention Program, or collective bargaining agreement, in effect that includes a Substance Abuse Prevention Program. Submit a copy of the Substance Abuse Prevention Program or collective bargaining agreement with the governing body, along with a cover letter certifying that their program meets the requirements of the Act, or, as an alternative, use Certificate 00 62 07 from these Specifications. This requirement applies to Contractors and Subcontractors.

25.02 Provide signed certification of compliance (Certificate 00 62 07) with the bid.

26 EMPLOYMENT OF ILLINOIS WORKERS

26.01 Comply with the provisions of the Employment of Illinois Workers on Public Works Act (30 ILCS 570/) as they may apply to this project.

26.02 If at the time the Agreement for this Project is executed, or if during the term of the Agreement, there is excessive unemployment in Illinois as defined in the Employment of Illinois Workers on Public Works Act, 30 ILCS 570/0.01 et seq., as two consecutive months of unemployment exceeding 5%, the Bidder agrees to employ Illinois laborers in accordance with the "Employment of Illinois Workers on Public Works Act". An "Illinois laborer" is defined as any person who has resided in Illinois for at least thirty (30) days and intends to become or remain an Illinois resident.

26.03 Provide signed certification of compliance (Certificate 00 62 08) with the bid.

27 UNCONTAMINATED SOIL CERTIFICATION AND DISPOSAL

27.01 Comply with the requirement of Sections 22.51(f)(2)(B) and 22.51a(d)(2)(B) of the Illinois Environmental Protection Act ([415 ILCS5/22.51(f)26(B)] and [415 ILCS5/22.51a(d)(2)(B)]) for the disposal of uncontaminated soils including uncontaminated soil mixed with other clean construction or demolition debris (CCDD) materials.

28 AUDIT AND ACCESS TO RECORDS

28.01 Any contract entered into by the loan recipient and any subagreement thereunder shall provide that representatives of the Agency will have access to the work whenever it is in preparation or progress and that the Contractor or Subcontractor will provide proper facilities for such access and inspection. Such contract or subagreement must also provide that the Agency or any authorized representative shall have access to any books, documents, papers, and records of the Contractor or Subcontractor which are pertinent to the project for the purpose of making audit, examination, excerpts, and transcriptions thereof.

29 DISCRIMINATION

29.01 Bidder shall not discriminate on the basis of race, color, national origin or sex in the performance of this contract. The contractor shall carry out applicable requirements of 40 CFR Part 33 in the award and administration of contracts awarded under EPA financial assistance agreements. Failure by the contractor to carry out these requirements is a material breach of this contract which may result in the termination of this contract or other legally available remedies.

30 PUBLIC WATER SUPPLY LOAN PROGRAM DOCUMENTS

30.01 The following documents attached to this Section describe the requirements of the loan program:

- Affirmative Action for Equal Employment Opportunity (E.O. 11246)
- USEPA Certification of Nonsegregated Facilities (Certificate 00 74 23.05)
- Nondiscrimination in Employment Form (Certificate 00 74 23.06)
- USEPA Certification Regarding Debarment, Suspension, and Other Responsibility Matters (Certificate 00 74 23.07)
- Construction Contracts of Loan Recipient and Other Sections from "Procedures For Issuing Loans From the Public Water Supply Loan Program" (Section 00 21 00.41, Pages 25 to 28)
- Disadvantaged Business Enterprise Participation Specifications (Section 00 21 00.41, Pages 29 to 32)
- Davis-Bacon Wage Act Compliance Guidance
- Use of American Iron and Steel
- Illinois Works Jobs Program Act Apprenticeship Initiative Information for Contractors

NOTICE OF REQUIREMENT FOR AFFIRMATIVE ACTION TO ENSURE EQUAL EMPLOYMENT OPPORTUNITY (EXECUTIVE ORDER 11246)

- 1. The Bidder's attention is called to the "Equal Opportunity Clause" and the Standard Federal Equal Employment Opportunity Construction Contract Specifications" set forth herein.
- 2. The goals and timetables for minority and female participation, expressed in percentage terms for the Bidder's aggregate workforce in each trade on all construction work in the covered area, are as follows:

Timetables	Goals for Minority Participation for Each Trade	Goals for Female Participation in Each Trade	
	20.9%	6.9%	

These goals are applicable to all the Bidder's construction work (whether or not it is federal or federally assisted) performed in the covered area.

The Bidder's compliance with the Executive Order and the regulations in 41 CFR Part 60-4 shall be based on its implementation of the Equal Opportunity Clause, specific affirmative action obligations required by the specifications set forth in 41 CFR 60-4.3(a), and its efforts to meet the goals established for the geographical area where the contract resulting from this solicitation is to be performed. The hours of minority and female employment and training must be substantially uniform throughout the length of the contract, and in each trade, and the Bidder shall make a good faith effort to employ minorities and women evenly on each of its projects. The transfer of minority or female employees or trainees from Contractor to Contractor or from project to project for the sole purpose of meeting the Bidder's goals shall be a violation of the contract, the Executive Order and the regulations in 41 CFR 60-4. Compliance with the goals will be measured against the total work hours performed.

- 3. The Bidder shall provide written notification to the director of the Office of Federal Contract Compliance Programs within 10 working days of award of any construction subcontract in excess of \$10,000 at any tier for construction work under the contract resulting from this solicitation. The notification shall list the name, address and telephone number of the subcontractor; employer identification number; estimated dollar amount of the subcontract; estimated starting and completion dates of the subcontract; and the geographical area in which the contract is to be performed.
- 4. As used in this notice, and in the contract resulting from this solicitation, the "covered area" is the City of Joliet, Will County in the State of Illinois.

41 CFR 60

60-4.1 Scope and Application.

This part applies to all contractors and subcontractors that hold any Federal or federally assisted construction contract in excess of \$10,000. The regulations in this part are applicable to all of a construction contractor's or subcontractor's construction employees who are engaged in on site construction including those construction employees who work on a non-Federal or nonfederally assisted construction site. This part also establishes procedures, which all Federal contracting officers and all applicants, as applicable, shall follow in soliciting for and awarding Federal or federally assisted construction contracts. Procedures also are established which administering agencies shall follow in making any grant, contract, loan, insurance, or guarantee involving federally assisted construction which is not exempt from the requirements of Executive Order 11246, as amended. In addition, this part applies to construction work performed by construction contractors and subcontractors for Federal nonconstruction contractors and subcontractors if the construction work is necessary in whole or in part to the performance of a nonconstruction contract or subcontract.

[43 FR 49254, OCT. 20, 1978; 43 FR 51404, NOV. 3, 1978]

60-4.2 Solicitations.

- (a) All Federal contracting officers and all applicants shall include the notice set forth in paragraph (d) of this section and the Standard Federal Equal Employment Opportunity Construction Contract Specifications set forth in § 60-4.3 of this part in all solicitations for offers and bids on all Federal and federally assisted construction contracts or subcontracts to be performed in geographical areas designated by the Director pursuant to § 60-4.6 of the part. Administering agencies shall require the inclusion of the notice set forth in paragraph (d) of this section and the specifications set forth in § 60-4.3 of this part as a condition of any grant, contract, subcontract, loan, insurance or guarantee involving federally assisted construction covered by this Part 60-4.
- (b) All nonconstruction contractors covered by Executive Order 11246 and the implementing regulations shall include the notice in paragraph (d) of this section in all construction agreements, which are necessary in whole or in part to the performance of the covered nonconstruction contract.
- (c) Contracting officers, applicants and nonconstruction contractors shall given written notice to the Director within 10 working days of award of a contract subject to these provisions. The notification shall include the name, address and telephone number of the contractor; employer identification number; dollar amount of the contract, estimated starting and completion dates of the contract; the contract number; and geographical area in which the contract is to be performed.
- (d) The following notice shall be included in, and shall be a part of, all solicitations for offers and bids on all Federal and federally assisted construction contracts or subcontracts in excess of \$10,000 to be performed in geographical areas designated by the Director pursuant to § 60-4.6 of this part (see 41 CFR 60-4.2(a)):

Notice of Requirement for Affirmative Action To Ensure Equal Employment Opportunity (Executive Order 11246)

- 1. The Offeror or Bidder's attention is called to the "Equal Opportunity Clause" and the "Standard Federal Equal Employment Specifications" set forth herein.
- 2. The goals and timetables for minority and female participation, expressed in percentage terms for the Contractor's aggregate workforce in each trade on all construction work in the covered area, are as follows:

Timetables	Goals for minority participation for each trade	Goals for female participation in each trade
	20.9%	6.9%

These goals are applicable to all the Contractor's construction work (whether or not it is Federal or federally assisted) performed in the covered area. If the contractor performs construction work in a geographical area located outside of the covered area, it shall apply the goals established for such geographical area where the work is actually performed. With regard to this second area, the contractor also is subject to the goals for both its federally involved and nonfederally involved construction.

The Contractor's compliance with the Executive Order and the regulations in 41 CFR Part 60-4 shall be based on its implementation of the Equal Opportunity Clause, specific affirmative action obligations required by the specifications set forth in 41 CFR 60-4.3(a), and its efforts to meet the goals. The hours of minority and female employment and training must be substantially uniform throughout the length of the contract, and in each trade, and the contractor shall make a good faith effort to employ minorities and women evenly on each of its projects. The transfer of minority or female employees or trainees from Contractor to Contractor or from project to project for the sole purpose of meeting the Contractor's goals shall be a violation of the contract, the Executive Order and the regulations in 41 CFR Part 60-4. Compliance with the goals will be measured against the total work hours performed.

- 3. The Contractor shall provide written notification to the Director of the Office of Federal Contract Compliance Programs within 10 working days of award of any construction subcontract in excess of \$10,000 at any tier for construction work under the contract resulting from this solicitation. The notification shall list the name, address and telephone number of the subcontractor; employer identification number of the subcontractor; estimated dollar amount of the subcontract; estimated starting and completion dates of the subcontract; and the geographical area in which the subcontract is to be performed.
- 4. As used in this Notice, and in the contract resulting from this solicitation, the "covered area" is the City of Joliet, Will County in the State of Illinois.

[43 FR 49254, OCT. 20, 1978; 43 FR 51401, NOV. 3, 1978, AS AMENDED AT 45 FR 65977, OCT. 3, 1980]

60-4.3 Equal Opportunity Clauses.

(a) The equal opportunity clause published at 41 CFR 60-1.4(a) of this chapter is required to be included in, and is part of, all nonexempt Federal contracts and subcontracts, including construction contracts and subcontracts. The equal opportunity clause published at 41 CFR 60-1.4(b) is required to be included in, and is a part of, all nonexempt federally assisted construction contracts and subcontracts. In addition to the clauses described above, all Federal contracting officers, all applicants and all nonconstruction contractors, as applicable, shall include the specifications set forth in this section in all Federal and federally assisted construction contracts in excess of \$10,000 to be performed in geographical areas designated by the Director pursuant to § 60-4.6 of this part and in construction subcontracts in excess of \$10,000 necessary in whole or in part to the performance of nonconstruction Federal contracts and subcontracts covered under the Executive order.

Standard Federal Equal Employment Opportunity Construction Contract Specifications (Executive Order 11246)

- **1.** As used in these specifications:
- a. "Covered area," means the geographical area described in the solicitation from which this contract resulted:
- **b.** "Director" means Director, Office of Federal Contract Compliance Programs, United States Department of Labor, or any person to whom the Director delegates authority;
- **c.** "Employer identification number" means the Federal Social Security number used on the Employer's Quarterly Federal Tax Return, U.S. Treasury Department Form 941.
- d. "Minority" includes:
- (i) Black (all persons having origins in any of the Black African racial groups not of Hispanic origin);
- (ii) Hispanic (all persons of Mexican, Puerto Rican, Cuban, Central or South American or other Spanish Culture or origin, regardless of race);
- (iii) Asian and Pacific Islander (all persons having origins in any of the original peoples of the Far East, Southeast Asia, the Indian Subcontinent, or the Pacific Islands); and
- (iv) American Indian or Alaskan Native (all persons having origins in any of the original peoples of North America and maintaining identifiable tribal affiliations through membership and participation or community identification).
- **2.** Whenever the Contractor, or any Subcontractor at any tier, subcontracts a portion of the work involving any construction trade, it shall physically include in each subcontract in excess of \$10,000 the provisions of these specifications and the Notice which contains the applicable goals for minority and female participation and which is set forth in the solicitations from which this contract resulted.

- **3.** If the Contractor is participating (pursuant to 41 CFR 60-4.5) in a Hometown Plan approved by the U.S. Department of Labor in the covered area either individually or through an association, its affirmative action obligations on all work in the Plan area (including goals and timetables) shall be in accordance with that Plan for those trades which have unions participating in the Plan. Contractors must be able to demonstrate their participation in and compliance with the provisions of any such Hometown Plan. Each Contractor or Subcontractor participating in an approved Plan is individually required to comply with its obligations under the EEO clause, and to make a good faith effort to achieve each goal under the Plan in each trade in which it has employees. The overall good faith performance by other Contractors or Subcontractors toward a goal in an approved Plan does not excuse any covered Contractor or Subcontractor's failure to take good faith efforts to achieve the Plan goals and timetables.
- **4.** The Contractor shall implement the specific affirmative action standards provided in paragraphs 7 a through p of these specifications. The goals set forth in the solicitation from which this contract resulted are expressed as percentages of the total hours of employment and training of minority and female utilization the Contractor should reasonably be able to achieve in each construction trade in which it has employees in the covered area. Covered Construction contractors performing construction work in geographical areas where they do not have a Federal or federally assisted construction contract shall apply the minority and female goals established for the geographical area where the work is being performed. Goals are published periodically in the Federal Register in notice form, and such notices may be obtained from any Office of Federal Contract Compliance Programs office or from Federal procurement contracting officers. The Contractor is expected to make substantially uniform progress in meeting its goals in each craft during the period specified.
- **5.** Neither the provisions of any collective bargaining agreement, nor the failure by a union with whom the Contractor has a collective bargaining agreement, to refer either minorities or women shall excuse the Contractor's obligations under these specifications, Executive Order 11246, nor the regulations promulgated pursuant thereto.
- **6.** In order for the nonworking training hours of apprentices and trainees to be counted in meeting the goals, the Contractor must employ such apprentices and trainees during the training period, and the Contractor must have made a commitment to employ the apprentices and trainees at the completion of their training, subject to the availability of employment opportunities. Trainees must be trained pursuant to training programs approved by the U.S. Department of Labor.
- **7.** The Contractor shall take specific affirmative actions to ensure equal employment opportunity. The evaluation of the Contractor's compliance with these specifications shall be based upon its effort to achieve maximum results from its actions. The Contractor shall document these efforts fully, and shall implement affirmative action steps at least as extensive as the following:
- **a.** Ensure and maintain a working environment free of harassment, intimidation, and coercion at all sites, and in all facilities at which the Contractor's employees are assigned to work. The Contractor, where possible, will assign two or more women to each construction project. The Contractor shall specifically ensure that all foremen, superintendents, and other on-site supervisory personnel are aware of and carry out the Contractor's obligation to maintain such a working environment, with specific attention to minority or female individuals working at such sites or in such facilities.

- **b.** Establish and maintain a current list of minority and female recruitment sources, provide written notification to minority and female recruitment sources and to community organizations when the Contractor or its unions have employment opportunities available, and maintain a record of the organizations' responses.
- **c.** Maintain a current file of the names, addresses and telephone numbers of each minority and female off-the-street applicant and minority or female referral from a union, a recruitment source or community organization and of what action was taken with respect to each such individual. If such individual was sent to the union hiring hall for referral and was not referred back to the Contractor by the union or, if referred, not employed by the Contractor, this shall be documented in the file with the reason therefore; along with whatever additional actions the Contractor may have taken.
- **d.** Provide immediate written notification to the Director when the union or unions with which the Contractor has a collective bargaining agreement has not referred to the Contractor a minority person or woman sent by the Contractor, or when the Contractor has other information that the union referral process has impeded the Contractor's efforts to meet its obligations.
- **e.** Develop on-the-job training opportunities and/or participate in training programs for the area which expressly include minorities and women, including upgrading programs and apprenticeship and trainee programs relevant to the Contractor's employment needs, especially those programs funded or approved by the Department of Labor. The Contractor shall provide notice of these programs to the sources compiled under 7b above.
- **f.** Disseminate the Contractor's EEO policy by providing notice of the policy to unions and training programs and requesting their cooperation in assisting the Contractor in meeting its EEO obligations; by including it in any policy manual and collective bargaining agreement; by publicizing it in the company newspaper, annual report, etc.; by specific review of the policy with all management personnel and with all minority and female employees at least once a year; and by posting the company EEO policy on bulletin boards accessible to all employees at each location where construction work is performed.
- **g.** Review, at least annually, the company's EEO policy and affirmative action obligations under these specifications with all employees having any responsibility for hiring, assignment, layoff, termination or other employment decisions including specific review of these items with onsite supervisory personnel such as Superintendents, General Foremen, etc., prior to the initiation of construction work at any job site. A written record shall be made and maintained identifying the time and place of these meetings, persons attending, subject matter discussed, and disposition of the subject matter.
- **h.** Disseminate the Contractor's EEO policy externally by including it in any advertising in the news media, specifically including minority and female news media, and providing written notification to and discussing the Contractor's EEO policy with other Contractors and Subcontractors with whom the Contractor does or anticipates doing business.
- i. Direct its recruitment efforts, both oral and written, to minority, female and community organizations, to schools with minority and female students and to minority and female recruitment and training organizations serving the Contractor's recruitment area and employment

needs. Not later than one month prior to the date for the acceptance of applications for apprenticeship or other training by any recruitment source, the Contractor shall send written notification to organizations such as the above, describing the openings, screening procedures, and tests to be used in the selection process.

- **j.** Encourage present minority and female employees to recruit other minority persons and women and, where reasonable, provide after school, summer and vacation employment to minority and female youth both on the site and in other areas of a Contractor's work force.
- **k.** Validate all tests and other selection requirements where there is an obligation to do so under 41 CFR Part 60-3.
- **I.** Conduct, at least annually, an inventory and evaluation at least of all minority and female personnel for promotional opportunities and encourage these employees to seek or to prepare for, through appropriate training, etc., such opportunities.
- **m.** Ensure that seniority practices, job classifications, work assignments and other personnel practices, do not have a discriminatory effect by continually monitoring all personnel and employment related activities to ensure that the EEO policy and the Contractor's obligations under these specifications are being carried out.
- **n.** Ensure that all facilities and company activities are nonsegregated except that separate or single-user toilet and necessary changing facilities shall be provided to assure privacy between the sexes.
- **o.** Document and maintain a record of all solicitations of offers for subcontracts from minority and female construction contractors and suppliers, including circulation of solicitations to minority and female contractor associations and other business associations.
- **p.** Conduct a review, at least annually, of all supervisors' adherence to and performance under the Contractor's EEO policies and affirmative action obligations.
- **8.** Contractors are encouraged to participate in voluntary associations, which assist in fulfilling one or more of their affirmative action obligations (7a through p). The efforts of a contractor association, joint contractor-union, contractor-community, or other similar group of which the contractor is a member and participant, may be asserted as fulfilling any one or more of its obligations under 7a through p of these Specifications provided that the contractor actively participates in the group, makes every effort to assure that the group has a positive impact on the employment of minorities and women in the industry, ensures that the concrete benefits of the program are reflected in the Contractor's minority and female workforce participation, makes a good faith effort to meet its individual goals and timetables, and can provide access to documentation which demonstrates the effectiveness of actions taken on behalf of the Contractor. The obligation to comply, however, is the Contractor's and failure of such a group to fulfill an obligation shall not be a defense for the Contractor's noncompliance.
- **9.** A single goal for minorities and a separate single goal for women have been established. The Contractor, however, is required to provide equal employment opportunity and to take affirmative

action for all minority groups, both male and female, and all women, both minority and non-minority. Consequently, the Contractor may be in violation of the Executive Order if a particular group is employed in a substantially disparate manner (for example, even though the Contractor has achieved its goals for women generally, the Contractor may be in violation of the Executive Order if a specific minority group of women is underutilized).

- **10.** The Contractor shall not use the goals and timetables or affirmative action standards to discriminate against any person because of race, color, religion, sex, or national origin.
- **11.** The Contractor shall not enter into any Subcontract with any person or firm debarred from Government contracts pursuant to Executive Order 11246.
- **12.** The Contractor shall carry out such sanctions and penalties for violation of these specifications and of the Equal Opportunity Clause, including suspension, termination and cancellation of existing subcontracts as may be imposed or ordered pursuant to Executive Order 11246, as amended, and its implementing regulations, by the Office of Federal Contract Compliance Programs. Any Contractor who fails to carry out such sanctions and penalties shall be in violation of these specifications and Executive Order 11246, as amended.
- **13.** The Contractor, in fulfilling its obligations under these specifications, shall implement specific affirmative action steps, at least as extensive as those standards prescribed in paragraph 7 of these specifications, so as to achieve maximum results from its efforts to ensure equal employment opportunity. If the Contractor fails to comply with the requirements of the Executive Order, the implementing regulations, or these specifications, the Director shall proceed in accordance with 41 CFR 60-4.8.
- 14. The Contractor shall designate a responsible official to monitor all employment related activity to ensure that the company EEO policy is being carried out, to submit reports relating to the provisions hereof as may be required by the Government and to keep records. Records shall at least include for each employee the name, address, telephone numbers, construction trade, union affiliation if any, employee identification number when assigned, social security number, race, sex, status (e.g., mechanic, apprentice trainee, helper, or laborer), dates of changes in status, hours worked per week in the indicated trade, rate of pay, and locations at which the work was performed. Records shall be maintained in an easily understandable and retrievable form; however, to the degree that existing records satisfy this requirement, contractors shall not be required to maintain separate records.
- **15.** Nothing herein provided shall be construed as a limitation upon the application of other laws, which establish different standards of compliance or upon the application of requirements for the hiring of local or other area residents (e.g., those under the Public Works Employment Act of 1977 and the Community Development Block Grant Program).
- (a) The notice set forth in 41 CFR 60-4.2 and the specifications set forth in 41 CFR 60-4.3 replace the New Form for Federal Equal Employment Opportunity Bid Conditions for Federal and Federally Assisted Construction published at 41 FR 32482 and commonly known as the Model Federal EEO Bid Conditions, and the New Form shall not be used after the regulations in 41 CFR Part 60-4 become effective.

[43 FR 49254, OCT. 20, 1978; 43 FR 51401, NOV. 3, 1978, AS AMENDED AT 45 FR 65978, OCT. 3, 1980]

60-4.4 Affirmative Action Requirements.

- (a) To implement the affirmative action requirements of Executive Order 11246 in the construction industry, the Office of Federal Contract Compliance Programs previously has approved affirmative action programs commonly referred to as "Hometown Plans", has promulgated affirmative action plans referred to as "Imposed Plans" and has approved "Special Bid Conditions" for high impact projects constructed in areas not covered by a Hometown or an Imposed Plan. All solicitations for construction contracts made after the effective date of the regulations in this part shall include the notice specified in § 60-4.2 of this part and the specifications in § 60-4.3 of this part in lieu of the Hometown and Imposed Plans including the Philadelphia Plan and Special Bid Conditions. Until the Director has issued an order pursuant to § 60-4.6 of this part establishing goals and timetables for minorities in the appropriate geographical areas or for a project covered by Special Bid Conditions, the goals and timetables for minorities to be inserted in the Notice required by 41 CFR 60-4.2 shall be the goals and timetables contained in the Hometown Plan, Imposed Plan or Special Bid Conditions presently covering the respective geographical area or project involved.
- **(b)** Signatories to a Hometown Plan (including heavy highway affirmative action plans) shall have 45 days from the effective date of the regulations in this part to submit under such a Plan (for the director's approval) goals and timetables for women and to include female representation on the Hometown Plan Administrative Committee. Such goals for female representation shall be at least as high as the goals established for female representation in the notice issued pursuant to 41 CFR 60-4.6. Failure of the signatories, within the 45-day period, to include female representation and to submit goals for women or a new plan, as appropriate, shall result in an automatic termination of the Office of Federal Contract Compliance Program's approval of the Hometown Plan. At any time the Office of Federal Contract Compliance Programs terminates or withdraws its approval of a Hometown Plan, or when the plan expires and another plan is not approved, the contractors signatory to the plan shall be covered automatically by the specifications set forth in § 60-4.3 of this part and by the goals and timetables established for that geographical area pursuant to § 60-4.6 of this part.

60-4.5 Hometown Plans

(a) A contractor participating, either individually or through an association, in an approved Hometown Plan (including heavy highway affirmative action plans) shall comply with its affirmative action obligations under Executive Order 11246 by complying with its obligations under the plan: *Provided*, That each contractor or subcontractor participating in an approved plan is individually required to comply with the equal opportunity clause set forth in 41 CFR 60-1.4; to make a good faith effort to achieve the goals for each trade participating in the plan in which it has employees; and that the overall good performance by other contractors or subcontractors toward a goal in an approved plan does not excuse any covered contractor's or subcontractor's failure to take good faith efforts to achieve the plan's goals and timetables. If a contractor is not participating in an approved Hometown Plan it shall comply with the specifications set forth in § 60-4.3 of this part and with the goals and timetables for the appropriate area as listed in the notice required by 41

CFR 60-4.2 with regard to that trade. For the purposes of this part 60-4, a contractor is not participating in a Hometown Plan for a particular trade if it:

- (1) Ceases to be signatory to a Hometown Plan covering that trade;
- (2) Is signatory to a Hometown Plan for that trade but is not party to a collective bargaining agreement for that trade;
- (3) Is signatory to a Hometown Plan for that trade but is party to a collective bargaining agreement with labor organizations, which are not or cease to be signatories to the same Hometown Plan for that trade:
- (4) Is signatory to a Hometown Plan for that trade but is party to a collective bargaining agreement with a labor organization for that trade but the two have not jointly executed a specific commitment to minority and female goals and timetables and incorporated the commitment in the Hometown Plan for that trade:
- (5) Is participating in a Hometown Plan for that trade which is no longer acceptable to the Office of Federal Contract Compliance Programs;
- **(6)** Is signatory to a Hometown Plan for that trade but is party to a collective bargaining agreement with a labor organization for that trade and the labor organization and the contractor have failed to make a good faith effort to comply with their obligations under the Hometown Plan for that trade.
- **(b)** Contractors participating in Hometown Plans must be able to demonstrate their participation and document their compliance with the provision of the Hometown Plan.

[43 FR 49254, OCT. 20, 1978; 43 FR 51401, NOV. 3, 1978]

60-4.6 Goals and Timetables.

The Director, from time to time, shall issue goals and timetables for minority and female utilization, which shall be based on appropriate workforce, demographic, or other relevant data and which shall cover construction projects or construction contracts performed in specific geographical areas. The goals, which shall be applicable to each construction trade in a covered contractor's or subcontractor's entire workforce which is working in the area covered by the goals and timetables, shall be published as notices in the Federal Register, and shall be inserted by the contracting officers and applicants, as applicable, in the Notice required by 41 CFR 60-4.2. Covered construction contractors performing construction work in geographical areas where they do not have a Federal or federally assisted construction contract shall apply the minority and female goals established for the geographical area where the work is being performed.

[45 FR 65978, OCT. 3, 1980]

60-4.7 Effect on Other Regulations.

The regulations in this part are in addition to the regulations contained in this chapter, which apply to construction contractors and subcontractors generally. See particularly, 41 CFR 60-1.4 (a), (b), (c), (d), and (e); 60-1.5; 60-1.7; 60-1.8; 60-1.26; 60-1.29; 60-1.30; 60-1.32; 60-1.41; 60-1.42; 60-1.43; and 41 CFR Part 60-3; Part 60-20; Part 60-30; Part 60-40; and Part 60-50.

60-4.8 Show Cause Notice.

If an investigation or compliance review reveals that a construction contractor or subcontractor has violated the Executive order, any contract clause, specifications or the regulations in this chapter and if administrative enforcement is contemplated, the Director shall issue to the contractor or subcontractor a notice to show cause which shall contain the items specified in paragraphs (i) through (iv) of 41 CFR 60-2.2(c)(1). If the contractor does not show good cause within 30 days, or in the alternative, fails to enter an acceptable conciliation agreement which includes where appropriate, make up goals and timetables, back pay, and seniority relief for affected class members, the OFCCP shall follow the procedure in 41 CFR 60-1.26(b): *Provided*, That where a conciliation agreement has been violated, no show cause notice is required prior to the initiation of enforcement proceedings.

[43 FR 49254, OCT. 20, 1978; 43 FR 51401, NOV. 3, 1978]

60-4.9 Incorporation by Operation of the Order.

By operation of the order, the equal opportunity clause contained in § 60-1.4, the Notice of Requirement for Affirmative Action to Ensure Equal Employment Opportunity (Executive Order 11246) contained in § 60-4.2, and the Standard Federal Equal Employment Opportunity Construction Contract Specifications (Executive Order 11246) contained in § 60-4.3 shall be deemed to be a part of every solicitation or of every contract and subcontract, as appropriate, required by the order and the regulations in this chapter to include such clauses whether or not they are physically incorporated in such solicitation or contract and whether or not the contract is written.

Construction Contracts of Loan Recipient and Other Sections From

"Procedures For Issuing Loans From the Public Water Supply Loan Program"

Section 662.420(b)(2) Negotiations of Contract Amendments (Change Orders)

- 1) When the loan recipient authorizes the contractor to add, delete, or revise the work within the general scope of the contract documents, or authorizes an adjustment in the contract price or contract time, the loan recipient shall submit a change order to the Agency.
- 2) For each change order, the loan recipient shall submit to the Agency for approval the following documentation:
 - i) one copy of the fully executed change order signed by the loan recipient, construction engineer, and the contractor; and
 - ii) a description of any changes, with justification for the changes.
- 3) Prior approval by the Agency of a change order is required when a change order results in:
 - i) alterations in design scope that require a modification to a construction permit; or
 - ii) an increase in the amount of loan funds needed to complete the project.
- 4) Failure to give timely notice of proposed project changes or action by the loan recipient that is not consistent with the Agency's determination on those changes may result in disallowance of loan participation for costs incurred that are attributable to the change.

Section 662.620(f) Required Construction Contract Provisions

Each construction contract shall include the following provisions:

- 1) Audit; access to records:
 - A) The contractor shall maintain books, records, documents and other evidence directly pertinent to performance on loan work consistent with generally accepted accounting principles. The contractor shall also maintain the financial information and data used by the contractor in the preparation or support of any cost submissions required under subsection (b) above, (Negotiation of Contract Amendments, Change Orders) and a copy of the cost summary submitted to the Owner. The Illinois Auditor General, the Owner, the Agency, or any of their duly authorized representatives shall have access to the books, records, documents, and other evidence for purposes of inspection, audit, examination, excerpts, transcriptions, and copying. The contractor shall provide facilities for access and inspection.
 - B) For a formally advertised, competitively awarded, fixed price contract, the contractor shall include access to records as required by subsection (a)(1)(A) for all negotiated change orders and contract amendments in excess of \$25,000 that affect the contract price. In the case of all other prime contracts, the contractor shall agree to include access to records required by subsection (a)(1)(A) in all

contracts and all tier subcontracts or change orders in excess of \$25,000 that are directly related to project performance.

- C) Audits conducted pursuant to this provision shall be consistent with generally accepted auditing standards.
- D) The contractor agrees to the disclosure of all information and reports resulting from access to records pursuant to subsection (a)(1)(A) above. When the audit concerns the contractor, the auditing agency shall afford the contractor an opportunity for an audit exit conference and an opportunity to comment on the pertinent portions of the draft audit report. The final audit report shall include the written comments, if any, of the audited parties.
- E) The records required by subsection (a)(1)(A) above shall be maintained and made available during performance of the work under this loan agreement and for three years after the date of final loan audit. In addition, records that relate to any dispute or litigation or the settlement of claims arising out of any performance, costs or items to which an audit exception has been taken, shall be maintained and made available for three years after the date of resolution of such dispute, appeal, litigation, claim or exception.
- F) The right of access will generally be exercised with respect to financial records under:
 - i) Negotiated prime contractors;
 - ii) Negotiated change orders or contract amendments in excess of \$25,000 affecting the price of any formally advertised, competitively awarded, fixed price contract; and
 - iii) Subcontracts or purchase orders under any contract other than a formally advertised, competitively awarded, fixed price contract.
- G) The right of access will generally not be exercised with respect to a prime contract, subcontract, or purchase order awarded after effective price competition. In any event, the right of access shall be exercised under any type of contract or subcontract:
 - i) With respect to records pertaining directly to contract performance, excluding any financial records of the contractor; and
 - ii) If there is any indication that fraud, gross abuse, or corrupt practices may be involved in the award or performance of the contract or subcontract.
- Covenant against contingent fees

The contractor shall warrant that no person or selling agency has been employed or retained to solicit or secure the contract upon an agreement or understanding for a

commission, percentage, brokerage, or contingent fee. For breach or violation of this warranty, the Owner shall have the right to annul the contract without liability or in its discretion to deduct from the contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

3) Wage provisions

The contractor shall pay prevailing wages in accordance with the Davis-Bacon Act (40 USC 276a through 276a-5).

4) DBE Requirements

Evidence that the contractor has taken affirmative steps, such as, but not limited to, a copy of the advertisement(s) and the record of negotiation in accordance with 40 CFR Part 33, to assure that disadvantaged businesses are used when possible as sources of supplies, equipment, construction and services.

5) Debarment or Suspension Provisions

The contract shall require the successful bidder(s) to submit a certification of compliance with federal Executive Order 12549 regarding debarment, suspension, and other responsibility matters.

Section 662.620(g) Subcontracts under Construction Contracts

The award or execution of all subcontracts by a prime contractor and the procurement and negotiation procedures used by the prime contractor shall comply with:

- 1) All applicable provisions of federal, state and local law;
- 2) All provisions of this Part regarding fraud and other unlawful or corrupt practices;
- 3) All provisions of this Part with respect to access to facilities, records and audit of records; and
- 4) all provisions of subsection (f)(5) that require a Certification Regarding Debarment, Suspension, and Other Responsibility Matters (EPA Form 5700-49) showing compliance with any controlling federal Executive Orders.

Section 662.620 (h) Contractor Bankruptcy

In the event of a contractor bankruptcy, the loan recipient shall notify the Agency and shall keep the Agency advised of any negotiations with the bonding company, including any proposed settlement. The Agency may participate in those negotiations and will advise the loan recipient of the impact of any proposed settlement to the loan agreement. The loan recipient shall be responsible for assuring that every appropriate procedure and incidental legal requirement is observed in advertising for bids and re-awarding a construction contract.

Section 662.640 (c) Remedies

All claims, counter-claims, disputes, and other matters in question between the recipient and the contractor arising out of, or relating to, a sub-agreement or its breach shall be decided by arbitration if the parties agree, or in a court of competent jurisdiction within the State.

Section 662.620 (i) Access

Every contract entered into by the loan recipient for construction work, and every subagreement, shall provide Agency representatives with access to the work. The contractor or subcontractor shall provide facilities for access and inspection.

I. <u>Disadvantaged Business Enterprise Policy</u>

- A. It is the policy of the State of Illinois to award a fair share of sub-agreements to disadvantaged businesses. In complying with this requirement, Bidders or Offerors are required to take affirmative steps to assure that disadvantaged businesses are used when possible as sources of supplies, equipment, construction, and services as explained herein.
- B. These specifications define the terms, conditions, and requirements of the State Revolving Fund Loan Program, and the City of Joliet policy and procedures for complying with these requirements.
- C. As required by the award conditions of USEPA's Assistance Agreement with IEPA, the fair share percentages are 5% for MBEs and 12% for WBEs.

II. <u>Pre-Contract Award Obligations</u>

- A. Bidders are required to advertise subcontracting opportunities and to negotiate with disadvantaged businesses prior to the day of the bid opening. Failure to document such affirmative efforts shall be deemed, relative to disadvantaged business compliance non-responsive.
- B. To establish a bid as responsible, the Bidder will be required to document the proposed utilization of disadvantaged businesses with letters of intent signed by the bidder and by the disadvantaged business listed in the bid. The documentation requirements are outlined in Section III.
- City of Joliet disadvantaged business policy clearly intends for Bidders to contact and encourage the participation of disadvantaged businesses prior to bid opening. Affirmative efforts (the written record of conscientious and honest communications between the bidder and disadvantaged business) must be initiated and completed by the bidder prior to bid opening. All bidders must document compliance with the requirements of the disadvantaged business policy.

III. Evaluation of Disadvantaged Business Utilization and Affirmative Efforts

- A. As a prerequisite to demonstrate compliance with the City of Joliet disadvantaged business policy, ALL bidders shall provide the following with its bid, except Item 2 CERTIFICATE 00 74 23.03 DBE FORM #4 (BIDDER'S LIST) that may be submitted later:
 - 1. <u>Completed and signed certification</u> from the bidder(s), attesting that the bidder will award no sub-agreements, including the procurement of equipment, materials, supplies and services, in the performance of this contract.

OR

- 1. "Certification of publication", or adequate proof of publication, including an actual copy of the newspaper advertisement from a daily, regional newspaper. Alternatively, Bidders may publish the advertisement in an established, online Bidder's clearinghouse such as the "Dodge Report (http://construction.com/dodge/)", "iSqFt (www.isqft.com/start/)", or equivalent. If an online advertisement is placed with the "Dodge Report" or an equivalent website, a screenshot of the advertisement along with the webpage address, and a payment receipt is required as documentation. The advertisement must run one day at least (16) days prior to bid opening. See the "Suggested Disadvantaged Business (DBE) Advertisement for Construction Contractors", which can be found on page 322 of this Section.
- List of all disadvantaged business enterprise (DBE) and non-DBEs that submitted proposals to the Bidder along with the date of the proposal. Names, addresses, phone number and/or e-mail are required. CERTIFICATE 00 74 23.03 DBE FORM #4 (BIDDER'S LIST) may be used for this purpose, but any list that contains the required information may be submitted. This list does not have to be submitted with the Bid. This list shall be submitted within five (5) business days after the receipt of Bids.
- 3. List of disadvantaged businesses not being utilized and justification for non-utilization (see CERTIFICATE 00 74 23.01 DBE FORM #1 (CONTRACTOR CERTIFICATION)).
- 4. If DBE subcontractors will be utilized for the project, a completed and signed copy of CERTIFICATE 00 74 23.02 DBE FORM #3 (SUBCONTRACTOR UTILIZATION) or equivalent report is needed from each DBE subcontractor.
- 5. If DBE subcontractors will be utilized for the project, a completed and signed certification from the Bidder(s), attesting that the Bidder has no controlling or dominating interest or conflict of interest with the disadvantaged business that will be utilized (Use CERTIFICATE 00 74 23.01 DBE FORM #1 (CONTRACTOR CERTIFICATION)).
- 6. In instances where the Bidder(s) does not receive any proposals from disadvantaged businesses prior to bid opening, the Bidder(s) must provide a written certification attesting that no proposals were received (Use CERTIFICATE 00 74 23.01 DBE FORM #1 (CONTRACTOR CERTIFICATION)).

<u>Failure to submit the documentation pursuant to the requirements of A (1. or 1.-6.) above may cause rejection of the bid as non-responsive.</u>

- B. The low, responsive bidder will be deemed responsible with respect to the disadvantaged business requirements if:
 - 1. The low, responsive bidder submits CERTIFICATE 00 74 23.01 DBE FORM #1 (CONTRACTOR CERTIFICATION) and CERTIFICATE 00 74

23.02 DBE FORM #3 (SUBCONTRACTOR UTILIZATION) with its Bid, and submits CERTIFICATE 00 74 23.03 DBE FORM #4 (BIDDER'S LIST) within five (5) business days after the receipt of Bids.

<u>Failure to submit the documentation pursuant to the requirements of B(1)</u> above may cause rejection of the bid as non-responsible.

Where the bidder is considered non-responsible under this subsection, the Owner will promptly advise the bidder/offerer, in writing, of the basis for the non-responsibility determination.

IV. Sanctions

- A. The City of Joliet may reject one or all bids where the information submitted by the bidder fails to objectively demonstrate compliance with the disadvantaged business requirements (i.e., failure to place the pre-bid advertisement by the bidder(s) at least sixteen (16) days prior to the day of the bid opening shall not be considered as objectively demonstrating compliance with the disadvantaged business requirements).
- B. Upon finding that any Party has not complied with the requirements of these specifications, <u>including misrepresenting a firm as a disadvantaged business</u>, any one or a combination of the following actions may be taken.
 - 1. Declare the bidder and/or subcontractor non-responsible and therefore ineligible for contract award.
 - 2. Disallow all contract costs associated with non-compliance.
 - 3. Refer any matter, which may be fraudulent to the Illinois Attorney General.
 - 4. Refer any matter, which may lead to criminal prosecution of a claim for funds, to the Illinois Attorney General.

V. Post-Contract Award Compliance

- A. As required by the award conditions of USEPA's Assistance Agreement with IEPA, all sub-agreements of the prime contractor must identify that the fair share percentages are 5% for MBEs and 12% for WBEs.
- B. After award of the prime contract copies of all disadvantaged business related subagreements between the prime contractor and subcontractors shall be submitted to the Owner.
- C. Subsequent to Bid Submission, any changes in previously reported disadvantaged businesses utilization shall be handled in accordance with Part 33.302(b-h). If the contractor fails to initiate such actions, the Owner may withhold payments and/or institute other appropriate sanctions.

SUGGESTED DISADVANTAGED BUSINESS ADVERTISEMENT FOR CONSTRUCTION CONTRACTORS*

NOTICE TO DISADVANTAGED BUSINESSES

		is
(Name of Company)	(Address of Company)	(Telephone)
seeking disadvantaged busine	esses for	
the		
	(Name of Loan Recipient)	
Project for subcontracting opp	portunities in the following areas:	
All disadvantaged businesses	should contact, <u>in writing</u> ,	
(certified letter, return receipt	requested),(Company Con	, to discuss the ntact Person)
subcontracting opportunities.	All negotiations must be complete	ed prior to the bid opening
date		
	(Date of Bid Opening)	

 The advertisement must clearly state the method of evaluating the proposals or quotations, and the relative importance attached to each criterion. Bidders must uniformly and objectively evaluate the proposals submitted by disadvantaged businesses in response to the advertisement based upon the evaluation criteria stated in the advertisement. The evaluation criteria must not be restrictive or exclusionary.

I. Wage Rate Requirements Under Section 1606 of the American Recovery and Reinvestment Act For Subrecipients That Are Governmental Entities

The following terms and conditions specify how recipients will assist USEPA in meeting its Davis-Bacon (DB) responsibilities when DB applies to USEPA awards of financial assistance under the Recovery Act with respect to IEPA and subrecipients that are governmental entities. If a subrecipient has questions regarding when DB applies, obtaining the correct DB wage determinations, DB provisions, or compliance monitoring, it may contact IEPA for guidance. The subrecipient may also obtain additional guidance from DOL's web site at http://www.dol.gov/esa/whd/recovery/.

1. Applicability of the Davis-Bacon (DB) prevailing wage requirements.

Davis-Bacon prevailing wage requirements apply to the construction, alteration, and repair activity of infrastructure, including all construction, alteration and repair activity involving waste water or drinking water treatment plants is subject to DB. If a subrecipient encounters a unique situation at a site that presents uncertainties regarding DB applicability, the subrecipient must discuss the situation with IEPA before authorizing work on that site.

2. Obtaining Wage Determinations.

- (a) Subrecipients shall obtain the wage determination for the locality in which a covered activity subject to DB will take place prior to issuing requests for bids, proposals, quotes or other methods for soliciting contracts (solicitation) for activities subject to DB. These wage determinations shall be incorporated into solicitations and any subsequent contracts. Prime contracts must contain a provision requiring that subcontractors follow the wage determination incorporated into the prime contract.
- (i) While the solicitation remains open, the subrecipient shall monitor https://beta.sam.gov on a weekly basis to ensure that the wage determination contained in the solicitation remains current. The subrecipient shall amend the solicitation of DOL issues a modification more than 10 days prior to the closing date (i.e. bid opening) for the solicitation. If DOL modifies or supersedes the applicable wage determination less than 10 days prior to the closing date, the subrecipient may request a finding from IEPA that there is not a reasonable time to notify interested contractors of the modification of the wage determination. IEPA will provide a report of its findings to the subrecipient.
- (ii) If the subrecipient does not award the contract within 90 days of the closure of the solicitation, any modifications or supersedes DOL makes to the wage determination contained in the solicitation shall be effective unless IEPA, at the request of the subrecipient, obtains an extension of the 90 day period from DOL pursuant to 29 CFR 1.6(c)(3)(iv). The subrecipient shall monitor https://beta.sam.gov on a weekly basis if it does not award the contract within 90 days of closure of the solicitation to ensure that wage determinations contained in the solicitation remain current.
- (b) If the subrecipient carries out activity subject to DB by issuing a task order, work assignment or similar instrument to an existing contractor (ordering instrument) rather than by publishing a

solicitation, the subrecipient shall insert the appropriate DOL wage determination from https://beta.sam.gov into the ordering instrument.

- (c) Subrecipient shall review all subcontracts subject to DB entered into by prime contractors to verify that the prime contractor has required its subcontractors to include the applicable wage determinations.
- (d) As provided in 29 CFR 1.6(f), DOL may issue a revised wage determination applicable to a subrecipient's contract after the award of a contract or the issuance of an ordering instrument if DOL determines that the subrecipient has failed to incorporate a wage determination or has used a wage determination that clearly does not apply to the contract or ordering instrument. If this occurs, the subrecipient shall either terminate the contract or ordering instrument and issue a revised solicitation or ordering instrument or incorporate DOL's wage determination retroactive to the beginning of the contract or ordering instrument by change order. The subrecipient's contractor must be compensated for any increases in wages resulting from the use of DOL's revised wage determination.

3. Contract and Subcontract provisions.

(a) The Recipient shall insure that the **subrecipient** (s) shall insert in full in any contract in excess of \$2,000 which is entered into for the actual construction, alteration and/or repair, including painting and decorating, of a public building or public work, or building or work financed in whole or in part from Federal funds or in accordance with guarantees of a Federal agency or financed from funds obtained by pledge of any contract of a Federal agency to make a loan, grant or annual contribution (except where a different meaning is expressly indicated), and which is subject to the labor standards provisions of any of the acts listed in § 5.1, the following clauses:

(1) Minimum wages.

(i) All laborers and mechanics employed or working upon the site of the work will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in § 5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein:

Provided, That the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

Subrecipient(s) may obtain wage determinations from the U.S. Department of Labor's web site, https://beta.sam.gov.

- (ii)(A) The **subrecipient**(s), on behalf of EPA, shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The EPA award official shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:
- (1) The work to be performed by the classification requested is not performed by a classification in the wage determination; and
- (2) The classification is utilized in the area by the construction industry; and
- (3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the **subrecipient**(s) agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the **subrecipient**(s) to the State award official. The State award official will transmit the report, to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the State award official or will notify the State award official within the 30-day period that additional time is necessary.
- (C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the and the **subrecipient**(s) do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the award official shall refer the questions, including the views of all interested parties and the recommendation of the State award official, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (D) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs (a)(1)(ii)(B) or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.

- (iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.
- (2) Withholding. The **subrecipient** (s), shall upon written request of the EPA Award Official or an authorized representative of the Department of Labor, withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work, all or part of the wages required by the contract, the (Agency) may, after written notice to the contractor, sponsor, applicant, or Owner, take such action as may be necessary to cause the suspension of any further payment, advance, or quarantee of funds until such violations have ceased.
- (3) Payrolls and basic records.
- (i) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work. Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

- (ii)(A) The contractor shall submit weekly, for each week in which any contract work is performed, a copy of all payrolls to the **subrecipient**, that is, the entity that receives the sub-grant or loan from the State capitalization grant recipient. Such documentation shall be available on request of the State recipient or EPA. As to each payroll copy received, the subrecipient shall provide written confirmation in a form satisfactory to the State indicating whether or not the project is in compliance with the requirements of 29 CFR 5.5(a)(1) based on the most recent payroll copies for the specified week. The payrolls shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on the weekly payrolls. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to the subrecipient (s) for transmission to the State or EPA if requested by EPA, the State, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the subrecipient (s).
- (B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
- (1) That the payroll for the payroll period contains the information required to be provided under § 5.5 (a)(3)(ii) of Regulations, 29 CFR part 5, the appropriate information is being maintained under § 5.5 (a)(3)(i) of Regulations, 29 CFR part 5, and that such information is correct and complete;
- (2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3:
- (3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
- (C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.
- (D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.

(iii) The contractor or subcontractor shall make the records required under paragraph (a)(3)(i) of this section available for inspection, copying, or transcription by authorized representatives of the State, EPA or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency or State may, after written notice to the contractor, sponsor, applicant, or Owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

(4) Apprentices and trainees--

- (i) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (ii) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal

certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

- (iii) Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.
- (5) Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.
- (6) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses contained in 29 CFR 5.5(a)(1) through (10) and such other clauses as the EPA determines may by appropriate, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.
- (7) Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.
- (8) Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.
- (9) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and **Subrecipient** (s), State, EPA, the U.S. Department of Labor, or the employees or their representatives.

- (10) Certification of eligibility.
- (i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

4. Contract Provision for Contracts in Excess of \$100,000.

- (a) Contract Work Hours and Safety Standards Act. The **subrecipient** shall insert the following clauses set forth in paragraphs (a)(1), (2), (3), and (4) of this section in full in any contract in an amount in excess of \$100,000 and subject to the overtime provisions of the Contract Work Hours and Safety Standards Act. These clauses shall be inserted in addition to the clauses required by Item 3, above or 29 CFF 4.6. As used in this paragraph, the terms laborers and mechanics include watchmen and guards.
- (1) Overtime requirements. No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.
- (2) Violation; liability for unpaid wages; liquidated damages. In the event of any violation of the clause set forth in paragraph (a)(1) of this section the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph (a)(1) of this section, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph (a)(1) of this section.
- (3) Withholding for unpaid wages and liquidated damages. The **subrecipient**, upon written request of the EPA Award Official or an authorized representative of the Department of Labor, shall withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for

unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.

- (4) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraph (a)(1) through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs (a)(1) through (4) of this section.
- (b) In addition to the clauses contained in Item 3, above, in any contract subject only to the Contract Work Hours and Safety Standards Act and not to any of the other statutes cited in 29 CFR 5.1, the **Subrecipient** shall insert a clause requiring that the contractor or subcontractor shall maintain payrolls and basic payroll records during the course of the work and shall preserve them for a period of three years from the completion of the contract for all laborers and mechanics, including guards and watchmen, working on the contract. Such records shall contain the name and address of each such employee, social security number, correct classifications, hourly rates of wages paid, daily and weekly number of hours worked, deductions made, and actual wages paid. Further, the **Subrecipient** shall insert in any such contract a clause providing hat the records to be maintained under this paragraph shall be made available by the contractor or subcontractor for inspection, copying, or transcription by authorized representatives of the (write the name of agency) and the Department of Labor, and the contractor or subcontractor will permit such representatives to interview employees during working hours on the job.

5. Compliance Verification

- (a) The subrecipient shall periodically interview a sufficient number of employees entitled to DB prevailing wages (covered employees) to verify that contractors or subcontractors are paying the appropriate wage rates. As provided in 29 CFR 5.6(a)(6), all interviews must be conducted in confidence. The subrecipient must use Standard Form 1445 or equivalent documentation to memorialize the interviews. Copies of the SF 1445 are available from USEPA on request.
- (b) The subrecipient shall establish and follow an interview schedule based on its assessment of the risks of noncompliance with DB posed by contractors or subcontractors and the duration of the contract or subcontract. At a minimum, the subrecipient must conduct interviews with a representative group of covered employees within two weeks of each contractor or subcontractor's submission of its initial weekly payroll data and two weeks prior to the estimated completion date for the contract or subcontract. Subrecipients must conduct more frequent interviews if the initial interviews or other information indicates that there is a risk that the contractor or subcontractor is not complying with DB. Subrecipients shall immediately conduct necessary interviews in response to an alleged violation of the prevailing wage requirements. All interviews shall be conducted in confidence.
- (c) The subrecipient shall periodically conduct spot checks of a representative sample of weekly payroll data to verify that contractors and subcontractors are paying the appropriate wage rates. The subrecipient shall establish and follow a spot check schedule based on its assessment of the risks of noncompliance with DB posed by contractors or subcontractors and the duration of the contract or subcontract. At a minimum, the subrecipient must spot check

payroll data within two weeks of each contractor or subcontractor's submission of its initial payroll data and two weeks prior to the completion date the contract or subcontract. Subrecipients must conduct more frequent spot checks if the initial spot check or other information indicates that there is a risk that the contractor or subcontractor is not complying with DB. In addition, during the examinations the subrecipient shall verify evidence of fringe benefit plans and payments thereunder by contractors and subcontractors who claim credit for fringe benefit contributions.

- (d) The subrecipient shall periodically review contractors and subcontractors use of apprentices and trainees to verify registration and certification with respect to apprenticeship and training programs approved by either the U.S. Department of Labor or a state, as appropriate, and that contractors and subcontractors are not using disproportionate numbers of, laborers, trainees and apprentices. These reviews shall be conducted in accordance with the schedules for spot checks and interviews described in Items (b) and (c) above.
- (e) Subrecipients must immediately report potential violations of the DB prevailing wage requirements to the USEPA DB contact listed above and to the appropriated DOL Wage and Hour District Office listed at http://www.dol.gov/esa/contacts/whd/america2.htm.

Use of American Iron and Steel

Sec. 436. (a)(1) None of the funds made available by a State water pollution control revolving fund as authorized by title VI of the Federal Water Pollution Control Act (33 U.S.C. 1381 et seq.) or made available by a drinking water treatment revolving loan fund as authorized by section 1452 of the Safe Drinking Water Act (42 U.S.C. 300j-12) shall be used for a project for the construction, alteration, maintenance, or repair of a public water system or treatment works unless all of the iron and steel products used in the project are produced in the United States.

- (2) In this section, the term `iron and steel products' means the following products made primarily of iron or steel: lined or unlined pipes and fittings, manhole covers and other municipal castings, hydrants, tanks, flanges, pipe clamps and restraints, valves, structural steel, reinforced precast concrete, and construction materials.
- (b) Subsection (a) shall not apply in any case or category of cases in which the Administrator of the Environmental Protection Agency (in this section referred to as the `Administrator') finds that-
 - (1) applying subsection (a) would be inconsistent with the public interest;
 - (2) iron and steel products are not produced in the United States in sufficient and reasonably available quantities and of a satisfactory quality; or
 - (3) inclusion of iron and steel products produced in the United States will increase the cost of the overall project by more than 25 percent.
- (c) If the Administrator receives a request for a waiver under this section, the Administrator shall make available to the public on an informal basis a copy of the request and information available to the Administrator concerning the request, and shall allow for informal public input on the request for at least 15 days prior to making a finding based on the request. The Administrator shall make the request and accompanying information available by electronic means, including on the official public Internet Web site of the Environmental Protection Agency.
- (d) This section shall be applied in a manner consistent with United States obligations under international agreements.
- (e) The Administrator may retain up to 0.25 percent of the funds appropriated in this Act for the Clean and Drinking Water State Revolving Funds for carrying out the provisions described in subsection (a)(1) for management and oversight of the requirements of this section.
- (f) This section does not apply with respect to a project if a State agency approves the engineering plans and specifications for the project, in that agency's capacity to approve such plans and specifications prior to a project requesting bids, prior to the date of the enactment of this Act

ALL BIDDERS:

Provide signed "Bidder Certification Regarding the Use of American Iron and Steel Products" (Certificate 00 62 33.13) with the bid.

Illinois Works Jobs Program Act – Apprenticeship Initiative Information for Contractors

Summary: The Illinois Works Jobs Program Act, 30 ILCS 559/Art. 20, is a statewide initiative to ensure that all Illinois residents have access to State capital projects, careers in the construction industry, and to provide contracting and employment opportunities to historically underrepresented populations in the construction industry. This will be accomplished through three programs created by the Illinois Works Jobs Program Act: the Illinois Works Apprenticeship Initiative, the Illinois Works Pre-Apprenticeship Program, and the Illinois Works Bid Credit information available Program. Additional is at the following website: https://www2.illinois.gov/dceo/WorkforceDevelopment/Pages/IllinoisWorksJobsProgramAct.asp Χ.

The goal of the Illinois Works Apprenticeship Initiative ("10% apprenticeship goal") is that for projects estimated to cost \$500,000 or more, apprentices will perform either 10% of the total labor hours actually worked in each prevailing wage classification or 10% of the estimated labor hours in each prevailing wage classification, whichever is less. The 10% apprenticeship goal applies to projects being paid for in whole or in part by appropriated capital funds to construct a public work either through a contract or grant issued by a State agency. This requirement also applies to IEPA loans. The \$500,000 threshold applies to the total project cost and NOT the total cost or value of an individual construction contract.

Certification: Apprentices are required to be US Dept. of Labor certified (not limited to preapprentice program graduates).

Applicability

If the project is estimated to received \$500,000 or more of State Capital Funding:

If the state's contribution to the project amount equals 50% or more of the cost for the project, the 10% apprenticeship requirement applies to all prevailing wage eligible work performed by contractors on the public works project.

If the state's contribution to the project is less than 50% of the cost for the project, the 10% apprenticeship requirement applies only to prevailing wage eligible work being funded from state funds.

If the project has less than \$500,000 of State Capital Funding, but the total estimated cost is \$500,000 or more:

If the state's contribution to the project amount equals 50% or more of the estimated cost for the project, the 10% apprenticeship requirement applies to all prevailing wage eligible work performed by contractors on the public works project.

If the state's contribution to the project is less than 50% of the estimated cost for the project, the 10% apprenticeship requirement does not apply.

If the Total Estimated Project Cost is less than \$500,000: The 10% apprenticeship requirement does not apply.

Waivers from the Requirements: If goals cannot be met, the state has discretion to grant waivers, grant reductions, or to hold public hearings on the issue. Factors to be considered include the scale of the project and whether the contractor or subcontractor seeking the action has previously requested reductions or waivers. A waiver form is available on the IL Department of Commerce & Economic Development (DCEO), Illinois Works Jobs Program Act website at: https://www2.illinois.gov/dceo/WorkforceDevelopment/Documents/Request%20for%20Waiver%20or%20Reduction%20of%20Public%20Works%20Project%20Apprenticeship%20Goal%20%20Accessible%20and%20Enabled.pdf. The form can also be obtained from IEPA.

Reporting: An example quarterly reporting form is attached. A fillable version is available on the DCEO website. Contractors should coordinate with the loan applicant and their consultant for further instructions regarding these forms. Loan applicants are ultimately responsible to report quarterly labor hours to the state, but coordination with their contractor is essential.

For general program questions, please contact the Illinois Works Office at: CEO.ILWorks@Illinois.gov.

END OF BIDDER INSTRUCTIONS

00 31 24.51

CLEAN CONSTRUCTION OR DEMOLITION DEBRIS (CCDD) - LPC 662

PART 1 - GENERAL

1.1 SUMMARY

A. This Section provides verification from the Owner that to their knowledge the site of the work has never been used for commercial or industrial purposes and the soils within the work site is presumed to be uncontaminated.

1.2 CERTIFICATION

A. Upon receipt of soil samples from the Contractor, the Owner will complete pH testing and provide a signed LPC-662 form for Contractor's use.

END OF SECTION

00 31 32.11

SUBSURFACE DRILLING INFORMATION

PART 1 - GENERAL

1.1 SUMMARY

A. This Section describes soils investigation at the site, and use of data resulting from that investigation.

1.2 SOILS INVESTIGATION REPORT

A. General:

- 1. A soils investigation report has not been prepared for the site of this Work.
- 2. Bidders should visit the site and acquaint themselves with existing conditions.
- 3. Prior to bidding, Bidders may make their own subsurface investigations to satisfy themselves as to site and subsurface conditions, but such investigations may be performed only under time schedules and arrangements approved in advance by the Engineer.

END OF SUBSURFACE DRILLING INFORMATION

00 41 00.41

BID FORM

ONE ORIGINAL BID SHALL BE SUBMITTED

To: City of Joliet
Mayor and City Council
150 W. Jefferson Street
Joliet, IL 60432
(hereinafter called Owner)

rom:						
		Company				
		Address				
Cit	ty	State	Zip Code			
(_)	Telephone		F:	-MAIL	

(hereinafter called Bidder)

- 1. The undersigned Bidder proposes and agrees, if this Bid is accepted, to enter into an Agreement with Owner in the form included in the Bidding Documents to furnish all labor, materials, tools, and services required for the construction of the Ridgewood Phase 3 Water Main Improvements for the City of Joliet, Will County, Illinois (Engineers' Job No. 210039.40), all in accordance with the Bidding Documents prepared by Baxter & Woodman, Inc., Consulting Engineers.
- 2. Bidder accepts all of the terms and conditions of the Advertisement for Bids and Bidder Instructions, including without limitation those dealing with the disposition of Bid Security. This Bid will remain open for 120 days after the date of Bid opening or for such longer period of time that Bidder may agree to in writing upon request of Owner. Bidder will sign and submit the Agreement with the Bonds and other documents required by the Bidding Documents within 10 days after the date of Owner's Notice of Award.
- 3. In submitting this Bid, Bidder represents, as set forth in the Agreement, that:
 - a. Bidder has examined copies of all the Bidding Documents.
 - b. Bidder is familiar with the nature and extent of the Bidding Documents, Work, site, locality, and all local conditions and legal and regulatory requirements that in any

- manner may affect cost, progress, performance, or furnishing of the Work, and has made such independent investigations as Bidder deems necessary.
- c. Bidder has obtained and carefully studied (or assumes responsibility for having done so) all additional or supplementary examinations, investigations, explorations, tests, studies and data concerning conditions (surface, subsurface and Underground Facilities) at or contiguous to the Site which may affect cost, progress, or performance of the Work or which relate to any aspect of the means, methods, techniques, sequences, and procedures of construction to be employed by Bidder, including applying the specific means, methods, techniques, sequences, and procedures of construction expressly required by the Bidding Documents to be employed by Bidder, and safety precautions and programs incident thereto.
- d. Bidder does not consider that any further examinations, investigations, explorations, tests, studies, or data are necessary for the determination of this Bid for performance of the Work at the price(s) bid and within the times and in accordance with the other terms and conditions of the Bidding Documents.
- e. Bidder is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Bidding Documents.
- f. Bidder has correlated the information known to Bidder, information and observations obtained from visits to the Site, reports and drawings identified in the Bidding Documents, and all additional examinations, investigations, explorations, tests, studies, and data with the Bidding Documents.
- g. Bidder has given Engineer written notice of all conflicts, errors, ambiguities, or discrepancies that Bidder has discovered in the Bidding Documents, and the written resolution thereof by Engineer is acceptable to Bidder.
- h. The Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for the Work for which this Bid is submitted.
- i. This Bid is genuine and not made in the interest or on behalf of any undisclosed person, firm or corporation, and is not submitted in conformity with any agreement or rules of any group, association, organization, or corporation; Bidder has not directly or indirectly induced or solicited any other Bidder to submit a false or sham Bid; Bidder has not solicited or induced any person, firm, or a corporation to refrain from bidding; and Bidder has not sought by collusion to obtain for itself any advantage over any other Bidder or over Owner.
- j. Bidder is not barred from contracting with the Owner as a result of a violation 720 ILCS 5/33 et seq.
- k. Bidder agrees that no less than the prevailing rate of wages under the Davis-Bacon Wage Act (40 USC 276a through 276a-5) as determined by the U.S. Department of

Labor shall be paid to all laborers, workmen and mechanics performing work under this contract.

- I. Bidder agrees to cooperate with all Davis-Bacon Wage Act compliance activities including employee interviews by Owner.
- m. Bidder complies with the provisions of the Employment of Illinois Workers on Public Works Act (30 ILCS 570/) as they may apply to this Project.
- n. Bidder will comply with the requirements of Sections 22.51(f)(2)(B) and 22.51a(d)(2)(B) of the Illinois Environmental Protection Act ([415 ILCS5/22.51(f)(2)(B)] and [415 ILCS5/22.51a(d)(2)(B)]) for the disposal of uncontaminated soils including uncontaminated soil mixed with other clean construction or demolition debris (CCDD) materials and has included any costs associated with compliance in the Bid.
- 4. Bidder submits the following Schedule of Unit Prices for the Work to be performed in accordance with the Bidding Documents and agrees that items of work not specifically mentioned in the Schedule which are necessary and required to complete the Work intended shall be done incidental to and as part of the items of work for which a unit price is given, and understands that no additional payment will be made for such incidental work.

Bidder agrees that the Owner may select any one or more Parts of this Bid, and if awarded the Contract for such Part or Parts of this Bid, the Bidder agrees to perform and complete the work at the Contract Unit Prices submitted herein.

	SCHEDULE OF UNIT PRICES				
	Approximate Unit				
No.	Pay Item	Qua	antity	Price	Amount
1	PRECONSTRUCTION VIDEO RECORDING	1	LSUM	Lump Sum	\$
2	TRAFFIC CONTROL AND PROTECTION	1	LSUM	Lump Sum	\$
3	GRANULAR TRENCH BACKFILL	2,400	CUYD	\$	\$
4	CONTROLLED LOW STRENGTH FLOWABLE FILL MIXTURE (CLSM)	100	CUYD	\$	\$
5	DUCTILE IRON WATER MAIN (OPEN CUT), 6-INCH	131	FOOT	\$	\$
6	DUCTILE IRON WATER MAIN (OPEN CUT), 8-INCH	4,145	FOOT	\$	\$
7	DUCTILE IRON WATER MAIN (OPEN CUT), 12-INCH	20	FOOT	\$	\$
8	MISCELLANEOUS FITTINGS	6,500	POUND	\$	\$
9	PROJECT MANAGEMENT	3	HOUR	\$	\$

	SCHEDULE	OF UNIT	PRICES		
No.	Pay Item	Approximate Quantity		Unit Price	Amount
10	WATER SERVICE CONNECTION TO WATER MAIN AND CURB STOP, 1- INCH	77	EACH	\$	\$
11	WATER SERVICE CONNECTION TO WATER MAIN AND CURB STOP, 1.5-INCH	1	EACH	\$	\$
12	WATER SERVICE LINE TYPE K - OPEN CUT, 1-INCH	490	FOOT	\$	\$
13	WATER SERVICE LINE TYPE K - OPEN CUT, 1.5-INCH	25	FOOT	\$	\$
14	WATER SERVICE LINE TYPE K - DIRECTIONAL DRILLING, 1-INCH	1,470	FOOT	\$	\$
15	WATER SERVICE LINE TYPE K - DIRECTIONAL DRILLING, 1.5-INCH	25	FOOT	\$	\$
16	WATER SERVICE LINE TYPE L, 3/4-INCH	20	FOOT	\$	\$
17	WATER SERVICE LINE TYPE L, 1-INCH	20	FOOT	\$	\$
18	WATER SERVICE LINE TYPE L, 1.5- INCH	20	FOOT	\$	\$
19	WATER SERVICE CONNECTION ON PRIVATE PROPERTY - BASEMENT OR CRAWL SPACE	1	EACH	\$	\$
20	WATER SERVICE CONNECTION ON PRIVATE PROPERTY - SLAB	1	EACH	\$	\$
21	UTILITY LOCATION - PRIVATE PROPERTY	1	EACH	\$	\$
22	WATER METER RELOCATION	1	EACH	\$	\$
23	WATER METER REPLACEMENT	1	EACH	\$	\$ \$
24	CONNECT TO WATER MAIN (NON- PRESSURE), 6-INCH	3	EACH		\$
25	CONNECT TO WATER MAIN (NON- PRESSURE), 8-INCH	5	EACH	\$	\$
26	CONNECT TO WATER MAIN (NON- PRESSURE), 12-INCH	2	EACH	\$	\$
27	CONNECT TO WATER MAIN (PRESSURE), 12-INCH X 8-INCH	4	EACH	\$	\$
28	GATE VALVÉ, 8-INCH	17	EACH	\$	\$
29	VALVE BOX	8	EACH	\$	\$
30	VALVE VAULT, 4-FOOT DIAMETER	9	EACH	\$	\$
31	VALVE VAULT, 5-FOOT DIAMETER	4	EACH	\$	\$
32	VALVE VAULT (ADDITIONAL DEPTH), 4-FOOT DIAMETER	5	VFT	\$	\$

	SCHEDULE	OF UNIT	PRICES		
No. Pay Item Approximate Unit Quantity Price A		Amount			
33	VALVE VAULT (ADDITIONAL DEPTH), 5-FOOT DIAMETER	5	VFT	\$	\$
	FIRE HYDRANT		EACH	\$	\$
	FIRE HYDRANT BARREL EXTENSION	5	VFT	\$	\$
_	LINE STOP, 6-INCH		EACH	\$	\$
	LINE STOP, 8-INCH		EACH	\$	\$
	LINE STOP, 12-INCH	2	EACH	\$ \$	\$
39	ABANDON EXISTING WATER MAINS AND APPURTENANCES	1	LSUM		\$
40	FILLING OF EXISTING WATER MAIN PIPE FOR ABANDONMENT	4	CUYD	\$	\$
41	SANITARY SEWER REMOVAL AND REPLACEMENT, 8-INCH	51	FOOT	\$	\$
42	STORM SEWER REMOVAL AND REPLACEMENT - 15-INCH	20	FOOT	\$	\$
43	STORM SEWER REMOVAL AND REPLACEMENT, 18-INCH	22	FOOT	\$	\$
44	INLET PROTECTION FILTER	39	EACH	\$	\$
45	TEMPORARY HOT-MIX ASPHALT PAVEMENT	310	SQYD	\$	\$
46	HOT-MIX ASPHALT REMOVAL AND REPLACEMENT, 2.5" BINDER 1.5" SURFACE	3,800	SQYD	\$	\$
47	HOT-MIX ASPHALT REMOVAL AND REPLACEMENT, 10" BINDER 2" SURFACE	110	SQYD	\$	\$
48	HOT-MIX ASPHALT DRIVEWAY REMOVAL AND REPLACEMENT, 3- INCH	150	SQYD	\$	\$
49	PCC DRIVEWAY REMOVAL AND REPLACEMENT, 6-INCH	150	SQYD	\$	\$
50	GRAVEL SHOULDER RESTORATION	150	FOOT	\$	\$
51	CONCRETE CURB AND GUTTER REMOVAL AND REPLACEMENT	400	FOOT		
52	PCC SIDEWALK REMOVAL AND REPLACEMENT	2,000	SQFT	\$	\$
53	DETECTABLE WARNINGS	10	SQFT	\$	\$
	PAVEMENT MARKING, 6-INCH (THERMOPLASTIC)	210	FOOT	\$	\$
55	PAVEMENT MARKING, 24-INCH (THERMOPLASTIC)	180	FOOT	\$	\$
56	RESTORATION OF LAWNS AND PARKWAYS	700	SQYD	\$	\$

	SCHEDULE OF UNIT PRICES				
	Approximate Unit				
No.	Pay Item	Qua	antity	Price	Amount
57	REMOVE AND REPLACE DRAIN	130	FOOT	\$	\$
	TILES, 6-INCH THROUGH 12-INCH				
58	ADJUSTING SANITARY SEWER	14	EACH	\$	\$
	SERVICE LINE				
59	DUST CONTROL	13	DAY	\$	\$
60	REMOVAL AND REPLACEMENT OF	48	CUYD	\$	\$
	UNSUITABLE MATERIAL				
61	SOIL TESTS FOR CONTAMINANTS	8	EACH	\$	\$
62	CONTAMINATED WASTE DISPOSAL	48	CUYD	\$	\$
63	EXPLORATORY EXCAVATION	7	EACH	\$	\$
64	TREE ROOT PRUNING	3	EACH	\$	\$
65	TREE TRIMMING	1	LSUM	Lump Sum	\$
66	WATER SERVICE LOCATE	1	LSUM	Lump Sum	\$
	TOTAL AMOUNT OF BID \$				

- 5. Bidder agrees the Work will be substantially completed within 60 calendar days after the Contract Time commences to run as provided in paragraph 2.03 of the General Conditions, and completed and ready for final payment in accordance with Paragraph 1.11 of the Supplementary Conditions within 90 calendar days after the Contract Time commences to run. The estimated Notice to Proceed date is March 1, 2022.
 - a. Bidder accepts the provisions of the Supplementary Conditions as to liquidated damages in the event of failure to complete the Work on time.

6.	•	rity in the form of (Certified Check or Bid Bond) in Percent of the Bid Amount.
7.	Bidder will be obtaining Performance agent or broker:	e and Payment Bonds through the following loca
	Name:	
	Address:	
	Telephone:	email:
,	District and action of the second flat and in Octob	00.40.00.44

- 8. Bidder submits all items listed in Section 00 43 93.41 Bid Submittal Checklist.
- 9. Terms used in this Bid which are defined in the Standard General Conditions of the Construction Contract included as part of the Contract Documents have the meanings assigned to them in the General Conditions.

10.	Bidder	acknowledges receipt of the following Addenda:	
	Adden <u>Num</u>		
11.	alterat	certifies that all iron and steel products used in the project for the construct ion, maintenance, or repair of a public water system are produced in the Un in compliance with Section 608 of the Water Resources Reform and Developm	ited
12.		omission of the Bid, Bidder certifies, and in the case of a Joint Bid each party theres as to his own organization, that in connection with the Bid:	reto
	a.	The prices in the Bid have been arrived at independently, without consultat communication, or agreement, for the purpose of restricting competition, as to matter relating to such prices with any other Bidder or with any competitor.	
	b.	Unless otherwise required by law, the prices which have been quoted in the have not knowingly been disclosed by the Bidder, prior to opening, directly indirectly to any other Bidder or to any competitor; and	
	C.	No attempt has been made or will be made by the Bidder to induce any of person or firm to submit or not to submit a Bid for the purpose of restric competition.	
13.	The pe	erson signing this Bid certifies that: (Check applicable box.)	
		He/She is the person in the Bidder's organization responsible within organization for the decision as to the prices being bid and that he/she has participated, and will not participate, in any action contrary to that above; or	
		He/She is not the person in the Bidder's organization responsible within organization for the decision as to the prices being bid but that he/she has be authorized to act as agent for the persons responsible for such decision certifying that such persons have not participated, and will not participate, in action contrary to that above, and as their agent shall so certify; and shall a certify that he/she has not participated, and will not participate, in any accontrary to that above.	een n in any also

14. Bidder is currently certified as an MBE or W Yes ☐ No ☐	/BE under EPA's DBE program?
Respectfully submitted, signed, and sealed this	day of, 20
(SEAL)	Bidder
_	Name - Title
ATTEST:	
Name - Title	

END OF BID FORM

00 43 13 BID SECURITY FORM (SAMPLE FORM)

KNOW ALL MEN E	BY THESE PRESENTS, th	nat we, the undersigned,
		as Principal, and
		as Surety, are hereby
held and firmly	bound unto	as
Owner in the penal	sum of	for the
payment of which, successors and as	·	le, we hereby jointly and severally bind ourselves,
Signed, this	day of	, 20
The Condition of th	e above obligation is such	that whereas the Principal has submitted to
		a certain BID,
attached hereto an	d hereby made a part here	eof to enter into a contract in writing, for the

NOW, THEREFORE,

- (a) If said BID shall be rejected, or
- (b) If said BID shall be accepted and the Principal shall execute and deliver a contract in the Form of Contract attached hereto (properly completed in accordance with said BID) and shall furnish a BOND for his faithful performance of said contract, and for the payment of all persons performing labor or furnishing materials in connection therewith, and shall in all other respects perform the agreement created by the acceptance of said BID, then this obligation shall be void, otherwise the same shall remain in force and effect; it being expressly understood and agreed that the liability of the Surety for any and all claims hereunder shall, in no event, exceed the penal amount of this obligation as herein stated.

The Surety, for value received, hereby stipulates and agrees that the obligations of said Surety and its BOND shall be in no way impaired or affected by any extension of the time within which the Owner may accept such bid; and said Surety does hereby waive notice of any such extension.

IN WITNESS WHEREOF, the Principal and the Surety have hereunto set their hands and seals, and such of them as are corporations have caused their corporate seals to be hereto affixed and these presents to be signed by their proper officers, the day and year first set forth above.

	(L.S.)
Principal	(=:0.)
Surety	
Ву:	

IMPORTANT-Surety companies executing BONDS must appear on the Treasury Department's most current list (Circular 570 as amended) and be authorized to transact business in the state where the project is located.

00 43 36

PROPOSED SUBCONTRACTORS FORM

TO: City of Joliet
Mayor and City Council
150 West Jefferson Street
Joliet, Illinois 60432
(hereinafter called Owner)

1. Pursuant to bidding requirements for the Work titled:

CITY OF JOLIET, ILLINOIS RIDGEWOOD PHASE 3 WATER MAIN IMPROVEMENTS CITY OF JOLIET CONTRACT NO. 2609-0122

for portions of the Work equaling or exceeding \$10,000, Bidder proposes to use the following Subcontractors. Except as otherwise approved by Owner, Bidder proposes to perform all other portions of the Work with his own forces:

2.	Portion of the Work:	Subcontractor name and address:

Portion of the Work:	Subcontractor name and address:
	-

USE ADDITIONAL SHEETS IF REQUIRED

END OF PROPOSED SUBCONTRACTORS FORM

00 43 93.41

BID SUBMITTAL CHECKLIST

BIDDER submits the following documents with this Bid:

- a. Form 00 41 00.41 "BID FORM".
- b. Required Bid Security (Certified Check or Bid Bond).
- c. City of Joliet Affidavits.
- CERTIFICATE 00 62 03 ILLINOIS DRUG FREE WORKPLACE ACT.
- e. CERTIFICATE 00 62 04 ILLINOIS CONTRACTOR CERTIFICATION.
- f. CERTIFICATE 00 62 05 ILLINOIS COMPLIANCE WITH SAFETY REQUIREMENTS.
- g. CERTIFICATE 00 62 06 ILLINOIS TAX DELINQUENCY/DEFAULT.
- h. CERTIFICATE 00 62 07 SUBSTANCE ABUSE PREVENTION PROGRAM.
- i. CERTIFICATE 00 62 08 EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORKS ACT.
- j. CERTIFICATE 00 62 33.13 BIDDER CERTIFICATION REGARDING THE USE OF AMERICAN IRON AND STEEL PRODUCTS
- k. CERTIFICATE 00 74 23.01 DBE FORM #1 (CONTRACTOR CERTIFICATION).
- I. CERTIFICATE 00 74 23.02 DBE FORM #3 (SUBCONTRACTOR UTILIZATION).
- m. CERTIFICATE 00 74 23.03 BIDDER CERTIFICATION FOR ADVERTISEMENT REGARDING SUBCONTRACTING OPPORTUNITIES FOR DISADVANTAGED BUSINESSES.
- n. CERTIFICATE 00 74 23.04 BIDDER CERTIFICATION IN COMPLIANCE WITH ARTICLE 33E-11 TO THE "CRIMINAL CODE OF 2012".
- o. CERTIFICATE 00 74 23.05 USEPA CERTIFICATION OF NONSEGREGATED FACILITIES.
- p. CERTIFICATE 00 74 23.06 NOTICE TO LABOR UNIONS OR OTHER ORGANIZATIONS OF WORKERS NONDISCRIMINATION IN EMPLOYMENT.

q. CERTIFICATE 00 74 23.07 USEPA CERTIFICATION REGARDING DEBARMENT, SUSPENSION, AND OTHER RESPONSIBILITY MATTERS.

END OF BID SUBMITTAL CHECKLIST

00 51 05

NOTICE OF INTENT TO AWARD

To:	
Project Description:	
rioject Description:	
The OWNER has considered the BID submit	ted by you for the above described WORK, in
response to its Advertisement for Bids, dated	
You are hereby notified that your BID will be ac Protection Agency (IEPA) approval, for items in the	ccepted, contingent upon Illinois Environmental ne amount of
You will be required by the Bidder Instruction required CONTRACTOR'S Performance BOND, within ten (10) calendar days from the date of the you.	•
Dated this day of	<u></u> .
	OWNER
	By:
	Title:

00 51 10

NOTICE OF AWARD

PROJECT Descript	ion:		
response to its Adv	ertisement for Bids date	ubmitted by you for the above described WOR	
20 and Bidder	Instructions .		
You are hereby	y notified that your BII	has been accepted for items in the amour	ıt of
CONTRACTOR'S F (10) calendar days If you fail to exect date of this Notice, OWNER'S accepta OWNER will be ent	Performance BOND, Pay from the date of this Not cute said Agreement and said OWNER will be e nce of your BID as aba itled to such other rights	ment BOND and certificates of insurance within ce to you. It to furnish said BONDS within ten (10) days from ntitled to consider all your rights arising out of indoned and as a forfeiture of your BID BOND. as may be granted by law. Jeed copy of this NOTICE OF AWARD to the OWN	ten the the The
Dated this	day of	, 20	
		(Owner) By	
		Title	
Recei		ANCE OF NOTICE OF AWARD is hereby acknowledged,	
оу			
his the	day of	, 20	
Ву			
T:41a			

NOTICE OF AWARD 00 51 10-1 IEPA-PWSLP

AGREEMENT

the		n the date stated below by and between the between inicipal Corporation (hereinafter "City") and
	•	(hereinafter "Contractor").
prop No. Con Baco	Contractor agrees to timely perform all water completion of the Ridgewood Phase 2609-0122 (hereinafter "Work"); and the tract Documents. The Contractor further on Wage Act (40 USC 276a through 276 are Illinois <i>Prevailing Wage Act</i> (820 ILCS)	.,,
part	of this Agreement as if recited at length	st of the following documents which are hereby made herein:
3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	Contract Documents not listed about Addenda IN WITNESS WHEREOF, the City	et Documents)
CIT	TY OF JOLIET,	
	Illinois Municipal Corporation,	Print name of Contractor
	James V. Capparelli City Manager	
Att	test:	Ву:
2	Christa M. Desiderio City Clerk	Print Name:
	Approved as to form:	Title:
	Sabrina Spano Corporation Counsel	

END OF AGREEMENT

AGREEMENT 00 52 00-1 (210039.40)

00 55 00

NOTICE TO PROCEED

To:	Date:
	Project:
	ORK in accordance with the Agreement dated
The date of substantial completion of all	WORK is therefore
20 The date of final completion of all WORK is	therefore, 20
	(Owner)
	Ву
	Title
ACCEPTANCE OF NOTICE	
Receipt of the above NOTICE TO PROCEED is hereby acknowledged by	
This the day of, 20	
Ву	
-	

00 60 01.41

REPRESENTATIONS, CERTIFICATIONS AND FORMS

00 61 13.13	PERFORMANCE BOND FORM
00 61 13.16	PAYMENT BOND FORM
00 61 19	MAINTENANCE BOND FORM
CERTIFICATE 00 62 03	ILLINOIS DRUG FREE WORKPLACE ACT
CERTIFICATE 00 62 04	ILLINOIS CONTRACTOR CERTIFICATION
CERTIFICATE 00 62 05	ILLINOIS COMPLIANCE WITH SAFETY REQUIREMENTS
CERTIFICATE 00 62 06	ILLINOIS TAX DELINQUENCY/DEFAULT
CERTIFICATE 00 62 07	SUBSTANCE ABUSE PREVENTION PROGRAM
CERTIFICATE 00 62 08	EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORKS ACT
CERTIFICATE 00 62 16	CERTIFICATE OF INSURANCE FORM
CERTIFICATE 00 62 16.13	ADDITIONAL INSUREDS
CERTIFICATE 00 62 16.16	ADDITIONAL INSUREDS
CERTIFICATE 00 62 16.19	ADDITIONAL INSUREDS
CERTIFICATE 00 62 16.21	ADDITIONAL INSUREDS
CERTIFICATE 00 62 33.13	BIDDER CERTIFICATION REGARDING THE USE OF AMERICAN IRON AND STEEL PRODUCTS
CERTIFICATE 00 62 76.01	SWORN STATEMENT
FORM 00 62 76.02	PARTIAL WAIVER OF LIEN
FORM 00 62 76.03	CONTRACTOR'S AFFIDAVIT
FORM 00 62 76.04	FINAL WAIVER OF LIEN
00 62 76.14	TAX EXEMPTION CERTIFICATE
00 62 79	STORED MATERIAL FORM

00 63 49	WORK CHANGE DIRECTIVE FORM
00 63 63	CHANGE ORDER FORM
CERTIFICATE 00 65 16	CERTIFICATE OF SUBSTANTIAL COMPLETION FORM
CERTIFICATE 00 65 19	CERTIFICATE OF COMPLETION FORM
00 65 36	WARRANTY FORM
00 65 36.03	CONTRACTOR'S AFFIDAVIT LETTER
CERTIFICATE 00 74 23.01	DBE FORM #1 (CONTRACTOR CERTIFICATION)
CERTIFICATE 00 74 23.02	DBE FORM #3 (SUBCONTRACTOR UTILIZATION)
CERTIFICATE 00 74 23.03	DBE FORM #4 (BIDDER'S LIST)
CERTIFICATE 00 74 23.04	BIDDER CERTIFICATION IN COMPLIANCE WITH ARTICLE 33 E TO THE "CRIMINAL CODE OF 1961"
CERTIFICATE 00 74 23.05	USEPA CERTIFICATION OF NONSEGREGATED FACILITIES
CERTIFICATE 00 74 23.06	NOTICE TO LABOR UNIONS OR OTHER ORGANIZATIONS OF WORKERS NONDISCRIMINATION IN EMPLOYMENT
CERTIFICATE 00 74 23.07	USEPA CERTIFICATION REGARDING DEBARMENT, SUSPENSION, AND OTHER RESPONSIBILITY MATTERS
00 74 23.52	WHISTLEBLOWERS RIGHTS POSTER (APPENDIX B)
00 74 23.55	EMPLOYEE RIGHTS UNDER THE DAVIS-BACON ACT POSTER (APPENDIX E)
00 74 23.57	U.S. DEPARTMENT OF LABOR FORM 1215 PAYROLL (APPENDIX G)
00 74 23.58	U.S. DEPARTMENT OF LABOR FORM 1445 LABOR STANDARDS INTERVIEW (APPENDIX H)

00 61 13.13

PERFORMANCE BOND FORM

Any singular reference to Contractor, Surety, Owner or other party shall be considered plural where applicable.

CONTRACTOR:		SURETY:	
OWNER:			
CONSTRUCTION CONTRACT:			
Date: Amount: Description:			
BOND:			
Date: Amount:			
CONTRACTOR AS PRINCIPAL: Company:	(Corp. Seal)	SURETY: Company:	(Corp. Seal)
Signature:		Signature:	

- 1. The Contractor and the Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors and assigns to the Owner for the performance of the Construction Contract, which is incorporated herein by reference.
- 2. If the Contractor performs the Construction Contract, the Surety and the Contractor shall have no obligation under this Bond, except to participate in conferences as provided in Subparagraph 3.1.
- 3. If there is no Owner Default, the Surety's obligation under this Bond shall arise after:
- 3.1 The Owner has notified the Contractor and the Surety at its address described in Paragraph 10 below, that the Owner is considering declaring a Contractor Default and has requested and attempted to arrange a conference with the Contractor and the Surety to be held not later than fifteen days after receipt of such notice to discuss methods of performing the Construction Contract. If the Owner, the Contractor and Surety agree, the Contractor shall be allowed a reasonable time to perform the Construction Contract, but such an agreement shall not waive the Owner's right, if any, subsequently to declare a Contractor Default: and
- 3.2 The Owner has declared a Contractor Default and formally terminated the Contractor's right to complete the contract. Such Contractor Default shall not be declared earlier than twenty days after the Contractor and the Surety have received notice as provided in Subparagraph 3.1: and
- 3.3 The Owner has agreed to pay the Balance of the Contract Price to the Surety in accordance with the terms of the Construction Contract or to a contractor selected to perform the Construction Contract in accordance with the terms of the contract with the Owner.
- 4. When the Owner has satisfied the conditions of Paragraph 3, the Surety shall promptly and at the Surety's expense take one of the following actions:
- 4.1 Arrange for the Contractor, with consent of the Owner, to perform and complete the Construction Contract: or
- 4.2 Undertake to perform and complete the Construction Contract itself, through its agents or through independent contractors; or
- 4.3 Obtain bids or negotiated proposals from qualified contractors acceptable to the Owner for a contract for performance and completion of the Construction Contract, arrange for a contract to be prepared for execution by the Owner and the contractor selected with the Owner's concurrence, to be secured with performance and payment bonds executed by a qualified surety equivalent to the bonds issued on the Construction Contract, and pay to the Owner the amount of damages as described in Paragraph 6 in excess of the Balance of the Contract Price incurred by the Owner resulting from the Contractor's default: or
- 4.4 Waive its right to perform and complete, arrange for completion, or obtain a new contractor and with reasonable promptness under the circumstances:
 - After investigation, determine the amount for which it may be liable to the Owner and, as soon as practicable after the amount is determined, tender payment therefor to the Owner: or
 - Deny liability in whole or in part and notify the Owner citing reasons therefor.
- 5. If the Surety does not proceed as provided in Paragraph 4 with reasonable promptness, the Surety shall be deemed to be in default on this Bond fifteen days after receipt of an additional written notice from the Owner to the Surety demanding that the Surety perform its obligations under this Bond, and the Owner shall be entitled to enforce any remedy available to the Owner. If the Surety proceeds as provided in Subparagraph 4.4, and the Owner refuses the payment tendered or the Surety has denied liability, in whole or in part, without further notice the Owner shall be entitled to enforce any remedy available to the Owner.

PERFORMANCE BOND FORM

- 6. After the Owner has terminated the Contractor's right to complete the Construction Contract, and if the Surety elects to act under Subparagraph 4.1, 4.2, or 4.3 above, then the responsibilities of the Surety to the Owner shall not be greater than those of the Contractor under the Construction Contract, and the responsibilities of the Owner to the Surety shall not be greater than those of the Owner under the Construction Contract. To the limit of the amount of this Bond, but subject to commitment by the Owner of the Balance of the Contract Price to mitigation of costs and damages on the Construction Contract, the Surety is obligated without duplication for:
- 6.1. The responsibilities of the Contractor for correction of defective work and completion of the Construction Contract"
- 6.2. Additional legal, design professional and delay costs resulting from the Contractor's Default, and resulting from the actions of failure to act of the Surety under Paragraph 4; and
- 6.3 Liquidated damages, or if no liquidated damages are specified in the Construction Contract, actual damages caused by delayed performance or non-performance of the Contractor.
- 7. The Surety shall not be liable to the Owner or others for obligations of the Contractor that are unrelated to the Construction Contract, and the Balance of the Contract Price shall not be reduced or set off on account of any such unrelated obligations. No right of action shall accrue on this Bond to any person or entity other than the Owner or its heirs, executors, administrators, or successors.
- 8. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders and other obligations.
- 9. Any proceeding, legal or equitable, under this Bond may be instituted in any court of competent jurisdiction in the location in which the work or part of the work is located and shall be instituted within two years after Contractor Default or within two years after the Contractor ceased working or within two years after the Surety refuses or fails to perform its obligations under this Bond, whichever occurs first. If the provisions of this Paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit shall be applicable.
- 10. Notice to the Surety, the Owner or the Contractor shall be mailed or delivered to the address shown on the signature page.
- 11. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement shall be deemed deleted herefrom and provisions conforming to such statutory or other legal requirement shall be deemed incorporated herein. The intent is that this Bond shall be construed as a statutory bond and not as a common law bond.
- Definitions.
- 12.1 Balance of the Contract Price: The total amount payable by the Owner to the Contractor under the Construction Contract after all proper adjustments have been made, including allowance to the Contractor of any amounts received or to be received by the Owner in settlement of insurance or other claims for damages to which the Contractor is entitled, reduced by all valid and proper payments made to or on behalf of the Contractor under the Construction Contract.
- 12.2 Construction Contract: The agreement between the Owner and the Contractor identified on the signature page, including all Contract Documents and changes thereto.
- 12.3 Contractor Default: Failure of the Contractor, which has neither been remedied nor waived, to perform or otherwise to comply with the terms of the Construction Contract.
- 12.4 Owner Default: Failure of the Owner, which has neither been remedied nor waived, to pay the Contractor as required by the Construction Contract or to perform and complete or comply with the other terms thereof.

FOR INFORMATION ONLY NAME, ADDR		
AGENT OR BROKER: OWNER'S REPRESENTATIVE (Architect, Engineer or other party):		

00 61 13.16

PAYMENT BOND FORM

Any singular reference to Contractor, Surety, Owner or other party shall be considered plural where applicable.

CONTRACTOR:	_	SURETY:	
OWNER:			
CONSTRUCTION CONTRACT:			
Date: Amount: Description:			
BOND:			
Date: Amount:			
CONTRACTOR AS PRINCIPAL: Company:	(Corp. Seal)	SURETY: Company:	(Corp. Seal)
Signature:		Signature:	

- 1. The Contractor and the Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors and assigns to the Owner to pay for labor, materials and equipment furnished for use in the performance of the Construction Contract, which is incorporated herein by reference.
- 2. With respect to the Owner, this obligation shall be null and void if the Contractor:
- 2.1 Promptly makes payment, directly or indirectly, for all sums due Claimants, and
- 2.2 Defends, indemnifies and holds harmless the Owner from all claims, demands, liens or suits by any person or entity who furnished labor, materials or equipment for use in the performance of the Construction Contract, provided the Owner has promptly notified the Contractor and the Surety (at the address described in Paragraph 12) of any claims, demands, liens or suits and tendered defense of such claims, demands, liens or suits to the Contractor and the Surety, and provided there is no Owner Default.
- 3. With respect to Claimants, this obligation shall be null and void if the Contractor promptly makes payment, directly or indirectly, for all sums
- 4. The Surety shall have no obligation to Claimants under this Bond until:
 - 4.1 Claimants who are employed by or have a direct contract with the Contractor have given notice to the Surety (at the address described in Paragraph 12) and sent a copy, or notice thereof, to the Owner, stating that a claim is being made under this Bond and, with substantial accuracy, the amount of the claim.
 - 4.2 Claimants who do not have a direct contract with the Contractor:
 - Have furnished written notice to the Contractor and sent a copy, or notice thereof, to the Owner, within 90 days after having last performed labor or last furnished materials or equipment included in the claim stating, with substantial accuracy, the amount of the claim and the name of the party to whom the materials were furnished or supplied or for whom the labor was done or performed: and
 - Have either received a rejection in whole or in part from the Contractor, or not received within 30 days of furnishing the above notice any communication from the Contractor by which the Contractor has indicated the claim will be paid directly or indirectly: and
 - 3. Not having been paid within the above 30 days, have sent a written notice to the Surety (at the address described in Paragraph 12) and sent a copy, or notice thereof, to the Owner, stating that a claim is being made under this Bond and enclosing a copy of the previous written notice furnished to the Contractor.
- 5. If a notice required by Paragraph 4 is given by the Owner to the Contractor or to the Surety, that is sufficient compliance.
- RESERVED
- 7. The Surety's total obligation shall not exceed the amount of this Bond, and the amount of this Bond shall be credited for any payments made in good faith by the Surety.
- 8. Amounts owed by the Owner to the Contractor under the Construction Contract shall be used for the performance of the
- Construction Contract and to satisfy claims, if any, under any Construction Performance Bond. By the Contractor furnishing and the

- Owner accepting this Bond, they agree that all funds earned by the Contractor in the performance of the Construction Contract are dedicated to satisfy obligations of the Contractor and the Surety under this bond, subject to the Owner's priority to use the funds for the completion of the work.
- 9. The Surety shall not be liable to the Owner, Claimants or others for obligations of the Contractor that are unrelated to the Construction Contract. The Owner shall not be liable for payment of any costs or expenses of any Claimant under this Bond.
- 10. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders and other obligations.
- 11. No suit or action shall be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the location in which the work is located or after the expiration of one year from the date (1) on which the Claimant gave the notice required by Subparagraph 4.1 or Clause 4.2 (iii), or (2) on which the last labor or service was performed by anyone or the last materials or equipment were furnished by anyone under the Construction Contract, whichever of (1) or (2) first occurs. If the provisions of this Paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit shall be applicable.
- 12. Notice to the Surety, the Owner or the Contractor shall be mailed or delivered to the address shown on the signature page. Actual receipt of notice by Surety, the Owner or the Contractor, however accomplished, shall be sufficient compliance as of the date received at the address shown on the signature page.
- 13. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement shall be deemed incorporated herein. The intent is, that this Bond shall be construed as a statutory bond and not as a common law bond.
- 14. Upon request by any person or entity appearing to be a potential beneficiary of this Bond, the Contractor shall promptly furnish a copy of this Bond or shall permit a copy to be made.

15. DEFINITIONS

- 15.1 Claimant: An individual or entity having a direct contract with the Contractor or with a subcontractor of the Contractor to furnish labor, materials or equipment for use in the performance of the Contract. The intent of this Bond shall be to include without limitations in the terms, "labor, materials or equipment" that part of water, gas, power, light, heat, oil, gasoline, telephone service or rental equipment used in the Construction Contract, architectural and engineering services required for performance of the work of the Contractor and the Contractor's subcontractors, and all other items for which a mechanic's lien may be asserted in the jurisdiction where the labor, materials or equipment were furnished.
- 15.2 Construction Contract: The agreement between the Owner and the Contractor identified on the signature page, including all Contract Documents and changes thereto.
- 15.3 Owner Default: Failure of the Owner, which has neither been remedied nor waived, to pay the Contractor as required by the Construction Contract or to perform and complete with the other terms thereof.

FOR INFORMATION ONLY NAME, AD	DDRESS AND TELEPHONE)		
AGENT OR BROKER:	NT OR BROKER: OWNER'S REPRESENTATIVE (Architect, Engineer or other party):		

00 61 19 MAINTENANCE BOND FORM (SAMPLE FORM)

MAINTENANCE/WARRANTY BOND

as Principal, and				
as Surety, are held and firmly bound u				
Illinois as Obligee,				
payment well and truly to be made we do ladministrators, successors, and assigns join	bind ou	rselves, our	and each of ou	
WHEREAS, the said Principal, Sta				
dated	_ for _			
guarantee for the period ofsaid job, by the Owner, against all defects apparent during said period, and WHEREAS, the said Contract has	s in wor	kmanship a	nd materials w	which may become
day of				
NOW, THEREFORE, THE CONDITION Principal shall indemnify the Obligee for a defective materials or workmanship water year(s) from and after this obligation shall be void, otherwise to a	ll loss t which	hat the Obli becomes a	gee may sustai pparent durin	n by reason of any g the period of
SIGNED, SEALED, AND DATED				
				(L.S.)
				(L.S.)
				(L.S.)
		Princ	ipal(s)	
		Surety	Name	
by				
				, Attorney-in-Fact

DEPARTMENT OF FINANCE PURCHASING DIVISION DAWN A. KOCHAN PURCHASING/CONTRACTS ADMINISTRATOR dkochan@joliet.gov



Prospective Bidders:

Enclosed are bid documents which may be of interest to your company.

Please note the date as to when these bids will be opened.

THIS BID DOCUMENT MUST BE RETURNED INTACT (THE SAME ORDER AS RECEIVED). FAILURE TO DO SO MAY INVALIDATE YOUR PROPOSAL.

ALL BIDS SHALL BE SUBMITTED IN AN OPAQUE, SEALED ENVELOPE TO THE CITY CLERK, 150 WEST JEFFERSON STREET, JOLIET, ILLINOIS 60432, PRIOR TO THE TIME AND DATE SET FORTH FOR BID OPENING IN THE NOTICE TO BIDDERS. EACH BID SHALL BE ADDRESSED TO THE CITY CLERK AND SHALL BEAR ON THE FACE OF THE ENVELOPE THE NAME OF THE BIDDER, AND A STATEMENT THAT IT IS A SEALED BID, TO BE OPENED FOR THE CONTRACT CONSIDERED, AT THE DATE AND HOUR SET FORTH IN THE INVITATION TO BID.

THE BIDDER SHALL ALSO STATE ON THE OUTSIDE OF THE ENVELOPE THAT HE/SHE IS IN RECEIPT OF ALL ADDENDUMS TO THE CONTRACT.

ANY QUESTIONS OR CLARIFICATIONS CONCERNING THESE SPECIFICATIONS SHOULD BE IN WRITING AND DIRECTED TO DAWN A. KOCHAN, CONTRACT ADMINISTRATOR AT PHONE NO. (815) 724-3925 OR (815) 724-3929 FAX.

ANY ORAL OR WRITTEN COMMENTS RECEIVED FROM ANY OTHER PERSON OTHER THAN THE CONTRACT ADMINISTRATOR, WILL NOT BE CONSIDERED AND ALSO, MAY INVALIDATE YOUR PROPOSAL.

THANK YOU FOR BIDDING.

MAYOR Bob O'Dekirk

COUNCILPERSONS
Joe Clement
Bettye Gavin
Cesar Guerrero
Larry E. Hug
Terry Morris
Pat Mudron
Jan Hallums Quillman
Sherry Reardon

CITY MANAGER
James V. Capparelli

PURCHASING/CONTRACTS ADMINISTRATOR Dawn A. Kochan

DEPARTMENT OF PUBLIC WORKS & UTILITIES

815-724-4200 Direct 815-723-7770 Fax



GENERAL CONTRACTOR

RE: 2022 CONSTRUCTION PROJECTS

Dear General Contractor:

The City of Joliet currently has specifications, which require ten (10%) percent minority participation in contracts over \$100,000.00. In an effort to track this information better, the City of Joliet will be requiring all GENERAL CONTRACTS to submit to a complete list of subcontractors they intend to use on the awarded project at the pre-construction meeting. The issuance of this sub-contractor list will help the City ensure that minority requirements are met as well as prevailing wages. The sub-contractor list should include the dollar amount or percentage of this contract for the work the sub-contractor is responsible. I.D.O.T. for BC 260-A is an example of such an acceptable form. In addition, the specification states that monthly certified payroll records are to be submitted to the Engineer. This will also be enforced for the upcoming construction season.

The City of Joliet appreciates your effort throughout the years and looks forward to working with you in the upcoming construction season.

Gregory P. Ruddy, P.E. Director of Public Works

Allison Swisher

Director of Public Utilities

Shon Sode

NOTICE

The City of Joliet has adopted a Responsible Bidder Ordinance (2-440). In addition to any other requirement, unless otherwise expressly stated in a bid solicitation or an award of contract, or applicable state or federal law or regulation states otherwise, all bidders must comply with the following requirements in order to submit a bid or be awarded a contract.

- The bidder must be a duly organized legal entity in good standing with the Illinois Secretary of State and in compliance with all laws prerequisite to doing business in the state and submit a copy of the Illinois Secretary of State Department of Business Services online records evidencing that the bidder has a current corporate annual report on file, if required. If the bidder is an individual, sole proprietorship, or partnership, this subsection shall not apply.
- 2. The bidder must have a valid Federal Tax Identification Number or Social Security Number (for individuals), documents evidencing current registration with the Illinois Department of Revenue and the Illinois Department of Employment Security (if bidder has employees).
- 3. The bidder must be an equal opportunity employer and comply with Section 2000(e) of Chapter 21, Title 42 of the United States Code and Executive Order 11246.
- 4. The bidder shall provide certificates of insurance indicating the following coverages if called for in the bid solicitation: general liability, workers compensation, completed operations, automobile, hazardous occupation, and product liability.
- 5. A statement that all employees are (i) covered under a current workers' compensation insurance policy, and (ii) properly classified under such policy. If the bidder is insured with a carrier, the evidence of workers' compensation insurance shall be a copy of the "Information Page" which includes the name and address of the insured, as well as class codes upon which the compensation premium is based and the total estimated remuneration per class code.
- 6. A statement of compliance with all provisions of the Illinois Prevailing Wage Act (820 ILCS 130), as amended, and all the rules and regulations therein, for the past five (5) years. Such statement shall also provide that the contractor has reviewed the Prevailing Wage Act or the federal Davis-Bacon

and Related Acts, has reviewed and agrees to pay the applicable prevailing wage rates attached to the bid specifications, and will strictly comply with the Prevailing Wage Act or federal Davis-Bacon and Related Acts and related requirements, including wages, medical and hospitalization insurance and retirement for those trades covered in the Act. A contractor who has been found, by the Illinois Department of Labor, to be in violation of the Prevailing Wage Act twice within a three (3) year period shall be deemed to not be Responsible Bidder for two (2) years from the date of the latest finding.

- 7. Certification of compliance with the Substance Abuse Prevention on Public Works Projects Act (820 ILCS 265/1, et seq.).
- 8. A certification that individuals who perform work on the public works project on behalf of the contractor are properly classified as either (i) an employee, or (ii) an independent contractor under all applicable state and federal laws and local ordinances.
- 9. Documents evidencing any professional or trade licenses required by law or local ordinance for any trade or specialty in which the contractor is seeking a contract award. Additionally, the contractor must disclose any suspension or revocation of such license held by the company, or of any director, officer, or manager of the company.
- 10. For construction contracts (construction of new facilities, renovation of existing facilities, or road construction projects), or demolition contracts, over \$25,000 (twenty five thousand dollars), evidence of participation in apprenticeship and training programs applicable to the work to be performed on the project, which are approved by and registered with the United States Department of Labor's Office of Apprenticeship, or its successor organization. The required evidence includes but is not limited to a copy of all applicable apprenticeship standards or Apprenticeship Agreement(s) for any apprentice(s) who will perform work on the public works project.
- Disclosure of the name and address of each subcontractor from whom the contractor has accepted a bid and/or intends to hire on any part of the project. Further, each such subcontractor shall be required to adhere to the requirements set forth herein as though it were bidding directly to the city. Each contractor shall submit all subcontractor information and supporting documentation to the city prior to the subcontractor commencing work on the project.

- 12. Unless otherwise agreed to in a collective bargaining agreement, the bidder shall certify that it provides, or contributes to, a health care plan for those employees working on the project and shall provide a copy of the health care plan upon request. The contributions toward a health care plan must be part of the employee's regular compensation, and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.
- 13. Unless otherwise agreed to in a collective bargaining agreement, the bidder shall certify that it contributes to an employee pension or retirement program, including but not limited to a 401k, a defined benefit plan or a similar plan for its field employees working on the project and shall provide a copy of the plan upon request. The contributions toward a pension or retirement program must be a part of the employee's regular compensation and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.
- 14. Certification that the bidder has not been subject to any determinations by a court or governmental agency for violations of federal, state, or local laws, including but not limited to serious, willful, or repeated violations of the Occupational Safety and Health Act (OSHA), violations of contracting or antitrust laws, tax or licensing laws, environmental laws, or the federal Davis-Bacon and Related Acts.
- 15. Whenever practical, the bidder should prequalify as a local bidder.
- 16. Any material changes to the contractor's status, at any time during the performance of the project, must be reported in writing to the city within fourteen (14) days of its occurrence. Failure to comply with this requirement is grounds for the contractor to be deemed a non-responsible bidder.
- 17. The submission requirements of this Ordinance also apply to subcontractors, except that the contractor shall submit all subcontractor submissions to the city prior to the subcontractor commencing work on the project. Failure of a subcontractor to submit the required information shall not disqualify the successful bidder from performing work on the project and shall not constitute a contractual default and/or breach by the successful bidder. However, the city may withhold all payments otherwise due for work performed by a subcontractor until the subcontractor submits the required information and the city approves such information. Further, where the city deems necessary, the city may also require that the successful bidder remove the subcontractor from the project and replace it with a responsive and responsible subcontractor.

CERTIFICATIONS/ATTACHMENTS

For each required certification or submission, the bidder shall comply with the appropriate subsection of the City of Joliet code of ordinances by initialing the page and/or attaching the required documentation to the page.

The certification pages should be initialed and submitted with the bid along with the required attachments.

There are 16 certification pages and a checklist that follows.

2-440(1) - The bidder must be a duly organized legal entity in good standing with the Illinois Secretary of State and in compliance with all laws prerequisite to doing business in the state and submit a copy of the Illinois Secretary of State Department of Business Services online records evidencing that the bidder has a current corporate annual report on file, if required. If the bidder is an individual, sole proprietorship, or partnership, this subsection shall not apply.

Illinois Secretary of State and in compliance with all laws prerequisite to doing business in the state.
Initials of bidder
Or:
I certify that the bidder is an individual, sole proprietor, or partnership.
Initials of bidder

Attachment: Illinois Secretary of State Department of Business Services online records evidencing that the bidder has a current corporate annual report on file (if required).

Mark attachment as "Attachment A"

2-440(2) - The bidder must have a valid Federal Tax Identification Number or Social Security Number (for individuals), documents evidencing current registration with the Illinois Department of Revenue and the Illinois Department of Employment Security (if bidder has employees).

Attachment: Documents evidencing current registration with the Illinois Department of Revenue and the Illinois Department of Employment Security (if bidder has employees).

Mark attachment as "Attachment B"

2-440(3) - The bidder must be an equal opportunity employer and comply with Section 2000(e) of Chapter 21, Title 42 of the United States Code and Executive Order 11246.

I certify that the bidder is an equal opportunity employer and is in compliance with Secti	on
2000(e) of Chapter 21, Title 42 of the United States Code and Executive Order 11246	

Initials of bidder

2-440(4) - The bidder shall provide certificates of insurance indicating the following coverages if called for in the bid solicitation: general liability, workers compensation, completed operations, automobile, hazardous occupation, and product liability.

Attachments: Certificates of insurance indicating the following coverages as called for in the bid solicitation.

Mark attachment as "Attachment C"

2-440(5) - A statement that all employees are (i) covered under a current workers' compensation insurance policy, and (ii) properly classified under such policy. If the bidder is insured with a carrier, the evidence of workers' compensation insurance shall be a copy of the "Information Page" which includes the name and address of the insured, as well as class codes upon which the compensation premium is based and the total estimated remuneration per class code.

I certify that all employees are (i) covered under a current workers' compensation insurance policy, and (ii) properly classified under such policy.

Initials of bidder

Attachment: "Information Page" which includes the name and address of the insured, as well as class codes upon which the compensation premium is based and the total estimated remuneration per class code (if applicable).

Mark attachment as "Attachment D"

2-440(6) - A statement of compliance with all provisions of the Illinois Prevailing Wage Act (820 ILCS 130), as amended, and all the rules and regulations therein, for the past five (5) years. Such statement shall also provide that the contractor has reviewed the Prevailing Wage Act or the federal Davis-Bacon and Related Acts, has reviewed and agrees to pay the applicable prevailing wage rates attached to the bid specifications, and will strictly comply with the Prevailing Wage Act or federal Davis-Bacon and Related Acts and related requirements, including wages, medical and hospitalization insurance and retirement for those trades covered in the Act. A contractor who has been found, by the Illinois Department of Labor, to be in violation of the Prevailing Wage Act twice within a three (3) year period shall be deemed to not be Responsible Bidder for two (2) years from the date of the latest finding.

I certify that the bidder is in compliance with all provisions of the Illinois Prevailing Wage Act (820 ILCS 130), as amended, and all the rules and regulations therein, and has been for the past five (5) years.

Initials of bidder

I certify that that the contractor has reviewed the Prevailing Wage Act or the federal Davis-Bacon and Related Acts, has reviewed and agrees to pay the applicable prevailing wage rates attached to the bid specifications, and will strictly comply with the Prevailing Wage Act or federal Davis-Bacon and Related Acts and related requirements, including wages, medical and hospitalization insurance and retirement for those trades covered in the Act.

Initials of bidder

2-440(7) - Certification of compliance with the Substance Abuse Prevention on Public Works Projects Act (820 ILCS 265/1, et seq.).

I certify that the bidder is in compliance with the S	ubstance Abuse Prevention on Public
Works Projects Act (820 ILCS 265/1, et seq.).	
	
Initials of bidder	

2-440(8) - A certification that individuals who perform work on the public works project on behalf of the contractor are properly classified as either (i) an employee, or (ii) an independent contractor under all applicable state and federal laws and local ordinances.

I certify that all individuals who perform work on the public works project on behalf of the contractor are properly classified as either (i) an employee, or (ii) an independent contractor under all applicable state and federal laws and local ordinances.

Initials of bidder

2-440(9) - Documents evidencing any professional or trade licenses required by law or local ordinance for any trade or specialty in which the contractor is seeking a contract award. Additionally, the contractor must disclose any suspension or revocation of such license held by the company, or of any director, officer, or manager of the company.

Attachments: Documents evidencing any professional or trade licenses required by law or local ordinance for any trade or specialty in which the contractor is seeking a contract award.

Mark attachment as "Attachment E"

Attachments: Disclosure of any suspension or revocation of such license held by the company, or of any director, officer, or manager of the company.

Mark attachment as "Attachment E1"

Or

I certify that the contractor has not been the subject of any suspension or revocation of such license held by the company, or of any director, officer, or manager of the company.

Initials of bidder

2-440(10) - For construction contracts (construction of new facilities, renovation of existing facilities, or road construction projects), or demolition contracts, over \$25,000 (twenty five thousand dollars), evidence of participation in apprenticeship and training programs applicable to the work to be performed on the project, which are approved by and registered with the United States Department of Labor's Office of Apprenticeship, or its successor organization. The required evidence includes but is not limited to a copy of all applicable apprenticeship standards or Apprenticeship Agreement(s) for any apprentice(s) who will perform work on the public works project.

Attachments: Evidence of participation in apprenticeship and training programs applicable to the work to be performed on the project, which are approved by and registered with the United States Department of Labor's Office of Apprenticeship, or its successor organization. The required evidence includes but is not limited to a copy of all applicable apprenticeship standards or Apprenticeship Agreement(s) for any apprentice(s) who will perform work on the public works project.

Mark attachment as "Attachment F"

2-440(11) - Disclosure of the name and address of each subcontractor from whom the contractor has accepted a bid and/or intends to hire on any part of the project. Further, each such subcontractor shall be required to adhere to the requirements set forth herein as though it were bidding directly to the city. Each contractor shall submit all subcontractor information and supporting documentation to the city prior to the subcontractor commencing work on the project.

Attachments: The name and address of each subcontractor from whom the contractor has accepted a bid and/or intends to hire on any part of the project.

Mark attachment as "Attachment G"

I certify and acknowledge that the bidder shall submit all subcontractor information	ation and
supporting documentation to the city prior to the subcontractor commencing wo	rk on the
project.	

Initials of bidder

2-440(12) - Unless otherwise agreed to in a collective bargaining agreement, the bidder shall certify that it provides, or contributes to, a health care plan for those employees working on the project and shall provide a copy of the health care plan upon request. The contributions toward a health care plan must be part of the employee's regular compensation, and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.

I certify that the bidder provides, or contributes to, a health care plan for those employees working on the project and shall provide a copy of the health care plan upon request. I further certify that these contributions are part of the employee's regular compensation, and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.

Initials of bidder

Or

I certify that the bidder is subject to a collective bargaining agreement which addresses issues regarding a health care plan.

Initials of bidder

2-440(13) - Unless otherwise agreed to in a collective bargaining agreement, the bidder shall certify that it contributes to an employee pension or retirement program, including but not limited to a 401k, a defined benefit plan or a similar plan for its field employees working on the project and shall provide a copy of the plan upon request. The contributions toward a pension or retirement program must be a part of the employee's regular compensation and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.

I certify that the bidder contributes to an employee pension or retirement program, including but not limited to a 401k, a defined benefit plan or a similar plan for its field employees working on the project and shall provide a copy of the plan upon request. These contributions toward a pension or retirement program are part of the employee's regular compensation and not merely part of the employee's compensation during the period of time for which the employee is performing work on the project.

Initials of bidder

Or

I certify that the bidder is subject to a collective bargaining agreement which addresses contributions to an employee pension or retirement program, including but not limited to a 401k, a defined benefit plan or a similar plan for its field employees working on the project and shall provide a copy of the plan upon request.

2-440(14) - Certification that the bidder has not been subject to any determinations by a court or governmental agency for violations of federal, state, or local laws, including but not limited to serious, willful, or repeated violations of the Occupational Safety and Health Act (OSHA), violations of contracting or antitrust laws, tax or licensing laws, environmental laws, or the federal Davis-Bacon and Related Acts.

I certify that the bidder has not been subject to any determinations by a court or governmental agency for violations of federal, state, or local laws, including but not limited to serious, willful, or repeated violations of the Occupational Safety and Health Act (OSHA), violations of contracting or antitrust laws, tax or licensing laws, environmental laws, or the federal Davis-Bacon and Related Acts.

2-440(b) - Any material changes to the contractor's status, at any time during the performance of the project, must be reported in writing to the city within fourteen (14) days of its occurrence. Failure to comply with this requirement is grounds for the contractor to be deemed a non-responsible bidder.

I certify and acknowledge that any material changes to the contractor's status, at any time during the performance of the project, must be reported in writing to the city within fourteen (14) days of its occurrence. Failure to comply with this requirement is grounds for the contractor to be deemed a non-responsible bidder.

2-440(d) - The submission requirements of this Ordinance also apply to subcontractors, except that the contractor shall submit all subcontractor submissions to the city prior to the subcontractor commencing work on the project. Failure of a subcontractor to submit the required information shall not disqualify the successful bidder from performing work on the project and shall not constitute a contractual default and/or breach by the successful bidder. However, the city may withhold all payments otherwise due for work performed by a subcontractor until the subcontractor submits the required information and the city approves such information. Further, where the city deems necessary, the city may also require that the successful bidder remove the subcontractor from the project and replace it with a responsive and responsible subcontractor.

I certify and acknowledge that the submission requirements of this Ordinance also apply to subcontractors. I further acknowledge that the city may withhold all payments otherwise due for work performed by a subcontractor until the subcontractor submits required information and the city approves such information. I further acknowledge that where the city deems necessary, the city may also require that the successful bidder remove the subcontractor from the project and replace it with a responsive and responsible subcontractor.

CERTIFICATION CHECKLIST

Initial bottom of page to certify that required certifications and attachments are complete.

Bidders/Contractor are required to certify to the following items:

2-440(1) – Good standing with S.O.S.	
2-440(3) - Equal opportunity employer	
2-440(5) – Workers' compensation coverage	
2-440(6) – Prevailing Wage Act	
2-440(7) – Substance Abuse Prevention on Public Works Act	
2-440(8) – Employees properly classified	
2-440(9) - Contractor properly licensed	
2-440(11) - Disclosure of subcontractor information	
2-440(12) – Employee health care plan	
2-440(13) – Employee retirement plan	
2-440(14) - OSHA compliance	
2-440(b) - Required reporting of change in status	
2-440(d) - Required updating of subcontractor information	
Bidders/Contractor are required to attach the following documents:	
2-440(1) – S.O.S. annual report (if required)	
2-440(2) – Registration with IDOR and IDES	
2-440(4) – Certificates of insurance	
2-440(5) - Information page regarding workers' compensation	
2-440(9) – Professional or trade licenses required	
2-440(9) – Disclosure of suspension or revocation of license	
2-440(10) - Apprenticeship standards or agreements (if required)	
2-440(11) – List of subcontractors	
I certify that all required certifications and attachments documents	s are included in the bid

ILLINOIS DRUG FREE WORKPLACE ACT

III. Comp. Stat. ch. 30-580

The Contractor certifies and agrees that it will provide a drug free workplace by:

- (a) Publishing a statement:
 - (1) Notifying employees that the unlawful manufacture, distribution, dispensing, possession or use of a controlled substance, including cannabis, is prohibited in the Contractor's workplace.
 - (2) Specifying the actions that will be taken against employees for violations of such prohibition.
 - (3) Notifying the employee that, as a condition of employment on such contract, the employee will:
 - (A) abide by the terms of the statement; and
 - (B) notify the employer of any criminal drug statute conviction for a violation occurring in the workplace no later than five (5) days after such conviction.
- (b) Establishing a drug free awareness program to inform employees about:
 - (1) the dangers of drug abuse in the workplace;
 - (2) the Contractor's policy of maintaining a drug free workplace;
 - (3) any available drug counseling, rehabilitation, and employee assistance programs; and
 - (4) the penalties that may be imposed upon an employee for drug violations.
- (c) Providing a copy of the statement required by subparagraph (a) to each employee engaged in the performance of the contract and to post the statement in a prominent place in the workplace.
- (d) notifying the contracting agency within ten (10) days after receiving notice under part (B) of paragraph (3) of subsection (a) above from an employee or otherwise receiving actual notice of such conviction.
- (e) Imposing a sanction on, or requiring the satisfactory participation in a drug abuse assistance or rehabilitation program by, any employee who is so convicted, as required by section 5 of the Drug Free Workplace Act.

- (f) Assisting employees in selecting a course of action in the event drug counseling, treatment, and rehabilitation is required and indicating that a trained referral team is in place.
- (g) Making a good faith effort to continue to maintain a drug free workplace through implementation of the Drug Free Workplace Act.

THE UNDERSIGNED AFFIRMS, UNDER PENALTIES OF PERJURY, THAT HE OR SHE IS AUTHORIZED TO EXECUTE THIS CERTIFICATION ON BEHALF OF THE DESIGNATED ORGANIZATION.

Printed Name of Contractor	
Signature of Authorized Representative	
Printed Name and Title	Date

ILLINOIS CONTRACTOR CERTIFICATION

that all statements herein made are made to make them, and that the statements co	on behalf of Con	•					
Contractor deposes, states, and certifies that Contractor is not barred from bidding and/or contracting with a unit of state or local government as a result of (I) any person holding an interest in this Contract in violation of Sections 11.1 et seq. of the Illinois Purchasing Act, 30 ILCS 505/11.1 et seq., (ii) the commission of an act in violation of Section 10.1 of the Illinois Purchasing Act, 30 ILCS 505/10.1; (iii) a default on an educational loan as provided in the Educational Loan Default Act, 5 ILCS 385/0.01 et seq.; (iv) a delinquency in the payment of any tax administered by the Illinois Department of Revenue unless contractor is contesting in accordance with the procedures established by the appropriate Revenue Act, its liability for the tax or the amount of tax, as set forth in 65 ILCS 5/11-42.1-1; or (v) a violation of either Section 33E-3 or Section 33-E-4 of Article 33E of the Criminal Code of 1961, 720 ILCS 5/33E-1 et seq.							
Contractor further deposes, states and continue to comply with the Illinois Human		•					
Dated this day of		, 20,					
Attest/Witness							
By:	Ву:						
Title:	Title:_						
Subscribed and sworn to before me this day of	, 20	My Commission Expires:					
Notary Public		SEAL					

ILLINOIS COMPLIANCE WITH SAFETY REQUIREMENTS

The undersigned, as a duly-authorized representative of the Contractor,	nereby ce	rtifies
to the (Owner)	_ Illinois,	that
the Contractor is fully aware of and able to comply with all Local, State, and Federal	eral Safety	Laws
and Regulations applicable for the construction of the Project known as (Project)		
	_·	
Name		
Company		
Date		

ILLINOIS TAX DELINQUENCY/DEFAULT

The	undersigned,	upon	being	first	duly	sworn,	hereby	certifies	to	the	(Owner)
							, t	hat			
(Contractor) is not currently	y delino	quent ir	the p	oayme	ent of an	y tax adn	ninistered	by o	or ow	ed to the
Illinois Dep	eartment of Rev	/enue,	of othe	erwise	in d	efault up	oon any	such tax	as o	defin	ed under
Chapter 24	, Section 11-42	1, <u>Illir</u>	ois Re	vised	Statut	es.					
Ву:	(Name of C	ontrac	tor)								
	(Tit	le)									
	and sworn to b day of			,	20	۱ .	My Comn	nission Ex	крire	s:	
	Notary Public					_		SE	AL		

SUBSTANCE ABUSE PREVENTION PROGRAM

The undersigned, upon being first d	uly sworn, l	nereby certifies to the (Owner)
that		,
		(Contractor) has
in place a written Substance Abuse Preven	tion Progra	m that meets or exceeds the requirements
of the State of Illinois P.A. 095-0635, or ha	as a collecti	ve bargaining agreement in effect dealing
with the subject matter of P.A. 095-0635.	The Contra	ctor and Subcontractors will file a copy of
the Substance Abuse Prevention Program	, or collect	ive bargaining agreement, with the Client
prior to any work being conducted on the p	roject.	
By:		
(Name of Contractor)	_	
(Title)		
Subscribed and sworn to before me this day of	_, 20	My Commission Expires:
Notary Public		SEAL

EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORKS ACT

This law comes into effect following two consecutive months of a state unemployment rate above 5 percent. More information about the Employment of Illinois Workers on Public Works Act can be found here: http://www.ilga.gov/legislation/ilcs/ilcs3.asp?ActID=549&ChapterID=7.

The Employment of IL Workers on Public Works Act requires the workforce on all public works projects to be comprised of a minimum of 90% Illinois residents.

The Illinois Department of Labor (IDOL) administers the Employment of Illinois Workers on Public Works Act, which was enacted to alleviate unemployment in Illinois by ensuring that most workers on public works projects live in the state. The requirement applies to all labor on public works projects or improvements. That includes public works projects that are funded in whole or in part with state funds or funds administered by the state of Illinois. Any public works project financed in whole or in part by federal funds administered by the State of Illinois is covered under the provisions of this act.

By signing below, the Contractor confirms awareness of this requirement.

(Typed Name & Title of Authorized Representative)	
(Signature of Authorized Representative)	(Date)
(Company/Organization)	

CERTIFICATE 00 62 16 CERTIFICATE OF INSURANCE (SAMPLE FORM)

-	
ACORD	

DATE (MM/DD/YYYY)

7	COND	EK	ш	ICATE OF LIAE	SILI	I Y INS	URANG	,E		
(THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.									
t	MPORTANT: If the certificate holder he terms and conditions of the policy pertificate holder in lieu of such endo	/, cer	ain p	policies may require an end	olicy(dorse	(les) must be ment. A sta	e endorsed. tement on th	If SUBROGATION IS W ils certificate does not d	AIVED confer r	, subject to ights to the
PR	DDUCER			9	CONTA	CT				
				П	PHONE			FAX		
				H	MAL.	o. Exti:		(A/C, No)		
				Ľ	ADDRE					$\overline{}$
						INS	SURER(S) AFFOR	RDING COVERAGE		NAIC #
					NSURE	R A: (Type in Fu	il Name, Carrier (Co.)		
INS	URED			<u> </u>	NSURE	RB; Shall have t	minimum rating of	A-VII		
				<u>.</u>	NSURE	RC:				
					NSURE	RD:				
					NSURE	RE:				
					NSURE	RF:				
CO	OVERAGES CER	RTIFIC	ATE	NUMBER:				REVISION NUMBER:		
	THIS IS TO CERTIFY THAT THE POLICIES				BEEN	I ISSUED TO	THE INSURED	NAMED ABOVE FOR TH	E POLIC	Y PERIOD
	NDICATED. NOTWITHSTANDING ANY RI	EQUIR	EME	NT, TERM OR CONDITION OF	F ANY	CONTRACT	OR OTHER D	OCUMENT WITH RESPEC	T TO W	VHICH THIS
	CERTIFICATE MAY BE ISSUED OR MAY EXCLUSIONS AND CONDITIONS OF SUCH							HEREIN IS SUBJECT TO	ALL TH	IE TERMS,
INS		ADDL	SUBR		LEN N	MMODAYYYY	POLICY EXP			
LTI	X COMMERCIAL GENERAL LIABILITY	INSD	WVD	POLICY NUMBER	-	(MINUSCHITTI)	(MMDD/YYYY)	LIMI	18	0.000.000
		1						EACH OCCURRENCE DAMAGE TO RENTED	\$	2,000,000
	CLAIMS-MADE X OCCUR	1						PREMISES (Ea occurrence) MED EXP (Any one person)	\$	
		1						PERSONAL & ADVINJURY	\$	2,000,000
		1					l l	PERSONAL A ADVINCORT	•	2,000,000
	GEN'L AGGREGATE LIMIT APPLIES PER:	1		0.4		AD.		GENERAL AGGREGATE	\$	2,000,000
	POLICY X PRO-	1		SA	ΔI	MPI		PRODUCTS - COMP/OP AGG	\$	2,000,000
<u> </u>	OTHER:	₩		07		V		COMPAND ON OF FURIT	\$	
	X ALLOWNED SCHEDULED ALTOS ALTOS							COMBINED SINGLE LIMIT (Ea ecodent)	\$	1,000,000
	X AUTOS AUTOS							BODILY INJURY (Per person)	\$	
	X HIRED AUTOS X NON-OWNED	1					-	BODILY INJURY (Per accident) \$	
	HIRED AUTOS A AUTOS							PROPERTY DAMAGE (Per audident)	\$	
⊢	++	+	<u> </u>		_				\$	
	X UMBRELLA LIAB X OCCUR	1		0.4		AD.	_	EACH OCCURRENCE	\$	5,000,000
	X EXCESS LIAB CLAIMS, MAD	E		54	ΔI	MPI		AGGREGATE	\$	5,000,000
_	DED RETENTIONS WORKERS COMPENSATION	₩	_	O/		VII .		000 - 000	\$	
	AND EMPLOYERS' LIABILITY							PER STATUTE X OTH- ER		
	ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED?	NIA	1				l l	E.L. EACH ACCIDENT	\$	1,000,000
	(Mandatory in NH) If yes, describe under	1						E.L. DISEASE - EA EMPLOYE	E \$	1,000,000
L	DESCRIPTION OF OPERATIONS below							E.L. DISEASE - POLICY LIMIT	\$	1,000,000
	Property Insurance – Minimum	1								
	amount – Initial Contract Prices	1								
	DESCRIPTION OF OPERATIONS / LOCATIONS	/ VEH	CLES	Attach ACORD 101, Additional Rema	ntos Sch	edule, if more spe	ice is required)			
	(Project Name and Name of Ad	ditio	al le	eurade)						
	(Froject Haine and Haine of No			isarcus,						
L										
CE	RTIFICATE HOLDER					C	ANCELLATI	ON		
						III D ANY OF	THE ABOVE T	ESCRIBED POLICIES BE C	AMORT	ED BEFORE
1	Owner's Name)							ESCRIBED POLICIES BE C OF, NOTICE WILL BE DELIV		
	Engineer's Name)							Y PROVISIONS.		
	Other Additional Insured's Name)									

CERTIFICATE 00 62 16.13 ADDITIONAL INSUREDS (SAMPLE FORM)

POLICY NUMBER: ()	COMMI LIABILI	ERCIAL GENERAL TY		
THIS ENDORSEMENT	CHANGES THE POL	ICY, PLEASE READ I	T CAREFULLY.		
ADDIT	IONAL INSURED - OW CONTRACTORS		l.		
This endorsement modified i	nsurance provided unde	er the following:			
COMMERCIAL GEN	ERAL LIABILITY COVE	RAGE PART			
	SCHEDU	LE			
Name of Person or Organiz	zations:				
(OWNER) (BAXTER & WOODMAN, IN (Others as required by the C	,				
(If no entry appears above, ithe Declarations as applicab			ment will be shown in		
WHO IS AN INSURED (Section II) is amended to include as an insured the person or organization shown in the Schedule, but only with respect to liability arising out of "your work" for that insured by or for you.					
		Authorized Represent	tative		
	This ₋	day of	, 20		

ADDITIONAL INSUREDS CERTIFICATE 00 62 16.13

CERTIFICATE 00 62 16.16 ADDITIONAL INSUREDS (SAMPLE FORM)

POLICY NUMBER	()		PROPERTY INSURANCE			
THIS ENDORS	SEMENT CHA	ANGES THE	E POLICY, PLEASE	READ IT CAREFULLY.			
ADDITIONAL INSURED - OWNERS, LESSEES OR CONTRACTORS (FORM B)							
This endorsement m	odified insura	nce provide	d under the following	g:			
PROPERTY	INSURANCE						
		SCI	HEDULE				
Name of Person or	Organization	is:					
(OWNER) (BAXTER & WOODN (Others as required by		ct Documen	ts)				
(If no entry appears the Declarations as a				endorsement will be shown in			
WHO IS AN INSURED (Section II) is amended to include as an insured the person or organization shown in the Schedule, but only with respect to liability arising out of "your work" for that insured by or for you.							
			Authorized R	epresentative			
			This day of	•			

CERTIFICATE 00 62 16.19 ADDITIONAL INSUREDS (SAMPLE FORM)

ADDITIONAL INSUREDS CERTIFICATE 00 62 16.19

CERTIFICATE 00 62 16.21 ADDITIONAL INSUREDS (SAMPLE FORM)

POLICY NUMBER ()	EXCESS/UMBRELLA LIABILITY						
THIS ENDORSEMENT CHANGES THE POLICY, PLEASE READ IT CAREFULLY.							
ADDITIONAL INSURED - OWNERS, LESSEES OR CONTRACTORS (FORM B)							
This endorsement modified insurance provided u	inder the following:						
EXCESS/UMBRELLA LIABILITY COVER	AGE PART						
SCHE	DULE						
Name of Person or Organizations:							
(OWNER) (BAXTER & WOODMAN, INC.) (Others as required by the Contract Documents)							
(If no entry appears above, information required the Declarations as applicable to this endorseme							
WHO IS AN INSURED (Section II) is amended to shown in the Schedule, but only with respect to I by or for you.							
	Authorized Representative						
Th	is day of, 20						

CERTIFICATE 00 62 33.13

BIDDER CERTIFICATION REGARDING THE USE OF AMERICAN IRON AND STEEL PRODUCTS

Ι,	, do nereby certify that:
	Name
1.	I am(title) of the (company, partnership, etc.) and have authority to execute this certification on behalf of the firm.
2.	This firm is aware that all iron and steel products used for this project must be produced in the United States per Section 436 (a) $-$ (f) of the Consolidated Appropriations Act, 2014.
3.	This firm is aware that the use of American iron and steel products applies to all projects for the construction, alteration, maintenance, or repair of publically owned treatment works (POTW) or public water systems.
4.	This firm understands the term "iron and steel products" refers to the following products made primarily of iron or steel: lined or unlined pipes and fittings, manhole covers and other municipal castings, hydrants, tanks, flanges, pipe clamps and restraints, valves, structural steel, reinforced precast concrete, and construction materials.
5.	I am aware that this requirement applies to all portions of the project that are subcontracted.
Name	of Firm
Signat	ure
Title _	
Date _	
Corpo	rata Seal (where appropriate)

END OF CERTIFICATE 00 62 33.13

Use of American Iron and Steel

Sec. 436. (a)(1) None of the funds made available by a State water pollution control revolving fund as authorized by title VI of the Federal Water Pollution Control Act (33 U.S.C. 1381 et seq.) or made available by a drinking water treatment revolving loan fund as authorized by section 1452 of the Safe Drinking Water Act (42 U.S.C. 300j-12) shall be used for a project for the construction, alteration, maintenance, or repair of a public water system or treatment works unless all of the iron and steel products used in the project are produced in the United States.

- (2) In this section, the term `iron and steel products' means the following products made primarily of iron or steel: lined or unlined pipes and fittings, manhole covers and other municipal castings, hydrants, tanks, flanges, pipe clamps and restraints, valves, structural steel, reinforced precast concrete, and construction materials.
- (b) Subsection (a) shall not apply in any case or category of cases in which the Administrator of the Environmental Protection Agency (in this section referred to as the `Administrator') finds that--
 - (1) applying subsection (a) would be inconsistent with the public interest;
 - (2) iron and steel products are not produced in the United States in sufficient and reasonably available quantities and of a satisfactory quality; or
 - (3) inclusion of iron and steel products produced in the United States will increase the cost of the overall project by more than 25 percent.
- (c) If the Administrator receives a request for a waiver under this section, the Administrator shall make available to the public on an informal basis a copy of the request and information available to the Administrator concerning the request, and shall allow for informal public input on the request for at least 15 days prior to making a finding based on the request. The Administrator shall make the request and accompanying information available by electronic means, including on the official public Internet Web site of the Environmental Protection Agency.
- (d) This section shall be applied in a manner consistent with United States obligations under international agreements.
- (e) The Administrator may retain up to 0.25 percent of the funds appropriated in this Act for the Clean and Drinking Water State Revolving Funds for carrying out the provisions described in subsection (a)(1) for management and oversight of the requirements of this section.
- (f) This section does not apply with respect to a project if a State agency approves the engineering plans and specifications for the project, in that agency's capacity to approve such plans and specifications prior to a project requesting bids, prior to the date of the enactment of this Act.

CERTIFICATE 00 62 76.01 SWORN STATEMENT (SAMPLE FORM)

State of	of	_}ss.					
County	of	_}					
	THE AFFIANT,				BEING FIRST DULY SV	VORN, ON OATH DE	POSES AND SAYS
THAT HE/	SHE IS					THAT	HAS
CONTRAC	T WITH		(Position)			, OWNER	FOR
						,	
ON THE F	(KIND OF OLLOWING DESCR		I SAID COUNTY, TO-WIT:				
AMOUNTS	NG MATERIALS FO	OR, AND HAVE DO	D CONTRACT, THE FOLLOWING I NE OR ARE DOING LABOR ON S MATERIALS OR LABOR AS STATE	AID IMPROVEMENT. THAT TI	HERE IS DUE AND TO	BECOME DUE THEI	M, RESPECTIVELY, THE
	1	2	3	4	5	6	7
NAME AN	ID Address	KIND OF WORK	ADJUSTED TOTAL CONTRACT INCL. EXTRAS AND CREDITS	TOTAL RETAINED INCLUDING THIS APPLICATION	NET PREVIOUSLY PAID	NET AMOUNT OF THIS PAYMENT	BALANCE TO BECOME DUE (INC. RETENTION)
OTAL							
	AMOUNT OF ORIGINAL CONTRACT \$ EXTRAS TO CONTRACT \$ TOTAL CONTRACT AND EXTRAS \$ CREDITS TO CONTRACT \$ ADJUSTED TOTAL CONTRACT \$			WORK COMPLETED TO DATE LESS % RETAINED \$			
to date.	It is understood that the total amount paid to date plus the amount requested in this application shall not exceed% of the cost of work compate.						ost of work completed
	I agree to furnis	sh Waivers of Lie	en for all materials under my co				
			9	Signed			
			<u>-</u>				
	Subscribadian	d sworn to hoforn	me thisday of		osition)		
	Jubscribeu aric	a Sworn to Delole	uay 01	20		Notar	v Public

00 62 76.02 PARTIAL WAIVER OF LIEN (SAMPLE FORM)

STATE OF ILLINOIS COUNTY OF	} ss	Gty # Loan #		
TO WHOM IT MAY CONCE WHEREAS the undersigned to furnish for the premises known as	has been employed by			
<u>-</u>				
of which	and in consideration of			is the owner.
whereof is hereby acknowl the Statutes of the State improvements thereon, and) Dolla edged do(es) hereby wain of Illinois, relating to I d on the monies or other of	ve and release any a Mechanic's liens, o considerations due o	n the above describer to become from the o	right of lien under ed premises and wner, on account
of labor or services, mater above described premises. Given under			•	dersigned for the
	y ofnand	ar	, 20	tills
				SEAL
				SEAL
Note: All waivers must be f corporate seal affixed and partnership name should b	title of officer signing wa	aiver should be set	forth; if waiver is for a	

00 62 76.03 CONTRACTOR'S AFFIDAVIT (SAMPLE FORM)

STATE OF ILLINOIS COUNTY OF	}ss			
TO WHOM IT MAY CO	NCERN:			
THE undersigned, being	g duly sworn, deposes and says to			
who is the Contractor for the			work o	n the
being located at owned by				. That the
total amount of the Contra	ct including extras is \$		on which he	has received payment of
unconditionally and that following are the names having contracts or sub thereof and the amount	ior to this payment. That a there is no claim either lega s of all parties who have furro-contracts for specific portiodue or to become due to eacid work according to plans an	I or equitable to d hished material or ns of said work or ch, and that the ite	efeat the validity of sa labor, or both, for said r for material entering	id waivers. That the work and all parties into the construction
NAMES	WHAT FOR	CONTRACT PRICE	AMOUNT PAID	BALANCE DUE
TOTAL LABOR AND I	MATERIAL TO COMPLETE			
become due to any pers	r contracts for said work out son for material, labor or othe work other than above stated	r work of any kind	•	
Signed	this day of		, 20	
	_			

00 62 76.04 FINAL WAIVER OF LIEN (SAMPLE FORM)

	ss s	O: "		
		Gty # Loan #		
O WHOM IT MAY CONC VHEREAS the undersigned o furnish	ed has been employed			
or the premises known as	3			
f which		-		is the owner
f which THE undersigned, for (\$	r and in consideration	ofDollars. and	d other good and val	
	,	Dollars, ari	d other good and van	uable considerations, the receipt
whereof is hereby acknow	wledged do(es) hereby	y waive and release	any and all lien or c	uable considerations, the receipt aim or right of lien under
whereof is hereby acknown the Statutes of the Statimprovements thereon, a	wledged do(es) hereby te of Illinois, relating nd on the monies or o	y waive and release to Mechanic's lie	any and all lien or c ns, on the above of due or to become fro	uable considerations, the receipt aim or right of lien under described premises and m the owner, on account
whereof is hereby acknown the Statutes of the Statimprovements thereon, a of labor or services, mate	wledged do(es) hereby te of Illinois, relating nd on the monies or of erial, fixtures or appares.	y waive and release to Mechanic's lie ther considerations atus heretofore furr	any and all lien or cons, on the above of due or to become fromished to this date by	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the
whereof is hereby acknown the Statutes of the	wledged do(es) hereby te of Illinois, relating and on the monies or of erial, fixtures or appar es.	y waive and release to Mechanic's lie	any and all lien or cons, on the above of due or to become from his date by and seal	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the
whereof is hereby acknown the Statutes of the	wledged do(es) hereby te of Illinois, relating nd on the monies or of erial, fixtures or appares.	y waive and release to Mechanic's lie ther considerations atus heretofore furr	any and all lien or cons, on the above of due or to become fromished to this date by	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the
whereof is hereby acknown the Statutes of the	wledged do(es) hereby te of Illinois, relating and on the monies or of erial, fixtures or appar es.	y waive and release to Mechanic's lie ther considerations atus heretofore furr	any and all lien or cons, on the above of due or to become from his date by and seal	uable considerations, the receipt aim or right of lien under described premises and m the owner, on account the undersigned for the this
whereof is hereby acknown the Statutes of the Statimprovements thereon, a find the state of labor or services, mate above described premise Given under	wledged do(es) hereby te of Illinois, relating and on the monies or of erial, fixtures or appar es.	y waive and release to Mechanic's lie ther considerations atus heretofore furr	any and all lien or cons, on the above of due or to become from his date by and seal	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the this
whereof is hereby acknown the Statutes of the Statimprovements thereon, a of labor or services, mate above described premise Given under	wledged do(es) hereby te of Illinois, relating and on the monies or of erial, fixtures or appar es.	y waive and release to Mechanic's lie ther considerations atus heretofore furr	any and all lien or cons, on the above of due or to become from his date by and seal	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the this
whereof is hereby acknown the Statutes of the Statimprovements thereon, and of labor or services, materabove described premise Given under	wledged do(es) hereby te of Illinois, relating nd on the monies or of erial, fixtures or appar esh day of	y waive and release to Mechanic's lie ther considerations atus heretofore furn and aid. If waiver is for a	any and all lien or cons, on the above of due or to become from shed to this date by and seal, 20	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the this SEAL
whereof is hereby acknown the Statutes of the Statimprovements thereon, a of labor or services, mate above described premise Given under	wledged do(es) hereby te of Illinois, relating nd on the monies or of erial, fixtures or appar es. h day of e for the full amount pand title of officer signi	y waive and release to Mechanic's lie ther considerations atus heretofore furr and aid. If waiver is for a	any and all lien or cons, on the above of due or to become from shed to this date by and seal, 20	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the this SEAL SEAL ste name should be used, is for a partnership, the
whereof is hereby acknown the Statutes of the Statimprovements thereon, and of labor or services, materiabove described premise Given under	wledged do(es) hereby te of Illinois, relating nd on the monies or of erial, fixtures or appar es. h day of e for the full amount pand title of officer signi	y waive and release to Mechanic's lie ther considerations atus heretofore furr and aid. If waiver is for a	any and all lien or cons, on the above of due or to become from shed to this date by and seal, 20	uable considerations, the receipt laim or right of lien under described premises and m the owner, on account the undersigned for the this SEAL SEAL ste name should be used, is for a partnership, the

Verify that all of your Illinois Sales Tax Exemption Certificate information is correct

√ If not, contact us immediately.

✓ Do not discard - your Illinois Sales Tax Exemption Certificate is an important tax document that authorizes you to purchase tangible personal property for use or consumption tax-free.

OFFICIAL DOCUMENT State of Illinois - Department of Revenue OFFICIAL DOCUMENT **Illinois Sales Tax Exemption Certificate CITY OF JOLIET** 150 W JEFFERSON ST **JOLIET IL 60432-4148** Sales Tax Exemption Certificate Issue date: Sales Tax Exemption E99925631 02/10/2020 **Expiration date:** Organization type: **Governmental** 03/01/2025 This entity is authorized under the Retailers' Occupation Tax Act to purchase tangible personal property for use or consumption tax-free. Director OFFICIAL DOCUMENT - DO NOT DESTROY

00 62 79

STORED MATERIAL FORM

PROJECT: _				F		OOF
OWNER:						_
CONTRACTO	OR:					
Periodical Esti	mate for Partial Payment No)		<u> </u>		
For Period		to			Inclusively	
CONTRACT ITEM	DESCRIPTION		PREVIOUS VALUES	(\$) RECEIVED	(\$) USED	PRESENT VALUE (\$)

00 63 49

WORK CHANGE DIRECTIVE FORM

NO	
PROJECT:	DATE OF ISSUANCE:
OWNER:	ENGINEER: Baxter & Woodman, Inc. 8678 Ridgefield Road Crystal Lake, IL 60012
CONTRACTOR:	ENGINEER'S PROJECT NO:
You are directed to proceed promptly with the following Description:	change(s):
Purpose of Work Directive Change:	
Attachments:	
If a claim is made that the above change(s) have affected Change Order based thereon will involve one of the follows:	ed Contract Price or Contract Time, any claim for a owing methods of determining the effect of the change(s).
Method of determining change in Contract Price:	Method of determining change in Contract Time:
☐ Time and materials ☐ Unit prices ☐ Cost plus fixed fee	Contractor's recordsEngineer's recordsOther
Other Estimated increase(decrease) in Contract Price: \$ If the Change involves an increase, the estimated amount is not to be exceeded without further authorization.	Estimated increase(decrease) in Contract Time: days. If the Change involves an increase, the estimated amount is not to be exceeded without further authorization.
RECOMMENDED: BAXTER & WOODMAN, INC.	AUTHORIZED: OWNER
Ву	By
Name Title	Name Title

00 63 63

CHANGE ORDER FORM

PROJECT:		DATE OF ISSUANCE:	
OWNER:	ENGINEER:		Baxter & Woodman, Inc.
CONTRACTOR:		ENGINEER's Project No)
You are directed to make the following cha	anges in the Contrac	t Documents:	
Description:			
Purpose of Change Order:			
Attachments:			
CHANGE IN CONTRACT PRICE:		CHANGE IN CONTRACT	TIME:
Original Contract Price:	\$	Original Contract Time: Substantial Completion Completion	
Previous Change Orders: No to No	\$	Change from previous Ch	nange Orders:
Current Contract Price:	\$	Current Contract Time: Substantial Completio Completio	
Net increase/decrease of this Change Order:	\$	Net increase/decrease of	
Contract Price with this Change Order:	\$	Contract Time with this Cl Substantial Completio Completio	on:
NOTE: USE ANYTIME THE CHANGE Pursuant to 720 ILCS 5/33E-9, (1) the cir reasonably foreseeable at the time the cor signed, and (3) the change order is in the b	rcumstances said to ntract was signed, (2	o necessitate the change i 2) the change is germane to	n performance were not o the original contract as
PREPARED BY: AF	PPROVED:	ACCEPTE	D:
BAXTER & WOODMAN, INC.	Municipali	ty	Contractor
ByBy	Lean ant NI	By	In a set Nisses
Insert Name Construction Project Manager	Insert Na Insert T		Insert Name Insert Title
RECOMMENDED BY:			
Ву			
Insert Name Client Manager			

CERTIFICATE 00 65 16

CERTIFICATE OF SUBSTANTIAL COMPLETION FORM

PROJECT: ENG	GINEER'S Project No.:
CONTRACTOR: CO	NTRACT DATE:
This Certificate of Substantial Completion applies to all Work specified parts thereof: TO: [Owner]	under the Contract Documents or to the following
TO: [Owner]	
AND TO: [Contractor]	
-	
The Work to which this Certificate applies has been obse Contractor, and Engineer, and that Work is hereby declared the Contract Documents on	to be substantially complete in accordance with
A tentative list of items to be completed or corrected [follows inclusive, and failure to include an item does not alter the re Work in accordance with the Contract Documents. The items i by Contractor within calendar days of the about	sponsibility of the Contractor to complete all the n the tentative list shall be completed or corrected
The responsibilities between Owner and Contractor for securi and warranties shall be as follows:	ty, operation, safety, maintenance, heat, utilities,
RESPONSIBILITIES:	
Owner:	
Contractor:	

This Certificate does not constitute an acceptance of W is it a release of Contractor's obligation to complete the	ork not in acc	cordance with the Contract Documents nor ordance with the Contract Documents.
For each of the Foreign and Project Management of Office (Ma		00
Executed by Engineer's Project Manager and Client Ma	anager on	, 20
	BAXTER &	WOODMAN, INC.
	Ву:	
		Client Manager
	Ву:	
		Project Manager
Contractor accepts this Certificate of Substantial Comp	oletion on	, 20
	CONTRACT	OR NAME - ALL CAPS
	Ву:	
		Name Title
Owner accepts this Certificate of Substantial Completic	on on	, 20
	OWNER NA	ME - ALL CAPS
	Ву:	
		Name Title

The following documents are attached to and made a part of this Certificate:

CERTIFICATE 00 65 19

CERTIFICATE OF COMPLETION FORM

ENGINEER'S PROJECT NO.:	
PROJECT:	
CONSTRUCTION CONTRACT DATE:	
то	
AND TO	
The undersigned hereby gives notice that by Contractor under the Contract for the above provisions of the Contract and the terms and contract and the terms are the terms and contract and the terms are the terms and contract and the terms and contract and the terms are the terms and contract and the terms are the terms and the terms are the	
	BAXTER & WOODMAN, INC. CONSULTING ENGINEERS
	By:
Date:, 20	

The Notice on the front side of this paper is expressly made subject to the following terms and conditions:

- 1. Said Notice is given with the skill and care ordinarily used by members of the Engineering profession practicing under similar conditions at the same time and in the same locality.
- 2. Said Notice reflects and is an expression of the professional judgment of Engineer.
- 3. Said Notice is given as to the best of Engineer's knowledge, information and belief.
- 4. Said Notice is based entirely on the expressly limited scope of services Engineer has been employed by Owner to perform or furnish during construction of the Project (including observation of the Work) under Engineer's Agreement with Owner and under the Contract reference on the reverse hereof, and applies only to facts that are within Engineer's knowledge or could reasonably have been ascertained by Engineer as a result of carrying out the responsibilities specifically assigned to Engineer under Engineer's Agreement with Owner and the Contract referenced on the reverse hereof.
- 5. Said Notice is not a guarantee or warranty of Contractor's performance under the above-referenced Contract referenced on the reverse hereof nor an assumption of responsibility for any failure of Contractor to furnish and perform the Work thereunder in accordance with the Contract Documents.

00 65 36 WARRANTY FORM (SAMPLE FORM)

TO BE TRANSFERRED TO THE CONTRACTOR'S LETTERHEAD

Owner (see Bid Form)			
Subject:			
Dear:			
We, theguarantee for a period	of		, commencing
should any defect due to improp	per materials or w	g orkmanship develop du ood by us without	ring the period of the
This guarantee is for all work ex Section 01 61 01 of the Specific		ent separately guaranted	ed as called for under
Very truly yours,			
A B C, INC.			
:			
C:			

00 65 36.03 CONTRACTOR'S AFFIDAVIT LETTER (SAMPLE FORM)

TO BE TRANSFERRED TO THE CONTRACTOR'S LETTERHEAD

Owner (see Bid Form)	 _			_	
Subject:				_	
Dear:					
We,					
knowledge, the waivers of lien material for which a lien could be	erewith repro	esent release	es for all t	he labo	r and
Should any liens remain (Owne				•	
refund to thein discharging such liens, includi				igated t	o pay
Very truly yours,					
A B C, INC.					
:					
C:	 				
	 				

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

Prepared by

ENGINEERS JOINT CONTRACT DOCUMENTS COMMITTEE

and

Issued and Published Jointly by









PROFESSIONAL ENGINEERS IN PRIVATE PRACTICE $A\ Practice\ Division\ of\ the$ NATIONAL SOCIETY OF PROFESSIONAL ENGINEERS

Endorsed by



CONSTRUCTION SPECIFICATIONS INSTITUTE

Copyright © 2007 National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882 www.nspe.org

> American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474 www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723 www.asce.org

Associated General Contractors of America 2300 Wilson Boulevard, Suite 400, Arlington, VA 22201-3308 (703) 548-3118 www.agc.org

The copyright for this EJCDC document is owned jointly by the four EJCDC sponsoring organizations and held in trust for their benefit by NSPE.

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

		Page
Article 1 –	Definitions and Terminology	7
1.01	Defined Terms	
1.02	Terminology	
Article 2 –	Preliminary Matters	12
2.01	Delivery of Bonds and Evidence of Insurance	12
2.02	Copies of Documents	
2.03	Commencement of Contract Times; Notice to Proceed	
2.04	Starting the Work	
2.05	Before Starting Construction	
2.06	Preconstruction Conference; Designation of Authorized Representatives	
2.07	Initial Acceptance of Schedules	
Article 3 –	Contract Documents: Intent, Amending, Reuse	14
3.01	Intent	
3.02	Reference Standards	
3.03	Reporting and Resolving Discrepancies	
3.04	Amending and Supplementing Contract Documents	
3.05	Reuse of Documents	
3.06	Electronic Data	16
Article 4 –	Availability of Lands; Subsurface and Physical Conditions; Hazardous Environment	ıtal
	onditions; Reference Points	
4.01	Availability of Lands	16
4.02	Subsurface and Physical Conditions	17
4.03	Differing Subsurface or Physical Conditions	17
4.04	Underground Facilities	
4.05	Reference Points	19
4.06	Hazardous Environmental Condition at Site	20
Article 5 –	Bonds and Insurance	21
5.01	Performance, Payment, and Other Bonds	21
5.02	Licensed Sureties and Insurers	22
5.03	Certificates of Insurance	22
5.04	Contractor's Insurance	23
5.05	Owner's Liability Insurance	24
5.06	Property Insurance	24
5.07	Waiver of Rights	25
5.08	Receipt and Application of Insurance Proceeds	
5.09	Acceptance of Bonds and Insurance; Option to Replace	27
5.10	Partial Utilization, Acknowledgment of Property Insurer	27

Article 6 –	Contractor's Responsibilities	27
6.01	Supervision and Superintendence	27
6.02	Labor; Working Hours	27
6.03	Services, Materials, and Equipment	28
6.04	Progress Schedule	28
6.05	Substitutes and "Or-Equals"	28
6.06	Concerning Subcontractors, Suppliers, and Others	
6.07	Patent Fees and Royalties	
6.08	Permits	
6.09	Laws and Regulations	33
6.10	Taxes	33
6.11	Use of Site and Other Areas	33
6.12	Record Documents	34
6.13	Safety and Protection	34
6.14	Safety Representative	35
6.15	Hazard Communication Programs	35
6.16	Emergencies	
6.17	Shop Drawings and Samples	36
6.18	Continuing the Work	
6.19	Contractor's General Warranty and Guarantee	
6.20	Indemnification	
6.21	Delegation of Professional Design Services	39
Article 7 –	Other Work at the Site	40
7.01	Related Work at Site	40
7.02	Coordination	40
7.03	Legal Relationships	41
Article 8 –	Owner's Responsibilities	41
8.01	Communications to Contractor	41
8.02	Replacement of Engineer	41
8.03	Furnish Data	
8.04	Pay When Due	41
8.05	Lands and Easements; Reports and Tests	41
8.06	Insurance	42
8.07	Change Orders	42
8.08	Inspections, Tests, and Approvals	42
8.09	Limitations on Owner's Responsibilities	42
8.10	Undisclosed Hazardous Environmental Condition	42
8.11	Evidence of Financial Arrangements	42
8.12	Compliance with Safety Program	42
Article 9 –	Engineer's Status During Construction	42
9.01	Owner's Representative	42
9.02	Visits to Site	42
9.03	Project Representative	43
9.04		
9.05	Rejecting Defective Work	43
9.06	Shop Drawings, Change Orders and Payments	44

9.07	Determinations for Unit Price Work	44
9.08	Decisions on Requirements of Contract Documents and Acceptability of Work	44
9.09	Limitations on Engineer's Authority and Responsibilities	
9.10	Compliance with Safety Program	45
Article 10 –	Changes in the Work; Claims	45
	Authorized Changes in the Work	
10.02	Unauthorized Changes in the Work	45
10.03	Execution of Change Orders	46
10.04	Notification to Surety	46
10.05	Claims	46
Article 11 –	Cost of the Work; Allowances; Unit Price Work	47
11.01	Cost of the Work	47
11.02	Allowances	50
11.03	Unit Price Work	50
Article 12 –	Change of Contract Price; Change of Contract Times	51
12.01	Change of Contract Price	51
12.02	Change of Contract Times	52
12.03	Delays	52
Article 13 –	Tests and Inspections; Correction, Removal or Acceptance of Defective Work	53
13.01	Notice of Defects	53
13.02	Access to Work	53
13.03	Tests and Inspections	53
13.04	Uncovering Work	54
13.05	Owner May Stop the Work	54
13.06	Correction or Removal of Defective Work	55
13.07	Correction Period	55
13.08	Acceptance of Defective Work	56
13.09	Owner May Correct Defective Work	56
Article 14 –	Payments to Contractor and Completion	57
14.01	Schedule of Values	57
14.02	Progress Payments	57
14.03	Contractor's Warranty of Title	59
14.04	Substantial Completion	60
14.05	Partial Utilization	60
14.06	Final Inspection	61
14.07	Final Payment	61
	Final Completion Delayed	
14.09	Waiver of Claims	63
	Suspension of Work and Termination	
15.01	Owner May Suspend Work	63
15.02	Owner May Terminate for Cause	63
15.03	Owner May Terminate For Convenience	64
15.04	Contractor May Stop Work or Terminate	65

Article 16 –	- Dispute Resolution	65
	Methods and Procedures	
Article 17 –	- Miscellaneous	66
17.01	Giving Notice	66
17.02	Computation of Times	66
17.03	Cumulative Remedies	66
17.04	Survival of Obligations	66
17.05	Controlling Law	66
17.06	Headings	66

ARTICLE 1 – DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents and printed with initial capital letters, the terms listed below will have the meanings indicated which are applicable to both the singular and plural thereof. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
 - 1. *Addenda*—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
 - 2. *Agreement*—The written instrument which is evidence of the agreement between Owner and Contractor covering the Work.
 - 3. Application for Payment—The form acceptable to Engineer which is to be used by Contractor during the course of the Work in requesting progress or final payments and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
 - 4. *Asbestos*—Any material that contains more than one percent asbestos and is friable or is releasing asbestos fibers into the air above current action levels established by the United States Occupational Safety and Health Administration.
 - 5. *Bid*—The offer or proposal of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
 - 6. *Bidder*—The individual or entity who submits a Bid directly to Owner.
 - 7. *Bidding Documents*—The Bidding Requirements and the proposed Contract Documents (including all Addenda).
 - 8. *Bidding Requirements*—The advertisement or invitation to bid, Instructions to Bidders, Bid security of acceptable form, if any, and the Bid Form with any supplements.
 - Change Order—A document recommended by Engineer which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, issued on or after the Effective Date of the Agreement.
 - 10. *Claim*—A demand or assertion by Owner or Contractor seeking an adjustment of Contract Price or Contract Times, or both, or other relief with respect to the terms of the Contract. A demand for money or services by a third party is not a Claim.
 - 11. *Contract*—The entire and integrated written agreement between the Owner and Contractor concerning the Work. The Contract supersedes prior negotiations, representations, or agreements, whether written or oral.
 - 12. *Contract Documents*—Those items so designated in the Agreement. Only printed or hard copies of the items listed in the Agreement are Contract Documents. Approved Shop Drawings, other Contractor submittals, and the reports and drawings of subsurface and physical conditions are not Contract Documents.

- 13. *Contract Price*—The moneys payable by Owner to Contractor for completion of the Work in accordance with the Contract Documents as stated in the Agreement (subject to the provisions of Paragraph 11.03 in the case of Unit Price Work).
- 14. *Contract Times*—The number of days or the dates stated in the Agreement to: (i) achieve Milestones, if any; (ii) achieve Substantial Completion; and (iii) complete the Work so that it is ready for final payment as evidenced by Engineer's written recommendation of final payment.
- 15. *Contractor*—The individual or entity with whom Owner has entered into the Agreement.
- 16. Cost of the Work—See Paragraph 11.01 for definition.
- 17. *Drawings*—That part of the Contract Documents prepared or approved by Engineer which graphically shows the scope, extent, and character of the Work to be performed by Contractor. Shop Drawings and other Contractor submittals are not Drawings as so defined.
- 18. *Effective Date of the Agreement*—The date indicated in the Agreement on which it becomes effective, but if no such date is indicated, it means the date on which the Agreement is signed and delivered by the last of the two parties to sign and deliver.
- 19. *Engineer*—The individual or entity named as such in the Agreement.
- 20. *Field Order*—A written order issued by Engineer which requires minor changes in the Work but which does not involve a change in the Contract Price or the Contract Times.
- 21. General Requirements—Sections of Division 1 of the Specifications.
- 22. *Hazardous Environmental Condition*—The presence at the Site of Asbestos, PCBs, Petroleum, Hazardous Waste, or Radioactive Material in such quantities or circumstances that may present a substantial danger to persons or property exposed thereto.
- 23. *Hazardous Waste*—The term Hazardous Waste shall have the meaning provided in Section 1004 of the Solid Waste Disposal Act (42 USC Section 6903) as amended from time to time.
- 24. Laws and Regulations; Laws or Regulations—Any and all applicable laws, rules, regulations, ordinances, codes, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 25. *Liens*—Charges, security interests, or encumbrances upon Project funds, real property, or personal property.
- 26. *Milestone*—A principal event specified in the Contract Documents relating to an intermediate completion date or time prior to Substantial Completion of all the Work.
- 27. *Notice of Award*—The written notice by Owner to the Successful Bidder stating that upon timely compliance by the Successful Bidder with the conditions precedent listed therein, Owner will sign and deliver the Agreement.

- 28. *Notice to Proceed*—A written notice given by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work under the Contract Documents.
- 29. *Owner*—The individual or entity with whom Contractor has entered into the Agreement and for whom the Work is to be performed.
- 30. *PCBs*—Polychlorinated biphenyls.
- 31. *Petroleum*—Petroleum, including crude oil or any fraction thereof which is liquid at standard conditions of temperature and pressure (60 degrees Fahrenheit and 14.7 pounds per square inch absolute), such as oil, petroleum, fuel oil, oil sludge, oil refuse, gasoline, kerosene, and oil mixed with other non-Hazardous Waste and crude oils.
- 32. *Progress Schedule*—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising the Contractor's plan to accomplish the Work within the Contract Times.
- 33. *Project*—The total construction of which the Work to be performed under the Contract Documents may be the whole, or a part.
- 34. *Project Manual*—The bound documentary information prepared for bidding and constructing the Work. A listing of the contents of the Project Manual, which may be bound in one or more volumes, is contained in the table(s) of contents.
- 35. *Radioactive Material*—Source, special nuclear, or byproduct material as defined by the Atomic Energy Act of 1954 (42 USC Section 2011 et seq.) as amended from time to time.
- 36. *Resident Project Representative*—The authorized representative of Engineer who may be assigned to the Site or any part thereof.
- 37. *Samples*—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and which establish the standards by which such portion of the Work will be judged.
- 38. *Schedule of Submittals*—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements to support scheduled performance of related construction activities.
- 39. *Schedule of Values*—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 40. *Shop Drawings*—All drawings, diagrams, illustrations, schedules, and other data or information which are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work.
- 41. *Site*—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements for access thereto, and such other lands furnished by Owner which are designated for the use of Contractor.

- 42. *Specifications*—That part of the Contract Documents consisting of written requirements for materials, equipment, systems, standards and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable thereto.
- 43. *Subcontractor*—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work at the Site.
- 44. Substantial Completion—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion thereof.
- 45. Successful Bidder—The Bidder submitting a responsive Bid to whom Owner makes an award.
- 46. *Supplementary Conditions*—That part of the Contract Documents which amends or supplements these General Conditions.
- 47. *Supplier*—A manufacturer, fabricator, supplier, distributor, materialman, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or Subcontractor.
- 48. *Underground Facilities*—All underground pipelines, conduits, ducts, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or attachments, and any encasements containing such facilities, including those that convey electricity, gases, steam, liquid petroleum products, telephone or other communications, cable television, water, wastewater, storm water, other liquids or chemicals, or traffic or other control systems.
- 49. *Unit Price Work*—Work to be paid for on the basis of unit prices.
- 50. Work—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction, and furnishing, installing, and incorporating all materials and equipment into such construction, all as required by the Contract Documents.
- 51. Work Change Directive—A written statement to Contractor issued on or after the Effective Date of the Agreement and signed by Owner and recommended by Engineer ordering an addition, deletion, or revision in the Work, or responding to differing or unforeseen subsurface or physical conditions under which the Work is to be performed or to emergencies. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the change ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order following negotiations by the parties as to its effect, if any, on the Contract Price or Contract Times.

1.02 *Terminology*

A. The words and terms discussed in Paragraph 1.02.B through F are not defined but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.

B. Intent of Certain Terms or Adjectives:

1. The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Paragraph 9.09 or any other provision of the Contract Documents.

C. Day:

1. The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.

D. Defective:

- 1. The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
 - a. does not conform to the Contract Documents; or
 - b. does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
 - c. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 14.04 or 14.05).

E. Furnish, Install, Perform, Provide:

- 1. The word "furnish," when used in connection with services, materials, or equipment, shall mean to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
- 2. The word "install," when used in connection with services, materials, or equipment, shall mean to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
- 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, shall mean to furnish and install said services, materials, or equipment complete and ready for intended use.
- 4. When "furnish," "install," "perform," or "provide" is not used in connection with services, materials, or equipment in a context clearly requiring an obligation of Contractor, "provide" is implied.

F. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

ARTICLE 2 – PRELIMINARY MATTERS

- 2.01 Delivery of Bonds and Evidence of Insurance
 - A. When Contractor delivers the executed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner such bonds as Contractor may be required to furnish.
 - B. *Evidence of Insurance:* Before any Work at the Site is started, Contractor and Owner shall each deliver to the other, with copies to each additional insured identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance which either of them or any additional insured may reasonably request) which Contractor and Owner respectively are required to purchase and maintain in accordance with Article 5.

2.02 Copies of Documents

A. Owner shall furnish to Contractor up to ten printed or hard copies of the Drawings and Project Manual. Additional copies will be furnished upon request at the cost of reproduction.

2.03 Commencement of Contract Times; Notice to Proceed

A. The Contract Times will commence to run on the thirtieth day after the Effective Date of the Agreement or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Agreement. In no event will the Contract Times commence to run later than the sixtieth day after the day of Bid opening or the thirtieth day after the Effective Date of the Agreement, whichever date is earlier.

2.04 Starting the Work

A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work shall be done at the Site prior to the date on which the Contract Times commence to run.

2.05 Before Starting Construction

- A. *Preliminary Schedules:* Within 10 days after the Effective Date of the Agreement (unless otherwise specified in the General Requirements), Contractor shall submit to Engineer for timely review:
 - 1. a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract Documents;
 - 2. a preliminary Schedule of Submittals; and
 - 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work into component parts in sufficient detail to serve as the basis for progress payments during

performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.

2.06 Preconstruction Conference; Designation of Authorized Representatives

- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work and to discuss the schedules referred to in Paragraph 2.05.A, procedures for handling Shop Drawings and other submittals, processing Applications for Payment, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit instructions, receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.

2.07 Initial Acceptance of Schedules

- A. At least 10 days before submission of the first Application for Payment a conference attended by Contractor, Engineer, and others as appropriate will be held to review for acceptability to Engineer as provided below the schedules submitted in accordance with Paragraph 2.05.A. Contractor shall have an additional 10 days to make corrections and adjustments and to complete and resubmit the schedules. No progress payment shall be made to Contractor until acceptable schedules are submitted to Engineer.
 - 1. The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.
 - 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
 - 3. Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to component parts of the Work.

ARTICLE 3 – CONTRACT DOCUMENTS: INTENT, AMENDING, REUSE

3.01 Intent

- A. The Contract Documents are complementary; what is required by one is as binding as if required by all.
- B. It is the intent of the Contract Documents to describe a functionally complete project (or part thereof) to be constructed in accordance with the Contract Documents. Any labor, documentation, services, materials, or equipment that reasonably may be inferred from the Contract Documents or from prevailing custom or trade usage as being required to produce the indicated result will be provided whether or not specifically called for, at no additional cost to Owner.

C. Clarifications and interpretations of the Contract Documents shall be issued by Engineer as provided in Article 9.

3.02 Reference Standards

A. Standards, Specifications, Codes, Laws, and Regulations

- 1. Reference to standards, specifications, manuals, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, shall mean the standard, specification, manual, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Agreement if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
- 2. No provision of any such standard, specification, manual, or code, or any instruction of a Supplier, shall be effective to change the duties or responsibilities of Owner, Contractor, or Engineer, or any of their subcontractors, consultants, agents, or employees, from those set forth in the Contract Documents. No such provision or instruction shall be effective to assign to Owner, Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, any duty or authority to supervise or direct the performance of the Work or any duty or authority to undertake responsibility inconsistent with the provisions of the Contract Documents.

3.03 Reporting and Resolving Discrepancies

A. Reporting Discrepancies:

- 1. Contractor's Review of Contract Documents Before Starting Work: Before undertaking each part of the Work, Contractor shall carefully study and compare the Contract Documents and check and verify pertinent figures therein and all applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy which Contractor discovers, or has actual knowledge of, and shall obtain a written interpretation or clarification from Engineer before proceeding with any Work affected thereby.
- 2. Contractor's Review of Contract Documents During Performance of Work: If, during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) any standard, specification, manual, or code, or (c) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 6.16.A) until an amendment or supplement to the Contract Documents has been issued by one of the methods indicated in Paragraph 3.04.
- 3. Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.

B. Resolving Discrepancies:

- 1. Except as may be otherwise specifically stated in the Contract Documents, the provisions of the Contract Documents shall take precedence in resolving any conflict, error, ambiguity, or discrepancy between the provisions of the Contract Documents and:
 - a. the provisions of any standard, specification, manual, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference in the Contract Documents); or
 - b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).

3.04 Amending and Supplementing Contract Documents

- A. The Contract Documents may be amended to provide for additions, deletions, and revisions in the Work or to modify the terms and conditions thereof by either a Change Order or a Work Change Directive.
- B. The requirements of the Contract Documents may be supplemented, and minor variations and deviations in the Work may be authorized, by one or more of the following ways:
 - 1. A Field Order;
 - 2. Engineer's approval of a Shop Drawing or Sample (subject to the provisions of Paragraph 6.17.D.3); or
 - 3. Engineer's written interpretation or clarification.

3.05 Reuse of Documents

- A. Contractor and any Subcontractor or Supplier shall not:
 - 1. have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media editions; or
 - reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein shall preclude Contractor from retaining copies of the Contract Documents for record purposes.

3.06 Electronic Data

A. Unless otherwise stated in the Supplementary Conditions, the data furnished by Owner or Engineer to Contractor, or by Contractor to Owner or Engineer, that may be relied upon are limited to the printed copies (also known as hard copies). Files in electronic media format of text, data, graphics, or other types are furnished only for the convenience of the receiving party. Any conclusion or information obtained or derived from such electronic files will be at the user's sole

- risk. If there is a discrepancy between the electronic files and the hard copies, the hard copies govern.
- B. Because data stored in electronic media format can deteriorate or be modified inadvertently or otherwise without authorization of the data's creator, the party receiving electronic files agrees that it will perform acceptance tests or procedures within 60 days, after which the receiving party shall be deemed to have accepted the data thus transferred. Any errors detected within the 60-day acceptance period will be corrected by the transferring party.
- C. When transferring documents in electronic media format, the transferring party makes no representations as to long term compatibility, usability, or readability of documents resulting from the use of software application packages, operating systems, or computer hardware differing from those used by the data's creator.

ARTICLE 4 – AVAILABILITY OF LANDS; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS; REFERENCE POINTS

4.01 Availability of Lands

- A. Owner shall furnish the Site. Owner shall notify Contractor of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work. Owner will obtain in a timely manner and pay for easements for permanent structures or permanent changes in existing facilities. If Contractor and Owner are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times, or both, as a result of any delay in Owner's furnishing the Site or a part thereof, Contractor may make a Claim therefor as provided in Paragraph 10.05.
- B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which the Work is to be performed and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
- C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.

4.02 Subsurface and Physical Conditions

- A. *Reports and Drawings:* The Supplementary Conditions identify:
 - 1. those reports known to Owner of explorations and tests of subsurface conditions at or contiguous to the Site; and
 - 2. those drawings known to Owner of physical conditions relating to existing surface or subsurface structures at the Site (except Underground Facilities).
- B. Limited Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the "technical data" contained in such reports and drawings, but such reports and drawings are not Contract Documents. Such "technical data" is identified in the Supplementary Conditions. Except for such reliance on such "technical data," Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors with respect to:

- 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto; or
- 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
- 3. any Contractor interpretation of or conclusion drawn from any "technical data" or any such other data, interpretations, opinions, or information.

4.03 Differing Subsurface or Physical Conditions

- A. *Notice:* If Contractor believes that any subsurface or physical condition that is uncovered or revealed either:
 - 1. is of such a nature as to establish that any "technical data" on which Contractor is entitled to rely as provided in Paragraph 4.02 is materially inaccurate; or
 - 2. is of such a nature as to require a change in the Contract Documents; or
 - 3. differs materially from that shown or indicated in the Contract Documents; or
 - 4. is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 6.16.A), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except as aforesaid) until receipt of written order to do so.

- B. *Engineer's Review*: After receipt of written notice as required by Paragraph 4.03.A, Engineer will promptly review the pertinent condition, determine the necessity of Owner's obtaining additional exploration or tests with respect thereto, and advise Owner in writing (with a copy to Contractor) of Engineer's findings and conclusions.
- C. Possible Price and Times Adjustments:
 - 1. The Contract Price or the Contract Times, or both, will be equitably adjusted to the extent that the existence of such differing subsurface or physical condition causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:
 - a. such condition must meet any one or more of the categories described in Paragraph 4.03.A; and
 - b. with respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraphs 9.07 and 11.03.

- 2. Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times if:
 - a. Contractor knew of the existence of such conditions at the time Contractor made a final commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract; or
 - b. the existence of such condition could reasonably have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and contiguous areas required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such final commitment; or
 - c. Contractor failed to give the written notice as required by Paragraph 4.03.A.
- 3. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times, or both, a Claim may be made therefor as provided in Paragraph 10.05. However, neither Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors shall be liable to Contractor for any claims, costs, losses, or damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Contractor on or in connection with any other project or anticipated project.

4.04 *Underground Facilities*

- A. *Shown or Indicated:* The information and data shown or indicated in the Contract Documents with respect to existing Underground Facilities at or contiguous to the Site is based on information and data furnished to Owner or Engineer by the owners of such Underground Facilities, including Owner, or by others. Unless it is otherwise expressly provided in the Supplementary Conditions:
 - 1. Owner and Engineer shall not be responsible for the accuracy or completeness of any such information or data provided by others; and
 - 2. the cost of all of the following will be included in the Contract Price, and Contractor shall have full responsibility for:
 - a. reviewing and checking all such information and data;
 - b. locating all Underground Facilities shown or indicated in the Contract Documents;
 - c. coordination of the Work with the owners of such Underground Facilities, including Owner, during construction; and
 - d. the safety and protection of all such Underground Facilities and repairing any damage thereto resulting from the Work.

B. Not Shown or Indicated:

1. If an Underground Facility is uncovered or revealed at or contiguous to the Site which was not shown or indicated, or not shown or indicated with reasonable accuracy in the Contract Documents, Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith

(except in an emergency as required by Paragraph 6.16.A), identify the owner of such Underground Facility and give written notice to that owner and to Owner and Engineer. Engineer will promptly review the Underground Facility and determine the extent, if any, to which a change is required in the Contract Documents to reflect and document the consequences of the existence or location of the Underground Facility. During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.

2. If Engineer concludes that a change in the Contract Documents is required, a Work Change Directive or a Change Order will be issued to reflect and document such consequences. An equitable adjustment shall be made in the Contract Price or Contract Times, or both, to the extent that they are attributable to the existence or location of any Underground Facility that was not shown or indicated or not shown or indicated with reasonable accuracy in the Contract Documents and that Contractor did not know of and could not reasonably have been expected to be aware of or to have anticipated. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment in Contract Price or Contract Times, Owner or Contractor may make a Claim therefor as provided in Paragraph 10.05.

4.05 Reference Points

A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

4.06 Hazardous Environmental Condition at Site

- A. *Reports and Drawings:* The Supplementary Conditions identify those reports and drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at the Site.
- B. Limited Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the "technical data" contained in such reports and drawings, but such reports and drawings are not Contract Documents. Such "technical data" is identified in the Supplementary Conditions. Except for such reliance on such "technical data," Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors with respect to:
 - the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures of construction to be employed by Contractor and safety precautions and programs incident thereto; or
 - 2. other data, interpretations, opinions and information contained in such reports or shown or indicated in such drawings; or

- 3. any Contractor interpretation of or conclusion drawn from any "technical data" or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for any Hazardous Environmental Condition uncovered or revealed at the Site which was not shown or indicated in Drawings or Specifications or identified in the Contract Documents to be within the scope of the Work. Contractor shall be responsible for a Hazardous Environmental Condition created with any materials brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible.
- D. If Contractor encounters a Hazardous Environmental Condition or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, Contractor shall immediately: (i) secure or otherwise isolate such condition; (ii) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 6.16.A); and (iii) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 4.06.E.
- E. Contractor shall not be required to resume Work in connection with such condition or in any affected area until after Owner has obtained any required permits related thereto and delivered written notice to Contractor: (i) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work; or (ii) specifying any special conditions under which such Work may be resumed safely. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, or both, as a result of such Work stoppage or such special conditions under which Work is agreed to be resumed by Contractor, either party may make a Claim therefor as provided in Paragraph 10.05.
- F. If after receipt of such written notice Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of an adjustment in Contract Price or Contract Times as a result of deleting such portion of the Work, then either party may make a Claim therefor as provided in Paragraph 10.05. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 7.
- G. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition: (i) was not shown or indicated in the Drawings or Specifications or identified in the Contract Documents to be included within the scope of the Work, and (ii) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 4.06.G shall obligate Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.

- H. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 4.06.H shall obligate Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- I. The provisions of Paragraphs 4.02, 4.03, and 4.04 do not apply to a Hazardous Environmental Condition uncovered or revealed at the Site.

ARTICLE 5 – BONDS AND INSURANCE

- 5.01 Performance, Payment, and Other Bonds
 - A. Contractor shall furnish performance and payment bonds, each in an amount at least equal to the Contract Price as security for the faithful performance and payment of all of Contractor's obligations under the Contract Documents. These bonds shall remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 13.07, whichever is later, except as provided otherwise by Laws or Regulations or by the Contract Documents. Contractor shall also furnish such other bonds as are required by the Contract Documents.
 - B. All bonds shall be in the form prescribed by the Contract Documents except as provided otherwise by Laws or Regulations, and shall be executed by such sureties as are named in the list of "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Circular 570 (amended) by the Financial Management Service, Surety Bond Branch, U.S. Department of the Treasury. All bonds signed by an agent or attorney-in-fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority shall show that it is effective on the date the agent or attorney-in-fact signed each bond.
 - C. If the surety on any bond furnished by Contractor is declared bankrupt or becomes insolvent or its right to do business is terminated in any state where any part of the Project is located or it ceases to meet the requirements of Paragraph 5.01.B, Contractor shall promptly notify Owner and Engineer and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which shall comply with the requirements of Paragraphs 5.01.B and 5.02.

5.02 Licensed Sureties and Insurers

A. All bonds and insurance required by the Contract Documents to be purchased and maintained by Owner or Contractor shall be obtained from surety or insurance companies that are duly licensed or authorized in the jurisdiction in which the Project is located to issue bonds or insurance policies for the limits and coverages so required. Such surety and insurance companies shall also meet such additional requirements and qualifications as may be provided in the Supplementary Conditions.

5.03 Certificates of Insurance

- A. Contractor shall deliver to Owner, with copies to each additional insured and loss payee identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance requested by Owner or any other additional insured) which Contractor is required to purchase and maintain
- B. Owner shall deliver to Contractor, with copies to each additional insured and loss payee identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance requested by Contractor or any other additional insured) which Owner is required to purchase and maintain.
- C. Failure of Owner to demand such certificates or other evidence of Contractor's full compliance with these insurance requirements or failure of Owner to identify a deficiency in compliance from the evidence provided shall not be construed as a waiver of Contractor's obligation to maintain such insurance.
- D. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor.
- E. The insurance and insurance limits required herein shall not be deemed as a limitation on Contractor's liability under the indemnities granted to Owner in the Contract Documents.

5.04 Contractor's Insurance

- A. Contractor shall purchase and maintain such insurance as is appropriate for the Work being performed and as will provide protection from claims set forth below which may arise out of or result from Contractor's performance of the Work and Contractor's other obligations under the Contract Documents, whether it is to be performed by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable:
 - 1. claims under workers' compensation, disability benefits, and other similar employee benefit acts;
 - 2. claims for damages because of bodily injury, occupational sickness or disease, or death of Contractor's employees;
 - 3. claims for damages because of bodily injury, sickness or disease, or death of any person other than Contractor's employees;
 - 4. claims for damages insured by reasonably available personal injury liability coverage which are sustained:
 - a. by any person as a result of an offense directly or indirectly related to the employment of such person by Contractor, or
 - b. by any other person for any other reason;
 - 5. claims for damages, other than to the Work itself, because of injury to or destruction of tangible property wherever located, including loss of use resulting therefrom; and

6. claims for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance or use of any motor vehicle.

B. The policies of insurance required by this Paragraph 5.04 shall:

- 1. with respect to insurance required by Paragraphs 5.04.A.3 through 5.04.A.6 inclusive, be written on an occurrence basis, include as additional insureds (subject to any customary exclusion regarding professional liability) Owner and Engineer, and any other individuals or entities identified in the Supplementary Conditions, all of whom shall be listed as additional insureds, and include coverage for the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of all such additional insureds, and the insurance afforded to these additional insureds shall provide primary coverage for all claims covered thereby;
- include at least the specific coverages and be written for not less than the limits of liability provided in the Supplementary Conditions or required by Laws or Regulations, whichever is greater;
- 3. include contractual liability insurance covering Contractor's indemnity obligations under Paragraphs 6.11 and 6.20;
- 4. contain a provision or endorsement that the coverage afforded will not be canceled, materially changed or renewal refused until at least 30 days prior written notice has been given to Owner and Contractor and to each other additional insured identified in the Supplementary Conditions to whom a certificate of insurance has been issued (and the certificates of insurance furnished by the Contractor pursuant to Paragraph 5.03 will so provide);
- 5. remain in effect at least until final payment and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work in accordance with Paragraph 13.07; and
- 6. include completed operations coverage:
 - a. Such insurance shall remain in effect for two years after final payment.
 - b. Contractor shall furnish Owner and each other additional insured identified in the Supplementary Conditions, to whom a certificate of insurance has been issued, evidence satisfactory to Owner and any such additional insured of continuation of such insurance at final payment and one year thereafter.

5.05 Owner's Liability Insurance

A. In addition to the insurance required to be provided by Contractor under Paragraph 5.04, Owner, at Owner's option, may purchase and maintain at Owner's expense Owner's own liability insurance as will protect Owner against claims which may arise from operations under the Contract Documents.

5.06 Property Insurance

A. Unless otherwise provided in the Supplementary Conditions, Owner shall purchase and maintain property insurance upon the Work at the Site in the amount of the full replacement cost thereof

(subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). This insurance shall:

- 1. include the interests of Owner, Contractor, Subcontractors, and Engineer, and any other individuals or entities identified in the Supplementary Conditions, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, each of whom is deemed to have an insurable interest and shall be listed as a loss payee;
- 2. be written on a Builder's Risk "all-risk" policy form that shall at least include insurance for physical loss or damage to the Work, temporary buildings, falsework, and materials and equipment in transit, and shall insure against at least the following perils or causes of loss: fire, lightning, extended coverage, theft, vandalism and malicious mischief, earthquake, collapse, debris removal, demolition occasioned by enforcement of Laws and Regulations, water damage (other than that caused by flood), and such other perils or causes of loss as may be specifically required by the Supplementary Conditions.
- 3. include expenses incurred in the repair or replacement of any insured property (including but not limited to fees and charges of engineers and architects);
- 4. cover materials and equipment stored at the Site or at another location that was agreed to in writing by Owner prior to being incorporated in the Work, provided that such materials and equipment have been included in an Application for Payment recommended by Engineer;
- 5. allow for partial utilization of the Work by Owner;
- 6. include testing and startup; and
- 7. be maintained in effect until final payment is made unless otherwise agreed to in writing by Owner, Contractor, and Engineer with 30 days written notice to each other loss payee to whom a certificate of insurance has been issued.
- B. Owner shall purchase and maintain such equipment breakdown insurance or additional property insurance as may be required by the Supplementary Conditions or Laws and Regulations which will include the interests of Owner, Contractor, Subcontractors, and Engineer, and any other individuals or entities identified in the Supplementary Conditions, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, each of whom is deemed to have an insurable interest and shall be listed as a loss payee.
- C. All the policies of insurance (and the certificates or other evidence thereof) required to be purchased and maintained in accordance with this Paragraph 5.06 will contain a provision or endorsement that the coverage afforded will not be canceled or materially changed or renewal refused until at least 30 days prior written notice has been given to Owner and Contractor and to each other loss payee to whom a certificate of insurance has been issued and will contain waiver provisions in accordance with Paragraph 5.07.
- D. Owner shall not be responsible for purchasing and maintaining any property insurance specified in this Paragraph 5.06 to protect the interests of Contractor, Subcontractors, or others in the Work to the extent of any deductible amounts that are identified in the Supplementary Conditions. The risk of loss within such identified deductible amount will be borne by Contractor, Subcontractors,

- or others suffering any such loss, and if any of them wishes property insurance coverage within the limits of such amounts, each may purchase and maintain it at the purchaser's own expense.
- E. If Contractor requests in writing that other special insurance be included in the property insurance policies provided under this Paragraph 5.06, Owner shall, if possible, include such insurance, and the cost thereof will be charged to Contractor by appropriate Change Order. Prior to commencement of the Work at the Site, Owner shall in writing advise Contractor whether or not such other insurance has been procured by Owner.

5.07 Waiver of Rights

- A. Owner and Contractor intend that all policies purchased in accordance with Paragraph 5.06 will protect Owner, Contractor, Subcontractors, and Engineer, and all other individuals or entities identified in the Supplementary Conditions as loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) in such policies and will provide primary coverage for all losses and damages caused by the perils or causes of loss covered thereby. All such policies shall contain provisions to the effect that in the event of payment of any loss or damage the insurers will have no rights of recovery against any of the insureds or loss payees thereunder. Owner and Contractor waive all rights against each other and their respective officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them for all losses and damages caused by, arising out of or resulting from any of the perils or causes of loss covered by such policies and any other property insurance applicable to the Work; and, in addition, waive all such rights against Subcontractors and Engineer, and all other individuals or entities identified in the Supplementary Conditions as loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) under such policies for losses and damages so caused. None of the above waivers shall extend to the rights that any party making such waiver may have to the proceeds of insurance held by Owner as trustee or otherwise payable under any policy so issued.
- B. Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them for:
 - 1. loss due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other perils whether or not insured by Owner; and
 - 2. loss or damage to the completed Project or part thereof caused by, arising out of, or resulting from fire or other insured peril or cause of loss covered by any property insurance maintained on the completed Project or part thereof by Owner during partial utilization pursuant to Paragraph 14.05, after Substantial Completion pursuant to Paragraph 14.04, or after final payment pursuant to Paragraph 14.07.
- C. Any insurance policy maintained by Owner covering any loss, damage or consequential loss referred to in Paragraph 5.07.B shall contain provisions to the effect that in the event of payment of any such loss, damage, or consequential loss, the insurers will have no rights of recovery against Contractor, Subcontractors, or Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them.

- A. Any insured loss under the policies of insurance required by Paragraph 5.06 will be adjusted with Owner and made payable to Owner as fiduciary for the loss payees, as their interests may appear, subject to the requirements of any applicable mortgage clause and of Paragraph 5.08.B. Owner shall deposit in a separate account any money so received and shall distribute it in accordance with such agreement as the parties in interest may reach. If no other special agreement is reached, the damaged Work shall be repaired or replaced, the moneys so received applied on account thereof, and the Work and the cost thereof covered by an appropriate Change Order.
- B. Owner as fiduciary shall have power to adjust and settle any loss with the insurers unless one of the parties in interest shall object in writing within 15 days after the occurrence of loss to Owner's exercise of this power. If such objection be made, Owner as fiduciary shall make settlement with the insurers in accordance with such agreement as the parties in interest may reach. If no such agreement among the parties in interest is reached, Owner as fiduciary shall adjust and settle the loss with the insurers and, if required in writing by any party in interest, Owner as fiduciary shall give bond for the proper performance of such duties.

5.09 Acceptance of Bonds and Insurance; Option to Replace

A. If either Owner or Contractor has any objection to the coverage afforded by or other provisions of the bonds or insurance required to be purchased and maintained by the other party in accordance with Article 5 on the basis of non-conformance with the Contract Documents, the objecting party shall so notify the other party in writing within 10 days after receipt of the certificates (or other evidence requested) required by Paragraph 2.01.B. Owner and Contractor shall each provide to the other such additional information in respect of insurance provided as the other may reasonably request. If either party does not purchase or maintain all of the bonds and insurance required of such party by the Contract Documents, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage. Without prejudice to any other right or remedy, the other party may elect to obtain equivalent bonds or insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and a Change Order shall be issued to adjust the Contract Price accordingly.

5.10 Partial Utilization, Acknowledgment of Property Insurer

A. If Owner finds it necessary to occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work as provided in Paragraph 14.05, no such use or occupancy shall commence before the insurers providing the property insurance pursuant to Paragraph 5.06 have acknowledged notice thereof and in writing effected any changes in coverage necessitated thereby. The insurers providing the property insurance shall consent by endorsement on the policy or policies, but the property insurance shall not be canceled or permitted to lapse on account of any such partial use or occupancy.

ARTICLE 6 – CONTRACTOR'S RESPONSIBILITIES

6.01 Supervision and Superintendence

A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work

in accordance with the Contract Documents. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction. Contractor shall not be responsible for the negligence of Owner or Engineer in the design or specification of a specific means, method, technique, sequence, or procedure of construction which is shown or indicated in and expressly required by the Contract Documents.

B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who shall not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.

6.02 Labor; Working Hours

- A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall at all times maintain good discipline and order at the Site.
- B. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site shall be performed during regular working hours. Contractor will not permit the performance of Work on a Saturday, Sunday, or any legal holiday without Owner's written consent (which will not be unreasonably withheld) given after prior written notice to Engineer.

6.03 Services, Materials, and Equipment

- A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start-up, and completion of the Work.
- B. All materials and equipment incorporated into the Work shall be as specified or, if not specified, shall be of good quality and new, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications shall expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
- C. All materials and equipment shall be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.

6.04 Progress Schedule

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.07 as it may be adjusted from time to time as provided below.
 - 1. Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.07) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times. Such adjustments will comply with any provisions of the General Requirements applicable thereto.

2. Proposed adjustments in the Progress Schedule that will change the Contract Times shall be submitted in accordance with the requirements of Article 12. Adjustments in Contract Times may only be made by a Change Order.

6.05 Substitutes and "Or-Equals"

- A. Whenever an item of material or equipment is specified or described in the Contract Documents by using the name of a proprietary item or the name of a particular Supplier, the specification or description is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or-equal" item or no substitution is permitted, other items of material or equipment or material or equipment of other Suppliers may be submitted to Engineer for review under the circumstances described below.
 - 1. "Or-Equal" Items: If in Engineer's sole discretion an item of material or equipment proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, it may be considered by Engineer as an "or-equal" item, in which case review and approval of the proposed item may, in Engineer's sole discretion, be accomplished without compliance with some or all of the requirements for approval of proposed substitute items. For the purposes of this Paragraph 6.05.A.1, a proposed item of material or equipment will be considered functionally equal to an item so named if:
 - a. in the exercise of reasonable judgment Engineer determines that:
 - 1) it is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;
 - 2) it will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole; and
 - 3) it has a proven record of performance and availability of responsive service.
 - b. Contractor certifies that, if approved and incorporated into the Work:
 - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
 - 2) it will conform substantially to the detailed requirements of the item named in the Contract Documents.

2. Substitute Items:

- a. If in Engineer's sole discretion an item of material or equipment proposed by Contractor does not qualify as an "or-equal" item under Paragraph 6.05.A.1, it will be considered a proposed substitute item.
- b. Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is essentially equivalent to that named and an acceptable substitute therefor. Requests for review of proposed substitute items of material or equipment will not be accepted by Engineer from anyone other than Contractor.

- c. The requirements for review by Engineer will be as set forth in Paragraph 6.05.A.2.d, as supplemented by the General Requirements, and as Engineer may decide is appropriate under the circumstances.
- d. Contractor shall make written application to Engineer for review of a proposed substitute item of material or equipment that Contractor seeks to furnish or use. The application:
 - 1) shall certify that the proposed substitute item will:
 - a) perform adequately the functions and achieve the results called for by the general design,
 - b) be similar in substance to that specified, and
 - c) be suited to the same use as that specified;

2) will state:

- a) the extent, if any, to which the use of the proposed substitute item will prejudice Contractor's achievement of Substantial Completion on time,
- b) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item, and
- c) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty;

3) will identify:

- a) all variations of the proposed substitute item from that specified, and
- b) available engineering, sales, maintenance, repair, and replacement services; and
- 4) shall contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including costs of redesign and claims of other contractors affected by any resulting change.
- B. Substitute Construction Methods or Procedures: If a specific means, method, technique, sequence, or procedure of construction is expressly required by the Contract Documents, Contractor may furnish or utilize a substitute means, method, technique, sequence, or procedure of construction approved by Engineer. Contractor shall submit sufficient information to allow Engineer, in Engineer's sole discretion, to determine that the substitute proposed is equivalent to that expressly called for by the Contract Documents. The requirements for review by Engineer will be similar to those provided in Paragraph 6.05.A.2.
- C. *Engineer's Evaluation:* Engineer will be allowed a reasonable time within which to evaluate each proposal or submittal made pursuant to Paragraphs 6.05.A and 6.05.B. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No "or equal" or substitute will be ordered, installed or utilized until

Engineer's review is complete, which will be evidenced by a Change Order in the case of a substitute and an approved Shop Drawing for an "or equal." Engineer will advise Contractor in writing of any negative determination.

- D. *Special Guarantee:* Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- E. *Engineer's Cost Reimbursement*: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor pursuant to Paragraphs 6.05.A.2 and 6.05.B. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for making changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.
- F. *Contractor's Expense*: Contractor shall provide all data in support of any proposed substitute or "or-equal" at Contractor's expense.

6.06 Concerning Subcontractors, Suppliers, and Others

- A. Contractor shall not employ any Subcontractor, Supplier, or other individual or entity (including those acceptable to Owner as indicated in Paragraph 6.06.B), whether initially or as a replacement, against whom Owner may have reasonable objection. Contractor shall not be required to employ any Subcontractor, Supplier, or other individual or entity to furnish or perform any of the Work against whom Contractor has reasonable objection.
- B. If the Supplementary Conditions require the identity of certain Subcontractors, Suppliers, or other individuals or entities to be submitted to Owner in advance for acceptance by Owner by a specified date prior to the Effective Date of the Agreement, and if Contractor has submitted a list thereof in accordance with the Supplementary Conditions, Owner's acceptance (either in writing or by failing to make written objection thereto by the date indicated for acceptance or objection in the Bidding Documents or the Contract Documents) of any such Subcontractor, Supplier, or other individual or entity so identified may be revoked on the basis of reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor, Supplier, or other individual or entity, and the Contract Price will be adjusted by the difference in the cost occasioned by such replacement, and an appropriate Change Order will be issued. No acceptance by Owner of any such Subcontractor, Supplier, or other individual or entity, whether initially or as a replacement, shall constitute a waiver of any right of Owner or Engineer to reject defective Work.
- C. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of the Subcontractors, Suppliers, and other individuals or entities performing or furnishing any of the Work just as Contractor is responsible for Contractor's own acts and omissions. Nothing in the Contract Documents:
 - 1. shall create for the benefit of any such Subcontractor, Supplier, or other individual or entity any contractual relationship between Owner or Engineer and any such Subcontractor, Supplier or other individual or entity; nor

- 2. shall create any obligation on the part of Owner or Engineer to pay or to see to the payment of any moneys due any such Subcontractor, Supplier, or other individual or entity except as may otherwise be required by Laws and Regulations.
- D. Contractor shall be solely responsible for scheduling and coordinating the Work of Subcontractors, Suppliers, and other individuals or entities performing or furnishing any of the Work under a direct or indirect contract with Contractor.
- E. Contractor shall require all Subcontractors, Suppliers, and such other individuals or entities performing or furnishing any of the Work to communicate with Engineer through Contractor.
- F. The divisions and sections of the Specifications and the identifications of any Drawings shall not control Contractor in dividing the Work among Subcontractors or Suppliers or delineating the Work to be performed by any specific trade.
- G. All Work performed for Contractor by a Subcontractor or Supplier will be pursuant to an appropriate agreement between Contractor and the Subcontractor or Supplier which specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract Documents for the benefit of Owner and Engineer. Whenever any such agreement is with a Subcontractor or Supplier who is listed as a loss payee on the property insurance provided in Paragraph 5.06, the agreement between the Contractor and the Subcontractor or Supplier will contain provisions whereby the Subcontractor or Supplier waives all rights against Owner, Contractor, Engineer, and all other individuals or entities identified in the Supplementary Conditions to be listed as insureds or loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) for all losses and damages caused by, arising out of, relating to, or resulting from any of the perils or causes of loss covered by such policies and any other property insurance applicable to the Work. If the insurers on any such policies require separate waiver forms to be signed by any Subcontractor or Supplier, Contractor will obtain the same.

6.07 Patent Fees and Royalties

- A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If a particular invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights shall be disclosed by Owner in the Contract Documents.
- B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.

C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

6.08 Permits

A. Unless otherwise provided in the Supplementary Conditions, Contractor shall obtain and pay for all construction permits and licenses. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of opening of Bids, or, if there are no Bids, on the Effective Date of the Agreement. Owner shall pay all charges of utility owners for connections for providing permanent service to the Work.

6.09 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Except where otherwise expressly required by applicable Laws and Regulations, neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations.
- B. If Contractor performs any Work knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work. However, it shall not be Contractor's responsibility to make certain that the Specifications and Drawings are in accordance with Laws and Regulations, but this shall not relieve Contractor of Contractor's obligations under Paragraph 3.03.
- C. Changes in Laws or Regulations not known at the time of opening of Bids (or, on the Effective Date of the Agreement if there were no Bids) having an effect on the cost or time of performance of the Work shall be the subject of an adjustment in Contract Price or Contract Times. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment, a Claim may be made therefor as provided in Paragraph 10.05.

6.10 *Taxes*

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

6.11 *Use of Site and Other Areas*

- A. Limitation on Use of Site and Other Areas:
 - 1. Contractor shall confine construction equipment, the storage of materials and equipment, and the operations of workers to the Site and other areas permitted by Laws and Regulations, and

shall not unreasonably encumber the Site and other areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for any damage to any such land or area, or to the owner or occupant thereof, or of any adjacent land or areas resulting from the performance of the Work.

- 2. Should any claim be made by any such owner or occupant because of the performance of the Work, Contractor shall promptly settle with such other party by negotiation or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law.
- 3. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused by or based upon Contractor's performance of the Work.
- B. *Removal of Debris During Performance of the Work:* During the progress of the Work Contractor shall keep the Site and other areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris shall conform to applicable Laws and Regulations.
- C. Cleaning: Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site all tools, appliances, construction equipment and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.
- D. *Loading Structures:* Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent property to stresses or pressures that will endanger it.

6.12 Record Documents

A. Contractor shall maintain in a safe place at the Site one record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, and written interpretations and clarifications in good order and annotated to show changes made during construction. These record documents together with all approved Samples and a counterpart of all approved Shop Drawings will be available to Engineer for reference. Upon completion of the Work, these record documents, Samples, and Shop Drawings will be delivered to Engineer for Owner.

6.13 Safety and Protection

A. Contractor shall be solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations. Contractor shall take

all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury or loss to:

- 1. all persons on the Site or who may be affected by the Work;
- 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
- 3. other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- B. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection. Contractor shall notify owners of adjacent property and of Underground Facilities and other utility owners when prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property.
- C. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. The Supplementary Conditions identify any Owner's safety programs that are applicable to the Work.
- D. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.
- E. All damage, injury, or loss to any property referred to in Paragraph 6.13.A.2 or 6.13.A.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- F. Contractor's duties and responsibilities for safety and for protection of the Work shall continue until such time as all the Work is completed and Engineer has issued a notice to Owner and Contractor in accordance with Paragraph 14.07.B that the Work is acceptable (except as otherwise expressly provided in connection with Substantial Completion).

6.14 Safety Representative

A. Contractor shall designate a qualified and experienced safety representative at the Site whose duties and responsibilities shall be the prevention of accidents and the maintaining and supervising of safety precautions and programs.

6.15 Hazard Communication Programs

A. Contractor shall be responsible for coordinating any exchange of material safety data sheets or other hazard communication information required to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

6.16 *Emergencies*

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent threatened damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused thereby or are required as a result thereof. If Engineer determines that a change in the Contract Documents is required because of the action taken by Contractor in response to such an emergency, a Work Change Directive or Change Order will be issued.

6.17 Shop Drawings and Samples

A. Contractor shall submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals (as required by Paragraph 2.07). Each submittal will be identified as Engineer may require.

1. Shop Drawings:

- a. Submit number of copies specified in the General Requirements.
- b. Data shown on the Shop Drawings will be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide and to enable Engineer to review the information for the limited purposes required by Paragraph 6.17.D.

2. Samples:

- a. Submit number of Samples specified in the Specifications.
- b. Clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the submittal for the limited purposes required by Paragraph 6.17.D.
- B. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.

C. Submittal Procedures:

- 1. Before submitting each Shop Drawing or Sample, Contractor shall have:
 - a. reviewed and coordinated each Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;

- b. determined and verified all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect thereto;
- c. determined and verified the suitability of all materials offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
- d. determined and verified all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto.
- 2. Each submittal shall bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review and approval of that submittal.
- 3. With each submittal, Contractor shall give Engineer specific written notice of any variations that the Shop Drawing or Sample may have from the requirements of the Contract Documents. This notice shall be both a written communication separate from the Shop Drawings or Sample submittal; and, in addition, by a specific notation made on each Shop Drawing or Sample submitted to Engineer for review and approval of each such variation.

D. Engineer's Review:

- Engineer will provide timely review of Shop Drawings and Samples in accordance with the Schedule of Submittals acceptable to Engineer. Engineer's review and approval will be only to determine if the items covered by the submittals will, after installation or incorporation in the Work, conform to the information given in the Contract Documents and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.
- 2. Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction (except where a particular means, method, technique, sequence, or procedure of construction is specifically and expressly called for by the Contract Documents) or to safety precautions or programs incident thereto. The review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
- 3. Engineer's review and approval shall not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 6.17.C.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer's review and approval shall not relieve Contractor from responsibility for complying with the requirements of Paragraph 6.17.C.1.

E. Resubmittal Procedures:

1. Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous submittals.

6.18 *Continuing the Work*

A. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work shall be delayed or postponed pending resolution of any disputes or disagreements, except as permitted by Paragraph 15.04 or as Owner and Contractor may otherwise agree in writing.

6.19 Contractor's General Warranty and Guarantee

- A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer and its officers, directors, members, partners, employees, agents, consultants, and subcontractors shall be entitled to rely on representation of Contractor's warranty and guarantee.
- B. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
 - 1. abuse, modification, or improper maintenance or operation by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
 - 2. normal wear and tear under normal usage.
- C. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents shall be absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents or a release of Contractor's obligation to perform the Work in accordance with the Contract Documents:
 - 1. observations by Engineer;
 - 2. recommendation by Engineer or payment by Owner of any progress or final payment;
 - 3. the issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
 - 4. use or occupancy of the Work or any part thereof by Owner;
 - 5. any review and approval of a Shop Drawing or Sample submittal or the issuance of a notice of acceptability by Engineer;
 - 6. any inspection, test, or approval by others; or
 - 7. any correction of defective Work by Owner.

6.20 Indemnification

A. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the performance of the Work, provided that any such claim, cost, loss, or damage is

attributable to bodily injury, sickness, disease, or death, or to injury to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work or anyone for whose acts any of them may be liable.

- B. In any and all claims against Owner or Engineer or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 6.20.A shall not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- C. The indemnification obligations of Contractor under Paragraph 6.20.A shall not extend to the liability of Engineer and Engineer's officers, directors, members, partners, employees, agents, consultants and subcontractors arising out of:
 - 1. the preparation or approval of, or the failure to prepare or approve maps, Drawings, opinions, reports, surveys, Change Orders, designs, or Specifications; or
 - 2. giving directions or instructions, or failing to give them, if that is the primary cause of the injury or damage.

6.21 Delegation of Professional Design Services

- A. Contractor will not be required to provide professional design services unless such services are specifically required by the Contract Documents for a portion of the Work or unless such services are required to carry out Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. Contractor shall not be required to provide professional services in violation of applicable law.
- B. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of Contractor by the Contract Documents, Owner and Engineer will specify all performance and design criteria that such services must satisfy. Contractor shall cause such services or certifications to be provided by a properly licensed professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to Engineer.
- C. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications or approvals performed by such design professionals, provided Owner and Engineer have specified to Contractor all performance and design criteria that such services must satisfy.
- D. Pursuant to this Paragraph 6.21, Engineer's review and approval of design calculations and design drawings will be only for the limited purpose of checking for conformance with performance and

design criteria given and the design concept expressed in the Contract Documents. Engineer's review and approval of Shop Drawings and other submittals (except design calculations and design drawings) will be only for the purpose stated in Paragraph 6.17.D.1.

E. Contractor shall not be responsible for the adequacy of the performance or design criteria required by the Contract Documents.

ARTICLE 7 – OTHER WORK AT THE SITE

7.01 Related Work at Site

- A. Owner may perform other work related to the Project at the Site with Owner's employees, or through other direct contracts therefor, or have other work performed by utility owners. If such other work is not noted in the Contract Documents, then:
 - 1. written notice thereof will be given to Contractor prior to starting any such other work; and
 - 2. if Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times that should be allowed as a result of such other work, a Claim may be made therefor as provided in Paragraph 10.05.
- B. Contractor shall afford each other contractor who is a party to such a direct contract, each utility owner, and Owner, if Owner is performing other work with Owner's employees, proper and safe access to the Site, provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work, and properly coordinate the Work with theirs. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected. The duties and responsibilities of Contractor under this Paragraph are for the benefit of such utility owners and other contractors to the extent that there are comparable provisions for the benefit of Contractor in said direct contracts between Owner and such utility owners and other contractors.
- C. If the proper execution or results of any part of Contractor's Work depends upon work performed by others under this Article 7, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.

7.02 Coordination

- A. If Owner intends to contract with others for the performance of other work on the Project at the Site, the following will be set forth in Supplementary Conditions:
 - 1. the individual or entity who will have authority and responsibility for coordination of the activities among the various contractors will be identified;
 - 2. the specific matters to be covered by such authority and responsibility will be itemized; and

- 3. the extent of such authority and responsibilities will be provided.
- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.

7.03 Legal Relationships

- A. Paragraphs 7.01.A and 7.02 are not applicable for utilities not under the control of Owner.
- B. Each other direct contract of Owner under Paragraph 7.01.A shall provide that the other contractor is liable to Owner and Contractor for the reasonable direct delay and disruption costs incurred by Contractor as a result of the other contractor's wrongful actions or inactions.
- C. Contractor shall be liable to Owner and any other contractor under direct contract to Owner for the reasonable direct delay and disruption costs incurred by such other contractor as a result of Contractor's wrongful action or inactions.

ARTICLE 8 – OWNER'S RESPONSIBILITIES

- 8.01 *Communications to Contractor*
 - A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.
- 8.02 Replacement of Engineer
 - A. In case of termination of the employment of Engineer, Owner shall appoint an engineer to whom Contractor makes no reasonable objection, whose status under the Contract Documents shall be that of the former Engineer.
- 8.03 Furnish Data
 - A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 8.04 Pay When Due
 - A. Owner shall make payments to Contractor when they are due as provided in Paragraphs 14.02.C and 14.07.C.
- 8.05 Lands and Easements; Reports and Tests
 - A. Owner's duties with respect to providing lands and easements and providing engineering surveys to establish reference points are set forth in Paragraphs 4.01 and 4.05. Paragraph 4.02 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of subsurface conditions and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 8.06 Insurance
 - A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 5.

8.07 *Change Orders*

- A. Owner is obligated to execute Change Orders as indicated in Paragraph 10.03.
- 8.08 Inspections, Tests, and Approvals
 - A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 13.03.B.
- 8.09 Limitations on Owner's Responsibilities
 - A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 8.10 Undisclosed Hazardous Environmental Condition
 - A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 4.06.
- 8.11 Evidence of Financial Arrangements
 - A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract Documents.
- 8.12 *Compliance with Safety Program*
 - A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed pursuant to Paragraph 6.13.D.

ARTICLE 9 – ENGINEER'S STATUS DURING CONSTRUCTION

- 9.01 *Owner's Representative*
 - A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract Documents.
- 9.02 Visits to Site
 - A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe as an experienced and qualified design professional the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform

- generally to the Contract Documents. On the basis of such visits and observations, Engineer will keep Owner informed of the progress of the Work and will endeavor to guard Owner against defective Work.
- B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 9.09. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

9.03 Project Representative

A. If Owner and Engineer agree, Engineer will furnish a Resident Project Representative to assist Engineer in providing more extensive observation of the Work. The authority and responsibilities of any such Resident Project Representative and assistants will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in Paragraph 9.09. If Owner designates another representative or agent to represent Owner at the Site who is not Engineer's consultant, agent or employee, the responsibilities and authority and limitations thereon of such other individual or entity will be as provided in the Supplementary Conditions.

9.04 Authorized Variations in Work

A. Engineer may authorize minor variations in the Work from the requirements of the Contract Documents which do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. These may be accomplished by a Field Order and will be binding on Owner and also on Contractor, who shall perform the Work involved promptly. If Owner or Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, or both, and the parties are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment, a Claim may be made therefor as provided in Paragraph 10.05.

9.05 Rejecting Defective Work

A. Engineer will have authority to reject Work which Engineer believes to be defective, or that Engineer believes will not produce a completed Project that conforms to the Contract Documents or that will prejudice the integrity of the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Engineer will also have authority to require special inspection or testing of the Work as provided in Paragraph 13.04, whether or not the Work is fabricated, installed, or completed.

9.06 Shop Drawings, Change Orders and Payments

A. In connection with Engineer's authority, and limitations thereof, as to Shop Drawings and Samples, see Paragraph 6.17.

- B. In connection with Engineer's authority, and limitations thereof, as to design calculations and design drawings submitted in response to a delegation of professional design services, if any, see Paragraph 6.21.
- C. In connection with Engineer's authority as to Change Orders, see Articles 10, 11, and 12.
- D. In connection with Engineer's authority as to Applications for Payment, see Article 14.

9.07 Determinations for Unit Price Work

A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, subject to the provisions of Paragraph 10.05.

9.08 Decisions on Requirements of Contract Documents and Acceptability of Work

- A. Engineer will be the initial interpreter of the requirements of the Contract Documents and judge of the acceptability of the Work thereunder. All matters in question and other matters between Owner and Contractor arising prior to the date final payment is due relating to the acceptability of the Work, and the interpretation of the requirements of the Contract Documents pertaining to the performance of the Work, will be referred initially to Engineer in writing within 30 days of the event giving rise to the question.
- B. Engineer will, with reasonable promptness, render a written decision on the issue referred. If Owner or Contractor believes that any such decision entitles them to an adjustment in the Contract Price or Contract Times or both, a Claim may be made under Paragraph 10.05. The date of Engineer's decision shall be the date of the event giving rise to the issues referenced for the purposes of Paragraph 10.05.B.
- C. Engineer's written decision on the issue referred will be final and binding on Owner and Contractor, subject to the provisions of Paragraph 10.05.
- D. When functioning as interpreter and judge under this Paragraph 9.08, Engineer will not show partiality to Owner or Contractor and will not be liable in connection with any interpretation or decision rendered in good faith in such capacity.

9.09 Limitations on Engineer's Authority and Responsibilities

- A. Neither Engineer's authority or responsibility under this Article 9 or under any other provision of the Contract Documents nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer shall create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws

- and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Paragraph 14.07.A will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals that the results certified indicate compliance with, the Contract Documents.
- E. The limitations upon authority and responsibility set forth in this Paragraph 9.09 shall also apply to the Resident Project Representative, if any, and assistants, if any.

9.10 Compliance with Safety Program

A. While at the Site, Engineer's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Engineer has been informed pursuant to Paragraph 6.13.D.

ARTICLE 10 - CHANGES IN THE WORK; CLAIMS

10.01 Authorized Changes in the Work

- A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work by a Change Order, or a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved which will be performed under the applicable conditions of the Contract Documents (except as otherwise specifically provided).
- B. If Owner and Contractor are unable to agree on entitlement to, or on the amount or extent, if any, of an adjustment in the Contract Price or Contract Times, or both, that should be allowed as a result of a Work Change Directive, a Claim may be made therefor as provided in Paragraph 10.05.

10.02 Unauthorized Changes in the Work

A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents as amended, modified, or supplemented as provided in Paragraph 3.04, except in the case of an emergency as provided in Paragraph 6.16 or in the case of uncovering Work as provided in Paragraph 13.04.D.

10.03 Execution of Change Orders

- A. Owner and Contractor shall execute appropriate Change Orders recommended by Engineer covering:
 - 1. changes in the Work which are: (i) ordered by Owner pursuant to Paragraph 10.01.A, (ii) required because of acceptance of defective Work under Paragraph 13.08.A or Owner's correction of defective Work under Paragraph 13.09, or (iii) agreed to by the parties;
 - 2. changes in the Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive; and
 - 3. changes in the Contract Price or Contract Times which embody the substance of any written decision rendered by Engineer pursuant to Paragraph 10.05; provided that, in lieu of executing any such Change Order, an appeal may be taken from any such decision in accordance with the provisions of the Contract Documents and applicable Laws and Regulations, but during any such appeal, Contractor shall carry on the Work and adhere to the Progress Schedule as provided in Paragraph 6.18.A.

10.04 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

10.05 *Claims*

- A. *Engineer's Decision Required*: All Claims, except those waived pursuant to Paragraph 14.09, shall be referred to the Engineer for decision. A decision by Engineer shall be required as a condition precedent to any exercise by Owner or Contractor of any rights or remedies either may otherwise have under the Contract Documents or by Laws and Regulations in respect of such Claims.
- B. *Notice:* Written notice stating the general nature of each Claim shall be delivered by the claimant to Engineer and the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto. The responsibility to substantiate a Claim shall rest with the party making the Claim. Notice of the amount or extent of the Claim, with supporting data shall be delivered to the Engineer and the other party to the Contract within 60 days after the start of such event (unless Engineer allows additional time for claimant to submit additional or more accurate data in support of such Claim). A Claim for an adjustment in Contract Price shall be prepared in accordance with the provisions of Paragraph 12.01.B. A Claim for an adjustment in Contract Times shall be prepared in accordance with the provisions of Paragraph 12.02.B. Each Claim shall be accompanied by claimant's written statement that the adjustment claimed is the entire adjustment to which the claimant believes it is entitled as a result of said event. The opposing party shall submit any response to Engineer and the claimant within 30 days after receipt of the claimant's last submittal (unless Engineer allows additional time).

- C. *Engineer's Action*: Engineer will review each Claim and, within 30 days after receipt of the last submittal of the claimant or the last submittal of the opposing party, if any, take one of the following actions in writing:
 - 1. deny the Claim in whole or in part;
 - 2. approve the Claim; or
 - 3. notify the parties that the Engineer is unable to resolve the Claim if, in the Engineer's sole discretion, it would be inappropriate for the Engineer to do so. For purposes of further resolution of the Claim, such notice shall be deemed a denial.
- D. In the event that Engineer does not take action on a Claim within said 30 days, the Claim shall be deemed denied.
- E. Engineer's written action under Paragraph 10.05.C or denial pursuant to Paragraphs 10.05.C.3 or 10.05.D will be final and binding upon Owner and Contractor, unless Owner or Contractor invoke the dispute resolution procedure set forth in Article 16 within 30 days of such action or denial.
- F. No Claim for an adjustment in Contract Price or Contract Times will be valid if not submitted in accordance with this Paragraph 10.05.

ARTICLE 11 – COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

11.01 *Cost of the Work*

- A. Costs Included: The term Cost of the Work means the sum of all costs, except those excluded in Paragraph 11.01.B, necessarily incurred and paid by Contractor in the proper performance of the Work. When the value of any Work covered by a Change Order or when a Claim for an adjustment in Contract Price is determined on the basis of Cost of the Work, the costs to be reimbursed to Contractor will be only those additional or incremental costs required because of the change in the Work or because of the event giving rise to the Claim. Except as otherwise may be agreed to in writing by Owner, such costs shall be in amounts no higher than those prevailing in the locality of the Project, shall not include any of the costs itemized in Paragraph 11.01.B, and shall include only the following items:
 - 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor. Such employees shall include, without limitation, superintendents, foremen, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work shall be apportioned on the basis of their time spent on the Work. Payroll costs shall include, but not be limited to, salaries and wages plus the cost of fringe benefits, which shall include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, bonuses, sick leave, vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, shall be included in the above to the extent authorized by Owner.
 - 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts shall accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts shall accrue to

Owner. All trade discounts, rebates and refunds and returns from sale of surplus materials and equipment shall accrue to Owner, and Contractor shall make provisions so that they may be obtained.

- 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, who will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee shall be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 11.01.
- 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed for services specifically related to the Work.
- 5. Supplemental costs including the following:
 - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
 - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, and hand tools not owned by the workers, which are consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.
 - c. Rentals of all construction equipment and machinery, and the parts thereof whether rented from Contractor or others in accordance with rental agreements approved by Owner with the advice of Engineer, and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs shall be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts shall cease when the use thereof is no longer necessary for the Work.
 - d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
 - e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
 - f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of property insurance established in accordance with Paragraph 5.06.D), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses shall include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses shall be included in the Cost of the Work for the purpose of determining Contractor's fee.

- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as telegrams, long distance telephone calls, telephone service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance Contractor is required by the Contract Documents to purchase and maintain.
- B. Costs Excluded: The term Cost of the Work shall not include any of the following items:
 - 1. Payroll costs and other compensation of Contractor's officers, executives, principals (of partnerships and sole proprietorships), general managers, safety managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 11.01.A.1 or specifically covered by Paragraph 11.01.A.4, all of which are to be considered administrative costs covered by the Contractor's fee.
 - 2. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
 - 3. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
 - 4. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.
 - 5. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraphs 11.01.A.
- C. *Contractor's Fee:* When all the Work is performed on the basis of cost-plus, Contractor's fee shall be determined as set forth in the Agreement. When the value of any Work covered by a Change Order or when a Claim for an adjustment in Contract Price is determined on the basis of Cost of the Work, Contractor's fee shall be determined as set forth in Paragraph 12.01.C.
- D. *Documentation:* Whenever the Cost of the Work for any purpose is to be determined pursuant to Paragraphs 11.01.A and 11.01.B, Contractor will establish and maintain records thereof in accordance with generally accepted accounting practices and submit in a form acceptable to Engineer an itemized cost breakdown together with supporting data.

11.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances:

1. Contractor agrees that:

- a. the cash allowances include the cost to Contractor (less any applicable trade discounts) of
 materials and equipment required by the allowances to be delivered at the Site, and all
 applicable taxes; and
- b. Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment on account of any of the foregoing will be valid.

C. Contingency Allowance:

- 1. Contractor agrees that a contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor on account of Work covered by allowances, and the Contract Price shall be correspondingly adjusted.

11.03 Unit Price Work

- A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.
- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Determinations of the actual quantities and classifications of Unit Price Work performed by Contractor will be made by Engineer subject to the provisions of Paragraph 9.07.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Owner or Contractor may make a Claim for an adjustment in the Contract Price in accordance with Paragraph 10.05 if:
 - 1. the quantity of any item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
 - 2. there is no corresponding adjustment with respect to any other item of Work; and
 - Contractor believes that Contractor is entitled to an increase in Contract Price as a result of
 having incurred additional expense or Owner believes that Owner is entitled to a decrease in
 Contract Price and the parties are unable to agree as to the amount of any such increase or
 decrease.

ARTICLE 12 – CHANGE OF CONTRACT PRICE; CHANGE OF CONTRACT TIMES

12.01 Change of Contract Price

- A. The Contract Price may only be changed by a Change Order. Any Claim for an adjustment in the Contract Price shall be based on written notice submitted by the party making the Claim to the Engineer and the other party to the Contract in accordance with the provisions of Paragraph 10.05.
- B. The value of any Work covered by a Change Order or of any Claim for an adjustment in the Contract Price will be determined as follows:
 - 1. where the Work involved is covered by unit prices contained in the Contract Documents, by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 11.03); or
 - 2. where the Work involved is not covered by unit prices contained in the Contract Documents, by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 12.01.C.2); or
 - 3. where the Work involved is not covered by unit prices contained in the Contract Documents and agreement to a lump sum is not reached under Paragraph 12.01.B.2, on the basis of the Cost of the Work (determined as provided in Paragraph 11.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 12.01.C).
- C. Contractor's Fee: The Contractor's fee for overhead and profit shall be determined as follows:
 - 1. a mutually acceptable fixed fee; or
 - 2. if a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
 - a. for costs incurred under Paragraphs 11.01.A.1 and 11.01.A.2, the Contractor's fee shall be 15 percent;
 - b. for costs incurred under Paragraph 11.01.A.3, the Contractor's fee shall be five percent;
 - c. where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 12.01.C.2.a and 12.01.C.2.b is that the Subcontractor who actually performs the Work, at whatever tier, will be paid a fee of 15 percent of the costs incurred by such Subcontractor under Paragraphs 11.01.A.1 and 11.01.A.2 and that any higher tier Subcontractor and Contractor will each be paid a fee of five percent of the amount paid to the next lower tier Subcontractor;
 - d. no fee shall be payable on the basis of costs itemized under Paragraphs 11.01.A.4, 11.01.A.5, and 11.01.B;
 - e. the amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in cost will be the amount of the actual net decrease in cost plus a deduction in Contractor's fee by an amount equal to five percent of such net decrease; and

f. when both additions and credits are involved in any one change, the adjustment in Contractor's fee shall be computed on the basis of the net change in accordance with Paragraphs 12.01.C.2.a through 12.01.C.2.e, inclusive.

12.02 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Claim for an adjustment in the Contract Times shall be based on written notice submitted by the party making the Claim to the Engineer and the other party to the Contract in accordance with the provisions of Paragraph 10.05.
- B. Any adjustment of the Contract Times covered by a Change Order or any Claim for an adjustment in the Contract Times will be determined in accordance with the provisions of this Article 12.

12.03 Delays

- A. Where Contractor is prevented from completing any part of the Work within the Contract Times due to delay beyond the control of Contractor, the Contract Times will be extended in an amount equal to the time lost due to such delay if a Claim is made therefor as provided in Paragraph 12.02.A. Delays beyond the control of Contractor shall include, but not be limited to, acts or neglect by Owner, acts or neglect of utility owners or other contractors performing other work as contemplated by Article 7, fires, floods, epidemics, abnormal weather conditions, or acts of God.
- B. If Owner, Engineer, or other contractors or utility owners performing other work for Owner as contemplated by Article 7, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times, or both. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times.
- C. If Contractor is delayed in the performance or progress of the Work by fire, flood, epidemic, abnormal weather conditions, acts of God, acts or failures to act of utility owners not under the control of Owner, or other causes not the fault of and beyond control of Owner and Contractor, then Contractor shall be entitled to an equitable adjustment in Contract Times, if such adjustment is essential to Contractor's ability to complete the Work within the Contract Times. Such an adjustment shall be Contractor's sole and exclusive remedy for the delays described in this Paragraph 12.03.C.
- D. Owner, Engineer, and their officers, directors, members, partners, employees, agents, consultants, or subcontractors shall not be liable to Contractor for any claims, costs, losses, or damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Contractor on or in connection with any other project or anticipated project.
- E. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delays within the control of Contractor. Delays attributable to and within the control of a Subcontractor or Supplier shall be deemed to be delays within the control of Contractor.

ARTICLE 13 – TESTS AND INSPECTIONS; CORRECTION, REMOVAL OR ACCEPTANCE OF DEFECTIVE WORK

13.01 Notice of Defects

A. Prompt notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor. Defective Work may be rejected, corrected, or accepted as provided in this Article 13.

13.02 Access to Work

A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and governmental agencies with jurisdictional interests will have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply therewith as applicable.

13.03 Tests and Inspections

- A. Contractor shall give Engineer timely notice of readiness of the Work for all required inspections, tests, or approvals and shall cooperate with inspection and testing personnel to facilitate required inspections or tests.
- B. Owner shall employ and pay for the services of an independent testing laboratory to perform all inspections, tests, or approvals required by the Contract Documents except:
 - 1. for inspections, tests, or approvals covered by Paragraphs 13.03.C and 13.03.D below;
 - 2. that costs incurred in connection with tests or inspections conducted pursuant to Paragraph 13.04.B shall be paid as provided in Paragraph 13.04.C; and
 - 3. as otherwise specifically provided in the Contract Documents.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.
- D. Contractor shall be responsible for arranging and obtaining and shall pay all costs in connection with any inspections, tests, or approvals required for Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work; or acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work. Such inspections, tests, or approvals shall be performed by organizations acceptable to Owner and Engineer.
- E. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation.

F. Uncovering Work as provided in Paragraph 13.03.E shall be at Contractor's expense unless Contractor has given Engineer timely notice of Contractor's intention to cover the same and Engineer has not acted with reasonable promptness in response to such notice.

13.04 Uncovering Work

- A. If any Work is covered contrary to the written request of Engineer, it must, if requested by Engineer, be uncovered for Engineer's observation and replaced at Contractor's expense.
- B. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, furnishing all necessary labor, material, and equipment.
- C. If it is found that the uncovered Work is defective, Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and Owner shall be entitled to an appropriate decrease in the Contract Price. If the parties are unable to agree as to the amount thereof, Owner may make a Claim therefor as provided in Paragraph 10.05.
- D. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, or both, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, Contractor may make a Claim therefor as provided in Paragraph 10.05.

13.05 *Owner May Stop the Work*

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, Owner may order Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work shall not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

13.06 Correction or Removal of Defective Work

A. Promptly after receipt of written notice, Contractor shall correct all defective Work, whether or not fabricated, installed, or completed, or, if the Work has been rejected by Engineer, remove it from the Project and replace it with Work that is not defective. Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or removal (including but not limited to all costs of repair or replacement of work of others).

B. When correcting defective Work under the terms of this Paragraph 13.06 or Paragraph 13.07, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.

13.07 Correction Period

- A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the terms of any applicable special guarantee required by the Contract Documents) or by any specific provision of the Contract Documents, any Work is found to be defective, or if the repair of any damages to the land or areas made available for Contractor's use by Owner or permitted by Laws and Regulations as contemplated in Paragraph 6.11.A is found to be defective, Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:
 - 1. repair such defective land or areas; or
 - 2. correct such defective Work; or
 - 3. if the defective Work has been rejected by Owner, remove it from the Project and replace it with Work that is not defective, and
 - 4. satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others or other land or areas resulting therefrom.
- B. If Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. All claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others) will be paid by Contractor.
- C. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- D. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this Paragraph 13.07, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.
- E. Contractor's obligations under this Paragraph 13.07 are in addition to any other obligation or warranty. The provisions of this Paragraph 13.07 shall not be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

13.08 Acceptance of Defective Work

A. If, instead of requiring correction or removal and replacement of defective Work, Owner (and, prior to Engineer's recommendation of final payment, Engineer) prefers to accept it, Owner may do so. Contractor shall pay all claims, costs, losses, and damages (including but not limited to all

fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness) and for the diminished value of the Work to the extent not otherwise paid by Contractor pursuant to this sentence. If any such acceptance occurs prior to Engineer's recommendation of final payment, a Change Order will be issued incorporating the necessary revisions in the Contract Documents with respect to the Work, and Owner shall be entitled to an appropriate decrease in the Contract Price, reflecting the diminished value of Work so accepted. If the parties are unable to agree as to the amount thereof, Owner may make a Claim therefor as provided in Paragraph 10.05. If the acceptance occurs after such recommendation, an appropriate amount will be paid by Contractor to Owner.

13.09 Owner May Correct Defective Work

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace rejected Work as required by Engineer in accordance with Paragraph 13.06.A, or if Contractor fails to perform the Work in accordance with the Contract Documents, or if Contractor fails to comply with any other provision of the Contract Documents, Owner may, after seven days written notice to Contractor, correct, or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 13.09, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, take possession of Contractor's tools, appliances, construction equipment and machinery at the Site, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this Paragraph.
- C. All claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 13.09 will be charged against Contractor, and a Change Order will be issued incorporating the necessary revisions in the Contract Documents with respect to the Work; and Owner shall be entitled to an appropriate decrease in the Contract Price. If the parties are unable to agree as to the amount of the adjustment, Owner may make a Claim therefor as provided in Paragraph 10.05. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.
- D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 13.09.

ARTICLE 14 – PAYMENTS TO CONTRACTOR AND COMPLETION

14.01 Schedule of Values

A. The Schedule of Values established as provided in Paragraph 2.07.A will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments on account of Unit Price Work will be based on the number of units completed.

14.02 Progress Payments

A. Applications for Payments:

- 1. At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents. If payment is requested on the basis of materials and equipment not incorporated in the Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment shall also be accompanied by a bill of sale, invoice, or other documentation warranting that Owner has received the materials and equipment free and clear of all Liens and evidence that the materials and equipment are covered by appropriate property insurance or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.
- Beginning with the second Application for Payment, each Application shall include an
 affidavit of Contractor stating that all previous progress payments received on account of the
 Work have been applied on account to discharge Contractor's legitimate obligations associated
 with prior Applications for Payment.
- 3. The amount of retainage with respect to progress payments will be as stipulated in the Agreement.

B. Review of Applications:

- 1. Engineer will, within 10 days after receipt of each Application for Payment, either indicate in writing a recommendation of payment and present the Application to Owner or return the Application to Contractor indicating in writing Engineer's reasons for refusing to recommend payment. In the latter case, Contractor may make the necessary corrections and resubmit the Application.
- 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules, that to the best of Engineer's knowledge, information and belief:
 - a. the Work has progressed to the point indicated;
 - b. the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial

- Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 9.07, and any other qualifications stated in the recommendation); and
- c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
- 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
 - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract Documents; or
 - b. there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.
- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
 - a. to supervise, direct, or control the Work, or
 - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or
 - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work, or
 - d. to make any examination to ascertain how or for what purposes Contractor has used the moneys paid on account of the Contract Price, or
 - e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 14.02.B.2. Engineer may also refuse to recommend any such payment or, because of subsequently discovered evidence or the results of subsequent inspections or tests, revise or revoke any such payment recommendation previously made, to such extent as may be necessary in Engineer's opinion to protect Owner from loss because:
 - a. the Work is defective, or completed Work has been damaged, requiring correction or replacement;
 - b. the Contract Price has been reduced by Change Orders;
 - c. Owner has been required to correct defective Work or complete Work in accordance with Paragraph 13.09; or
 - d. Engineer has actual knowledge of the occurrence of any of the events enumerated in Paragraph 15.02.A.

C. Payment Becomes Due:

1. Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended will (subject to the provisions of Paragraph 14.02.D) become due, and when due will be paid by Owner to Contractor.

D. Reduction in Payment:

- 1. Owner may refuse to make payment of the full amount recommended by Engineer because:
 - a. claims have been made against Owner on account of Contractor's performance or furnishing of the Work;
 - b. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens;
 - c. there are other items entitling Owner to a set-off against the amount recommended; or
 - d. Owner has actual knowledge of the occurrence of any of the events enumerated in Paragraphs 14.02.B.5.a through 14.02.B.5.c or Paragraph 15.02.A.
- 2. If Owner refuses to make payment of the full amount recommended by Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, when Contractor remedies the reasons for such action.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld shall be treated as an amount due as determined by Paragraph 14.02.C.1 and subject to interest as provided in the Agreement.

14.03 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials, and equipment covered by any Application for Payment, whether incorporated in the Project or not, will pass to Owner no later than the time of payment free and clear of all Liens.

14.04 Substantial Completion

- A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete (except for items specifically listed by Contractor as incomplete) and request that Engineer issue a certificate of Substantial Completion.
- B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.

- C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a tentative certificate of Substantial Completion which shall fix the date of Substantial Completion. There shall be attached to the certificate a tentative list of items to be completed or corrected before final payment. Owner shall have seven days after receipt of the tentative certificate during which to make written objection to Engineer as to any provisions of the certificate or attached list. If, after considering such objections, Engineer concludes that the Work is not substantially complete, Engineer will, within 14 days after submission of the tentative certificate to Owner, notify Contractor in writing, stating the reasons therefor. If, after consideration of Owner's objections, Engineer considers the Work substantially complete, Engineer will, within said 14 days, execute and deliver to Owner and Contractor a definitive certificate of Substantial Completion (with a revised tentative list of items to be completed or corrected) reflecting such changes from the tentative certificate as Engineer believes justified after consideration of any objections from Owner.
- D. At the time of delivery of the tentative certificate of Substantial Completion, Engineer will deliver to Owner and Contractor a written recommendation as to division of responsibilities pending final payment between Owner and Contractor with respect to security, operation, safety, and protection of the Work, maintenance, heat, utilities, insurance, and warranties and guarantees. Unless Owner and Contractor agree otherwise in writing and so inform Engineer in writing prior to Engineer's issuing the definitive certificate of Substantial Completion, Engineer's aforesaid recommendation will be binding on Owner and Contractor until final payment.
- E. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the tentative list.

14.05 Partial Utilization

- A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:
 - 1. Owner at any time may request Contractor in writing to permit Owner to use or occupy any such part of the Work which Owner believes to be ready for its intended use and substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 14.04.A through D for that part of the Work.
 - 2. Contractor at any time may notify Owner and Engineer in writing that Contractor considers any such part of the Work ready for its intended use and substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
 - 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 14.04 will apply with

- respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.
- 4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 5.10 regarding property insurance.

14.06 Final Inspection

A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.

14.07 Final Payment

A. Application for Payment:

- 1. After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, marked-up record documents (as provided in Paragraph 6.12), and other documents, Contractor may make application for final payment following the procedure for progress payments.
- 2. The final Application for Payment shall be accompanied (except as previously delivered) by:
 - a. all documentation called for in the Contract Documents, including but not limited to the evidence of insurance required by Paragraph 5.04.B.6;
 - b. consent of the surety, if any, to final payment;
 - c. a list of all Claims against Owner that Contractor believes are unsettled; and
 - d. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of or Liens filed in connection with the Work.
- 3. In lieu of the releases or waivers of Liens specified in Paragraph 14.07.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (i) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (ii) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien.

B. Engineer's Review of Application and Acceptance:

1. If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment and accompanying documentation

as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract Documents have been fulfilled, Engineer will, within ten days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of payment and present the Application for Payment to Owner for payment. At the same time Engineer will also give written notice to Owner and Contractor that the Work is acceptable subject to the provisions of Paragraph 14.09. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment.

C. Payment Becomes Due:

1. Thirty days after the presentation to Owner of the Application for Payment and accompanying documentation, the amount recommended by Engineer, less any sum Owner is entitled to set off against Engineer's recommendation, including but not limited to liquidated damages, will become due and will be paid by Owner to Contractor.

14.08 Final Completion Delayed

A. If, through no fault of Contractor, final completion of the Work is significantly delayed, and if Engineer so confirms, Owner shall, upon receipt of Contractor's final Application for Payment (for Work fully completed and accepted) and recommendation of Engineer, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance to be held by Owner for Work not fully completed or corrected is less than the retainage stipulated in the Agreement, and if bonds have been furnished as required in Paragraph 5.01, the written consent of the surety to the payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by Contractor to Engineer with the Application for such payment. Such payment shall be made under the terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

14.09 Waiver of Claims

- A. The making and acceptance of final payment will constitute:
 - 1. a waiver of all Claims by Owner against Contractor, except Claims arising from unsettled Liens, from defective Work appearing after final inspection pursuant to Paragraph 14.06, from failure to comply with the Contract Documents or the terms of any special guarantees specified therein, or from Contractor's continuing obligations under the Contract Documents; and
 - 2. a waiver of all Claims by Contractor against Owner other than those previously made in accordance with the requirements herein and expressly acknowledged by Owner in writing as still unsettled.

ARTICLE 15 – SUSPENSION OF WORK AND TERMINATION

15.01 Owner May Suspend Work

A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by notice in writing to Contractor and Engineer which will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be granted an adjustment in the Contract Price or an extension of the Contract Times, or both, directly attributable to any such suspension if Contractor makes a Claim therefor as provided in Paragraph 10.05.

15.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will justify termination for cause:
 - 1. Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment or failure to adhere to the Progress Schedule established under Paragraph 2.07 as adjusted from time to time pursuant to Paragraph 6.04);
 - 2. Contractor's disregard of Laws or Regulations of any public body having jurisdiction;
 - 3. Contractor's repeated disregard of the authority of Engineer; or
 - 4. Contractor's violation in any substantial way of any provisions of the Contract Documents.
- B. If one or more of the events identified in Paragraph 15.02.A occur, Owner may, after giving Contractor (and surety) seven days written notice of its intent to terminate the services of Contractor:
 - 1. exclude Contractor from the Site, and take possession of the Work and of all Contractor's tools, appliances, construction equipment, and machinery at the Site, and use the same to the full extent they could be used by Contractor (without liability to Contractor for trespass or conversion);
 - 2. incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere; and

- 3. complete the Work as Owner may deem expedient.
- C. If Owner proceeds as provided in Paragraph 15.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Owner arising out of or relating to completing the Work, such excess will be paid to Contractor. If such claims, costs, losses, and damages exceed such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this Paragraph, Owner shall not be required to obtain the lowest price for the Work performed.
- D. Notwithstanding Paragraphs 15.02.B and 15.02.C, Contractor's services will not be terminated if Contractor begins within seven days of receipt of notice of intent to terminate to correct its failure to perform and proceeds diligently to cure such failure within no more than 30 days of receipt of said notice.
- E. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue. Any retention or payment of moneys due Contractor by Owner will not release Contractor from liability.
- F. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 5.01.A, the termination procedures of that bond shall supersede the provisions of Paragraphs 15.02.B and 15.02.C.

15.03 Owner May Terminate For Convenience

- A. Upon seven days written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
 - 1. completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
 - expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses;
 - 3. all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) incurred in settlement of terminated contracts with Subcontractors, Suppliers, and others; and
 - 4. reasonable expenses directly attributable to termination.
- B. Contractor shall not be paid on account of loss of anticipated profits or revenue or other economic loss arising out of or resulting from such termination.

- A. If, through no act or fault of Contractor, (i) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (ii) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (iii) Owner fails for 30 days to pay Contractor any sum finally determined to be due, then Contractor may, upon seven days written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the Contract and recover from Owner payment on the same terms as provided in Paragraph 15.03.
- B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, seven days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The provisions of this Paragraph 15.04 are not intended to preclude Contractor from making a Claim under Paragraph 10.05 for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this Paragraph.

ARTICLE 16 – DISPUTE RESOLUTION

16.01 Methods and Procedures

- A. Either Owner or Contractor may request mediation of any Claim submitted to Engineer for a decision under Paragraph 10.05 before such decision becomes final and binding. The mediation will be governed by the Construction Industry Mediation Rules of the American Arbitration Association in effect as of the Effective Date of the Agreement. The request for mediation shall be submitted in writing to the American Arbitration Association and the other party to the Contract. Timely submission of the request shall stay the effect of Paragraph 10.05.E.
- B. Owner and Contractor shall participate in the mediation process in good faith. The process shall be concluded within 60 days of filing of the request. The date of termination of the mediation shall be determined by application of the mediation rules referenced above.
- C. If the Claim is not resolved by mediation, Engineer's action under Paragraph 10.05.C or a denial pursuant to Paragraphs 10.05.C.3 or 10.05.D shall become final and binding 30 days after termination of the mediation unless, within that time period, Owner or Contractor:
 - 1. elects in writing to invoke any dispute resolution process provided for in the Supplementary Conditions; or
 - 2. agrees with the other party to submit the Claim to another dispute resolution process; or
 - 3. gives written notice to the other party of the intent to submit the Claim to a court of competent jurisdiction.

ARTICLE 17 – MISCELLANEOUS

17.01 Giving Notice

- A. Whenever any provision of the Contract Documents requires the giving of written notice, it will be deemed to have been validly given if:
 - 1. delivered in person to the individual or to a member of the firm or to an officer of the corporation for whom it is intended; or
 - 2. delivered at or sent by registered or certified mail, postage prepaid, to the last business address known to the giver of the notice.

17.02 Computation of Times

A. When any period of time is referred to in the Contract Documents by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

17.03 Cumulative Remedies

A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract Documents. The provisions of this Paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.

17.04 Survival of Obligations

A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract Documents, as well as all continuing obligations indicated in the Contract Documents, will survive final payment, completion, and acceptance of the Work or termination or completion of the Contract or termination of the services of Contractor.

17.05 Controlling Law

A. This Contract is to be governed by the law of the state in which the Project is located.

17.06 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

ARTICLES 18, 19 AND 20 ADDED IN THE SUPPLEMENTARY CONDITIONS

00 73 00.41

SUPPLEMENTARY CONDITIONS

PART 1 - GENERAL

1.1 SUPPLEMENTARY CONDITIONS

- A. These Supplementary Conditions modify, change, delete from or add to the "Standard General Conditions of the Construction Contract" EJCDC No. C-700, 2007 edition. Where any Article of the General Conditions is modified, or any Paragraph, Subparagraph, or Clause thereof is modified or deleted by these Supplementary Conditions, the unaltered provisions of that Article, Paragraph, Subparagraph, or Clause shall remain in effect.
- B. The General Conditions also may be supplemented elsewhere in the Contract Documents by provisions located in, but not necessarily limited to Division 01 of the Specifications.

1.2 ARTICLE 1 – DEFINITIONS AND TERMINOLOGY

A. SC-1

 The terms used in these Supplementary Conditions which are defined in the Standard General Conditions of the Construction Contract have the identical meaning assigned to them in said General Conditions.

B. SC-(1.01A.17)

- 1. The term "Drawings" and the term "Plans" shall be considered synonymous whenever and wherever used in the Contract Documents.
- 2. The following Drawings are part of the Contract Documents:

Drawings titled: Ridgewood Phase 3 Water Main Improvements

Project No.: 210039.40

Owner: City of Joliet, Illinois

Sheets: Consisting of 20 sheets prepared by Baxter & Woodman, Inc.

1.3 ARTICLE 2 - PRELIMINARY MATTERS

A. SC-2.01.B

1. Delete paragraph 2.01.B in its entirety and substitute the following:

2.01.B When Contractor delivers the executed Agreements to Owner, Contractor shall also deliver to Owner, with copies to each additional insured indicated in paragraphs 5.03 through 5.06, certificates of insurance which Contractor is required to purchase and maintain in accordance with paragraphs 5.03 and 5.04.

- B. SC-2.02
 - 1. Delete paragraph 2.02 in its entirety and substitute the following:
 - 2.02 "Engineer will provide an electronic copy of the Drawings and Project Manual to the Contractor at the Preconstruction Conference."
- C. SC-2.03
 - 1. Under paragraph 2.03, delete the last sentence in its entirety.
- D. SC-2.05.A
 - Delete paragraph 2.05.A in its entirety.
- E. SC-2.05.A.1 through 2.05.A.3
 - 1. Delete paragraphs 2.05.A.1 through 2.05.A.3 inclusive in their entirety.
- F. SC-2.06
 - 1. Under paragraph 2.06, change "paragraph 2.05.A" to "paragraph 2.07".
- G. SC-2.07
 - 1. Delete paragraph 2.07 in its entirety and substitute the following:
 - 2.07 Prior to submission of the first Application for Payment, but no later than 30 calendar days after Contract Times commence, Contractor shall submit to Engineer for review and approval:
 - A. A progress schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract Documents;
 - B. A schedule of Shop Drawings and Sample submittal which will list each required submittal and the times for submitting, reviewing, and processing such submittal;
 - 1. The schedule for shop drawings shall show all submittals complete before 25 percent of Contract Price has been paid to Contractor.
 - 2. The schedule for maintenance manuals shall show all submittals complete before 50 percent of Contract Price has been paid to Contractor.
 - 3. Failure to meet either of these milestones will result in all future Pay Application Payments being delayed until such time as the milestone has been met in the opinion of the Engineer.
 - C. A schedule of values for all of the Work which will include quantities and prices of items aggregating the Contract Price and will subdivide the Work into component parts in sufficient detail to serve as the basis for progress payments during construction. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work; and

D. A schedule of progress payments Contractor anticipates will be earned during the course of the Work.

No progress payment shall be made to Contractor until the schedules are submitted to and acceptable to Engineer as provided below. The progress schedule will be acceptable to Engineer as providing an orderly progression of the Work to completion within any specified Milestones and the Contract Times, but such acceptance will neither impose on Engineer responsibility for the sequencing, scheduling or progress of the Work nor interfere with or relieve Contractor from Contractor's full responsibility therefor. Contractor's schedule of Shop Drawing and Sample submissions will be acceptable to Engineer as providing a workable arrangement for reviewing and processing the required submittals. Contractor's schedule of values will be acceptable to Engineer as to form and substance.

1.4 ARTICLE 4 - AVAILABILITY OF LANDS; SUBSURFACE AND PHYSICAL CONDITIONS; REFERENCE POINTS

A. SC-4.01.A

1. Under paragraph 4.01.A, third sentence, insert the words "and temporary construction easements shown on the Drawings" after the word "facilities".

B. SC-4.02.A

- 1. Delete paragraph 4.02.A in its entirety and replace it with the following:
 - A. Reports and Drawings:
 - Section 00 31 32.13 identifies those reports known to Owner of explorations and tests of subsurface conditions at or contiguous to the Site; and
- 2. The Supplementary Conditions identify those drawings known to Owner of physical conditions relating to existing surface or subsurface structures at the Site (except Underground Facilities).

C. SC-4.02.B

1. Under paragraph 4.02.B, delete the second sentence "Such technical data is identified in the Supplementary Conditions".

D. SC-4.03

- 1. Delete paragraph 4.03A in its entirety and substitute the following:
 - A. Notification by Contractor, and processing by the Engineer and Owner of Contractor's claims of differing site conditions shall be done in conformance with P.A. 91-0647.

E. SC-4.05.A

- 1. Under paragraph 4.05.A., delete first sentence and substitute the following:
 - A. Owner shall provide land surveys necessary to establish right-of-way, easements and property lines. Engineer will provide base lines, bench marks and reference points which in Engineer's judgment are necessary

- to enable Contractor to proceed with the Work. Contractor shall provide all stakes, markers, labor and assistance required by Engineer.
- 2. Under paragraph 4.05.A, last sentence, insert the words "and pay" between the words "responsible" and "for".

1.5 ARTICLE 5 - BONDS AND INSURANCE

- A. SC-5.02 through 5.10
 - 1. Delete paragraphs 5.02 through 5.10, inclusive in their entirety and substitute the following:
 - 5.02 Licensed Sureties and Insurers; Insurance Policies:
 - A. All bonds and insurance required by the Contract Documents to be purchased and maintained by Contractor shall be obtained from surety or insurance companies that are fully licensed or authorized in the jurisdiction in which the Project is located.
 - 5.03 Certificates of Insurance and Endorsements to Insurance Policies:
 - In addition to delivering certificates of insurance in accordance with SC Α. paragraph 2.01.B, Contractor shall also deliver to Owner, with copies to each additional insured, copies of all endorsements to the insurance policies which Contractor is required to purchase and maintain in accordance with paragraphs 5.04 and 5.06. All certificates of insurance and endorsements must be found acceptable by the Engineer prior to the Agreement being signed by the Owner. Certificate of Insurance and endorsements shall be fully completed, signed and delivered in accordance with the requirements of Article 5. Samples of Certificate of Insurance Form, and Additional Insureds endorsements, acceptable to Owner, are included in Sections 00 62 16, 00 62 16.13, 00 62 16.16, 00 62 16.19 and 00 62 16.21. Other acceptable endorsements for general liability endorsements may include a combination of CG 20 10 and CG 20 37 endorsements. Endorsements or general liability endorsements shall not exclude supervisory and inspection services.

5.04 Contractor's Insurance:

A. Contractor shall purchase and maintain such liability and other insurance as is appropriate for the Work being performed and furnished and as will provide protection from claims set forth below which may arise out of or result from Contractor's performance and furnishing of the Work and Contractor's other obligations under the Contract Documents, whether it is to be performed or furnished by Contractor, by any Subcontractor, or Supplier, by anyone directly or indirectly employed by any of them to perform or furnish any of the Work or by anyone for whose acts any of them may be liable:

- 1. <u>Workers' Compensation</u> Workers' Compensation in accordance with the laws of the State, but not less than:
 - E.L. \$1,000,000 each accident.
 - E.L. \$1,000,000 each employee.
 - E.L. \$1,000,000 policy limit.

2. General Liability

- (a) Bodily Injury Liability and Property Damage limit: \$1,000,000.
- (b) Medical Expense limit: \$10,000.
- (c) Personal and Advertising Injury limit: \$1,000,000.
- (d) General Aggregate limit (other than P-CO): \$2,000,000.
- (e) Products-Completed Operations Aggregate limit: \$1,250,000.

3. <u>Automobile Liability</u>

- (a) \$1,000,000 Bodily Injury and Property Damage (Combined Single Limit).
- (b) Coverage shall include hired and non-owned automobiles.
- Umbrella Liability Umbrella Liability coverage in an amount not less than \$5,000,000. Such coverage shall include, but not be limited to, excess coverage for the Workers' Compensation, General Liability, and Automobile Liability policies.

The policies of insurance so required by this paragraph 5.04 to be purchased and maintained shall:

- 5. Be furnished by insurers with A. M. Best Company rating of at least A-(Excellent), and a financial size category of VIII or greater.
- 6. With respect to general liability and umbrella liability insurance required by paragraph 5.04A, include as additional insured Owner, Engineer, and Engineer's Consultants, all of whom shall be listed on such policy by name as additional insureds through an endorsement thereto which provides for no different coverage to the additional insureds than to Contractor, and include coverage for the respective officers, directors, employees, agents and other consultants of each and any of such additional insureds. The additional insured endorsements shall provide the following:
 - (a) that the coverage afforded the additional insureds will be primary and non-contributory insurance for the additional insureds with respect to claims arising out of operations performed by or on behalf of the Contractor;
 - (b) that coverage afforded the additional insureds shall not exclude claims asserted by Contractor's employees;
 - (c) that if the additional insureds have other insurance which is applicable to the loss, such other insurance will be on an excess or contingent basis;

- (d) that the amount of Contractor's liability under the insurance policy will not be reduced by the existence of such other insurance.
- 7. Include contractual liability insurance covering Contractor's indemnity obligations under paragraphs 6.07, 6.11, and 6.20, including, but not limited to, all fees and charges of engineers, architects, attorneys and other professionals, and all court, arbitration, or other dispute resolution costs;
- 8. Remain in effect at least until final payment and at all times thereafter when Contractor may be correcting, removing or replacing defective work in accordance with paragraph 13.06; and
- 9. With respect to completed operations insurance remain in effect for at least two years after final payment (and Contractor shall furnish Owner and any other additional insured to whom an insurance policy has been furnished evidence satisfactory to Owner and any such additional insured of continuation of such insurance at final payment and one year thereafter).

5.05 Section Reserved.

5.06 Section Reserved.

5.07 Cancellation or Change:

All the policies of insurance required to be purchased and maintained by Contractor in accordance with paragraphs 5.04 shall contain a provision or endorsement that the coverage afforded will not be cancelled or materially changed or renewal refused until at least 30 days' prior written notice has been given to Owner, Engineer, and to each other additional insured to whom an insurance policy has been furnished.

5.08 Additional Insureds:

Additional insureds referenced in paragraphs 5.04 and 5.07 shall be as follows:

Owner: City of Joliet, Illinois

Address: 150 West Jefferson Street, Joliet, Illinois 60432

Engineer: Baxter & Woodman, Inc.

Address: 8678 Ridgefield Road, Crystal Lake, Illinois 60012

1.6 ARTICLE 6 - CONTRACTOR'S RESPONSIBILITIES

A. SC-6.02.B

 Under paragraph 6.02.B, add: The regular working hours are between 7:00 AM and 5:00 PM, Monday through Friday. In the event Contractor works more than eight hours on any weekday, or works anytime on Saturdays, Sundays, or holidays, during which time the Engineer, Resident Project Representative, or assistants are required to be present, Owner shall pay the cost for such overtime engineering services and shall deduct such cost from payments due Contractor. Overtime engineering services shall be charged at Engineer's standard hourly rates applied on a time and one-half basis for all time over eight hours on any single working day and for all hours on Saturday, and on a double time basis for all Sunday and holiday hours. If the amount due Contractor is not sufficient to cover the cost of overtime engineering services, Contractor shall reimburse Owner in the amount necessary to cover such costs. Legal holidays include:

New Years Day
Independence Day
Thanksgiving Day

Memorial Day
Labor Day
Christmas Day

If the legal holiday falls on Saturday, all hours worked on the preceding Friday and/or the Saturday will be considered as holiday hours. If the legal holiday falls on Sunday, all hours worked on the following Monday will be considered as holiday hours.

B. SC-6.06.A

1. Under paragraph 6.06.A, add: Any person employed by Contractor or Subcontractors who does not perform his work in a proper and skillful manner, or who is intemperate, disorderly, or otherwise objectionable, shall, at the written request of Owner, be forthwith removed from the project site and shall not be employed again in any portion of the Work without written consent of Owner.

C. SC-6.06.B

1. Under paragraph 6.06.B, add: Contractor shall identify all Subcontractors, major Suppliers and other persons or organizations providing principal items of work, material, and equipment. Contractor shall within ten working days of the date on the Notice of Award identify and submit in writing to the Engineer for Owner acceptance the names, addresses, and telephone numbers of all Subcontractors, Major Suppliers, and other persons or organizations providing principal items of work, material, and equipment.

D. SC-6.19.A

- 1. Delete paragraph 6.19.A in its entirety and substitute the following:
 - 6.19.A Contractor shall execute and deliver to Owner, before the final payment will be issued, a written warranty which guarantees that all work is in accordance with the Contract Documents and will not be defective. This warranty shall guarantee all work for a period of three years from the date of acceptance of the Work and final payment by Owner, except for equipment, motors, electrical controls, and other mechanical devices which shall be guaranteed for a period of two years from the date of acceptance and use of each item of equipment by Owner unless a different guarantee period of time is specified under other parts of the Contract Documents.

If within these guarantee periods or such longer period of time as may be prescribed by the Contract Documents, any work is found to be defective, Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions, either correct such defective work, or, if it has been rejected by Owner, remove it from the site and replace it with nondefective work. If Contractor does not promptly comply with the terms of such instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective work corrected or the rejected work removed and replaced, and all direct and indirect costs of such repair and/or replacement of work, including compensation for additional professional services, shall be paid or reimbursed to Owner by Contractor.

Contractor shall furnish a maintenance/warranty bond in an amount equal to five percent (5%) of the Contract Price, but not less than \$10,000, by a surety satisfactory to Owner to guarantee Contractor's warranty to repair or replace defective work. The warranty bond shall be in addition to Contractor's contract Performance-Payment Bond, and shall be delivered to Owner prior to final payment to Contractor for the Work.

The warranty bond shall be in addition to Contractor's contract Performance-Payment Bond, and shall be delivered to Owner prior to final payment to Contractor for the Work.

E. SC-6.20.A

1. Under paragraph 6.20.A, add:

In addition, Contractor shall indemnify, hold harmless, and pay for the defense of Owner and Engineer from and against claims, losses, or damages in regard to any act or failure to act by Owner or Engineer in connection with general supervision, observation and/or coordination of Contractor's operations.

Contractor shall, at its own expense, appear, defend, and pay all fees of attorneys and all costs and other expenses arising therefrom or incurred in connection therewith; and, if any judgments shall be rendered against any individual or entity indemnified hereunder in any such action, Contractor shall, at its own expense, satisfy and discharge same. Contractor expressly understands and agrees that any Letter of Credit or insurance protection required by the Contract, or otherwise provided by Contractor, shall in no way limit the responsibility to indemnify, keep, and save harmless, and defend any individual or entity indemnified hereunder as herein provided.

F. SC-6.20.C

- 1. Delete paragraphs 6.20.C.1 and 6.20.C.2 and replace them with the following:
 - 1. The preparation of Drawings, Specifications, or Property Surveys.
- 2. After paragraph 6.20.C, add the following:
- D. For any matter for which Owner and Engineer are indemnified under Paragraph 6.20.A, Contractor shall pay for Owner's and Engineer's reasonable

defense, including, but not limited to, all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs or awards until Owner or Engineer are found negligent. If Owner or Engineer are found negligent, Owner or Engineer shall reimburse Contractor for the prorata extent of Owner's or Engineer's negligence for the cost of Owner's or Engineer's reasonable defense.

E. In Paragraphs 6.20.A. through D. as may be amended by the Supplementary Conditions, Contractor shall indemnify the additional insureds listed in paragraph 5.08.:

G. SC-6.21

- 1. Under paragraph 6.21, add two new subparagraphs:
 - F. The design professional providing the design calculations and design drawings shall be licensed in the State of Illinois.
 - G. The design calculation and design drawings are not shop drawings, but shall be submitted to Engineer separately along with the required shop drawings for the system, material, or equipment specified. These calculations will be forwarded to Owner for its records.

H. SC 6.22

- 1. Add new paragraph 6.22 as follows:
 - 6.22 Construction Debris Manifest: Prepare documentation identifying the hauler, generator, place of origin of debris or soil, the weight or volume of debris or soil, and the location, owner, and operator of the facility where debris or soil was transferred, disposed, recycled or treated. This documentation must be maintained by the Contractor for three (3) years: (IL Public Act 90-761).
- 1.7 ARTICLE 8 OWNER'S RESPONSIBILITIES
 - A. SC-8.06
 - 1. Delete paragraph 8.06 in its entirety.
- 1.8 ARTICLE 9 ENGINEER'S STATUS DURING CONSTRUCTION
 - A. SC-9.03
 - 1. Under paragraph 9.03.A., delete the second sentence.

2. Under paragraph 9.03A add the following:

A LISTING OF THE DUTIES, RESPONSIBILITIES AND LIMITATIONS OF AUTHORITY OF THE RESIDENT PROJECT REPRESENTATIVE (RPR).

A. General

RPR is Engineer's agent at the site, will act as directed by and under the supervision of Engineer, and will confer with Engineer regarding RPR's dealings in matters pertaining to the on-site work shall in general be with Engineer and Contractor keeping Owner advised as necessary. RPR's dealings with subcontractors will only be through or with the full knowledge and approval of Contractor. RPR shall generally communicate with Owner with the knowledge of and under the direction of Engineer.

B. Duties and Responsibilities of RPR

- 1. Schedules: Review the progress schedule, schedule of the Shop Drawing submittals and schedule of values prepared by Contractor and consult with Engineer concerning acceptability.
- Conferences and Meetings: Attend meetings with Contractor, such as preconstruction conferences, progress meetings, job conferences and other project-related meetings, and prepare and circulate copies of minutes thereof.

3. Liaison:

- a. Serve as Engineer's liaison with Contractor, working principally through Contractor's superintendent and assist in understanding the intent of the Contract Documents; and assist Engineer in serving as Owner's liaison with Contractor when Contractor's operations affect Owner's on-site operations.
- b. Assist in obtaining from Owner additional details or information, when required for proper execution of the Work.

4. Shop Drawings and Samples:

- a. Record date of receipt of Shop Drawings and samples.
- Receive samples which are furnished at the site by Contractor, and notify Engineer of availability of samples for examination.
- c. Advise Engineer and Contractor of the commencement of any Work requiring a Shop Drawing or sample if the submittal has not been approved by Engineer.

- 5. Review of Work, Rejection of Defective Work, Inspections and Tests:
 - a. Conduct on-site observations of the Work in progress to assist Engineer in determining if the Work is in general proceeding in accordance with the Contract Documents.
 - b. Report to Engineer whenever RPR believes that any Work is unsatisfactory, faulty or defective or does not conform to the Contract Documents, or has been damaged, or does not meet the requirements of any inspection, test or approval required to be made: and advise Engineer of Work that RPR believes should be corrected or rejected or uncovered for observation, or requires special testing, inspection and approval.
 - c. Verify that tests, equipment and systems startups and operating and maintenance training are conducted in the presence of appropriate personnel, and that Contractor maintains adequate records thereof; and observe, record and report to Engineer appropriate details relative to the test procedures and startups.
 - d. Accompany visiting inspectors representing public or other agencies having jurisdiction over the Project, record the results of these inspections and report to Engineer.
- 6. Interpretation of Contract Documents: Report to Engineer when clarifications and interpretations of the Contract Documents are needed and transmit to Contractor clarifications and interpretations as issued by Engineer.
- Modifications: Consider and evaluate Contractor's suggestions for modifications in Drawings or Specifications and report with RPR's recommendations to Engineer. Transmit to Contractor decisions as issued by Engineer.

8. Records:

- a. Maintain at the job site orderly files for correspondence, reports of job conference, Shop Drawings and samples, reproductions of original Contract Documents including all Work Directive Changes, Addenda, Change Orders, Field Orders, additional Drawings issued subsequent to the execution of the Contract, Engineer's clarifications and interpretations of the Contract Documents, progress reports, and other Project related documents.
- b. Complete a Daily Report recording Contractor hours on the job site, weather conditions, data relative to questions or Work Directive Changes, Change Orders or changed conditions, list of job site visitors, daily activities, decisions, observations in general, and specific observations in more detail as in the case of observing test procedures; and send original to Engineer.

c. Record names, address and telephone numbers of all Contractors, subcontractors and major suppliers of materials and equipment.

9. Reports:

- a. Furnish Engineer periodic reports as required of progress of the Work and the Contractor's compliance with the progress scheduled and schedule of Shop Drawings and sample submittals.
- b. Consult with Engineer in advance of schedule major tests, inspections or start of important phases of the Work.
- c. Draft Field Orders, obtain backup material from Contractor and recommend to Engineer Change Orders and Work Directive Changes. Furnish Engineer copies of all Field Orders.
- d. Report immediately to Engineer and Owner upon occurrence of any accident.
- 10. Payment Requests: Review applications for payment with Contractor for compliance with the established procedure for their submission and forward with recommendations to Engineer, noting particularly the relationship of the payment requested to the schedule of values. Work completed and materials and equipment delivered at the site but not incorporated in the Work.
- 11. Certificates, Operation and Maintenance Manuals: During the course of the Work, verify that certificates, operation and maintenance manuals and other data required to be assembled and furnished by Contractor are applicable to the items actually installed and in accordance with the Contract Documents, and have this material delivered to Engineer for review and forwarding to Owner prior to final payment for the Work.

12. Completion:

- a. Before Engineer issues a Certificate of Substantial Completion, prepare and furnish to the Engineer a list of observed items requiring completion or correction.
- b. Conduct final inspection in the company of Engineer, Owner, and Contractor and prepare and furnish to the Engineer a final list of items to be completed or corrected.
- Observe that all items on final list have been completed or corrected and make recommendations to Engineer concerning acceptance.

C. Limitations of Authority

Resident Project Representative:

- Shall not authorize any deviation from the Contract Documents or substitution of materials or equipment, unless authorized by Engineer.
- 2. Shall not exceed limitations of Engineer's authority as set forth in the Agreement or the Contract Documents.
- 3. Shall not undertake any of the responsibilities of Contractor, subcontractors or Contractor's superintendent.
- 4. Shall not advise on, issue directions relative to or assume control over any aspect of the means, methods, techniques, sequences or procedures or construction unless such advice or directions are specifically required by the Contract Documents.
- 5. Shall not advise on, issue directions relative or assume control over safety precautions and programs in connection with the Work.
- 6. Shall not accept Shop Drawing or sample submittals from anyone other than Contractor.
- 7. Shall not authorize Owner to occupy the Project in whole or in part.
- 8. Shall not participate in specialized field or laboratory tests or inspections conducted by others except as specifically authorized by Engineer.

1.9 ARTICLE 12 - CHANGE OF CONTRACT PRICE; CHANGE OF CONTRACT TIMES

A. SC-12.04 through 12.06

1. Add the following paragraphs:

12.04 Start and Completion Times

The date of beginning and the time for completion of the Work are essential conditions of the Agreement and the Work required shall be commenced on a date specified in the Notice to Proceed.

12.05 Time for Completion

Contractor shall proceed with the Work at such rate of progress to insure full completion within the Contract Times. It is expressly understood and agreed, by and between the Contractor and the Owner, that the Contract Times for the completion of the Work described herein is a reasonable time, taking into consideration the adverse weather conditions for the season, or seasons, involved and other factors prevailing in the locality of the Work.

12.06 Liquidated Damages

Contractor understands that time is of the essence and that Owner will suffer financial loss if the Work is not completed within the times or by the dates specified in the Bid Form, plus any extensions thereof allowed in accordance with Article 12 of the General Conditions. Contractor also recognizes the delays, expense, and difficulties involved in proving in a legal or arbitration proceeding the actual loss suffered by Owner if the Work is not completed on time. Accordingly, instead of requiring any such proof, Contractor shall pay Owner as liquidated damages for delay (but not as a penalty) \$1,200 for each day that expires after the time or date specified in the Bid Form for Substantial Completion until the Work is substantially complete. After Substantial Completion, if Contractor shall neglect, refuse or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by Owner, Contractor shall pay Owner \$600 for each day that expires after the time or date specified in the Bid Form for completion and readiness for final payment. Liquidated damages accrued after the Substantial Completion date will be withheld from the project retention until the work is substantially completed. Liquidated damages accrued after the final completion date will also be withheld from the project retention until final completion is achieved. If the amount of project retention is not sufficient enough to cover the amount of accrued liquidated damages for substantial and final completion, further payments to the Contractor will be withheld until to the project is substantially and final completed.

12.07 Engineering Costs after Date for Substantial Completion Contractor understands the Owner will have part-time Resident Project Representative labor costs and expenses after the date established for Substantial Completion. The Contractor understands and agrees to pay the difference between part-time and full-time Resident Project Representative labor costs, expenses, and overtime engineering costs, if any, required after the date established for Substantial Completion.

Owner will pay the cost for such engineering services, overtime engineering services, and out-of-pocket and mileage expenses; and shall deduct such cost from payments due Contractor. Overtime engineering services shall be charged at Engineer's standard hourly rates applied on a time and one-half basis for all time over eight hours on any single working day and for all hours on Saturday, and on a double time basis for all Sunday and Legal Holiday hours.

If the amount due Contractor is not sufficient to cover the cost of overtime engineering services, Contractor shall reimburse Owner in the amount necessary to cover such costs.

12.08 Engineering Costs after Date for Final Completion Contractor understands the Owner will be required to pay for additional engineering time and expenses if the Work is not complete on or before the date established for Final Completion. The Contractor understands and agrees to pay for all engineering costs, including labor and expenses for full-time Resident Project Representative, including overtime engineering costs, required after the date established for Final Completion.

Owner will pay the cost for such engineering services, overtime engineering services, and out-of-pocket and mileage expenses; and shall deduct such cost from payments due Contractor. Overtime engineering services shall be charged at Engineer's standard hourly rates applied on a time and one-half basis for all time over eight hours on any single working day and for all hours on Saturday, and on a double time basis for all Sunday and Legal Holiday hours. If the amount due Contractor is not sufficient to cover the cost of overtime engineering services, Contractor shall reimburse Owner in the amount necessary to cover such costs.

- 1.10 ARTICLE 13 TESTS AND INSPECTIONS; CORRECTION, REMOVAL OR ACCEPTANCE OF DEFECTIVE WORK
 - A. SC-13.10
 - 1. Add new paragraph 13.10 as follows:
 - 13.10 Notification and Time Limit for Repairs:
 - A. Contractor shall be responsible for the proper and safe protection of his work at all times during construction and also during the three-year guarantee period after the acceptance of the completed work by Owner. Contractor shall provide, erect, and maintain barricades, red flags, and torches and lights at all places where work is in progress, and wherever else required by Owner.
 - B. Contractor shall maintain an emergency phone number where he/she can be notified at any time, Sundays and holidays included, of an emergency condition due to the work which requires immediate repair or protection. Upon such notification by Owner, Contractor shall be given a two-hour time limit to provide whatever barricades, flags, torches and lights are required to mark and protect the hazard. If Contractor fails to provide this protection within the two-hour period from time of notification, Owner will provide the necessary protection and deduct the sum of \$200.00 for each occurrence from the monies due and payable to Contractor for completed work.
 - C. Also, upon notification by Owner, Contractor shall be given a 24-hour time limit to begin to make any repairs to the Work as deemed necessary by Owner. If Contractor fails to proceed with necessary repairs within the 24-hour notification period, Owner will make the necessary repairs to the Work and deduct the cost of labor and materials, including engineering costs, for each repair incident from the monies due and payable to Contractor for completed work.

1.11 ARTICLE 14 - PAYMENTS TO CONTRACTOR AND COMPLETION

A. SC-14.01

- 1. Add the following paragraph after paragraph 14.01.A:
 - B. Contractor shall submit revisions to the initial schedule of progress payments whenever actual outlays for the Work vary beyond -5 percent and +10 percent from the schedule, as determined by Engineer.

B. SC-14.02

1. Under paragraph 14.02.A.1, delete the remainder of the first sentence after "(but not more than one a month)" and insert the following:

Contractor shall submit to Engineer for review an original plus four duplicate copies of each Application for Payment and each copy shall be accompanied by a "Sworn Statement For Contractor And Subcontractor To Owner" on a pre-printed or computer generated form similar to Certificate 00 62 76.01.

2. Delete paragraph 14.02.A.3, and substitute the following:

Periodic partial payments shall be for the value of the completed work less a retained amount of 10 percent of the value of completed work as approved by Engineer until construction is 50 percent complete, after which no additional amount will be retained if Contractor is making progress to Owner's satisfaction and there is no specific cause for withholding 10 percent of the total value of completed work. When the project is substantially complete and available for Owner's operational or beneficial occupancy, the retained amount shall be reduced to only that amount estimated by Engineer as necessary to assure completion of the Work, unless, in the opinion of the Engineer and Owner, Substantial and Final Completion will not be achieved by the designated completion dates established by the Notice to Proceed, as described by Article 12.06 of the Supplementary Conditions. The final payment, including the retained amount, shall be payable within 30 days after the completion of the Work, approval by Engineer and acceptance by Owner. The acceptance of the final payment by Contractor shall be considered to be a waiver of all claims against Owner under the Agreement.

C. SC-14.02.C

- 1. Delete paragraph 14.02.C.1. in its entirety, and replace with the following:
 - "1. Following presentation of each Application for Payment to Owner with Engineer's recommendation, the amount recommended (subject to the provisions of Paragraph 14.02.D) will be submitted to the Illinois Environmental Protection Agency (IEPA) for State Revolving Fund monies.

Upon receipt of funds from the IEPA, the Payment will become due, and when due will be paid by Owner to Contractor."

- D. SC-14.03
 - 1. Under paragraph 14.03.A, add the following:
 - B. Contractor shall procure from each Subcontractor and Supplier of material or labor a waiver of any claim which they may have under the mechanics lien laws of the state in which the Work is located, to insure Owner immunity from mechanics liens on account of anything which is done by Contractor or his Subcontractors in carrying out the Agreement and any work orders for additions thereto, all as a condition of any payment by Owner. Any payments made by Owner without requiring compliance with this paragraph shall not be construed as a waiver by Owner of the right to require compliance with this paragraph as a condition of later payments. Contractor shall furnish with his final Application for Payment a complete release of all liens arising out of this contract, or receipts in full in lieu thereof and an affidavit that the releases and receipts include all labor and material for which a lien could be filed.

E. SC-14.07

1. Under paragraph 14.07.C, change "Thirty" to "Sixty".

1.12 ARTICLE 18 - EQUAL EMPLOYMENT OPPORTUNITY - AFFIRMATIVE ACTION NOTICE

18.01 Equal Employment

- A. Contractor shall comply with the following Equal Employment Opportunity Affirmative Action Clause required by the Illinois Department of Human Rights:
 - 1. In the event the Contractor's noncompliance with any provision of this Equal Employment Opportunity Affirmative Action Clause, the Illinois Human Rights Act or the Department of Human Rights Rules and Regulations for Public Contracts, the Contractor may be declared nonresponsible and therefore ineligible for future contracts or subcontracts with the State of Illinois or any of its political subdivisions or municipal corporations, and the contract may be cancelled or voided in whole or in part, and such other sanctions or penalties may be imposed or remedies invoked as provided by statute or regulation. During the performance of the Work under this Agreement, the Contractor agrees as follows:
 - a. That it will not discriminate against any employee or applicant for employment because of race, color, religion, sex, national origin, ancestry, physical or mental handicap unrelated to ability, or an unfavorable discharge from military service; and further, that it will examine all job classifications to determine if minority persons or women are underutilized and will take appropriate affirmative action to rectify any such underutilization.
 - b. That, if it hires additional employees in order to perform the Work under this Agreement, or any portion hereof, it will determine the availability (in accordance with the Department's Rules and Regulations for Public Contracts) of minorities and women in the area(s) from which it may

- reasonably recruit and it will hire for each job classification for which employees are hired in such a way that minorities and women are not underutilized.
- c. That it will have written sexual harassment policies that shall include, at a minimum, the following information: the illegality of sexual harassment; the definition of sexual harassment under State law; a description of sexual harassment, utilizing examples; the Contractor's internal complaint process including penalties; the legal recourse, investigative and complaint process available through the Department of Human Rights and the Illinois Human Rights Commission; directions on how to contact these State agencies; and protection against retaliation as provided by Section 6-101 of the Illinois Human Rights Act.
- d. That, in all solicitations or advertisements for employees placed by it or on its behalf, it will state all applicants will be afforded equal opportunity without discrimination because of race, color, religion, sex, national origin, ancestry, physical or mental handicap unrelated to ability, or an unfavorable discharge from military service.
- e. That it will send to each labor organization or representative of workers with which it has or is bound by a collective bargaining or other agreement or understanding, a notice advising such labor organization or representative of the Contractor's obligations under the Illinois Human Rights Act and the Department's Rules and Regulations for Public Contracts. If any such labor organization or representative fails or refuses to cooperate with the Contractor in its efforts to comply with such Act and Rules and Regulations, the Contractor will promptly so notify the Illinois Department of Human Rights and the contracting agency and will recruit employees from other sources when necessary to fulfill its obligations thereunder.
- f. That it will submit reports as required by the Illinois Department of Human Rights' Rules and Regulations, furnish all relevant information as may from time to time be requested by the Department or the contracting agency, and in all respects comply with the Illinois Human Rights Act and the Department's Rules and Regulations.
- g. That it will permit access to all relevant books, records, accounts and work sites by personnel of the contracting agency and the Illinois Department of Human Rights for purposes of investigation to ascertain compliance with the Illinois Human Rights Act and the Department's Rules and Regulations.
- h. That it will include verbatim or by reference the provisions of Paragraphs a through h of this clause in every performance subcontract as defined in Section 1.1(17)(b) of the Department's Rules and Regulations so that such provisions will be binding upon every such Subcontractor; and that it will also so include the provisions of Paragraphs a, f, g, and h in every supply subcontract as defined in Section 1.1(17)(a) of the Department's Rules and Regulations so that such provisions will be binding upon every such Subcontractor. In the same manner as with other provisions of this Agreement, the Contractor will be liable for compliance with applicable provisions of this clause by all its Subcontractors, and further, it will

promptly notify the contracting agency and the Illinois Department of Human Rights in the event any Subcontractor fails or refuses to comply therewith. In addition, no Contractor will utilize any Subcontractor declared by the Department to be nonresponsible and therefore ineligible for contracts or subcontracts with the State of Illinois or any of its political subdivisions or municipal corporations.

2. Contractor and Subcontractor shall in turn include this Equal Employment Opportunity - Affirmative Action Clause in each of its subcontracts verbatim or by reference so that the provisions of Paragraphs a through h of said clause will binding upon Subcontractors of every tier; provided, however, that only Paragraphs a, f, g, and h need be included in every subcontract as defined in Section 2.10(a) of the Rules and regulations of the Illinois Department of Human Rights.

1.13 ARTICLE 19 – PREVAILING WAGE RATES

- A. SC-19.01 Prevailing Wage Rates
 - Contractor shall pay prevailing wages:
 - In accordance with the Davis-Bacon Wage Act (40 USC 276a through 276a-5) or Illinois Prevailing Wage Act (820 ILCS 130), whichever is greater.
 - b. At the rates determined by the United States Department of Labor as being not less than those prevailing under the Davis-Bacon Wage Act Provisions.
 - 2. If the Department of Labor revises prevailing wage rates, the revised prevailing wage rates on the Department of Labor's website (https://beta.sam.gov) shall apply to this contract and the Contractor will not be allowed additional compensation on account of said revisions. The Contractor shall review the wage rates applicable to the work of the contract at regular intervals to ensure timely payment of current wage rates. The Contractor agrees no additional notice is required. The Contractor shall be responsible to notify each subcontractor of the wage rates set forth in this contract and any revisions thereto.
 - 3. It shall be mandatory upon the Contractor or construction manager to whom a contract for public works is awarded to post, at a location on the project site of the public works that is easily accessible to the workers engaged on the project, the prevailing wage rates for each craft or type of work or mechanic needed to execute the contract or project or work to be performed.
 - 4. Contractor shall provide certified payrolls each week to demonstrate compliance with regulations.

1.14 ARTICLE 20 – STEEL PRODUCTS PROCUREMENT

- A. SC-20.01 Steel Products Procurement
 - 1. Contractor shall comply with the provisions of Illinois Steel Products Procurement Act (30 ILCS 565/) as it may be amended from time to time.

1.15 ARTICLE 21 - EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORKS ACT

A. SC-21.01 Comply with the provisions of the Employment of Illinois Workers on Public Works Act (30 ILCS 570/) as they may apply to this Project.

1.16 ARTICLE 22 – UNCONTAMINATED SOIL CERTIFICATION AND DISPOSAL

- A. SC-22.01 Comply with the requirements of Sections 22.51(f)(2)(B) and 22.51a(d)(2)(B) of the Illinois Environmental Protection Act ([415 ILCS5/22.51(f)(2)(B)] and [415 ILCS5/22.51a(d)(2)(B)]) for the disposal of uncontaminated soils including uncontaminated soil mixed with other clean construction or demolition debris (CCDD) materials.
- B. SC-22.02 Soils which cannot be certified as uncontaminated or soils that are found to be contaminated during the course of the Work shall be tested and disposed as required for Contaminated Waste Disposal.

END OF SUPPLEMENTARY CONDITIONS

DAVIS-BACON ACT WAGE DETERMINATION SCHEDULE

The Bidder is responsible for obtaining the most recent Davis-Bacon Act Wage Determination from the following website:

https://beta.sam.gov

CERTIFICATE 00 74 23.01

DBE FORM #1 (CONTRACTOR CERTIFICATION)



Illinois Environmental Protection Agency

1021 North Grand Avenue East • P.O. Box 19276 • Springfield • Illinois • 62794-9276 • (217) 782-3397

Disadvantaged Business Enterprise (DBE) Program DBE Form #1 - Contractor Certification

(To be completed by all Prime Contractors)

Plea	se check the appropriate	boxes that apply and cor	mplete the information	on on the bottom of the form.			
	This firm will award no subcontracts (including in the procurement of equipment, supplies, or services), in the performance of this contract.						
	This firm advertised for DBE subcontractors according to the good faith efforts outlined in the IEPA DBE Guidance Document. Attach proof of publication in compliance with pertinent provisions of Disadvantaged Business Enterprise Participation Specifications in Section 00 21 00.41, BIDDER INSTRUCTIONS.						
	This firm received proposals from DBE(s) that will not be utilized. A list of the DBEs not hired, along with their address, phone number, and reason(s) for non-utilization, is below.						
	Name of DBE	Address	Phone	Reason for Non-Utilization			
	This firm did not receive	any inquiries from DBEs.					
poss	tify that the above is true	. I further certify that this in ownership or conflict o	s firm and its partne	rs, directors, and officers do not er authority to control the DBE to			
Sigr	nature:	Na	ame:				
	Date:		Title:				
		Comp	any:				
DBE-1 R	ev. 5/2017	Contractor Certification	ation				

END OF CERTIFICATE 00 74 23.01

DBE FORM #1 (CONTRACTOR CERTIFICATION)
CERTIFICATE 00 74 23.01 (210039.40) IEPA-PWSLP



DBE-3 Rev. 5/2017

DBE FORM #3 (SUBCONTRACTOR UTILIZATION)

Illinois Environmental Protection Agency

1021 North Grand Avenue East • P.O. Box 19276 • Springfield • Illinois • 62794-9276 • (217) 782-3397

Disadvantaged Business Enterprise (DBE) Program DBE Form #3 - Subcontractor Utilization

(Only complete this form if DBE subcontractors or sub-consultants will be working on the project)

This form is intended to capture the DBE subcontractor's description of work to be performed and the price of the work submitted to the prime contractor. All subcontractors must complete this form, and it must be included in the prime contractor's bid package.

Subcontractor Name	Project Name
Contact Person's Name & Title	
Address	
Telephone	Email
DBE Certified By	Select One: MBE WBE SB DBE
Prime Contractor Name	
Type of Work to be Performed	Cost Estimate of Work

I certify under penalty of perjury that the forgoing statements are true and correct. Signing this form does not signify a commitment to using the subcontractors above. I am aware that in the event of a replacement of a subcontractor, I will adhere to the replacement requirements set forth in40 C.F.R. Part 33 Section 33.302(c).

Prime Contractor Signature	Printed Name
Date	Title
Subcontractor Signature	Printed Name
Date	Title

Subcontractor Utilization

END OF CERTIFICATE 00 74 23.02

DBE FORM #3 (SUBCONTRACTOR UTILIZATION)
CERTIFICATE 00 74 23.02 IEPA-PWSLP

CERTIFICATE 00 74 23.03 DBE FORM #4 (BIDDER'S LIST)



Illinois Environmental Protection Agency

1021 North Grand Avenue East • P.O. Box 19276 • Springfield • Illinois • 62794-9276 •(217) 782-3397

Disadvantaged Business Enterprise (DBE) Program DBE Form #4 - Bidders List

(List all DBE and non-DBE proposals submitted to Bidder)

Company Name & Contact Person	Address	Phone Number with area code	Email	Proposed Work (supplies, paint, etc.)	DB	E Status
					○ MBE	() WBE
					SBE	O DBE
					□ check if h	nired
					○ MBE	WBE
					SBE	O DBE
					□ check if h	nired
					○ MBE	○ WBE
					SBE	O DBE
					□ check if h	nired
					○ MBE	○ WBE
					SBE	O DBE
					□ check if h	nired

OSE4 Rev. 5/2017 Bidders List

END OF CERTIFICATE 00 74 23.03

DBE FORM #4 (BIDDER'S LIST)
CERTIFICATE 00 74 23.03 IEPA-PWSLP

BIDDER CERTIFICATION IN COMPLIANCE WITH ARTICLE 33E-11 TO THE "CRIMINAL CODE OF 2012"

I,		, do here	by certify	y that:	
	Name		·		
1.	I amPosition	of the			
	Position		Nam	e of Firm	
	and have authority to execu	ite this certification on bel	nalf of the	e firm;	
2.	This firm is not barred from Rotating violation as set for ILCS 5/33E-11]."				
Name	e of Firm				_
Signa	ture				_
Title _					_
Date _					_
Corpo	orate Seal (where appropriate	·)			
On thi	is day of		20	before me appeare	d
(name	e)	to me personall	y known,	who, being duly sworn,	did
execu	ite the foregoing affidavit, and	d did state that he or she v	vas prope	erly authorized by (Name	of
Firm)			to execut	e the affidavit and did so	as
his or	her free act and deed.				
Notar	y Public	Commis	ssion Exp	oires	
Notar	y Seal				

END OF CERTIFICATE 00 74 23.04

U.S. ENVIRONMENTAL PROTECTION AGENCY CERTIFICATION OF NONSEGREGATED FACILITIES

(Applicable to federally assisted construction contracts and related subcontracts exceeding \$10,000 which are not exempt from the Equal Opportunity clause.)

The federally assisted construction contractor certifies that he does not maintain or provide for his employees any segregated facilities at any of his establishments, and that he does not permit his employees to perform their services at any location, under his control, where segregated facilities are maintained. The federally assisted construction contractor certifies further that he will not maintain or provide for his employees any segregated facilities at any of his establishments, and that he will not permit his employees to perform their services at any location, under his control, where segregated facilities are maintained. The federally assisted construction contractor agrees that a breach of this certification is a violation of the Equal Opportunity clause in this contract. As used in this certification, the term "segregated facilities" means any waiting rooms, work areas, rest rooms and other storage or dressing areas, parking lots, drinking fountains, recreation or entertainment areas, transportation, and housing facilities provided for employees which are segregated by explicit directive or are in fact segregated on the basis of race, creed, color, or national origin, because of habit, local custom or otherwise. The federally assisted construction contractor agrees that (except where he has obtained identical certifications from proposed subcontractors for specific time periods) he will obtain identical certifications from proposed subcontractors prior to the award of subcontracts exceeding \$10,000 which are not exempt from the provisions of the Equal Opportunity clause, and that he will retain such certification in his files.

Signature	Date		
Name and Title of Signer	(Please type)		
Firm Name			

NOTE: The penalty for making false statements in offers is prescribed in 18 U.S.C. 1001.

END OF CERTIFICATE 00 74 23.05

NOTICE TO LABOR UNIONS OR OTHER ORGANIZATIONS OF WORKERS NONDISCRIMINATION IN EMPLOYMENT

Го:
(Name of union or organization of workers)
The undersigned currently holds contract(s) with(Name of applicant)
involving funds or credit of the U.S. Government or (a) subcontract(s) with a prime contractor holding such contract(s).
You are advised that under the provisions of the above contracts(s) or subcontract(s) and in accordance with Executive Order 11246, as amended, dated September 24, 1965, as amended, the undersigned is obliged not to discriminate against any employee or applicant for employment because of race, color, creed or national origin. This obligation not to discriminate in employment includes, but is not limited to, the following:
HIRING, PLACEMENT, UPGRADING, TRANSFER OR DEMOTION, RECRUITMENT,
ADVERTISING, OR SOLICITATION FOR EMPLOYMENT, TRAINING DURING
EMPLOYMENT, RATES OF PAY OR OTHER FORMS OF COMPENSATION, SELECTION
FOR TRAINING INCLUDING APPRENTICESHIP, LAYOFF OR TERMINATION.
This notice is furnished you pursuant to the provisions of the above contract(s) or subcontracts(s) and Executive Order 11246, as amended.
Copies of this notice will be posted by the undersigned in conspicuous places available to employees or applicants for employment.
(Contractor or Subcontractor)
(Date)

END OF CERTIFICATE 00 74 23.06

EPA Project C	ontrol#

United States Environmental Protection Agency Washington, DC 20460

Certification Regarding Debarment, Suspension and Other Responsibility Matters

The prospective participant to the best of its knowledge and belief that it and its principles:

- (a) Are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from covered transactions by any Federal department or agency;
- (b) Have not within a three year period preceding this proposal bee convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to retain, or performing a public (Federal, State or Local) transaction or contract under a public transaction: violation of Federal or State antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property.
- (c) Are not presently indicted for or otherwise criminally or civilly charged by a government entity (Federal, State or Local) with commission of any of the offenses enumerated in paragraph (1) (b) of this certification; and
- (d) Have not within a three-year period preceding this application/proposal had one or more public transactions (Federal, State or Local) terminated for cause or default.

I understand that a false statement on this certification may be ground for rejection of this proposal or termination of the award. In addition, under 18 USC Sec. 1001 a false statement may result in fine of up to \$10,000 or imprisonment for up to 5 years, or both.

(Typed Name & Title of Authorized Representative)	
(Signature of Authorized Representative)	(Date)
I am unable to certify the above statements. N	My explanation is attached.

CERTIFICATE 00 74 23.07-1 IEPA-PWSLP

EPA Form 5700-49 (11-88)

Instructions

Under executive order 12549, an individual or organization debarred or excluded from participation in Federal assistance or benefit programs may not receive any assistance award under a Federal program or subprogram hereunder for \$25,000 or more.

Accordingly, each prospective recipient of an EPA grant, loan, or cooperative agreement and any contract or sub-agreement participant hereunder must complete the attached certification or provide an explanation why they cannot. For further details see 40CRF 32.510. Participants' responsibilities, in the attached regulation.

Where to submit

The prospective EPA grant, loan or cooperative agreement recipient must return the signed certification or explanation with its application to the appropriate EPA Headquarters or Regional Office, as required in the application instructions.

A prospective prime contractor must submit a completed certification or explanation to the prime contractor fro the project.

Each prospective subcontractor must submit a completed certification or explanation to the prime contractor for the project.

How to obtain forms:

EPA includes the certification form, instructions, and a copy of its implementing regulation (40 CRF Part 32) in each application kit. Applicants may reproduce these materials as needed to provide them to their prospective prime contractor, who, in turn may reproduce and provide them to prospective subcontractors.

Additional copies/assistance may be requested from:

Compliance Branch Grants Administration Division (PM-216F) U.S. Environmental Protection Agency 401 M Street, SW Washington, DC 20460 (Telephone: 202-475-8025)

END OF CERTIFICATE 00 74 23.07

EPA Form 5700-49 (11-88)

Know Your Rights Under the Recovery Act!

Did you know?

The American Recovery and Reinvestment Act of 2009 ¹ provides protections for certain employees of non-federal employers who make specified disclosures relating to possible fraud, waste and/or abuse or Recovery Act funds.

Who is protected?

Employees of non-federal employers receiving recovery funds. This includes State and local governments, contractors, subcontractors, grantees or professional membership organizations acting in the interest of recovery fund recipients.

How are Whistleblowers Protected?

You cannot be discharged, demoted or otherwise discriminated against as a reprisal for making a protected disclosure.

What types of disclosures are protected?

The disclosure must be made by the employee to the Recovery Accountability and Transparency Board, an Inspector General, the Comptroller General, a member of Congress, a state or federal regulatory or law enforcement agency, a person with supervisory authority over the employee, a court or grand jury, or the head of a federal agency or his/her representatives.

The disclosure must involve information that the employee believes is evidence of:

- gross mismanagement of an agency contract or grant relating to recovery funds;
- a gross waste of recovery funds;
- a substantial and specific danger to public health or safety related to the implementation or use of recovery funds;
- an abuse of authority related to the implementation or use of recovery funds; or
- a violation of law, rule, or regulation related to an agency contract or grant awarded or issued relating to recovery funds.

Take Action!

Log on to Recovery.gov for more information about your rights and details on how to report at www.recovery.gov.

¹ Section 1553 of Division A, Title XV of the American Recovery and Reinvestment Act of 2009, P.L. 111-5

EMPLOYEE RIGHTS UNDER THE DAVIS-BACON ACT

FOR LABORERS AND MECHANICS EMPLOYED ON FEDERAL OR FEDERALLY ASSISTED CONSTRUCTION PROJECTS

THE UNITED STATES DEPARTMENT OF LABOR WAGE AND HOUR DIVISION

PREVAILING WAGES

You must be paid not less than the wage rate listed in the Davis-Bacon Wage Decision posted with this Notice for the work you perform.

OVERTIME

You must be paid not less than one and one-half times your basic rate of pay for all hours worked over 40 in a work week. There are few exceptions.

ENFORCEMENT

Contract payments can be withheld to ensure workers receive wages and overtime pay due, and liquidated damages may apply if overtime pay requirements are not met. Davis-Bacon contract clauses allow contract termination and debarment of contractors from future federal contracts for up to three years. A contractor who falsifies certified payroll records or induces wage kickbacks may be subject to civil or criminal prosecution, fines and/or imprisonment.

APPRENTICES

Apprentice rates apply only to apprentices properly registered under approved Federal or State apprenticeship programs.

PROPER PAY

If you do not receive proper pay, or require further information on the applicable wages, contact the Contracting Officer listed below:



or contact the U.S. Department of Labor's Wage and Hour Division.



WWW.WAGEHOUR.DOL.GOV

U.S. Department of Labor | Employment Standards Administration | Wage and Hour Division

WH 1321(Revised April 2009)

U.S. Department of Labor Wage and Hour Division		(For Contracto	or's Op	PAYROLL Contractor's Optional Use; See Instructions at www.dol.gov/whd/forms/wh347instr.htm)	PAYROLL tructions at	L t www.d	ol.gov/who	/forms/w	า347instr) (j		U.S. Wage and Hour Division	d Hour Div	vision
NAME OF CONTRACTOR OR SUBCONTRACTOR	OTOR	Persons are not i	paunba	Persons are not required to respond to the collection of information unless it displays a currently valid OMB control number. ADDRESS ADDRESS	ntormation ADDRESS	unless it dis	olays a current	ly valid OMB	control numb	er.		OMB No.: Expires: 1	Rev. Dec. 2008 OMB No.: 1215- Expires: 12/31/	1215-0149 12/31/2011
PAYROLL NO.		FOR WEEK ENDING			PROJECT	PROJECT AND LOCATION	No			PRO	PROJECT OR CONTRACT NO	RACT NO.		
(1)	SNG B	ව	13	(4) DAY AND DATE	(5)	9	6	,		(8) DEDUCTIONS	SN			6
NAME AND INDIVIDUAL IDENTIFYING NUMBER (e.g., LAST FOUR DIGITS OF SOCIAL SECURITY NUMBER) OF WORKER	NO. OF EXEMPTIC	WORK	80.10	HOURS WORKED EACH DAY	TOTAL	RATE OF PAY	GROSS AMOUNT EARNED	FICA	WITH- HOLDING TAX		OTHER	TOTAL ER DEDUCTIONS		NET WAGES PAID FORWEEK
			0				/			2				Î
			w											
			0				/							
	-		ø						Y					T.
			0		Ti		/				/-	-1		Î
			w				/							
			0											
			w					1)		12			'n	Î
			0				/		4.				eç V	ř
			ω				/							
			0				/							
			w				/			3			8	0
			0				/							
			w				\							
			0				/							
			v				/							
While completion of Form WH 347 is optional, it is mandatory for covered contractors and subcontractors performing work on Federally financed or assisted construction contracts to respond to the information contractors and subcontractors with contractors and subcontractors with contractors and subcontractors with contractors and subcontractors with contractors or performance or assisted construction contractor for the wages pale and each employee during the record performance or contractors or and infancing the construction project, accompanied by a signed 'Statement of Contracting particle subcontractors to subcontractors to subcontractors to appreciate agents' contracting for or financing the construction project, accompanied by a signed 'Statement of Contracting nation payors to a signed subcontractor or contractors to subcontractors and subcontractors to subcontractors to subcontractors and subcontractors and subcontractors are contractors and subcontractors to subcontractors are contractors and subcontractors and subcontractors and subcontractors are contractors and subcontractors and subcontractors are contractors and subcontractors and subcontractors are subcontractors and subcontractors areas and subcontractors are subcontractors and subcontractors are s	ory for cover ning work on ly a copy of Bacon preva	ed contractors and sub Federally financed or a all payrolls to the Feder illing wage rate for the v	contractors ssisted co al agency vork perfor	tractors and subcontractors performing work on Federally financed or assisted construction contracts to respond to the information collection conflated by The Coppeland Act all manced or assisted contraction or contracts by Thirais weekly as statement with respect to the wages paid ace and make acen employee during the preveding week. "U.S. Department of Contraction or statement and complete and make ach above acen to the contraction project, accomplemed by a signed "Statement of Complement instanting are correct and complete and make ach above acen acen to complete and make ach above acen acen to the work performed. Do L and federal contracting agencies receiving this information review the information to determine that employees have received legally required wages and fininge benefits.	nced or assis kly a statem struction proj agencies rec	ted construct ant with respe ect, accompa aiving this info	on contracts to re ct to the wages p nied by a signed'	spond to the in aid each emplo Statement of C e information t	formation colle fee during the ompliance" inc	ction contained in preceding week." Icating that the pa at employees have	29 C.F.R. §§ 3 U.S. Departm ayrolls are corre	3, 5.5(a). The Co ent of Labor (DO ct and complete y required wages	peland Act L) regulation and that each	is at th laborer benefits.
				Public Burden Statement	E									
We estimate that is will take an average of 55 minutes to complete this collection, including time for fevrewing instructions, searching externing and manifestion including time for information if you have any other aspect of this collection, including suggestions for reducing this burden, send them to the Administrator, Wage and Hour Division, U.S. Department of Labor, Room \$3502, 200 Constitution Averue, N.W. Washington, D.C. 20210	implete this ct of this co	collection, including tim llection, including sugge	e tor revie stions for	wng instructions, searching existin educing this burden, send them to	g data source the Administ	rs, gathering s rator, Wage a	nd maintaining tr nd Hour Division,	ie data needed U.S. Departme	and completin int of Labor, Re	ig and reviewing ti oom 83502, 200 C	ne collection of Constitution Ave	Information, If youe, N.W.	n nave	
				(rever)										

(b) WHERE FRINGE BENEFITS ARE PAID IN CASH	 Each laborer or mechanic listed in the above referenced payroll has been paid, as indicated on the payroll, an amount not less than the sum of the applicable basic hourly wage rate plus the amount of the required fringe benefits as listed in the confradt, except as noted in section 4(c) below. 	EXCEPTION (CRAFT) EXPLANATION									SIGNATURE	THE WILLFUL FASFICATION OF ANY OF THE ABOVE STATEMENTS MAY SUBLECT THE CONTRACTOR OR SUBCONTRACTOR TO CIVIL OR CRIMINAL PROSECUTION, SEE SECTION 1001 OF TITLE 18 AND SECTION 231 OF TITLE 31 OF THE UNITED STATES CODE.
(b) WHERE FF	(Title)	on the	; that during the payroll period commencing on the g the day of	nalf of said from the full	actor) ons have been made either directly or indirectly seithe deductions as defined in Regulations, Part	inder the Copeland Act, as amended (48 Stat. 948,), and described below:		REWARKS:	quired to be submitted for the above period are nechanics contained therein are not less than the nation incorporated into the contract; that the conform with the work he performed.	period are duly registered in a bona fide ticeship agency recognized by the Bureau of Labor, or if no such recognized agency exists in a l'Training, United States Department of Labor.	NAME AND TITLE NAME AND TITLE	each laborer or mechanic listed in benefits as listed in the contract rograms for the benefit of such
Date	I. (Name of Signatory Party) do hereby state: (1) That I pay or supervise the payment of the persons employed by	(Contractor or Subcontractor)	(Building or Work) day of, and ending the full weekly wares earned that no rehates have	been or will be made either directly or indirectly to or on behalf of said	(Contractor or Subcontractor) weekly wages earned by any person and that no deductions have been made either directly or indirectly from the full wanes earned by any person other than permissible deductions as defined in Remitations. Part	3 (29 C.F.R. Subtitle A), issued by the Secretary of Labor under the Copeland Act. 63 Start. 108, 72 Stat. 967; 76 Stat. 357; 40 U.S.C. § 3145), and described below:			(2) That any payrolls otherwise under this contract required to be submitted for the above period are correct and complete, that the wage rates for laborers or mechanics contained therein are not less than the applicable wage rates contained in any wage determination incorporated into the contract; that the classifications set forth therein for each laborer or mechanic conform with the work he performed.	(3) That any apprentices employed in the above period are duly registered in a bona fide apprenticeship program registered with a State apprenticeship agency recognized by the Bureau of Apprenticeship and Training, United States Department of Labor, or if no such recognized agency exists in a State, are registered with the Bureau of Apprenticeship and Training, United States Department of Labor.	(4) That: (a) WHERE FRINGE BENEFITS ARE PAID TO APPROVED PLANS, FUNDS, OR PROGRAMS	in addition to the basic hourly wage rates paid to the above referenced payroll, payments of fringe have been or will be made to appropriate phanchouses except as noted in section Alcharda.

APPENDIX H

		LABOR STA	AND	ARDS INT	ERVIEW				
CONTRACT NUMBE	ER	90-04 - Marining Contrago (1994) - 1993	104-11-11-11-1		EMPLOY	EE INFORMA	TION		
				LAST NAME		FIRST NAMI			MI
NAME OF PRIME CO	ONTRACTOR			070557 4000					1
NAME OF EMPLOY	FR			STREET ADDR	E55				
WANTE OF EARL EOTH	-10			CITY			STATE	ZIP CODE	
	SUPE	RVISOR'S NAME							
LAST NAME		FIRST NAME	MI	WORK CLASSI	FICATION		WAGE RA	TE	
-		ACTIO) NI					CHECK	BELOW
		AOTIC	J14					YES	NO
Do you work ov	er 8 hours per	day?							
Do you work ov	er 40 hours pe	er week?							
Are you paid at	least time and	d a half for overtime hours?							
Are you receiving	ng any cash pa	ayments for fringe benefits require	ed b	y the posted w	vage determination	decision?			
WHAT DEDUCTION	S OTHER THAN	TAXES AND SOCIAL SECURITY ARE MA	ADE F	ROM YOUR PAY	?				
	S DID YOU WORK	ON YOUR LAST WORK DAY BEFORE			TOOLS	YOU USE			
THIS INTERVIEW?									
DATE OF LAST WO	RK DAY BEFORE	EINTERVIEW (YYMMDD)							
DATE YOU BEGAN	WORK ON THIS	PROJECT (YYMMDD)							
		THE ABOVE IS CORRE	FOT T	O THE BEST OF	MAY KNIOVAS EDOE				
EMPLOYEE'S SIGN.	ATURE	THE ABOVE IS CORRE	ECTI	O THE BEST OF	WIT KNOWLEDGE			DATE (Y	YMMDD)
	-								
INTERVIEWER	SIGNATURE			TYPED OR P	RINTED NAME			DATE (Y	/MMDD)
		INTERVI	EWE	R'S COMME	NTS				
WORK EMPLOYEE	WAS DOING WH	EN INTERVIEWED		ACTION (If	explanation is needed, ι	se comments	section)	YES	ИО
				IS EMPLOYE	EE PROPERLY CLASSII	FIED AND PAI	D?		
				ARE WAGE	RATES AND POSTERS	DISPLAYED?			
			BY PA	AYROLL CHE	CKER			1	
	ATION IN AGREE NO	EMENT WITH PAYROLL DATA?							
COMMENTS	NO								
COMMENTS									
			CH	IECKER	Transpire -				
LAST NAME		FIRST NAME		MI	JOB TITLE				
SIGNATURE					l			DATE (Y	YMMDD)
AUTHORIZED FOR	IOCAL PEDDOD	LICTION				STANDAG	D EOD	14445	E) / 10.00
Previous edition not		OUTION				STANDAF Prescribed by			

SECTION 01 22 29

MEASUREMENT AND PAYMENT

PART 1 - GENERAL

1.1 SUMMARY

- A. Payment will be made for the Pay Items listed herein and at the Contract Unit Prices unless otherwise indicated.
- B. Work for which a Pay Item is not listed will be considered incidental to the Contract and no additional compensation will be allowed.
- C. The Owner reserves the right to alter the Drawings, modify incidental work as may be necessary, and increase or decrease quantities of work to be performed including deduction or cancellation of any one or more of the Pay Items.
- D. Quantities necessary to complete the work as shown on the Drawings or as specified herein govern over those shown in the Bid Form. The Engineer will make corrections and interpretations as may be deemed necessary for fulfillment of the intent of the Contract Documents.
- E. The Engineer will make measurements and determinations as necessary to classify the work within Pay Items and determine the quantities for payment.
- F. If the Contractor's Application for Payment is not in accordance with the Engineer's determination of quantities for payment, the Application will be returned for correction and resubmittal.

1.2 PRECONSTRUCTION VIDEO RECORDING

A. Description:

- 1. The work of this Pay Item consists of video recording at the construction areas prior to start of construction.
- 2. This Pay Item includes providing separate video and photographs of highway rights-of-way, prior to construction, at locations where pipelines are installed parallel to or across highways and the highway authorities require this documentation as a condition of the permit.
- 3. This Pay item includes providing video in tape format, and/or digital disk format if required by Owner or permitting agency.

B. Basis of Payment:

 The work will be paid for at the Contract Lump Sum Price for PRECONSTRUCTION VIDEO RECORDING.

1.3 TRAFFIC CONTROL AND PROTECTION

A. Description:

- 1. The work of this item consists of furnishing, installing, relocating, and maintaining traffic control devices for temporary control of vehicular and pedestrian traffic during construction. The following work items are associated with and shall be considered incidental to, this work:
 - a. Furnish, install, maintain, relocate, and remove traffic control devices.
 - b. Furnish and use temporary flaggers.
 - c. Furnish, place, and remove devices utilized for detours and construction safety.
 - d. Provide plans and schedules for detour and construction signage at Pre-construction Meeting for review and approval by Owner.
 - e. Provide and maintain temporary pavement markings (TAPE) until placement of final pavement markings.
 - f. Remove temporary pavement markings (TAPE) prior to installation of bituminous binder or surface courses.

B. Measurement:

 This work will be measured as a lump sum, paid as a direct percentage of total project completion.

C. Basis of Payment:

1. This work will be paid for at the Contract Lump Sum Price for TRAFFIC CONTROL AND PROTECTION.

1.4 GRANULAR TRENCH BACKFILL

- This work shall conform to Section 208 of the "Standard Specifications". This item shall consist of furnishing, placing, compacting and transporting course aggregate gradation CA-6 or CA-7 for backfilling material for all trenches made within the roadway, driveways, sidewalks, and as directed by the Engineer. All compaction shall be not less than 85 percent optimum, modified proctor. The material shall conform to Article 1004.5. This item also includes the disposal of surplus excavated material which is replaced by trench backfill. Such disposal shall be made in accordance with Article 202.3 of the "Standard Specifications".
- 2. The Contractor shall maintain trenches flush with existing surfaces until permanent patches are installed or roadway excavation has begun. No additional compensation will be provided to the Contractor for trench maintenance.
- 3. Trench backfill quantities are measured from a point 12" above the top of pipe.

1. The work will be paid for at the Contract Unit Price per cubic yard for GRANULAR TRENCH BACKFILL. Quantities incorporated into this item shall be measured by either the Trench Backfill table provided in the technical specifications for water main installation or the Trench Backfill table published by the State of Illinois, Department of Transportation, Division of Highways, Bureau of Construction for sewer replacements.

VOL	JME O	F TRENC	Н ВАСКЕ	FILL (CU Y	/D) PER I	LINEAL F	OOT OF	WATER I	MAIN
Nominal Pipe S	ize	4"	6"	8"	10"	12"	14"	16"	24" Casing
	2.0	0.08	0.08	0.09	0.09	0.10	0.11	0.11	0.13
	2.5	0.11	0.12	0.13	0.14	0.15	0.16	0.17	0.20
	3.0	0.15	0.16	0.17	0.18	0.20	0.21	0.22	0.38
	3.5	0.18	0.20	0.21	0.23	0.38	0.40	0.42	0.47
	4.0	0.22	0.40	0.42	0.44	0.46	0.48	0.50	0.56
	4.5	0.45	0.47	0.49	0.51	0.54	0.56	0.58	0.65
Average Depth to Top of Pipe	5.0	0.51	0.53	0.56	0.59	0.61	0.64	0.66	0.75
	5.5	0.57	0.60	0.63	0.66	0.69	0.72	0.75	0.84
	6.0	0.63	0.67	0.70	0.73	0.76	0.80	0.83	0.93
	6.5	0.70	0.73	0.77	0.80	0.84	0.88	0.91	1.02
	7.0	0.76	0.80	0.84	0.88	0.92	0.95	0.99	1.12
	7.5	0.82	0.87	0.91	0.95	0.99	1.03	1.08	1.21
	8.0	0.89	0.93	0.98	1.02	1.07	1.11	1.16	1.30

1.5 CONTROLLED LOW STRENGTH FLOWABLE FILL MIXTURE (CLSM)

- 1. The work of this Pay Item consists of backfilling trenches and excavation pits within Will County Right-of-Way, from the top of the pipe covering material to a point 12 inches below finished pavement, and extending 2 feet beyond edge of pavement, complete in place, including installation and removal of forms; and cleanup.
- 2. CLSM shall be used for all excavations within pavement and within 2 feet of pavement (e.g. street, driveway, etc.) within the Will County Right-of-Way unless the trenching operations is more than 50 LF (or 500 SF).
- 3. A maximum width of 6 feet is allowed for the open cut water main installation. Excess CLSM needed for open cut trenching for the water main beyond this limit is considered incidental to this Pay Item.

- 4. The aggregate required above the controlled low strength material will not be paid for and is considered incidental to this Pay Item.
- 5. CA-6 may be used in lieu of CLSM within Will County Right-of-Way if the trenching operation is more than 50 LF or 500 SF.

B. Measurement:

 Controlled low strength material will be measured in cubic yards, based on approved load tickets.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per cubic yard for CONTROLLED LOW STRENGTH FLOWABLE FILL MIXTURE (CLSM).

1.6 DUCTILE IRON WATER MAIN (OPEN CUT)

- The work of this Pay Item consists of water main pipe and restrained joint type (RJT) water main pipe complete in place, including sawcutting, and removal and disposal of existing pavements; excavation; removal and disposal of waste excavated materials; protection, replacement, or repair of utilities; trench dewatering, including erosion and sedimentation control methods and devices to provide protection to environment from all pumping operations; installation of pipe; bracing; bedding and covering of pipe; trench backfilling with and compaction of excavated materials; testing; disinfection, finish grading; but not including backfilling with granular backfill materials.
- This Pay Item includes all ductile iron pipe restrained joint type (RJT) fittings complete in place as shown on the Drawings. Additional fittings not shown on the Drawings shall be paid for as MISCELLANEOUS FITTINGS.
- 3. This Pay Item includes RJT pipe on each side of a fitting or valve to be restrained to the length indicated on the Drawings, or if not shown on the Drawings, to a distance indicated on the Restrained Joint Type Pipe Length table below or a minimum distance of not less than two factory length pipes on each side of a fitting or valve to be restrained.
 - A restrained joint between two sections of pipe serves to restrain both pipe sections, and both sections of pipe would be considered RJT pipe.
- 4. Installing new mains in excess of 5 feet of cover in order to cross existing mains, provide for future improvements or cross below sewer lines is considered incidental to the installation of the water main, and no addition to the contract will be allowed.
- 5. This Pay Item includes polyethylene wrapping of all pipe.
- 6. Removal and replacement of material, including unsuitable material, to a depth of one foot below the bottom of the pipe barrel is considered incidental to construction and no addition to the contract will be allowed.
- 7. Overexcavation, and removal and replacement of unsuitable materials with CA 1 greater than one foot below the bottom of the pipe barrel will be paid for in a separate Pay Item.

8. This Pay Item includes pipe insulation at locations indicated on the Drawings and as directed by the Engineer.

B. Measurement:

- 1. The work will be measured in lineal feet along the centerline of the pipe.
- 2. The measurement will go through fittings and valves.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per lineal foot for DUCTILE IRON WATER MAIN (OPEN CUT) of the pipe sizes and type, regardless of depth.

	Restraine	ed Joint Type	e Pipe Lengt	h (Feet)		
Fitting True			Diam	eter (in)		
Fitting Type	4"	6"	8"	10"	12"	16"
Horizontal Bend						
11.25°	5	5	5	5	5	5
22.5°	5	5	5	5	10	10
45°	5	10	10	10	15	15
90°	15	20	20	25	30	40
Vertical Bend						
45°	20	25	35	40	45	60
Tee	5	5	5	5	5	5
Valve (Operating Pressure)	20	30	35	45	50	65
Reducer						
6" Nominal Pipe	30	-	-	-	-	-
8" Nominal Pipe	55	35	-	-	-	-
10" Nominal Pipe	75	55	30	-	-	-
12" Nominal Pipe	90	80	55	35	-	-
16" Nominal Pipe	125	115	100	80	60	-

1.7 MISCELLANEOUS FITTINGS

A. Description:

The work of this Pay Item consists of any additional fittings necessary to alter or adjust the water main as directed by the Engineer. All fittings shown or inferred by the plans shall be incidental to the cost per lineal foot of DUCTILE IRON WATER MAIN (OPEN CUT). Any fitting indicated on the plans which is deleted will be credited against any claim for payment under MISCELLANEOUS FITTINGS, pound for pound. The Owner shall not in any case request a net credit under this provision.

1. The work will be paid for at the contract unit price per pound for MISCELLANEOUS FITTINGS.

1.8 PROJECT MANAGEMENT

A. Description:

- The work of this Pay Item shall include all supervision work associated with coordinating and performing the water service installation at a given property. This item is only applicable when an existing lead service is discovered and where water service work on private property is required. A three hour allotment will be allowed for each property. This work will include contacting the owner, meeting with the property owner to layout the installation, obtaining property owner signature on City agreement form, payment plan, no participation form or sample refusal form, providing City with sketch of proposed and as-built installation, scheduling the work and any follow-up meetings to explain the installation. No additional compensation will be allowed for the management of the project.
- 2. Photographs or video shall be taken at each property prior to the start of construction work to document the pre-existing conditions.
- 3. Monthly invoices should provide a breakdown of items and cost per property. Costs associated with preparing this breakdown shall be incidental to the contract.

B. Basis of Payment:

This work will be paid for 1 hour for each property where the property declines to participate and for up to 3 hours for each property where a water service is installed on private property. This work will be paid for at the Contract Unit Price per hour for PROJECT MANAGEMENT. No additional compensation will be allowed.

1.9 WATER SERVICE CONNECTION TO WATER MAIN AND CURB STOP

- 1. The work of this Pay Item consists of connecting water service lines to the water mains complete in place, including service saddles; corporation stops; curb stops; service boxes; all required fittings.
- 2. The tap into the existing public water main will be performed by the City of Joliet.
- 3. This Pay Item includes connecting to the curb stop within meter pits, where meter pits are encountered.
- 4. Water service connection trenches and pits shall be restored at the end of each work day.
- 5. Notify the City's Public Utilities Department if existing lead water services are discovered.
- 6. The work of this Pay Item includes all labor, equipment and materials required to remove the top section of existing b-boxes located by the Contractor and as required by the Engineer.

 The work will be paid for at the Contract Unit Price for each WATER SERVICE CONNECTION TO WATER MAIN AND CURB STOP of the service pipe sizes, and the item noted.

1.10 WATER SERVICE LINE TYPE K

A. Description:

- 1. The work of this Pay Item shall include furnishing and installing all materials required to lay water service piping in accordance to Sections 41-2.11c and 41-2.12 of the "Standard Specifications for Water and Sewer Construction in Illinois latest Edition". The work consists of water service pipe complete in place, by horizontal directional drilling method and open cut method, including excavation; bracing; bedding and covering of pipe; trench dewatering; trench backfilling with excavated materials; testing; disinfection; finish grading; removal and disposal of waste excavated materials; protection, and replacement or repair of utilities.
- 2. Embedment up to one foot above the pipe and earthen backfill shall be included in the cost per lineal foot of water service line
- Water service will be installed by horizontal directional drilling method under street pavements and for all rear yard service replacements, and open cut methods at connection points to water main and existing service pipe.
- 4. The open cut method includes backfilling with compacted granular backfill materials.
- 5. All service lines shall have a minimum depth of 54", unless otherwise approved by the Engineer.
- 6. Installing new services in excess of 5 feet of cover in order to cross existing mains, provide for future improvements or cross below sewer lines is considered incidental to the installation of the service, and no addition to the contract will be allowed.
- 7. Contractor may choose to install short side water services via open cut method. However, costs to replace the curb and sidewalk including trench backfill for short side services shall be considered incidental to the contract.

B. Measurement:

1. The work will be measured in lineal feet along the centerline of the pipe.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per lineal foot for WATER SERVICE TYPE K – OPEN CUT or WATER SERVICE LINE TYPE K – DIRECTIONAL DRILLING of the water service sizes specified.

1.11 WATER SERVICE LINE TYPE L

A. Description:

1. This work shall include furnishing and installing all materials required to connect existing interior plumbing to new water service location. All fittings required to complete this work shall be considered incidental.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per lineal foot for WATER SERVICE LINE TYPE L for the size specified.

1.12 WATER SERVICE CONNECTION ON PRIVATE PROPERTY – BASEMENT OR CRAWL SPACE

- This item shall include furnishing and installing all materials required to connect to the existing water service on interior of home up to the water meter where lead water services are discovered. The Contractor shall core drill the existing foundation wall to allow for penetration by the 1" copper water service pipe. The Contractor shall install a PVC sleeve in the foundation to protect the copper service pipe. The interior and exterior of the cored hole shall be completely sealed with hydraulic cement to prevent water infiltration. The Contractor shall exercise caution to prevent damaging the existing foundation. The Contractor shall be responsible for repairing any damage to the foundation caused by the coring operation. The Contractor shall be responsible for moving and restoring any items on the interior of the home that infringe upon the work. The Contractor is responsible for any modifications to the interior plumbing necessary to install new water service. New interior water shut-off valves (ball valves) shall be installed on either side of meter. If existing meter is determined to be in need of replacement, Contractor shall install new meter provided by Citv.
- 2. The Contractor shall be responsible for removing any debris generated by the work on the interior and exterior of the building and restore the area around the water service as nearly as practicable to its original condition.
- 3. This work shall also include abandoning the existing water service line in place. The existing plumbing shall be capped.
- 4. This work shall include all labor, equipment and materials required to excavate and crimp existing water service in order to complete interior water service connection at locations approved by Engineer where both the curb stop and interior shut-off are broken or where curb stop cannot be located. This work will be considered incidental.
- 5. This work shall consist of the removal and re-erection of fence segments as necessary to complete the work. The fence shall be removed so that all materials considered suitable by the Engineer for future use will be salvaged. The salvaged material shall be stored at locations and in a manner approved by the Engineer. Any of the material having salvage value and which has been damaged by the Contractor, shall be replaced

by the Contractor at his/her own expense with new material of the same kind. The salvaged fence shall be re-erected in accordance with Section 664 of the "Standard Specifications". This work will be considered incidental.

B. Basis of Payment:

 The work will be paid for at the Contract Unit Price for each WATER SERVICE CONNECTION ON PRIVATE PROPERTY – BASEMENT OR CRAWL SPACE.

1.13 WATER SERVICE CONNECTION ON PRIVATE PROPERTY – SLAB

- 1. This item shall include furnishing and installing all materials required to connect to the existing water service on interior of home up to the water meter where lead water services are discovered. The Contractor shall saw cut and remove a portion of the slab or core drill the slab to allow for penetration by the 1" copper water service pipe at the existing water service location. The Contractor shall install a PVC sleeve in the foundation when restoring the slab to protect the copper service pipe. The interior and exterior of the cored hole shall be completely sealed with hydraulic cement to prevent water infiltration. The Contractor shall exercise caution to prevent damaging the existing foundation. The Contractor shall be responsible for repairing any damage to the foundation caused by the work. The Contractor shall be responsible for moving and restoring any items on the interior of the home that infringe upon the work. The Contractor is responsible for any modifications to the interior plumbing necessary to install new water service. New interior water shut-off valves (ball valves) shall be installed on either side of meter. If existing meter is determined to be in need of replacement, Contractor shall install new meter provided by
- 2. The Contractor shall be responsible for removing any debris generated by the work on the interior and exterior of the building and restore the area around the water service as nearly as practicable to its original condition.
- 3. This work shall also include abandoning the existing water service line in place. The existing plumbing shall be capped.
- 4. This work shall include all labor, equipment and materials required to excavate and crimp existing water service in order to complete interior water service connection at locations approved by Engineer where both the curb stop and interior shut-off are broken or where curb stop cannot be located. This work will be considered incidental.
- 5. This work shall consist of the removal and re-erection of fence segments as necessary to complete the work. The fence shall be removed so that all materials considered suitable by the Engineer for future use will be salvaged. The salvaged material shall be stored at locations and in a manner approved by the Engineer. Any of the material having salvage value and which has been damaged by the Contractor, shall be replaced by the Contractor at his/her own expense with new material of the same

kind. The salvaged fence shall be re-erected in accordance with Section 664 of the "Standard Specifications". This work will be considered incidental.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each WATER SERVICE CONNECTION ON PRIVATE PROPERTY – SLAB.

1.14 UTILITY LOCATION – PRIVATE PROPERTY

A. Description:

 This work shall include all labor, equipment and materials associated with locating the existing water and sewer services on private property. Where locates are not provided for the water service, sewer service, or buffalo box, the Contractor shall complete locating services for means of installing new service line.

B. Basis of Payment:

 The work will be paid for at the Contract Unit Price for each UTILITY LOCATION – PRIVATE PROPERTY.

1.15 WATER METER RELOCATION

A. Description:

- 1. This work shall include all labor and materials required to relocate existing water meter to new water service entry point into home. Meter shall be located immediately upon entrance of the service pipe into the home approximately 3' off the ground. Meter shall be installed such that it is accessible for maintenance. Contractor to provide and install new shut-off valves (ball valves) on each side of the meter, drain connection between two shut-off valves, and #4 solid copper grounding wire between upstream and downstream with bronze ground clamps. Contractor shall install AMR wire to outside face of building. AMR (Automated Meter Reading) wire will be provided to Contractor by City. City staff will relocate AMR.
- 2. This pay item is only applicable when new water service location is required.

B. Basis of Payment:

 The work will be paid for at the Contract Unit Price for each WATER METER RELOCATION.

1.16 WATER METER REPLACEMENT

A. Description:

This work shall include all labor and materials required to replace existing water meter to new water service entry point into home. Meter shall be located immediately upon entrance of the service pipe into the home approximately 3' off the ground. Meter shall be installed such that it is

accessible for maintenance. Contractor to provide and install new shut-off valves (ball valves) on each side of the meter, drain connection between two shut-off valves, and #4 solid copper grounding wire between upstream and downstream with bronze ground clamps. Contractor shall install AMR wire to outside face of building. AMR (Automated Meter Reading) wire will be provided to Contractor by City. City staff will relocate AMR.

2. Ally, iPERL, and Accustream meters with digital LCD readers should not be replaced. Accustream meters with ICE register may be replaced only if not programmed for 8-digit read (see Appendix H). All brass type meters encountered shall be replaced.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each WATER METER REPLACEMENT.

1.17 CONNECT TO WATER MAIN (NON-PRESSURE)

A. Description:

- 1. The work of this Pay Item consists of connecting to existing water mains, including removal of existing plugs; cutting, removing, and plugging existing mains; all required fittings; thrust blocking; excavation; bracing; bedding and covering of pipe; trench dewatering; trench backfilling with excavated materials; testing; disinfection; finish grading; removal and disposal of waste excavated materials; protection, replacement, or repair of utilities; including backfilling with compacted granular backfill materials where required.
- 2. This Pay Item includes installation and removal of temporary thrust blocking and temporary plugs and coordination of shutdown of main with Owner and Engineer during final connection.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each CONNECT TO WATER MAIN (NON-PRESSURE) of the pipe sizes.

1.18 CONNECT TO WATER MAIN (PRESSURE)

A. Description:

1. The work of this Pay Item consists of pressure connections to existing water mains complete in place, including sawcutting, and removal and disposal of existing pavements; excavation; removal and disposal of waste excavated materials; trench dewatering, including erosion and sedimentation control for discharge resulting from all pumping operations; protection, replacement, or repair of utilities; tapping of pipe; tapping valves and tapping sleeves; all required fittings; bracing; testing corps and copper; bedding and covering of pipe; testing; disinfection; finish grading; and including backfilling with granular backfill materials; but not including valve vaults.

2. The City of Joliet Water Department shall do the actual tapping of the main. The Contractor will be responsible for calling 815-724-4230 to schedule an appointment with the City Water Department. Appointments are available on Tuesdays and Thursdays.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each CONNECT TO WATER MAIN (PRESSURE) of the pipe and valve sizes.

1.19 GATE VALVE

A. Description:

1. The work of this Pay Item consists of gate valves complete in place, but not including fire hydrant auxiliary valves or tapping valves.

B. Basis of Payment:

 The work will be paid for at the Contract Unit Price for each GATE VALVE of the valve sizes.

1.20 VALVE BOX

A. Description:

The work of this Pay Item consists of valve boxes, complete in place, including polyethylene wrapping of valve box; valve box stabilizer; but not including fire hydrant auxiliary valves.

B. Basis of Payment:

1. The work will be paid for at the contract Unit Price for each VALVE BOX.

1.21 VALVE VAULT

A. Description:

The work of this Pay Item consists of the installation of valve vaults during open cut water main installation complete in place as outlined in the technical specifications and on plan details, including sawcutting, removal and disposal of existing pavements; excavation in excess of that required for standard pipeline trench construction; removal and disposal of waste excavated materials; bracing, sheeting, and shoring; protection, replacement, or repair of utilities; dewatering, including erosion and sedimentation control methods and devices to provide protection to the environment from all pumping operations; vault bedding; monolithic type vault base with watertight flexible pipe connectors where pipes enter vault wall; manhole risers as required to provide proper depth; concentric cone or flat top cover; frame and cover adjusting rings; frame and cover; steps; backfilling with compacted excavated materials in parkways, and compacted granular materials under pavements; and including frame and cover adjustment to final grade at time of street or parkway restoration.

- 2. Provide eccentric cones only on vaults utilized for tapping of existing mains, or when indicated on the Drawings.
- 3. This Pay Item includes providing vaults of sufficient depth to provide the standard depth of cover indicated on the drawings, plus two feet, measured from the top of the water main to the top of the frame and cover. If depth of cover is 6 feet, vaults to a depth of 8 feet from top to main to top of frame and cover will be installed with no addition to the contract allowed. Where deeper vaults are installed, the cost for additional manhole risers required will be paid for in a separate Pay Item.
- 4. This Pay Item includes providing and installing bolt-down watertight frame and cover on all valve vaults.
- 5. This Pay Item includes providing exterior joint protection on all valve vaults.

1. The work will be paid for at the Contract Unit Price for each VALVE VAULT, of the size indicated.

1.22 VALVE VAULT (ADDITIONAL DEPTH)

A. Description:

- The work of this Pay Item consists of providing additional manhole risers for water valve vaults where the depth of cover is over the standard depth described in the pay item for VALVE VAULTS.
- 2. Depth of a standard valve vault is stated as within two vertical feet of the water main depth of cover, measured from top of main to top of cover, in other words, if standard depth of water main is indicated as 6 feet, a vault measuring 8 feet from top of main to top of frame and cover is a standard vault and no additional compensation will be allowed.

B. Measurement:

1. The work will be measured in vertical feet from the top of the water main to the top of the frame and cover, and the amount over the depth of cover plus 2 feet will be the amount of measurement, in even feet.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per vertical foot of VALVE VAULT (ADDITIONAL DEPTH), of the valve vault diameter indicated.

1.23 FIRE HYDRANT

A. Description:

 The work of this Pay Item consists of fire hydrants complete in place, including excavation; bracing; bedding and covering of pipe; auxiliary gate valve and valve box; thrust blocking; hydrant barrel drain washed stone pocket; support; trenching dewatering; trench backfilling with excavated materials; testing; disinfection; finish grading; removal and disposal of waste excavated materials; protection, replacement, or repair of utilities;

- and including backfilling with granular backfill materials where connecting pipe is below pavement.
- 2. This Pay Item includes polyethylene wrapping of connecting pipe, auxiliary valve, and hydrant.
- This Pay Item includes providing a fire hydrant of sufficient height to
 position the hydrant nozzles and break-away flange to the elevations
 indicated in the technical specifications and plan detail. Contractor will
 determine the proper "bury" height based on water main size and depth of
 bury.
- 4. Where depth of water main is required to be greater than standard bury depth, fire hydrant extensions will be paid for as FIRE HYDRANT BARREL EXTENSION.

 The work will be paid for at the Contract Unit Price for each FIRE HYDRANT.

1.24 FIRE HYDRANT BARREL EXTENSION

A. Description:

 The work of this Pay Item consists of providing barrel extensions for fire hydrants when existing or future final grade conditions result in the water main being deeper than the specified depth of bury, including determining the appropriate extension height and extension necessary, and installation of extension.

B. Measurement:

- The work will be measured in vertical foot of extension, determined by the amount added to the fire hydrant bury dimensions outlined in Pay Item FIRE HYDRANT.
- 2. Extensions utilized to obtain the proper amount "bury" on fire hydrants when the water main is not deeper than the specified depth of bury will not be paid for.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for vertical foot of FIRE HYDRANT BARREL EXTENSION, to the nearest foot.

1.25 LINE STOP

A. Description:

The work of this Pay Item consists of the installation of a line stop in an existing water main complete, including locating of existing main; sawcutting, and removal and disposal of existing pavements; excavation; removal and disposal of waste excavated materials; protection, repair, or replacement of existing utilities; dewatering, including erosion and sedimentation control methods and devices to provide protection to the environment from all pumping operations; sheeting; shoring; tapping of

- pipe to install temporary line stop bladder; installation of line stop; and temporary fencing, barricades, and other items needed to provide traffic control and protection and to protect excavation while new valve is being installed.
- 2. The work of this Pay Item includes removal of the line stop bladder, and capping of the tapping valve once new main valve is in place; backfilling of the excavation with compacted granular trench backfill material; but not including surface restoration.

 The work will be paid for at the Contract Unit Price for each LINE STOP, of the size indicated.

1.26 ABANDONMENT OF EXISTING WATER MAINS AND APPURTENANCES

- 1. The work of this Pay Item consists of the abandonment of existing water main, valves, vaults, and fire hydrants including installing plugs and/or caps on existing mains where portions of the existing main are to be abandoned after the new main is in service; coordination of shut-down of mains with Owner and Engineer; excavation and re-excavation of locations; removal of fire hydrant and auxiliary valve box; removal of valve box; removal of frame and cone section of valve vault; protection, replacement, or repair of utilities; removal and disposal of waste excavated materials; cutting of mains; installation of plugs or caps; concrete thrusts blocks; backfilling and compaction of excavated materials or granular trench backfill materials in pavements.
- 2. This work shall include all labor, materials and equipment necessary to remove an existing fire hydrant and auxiliary valve box. Contractor shall excavate a minimum of 3' and saw cut and remove riser pipe.
- 3. This work shall include all labor, materials, and equipment necessary to remove an existing valve box in its entirety as shown on the plans. Contractor shall be responsible for off-site disposal of valve box.
- 4. This work shall include all labor, materials, and equipment necessary to remove the frame and cone section of the vault. Remainder of the vault is to be filled with trench backfill which will be considered incidental to this item. Contractor shall be responsible for off-site disposal of valve vault.
- 5. Deliver valves, valve boxes, fire hydrants, and frames and lids to the Owner at 921 Washington Street, Joliet, Illinois.
- 6. Installation of concrete plugs in abandoned water mains or crimping of water services at the edge of the trench area is considered incidental to the installation of the new water main.
- 7. Removal and disposal of existing water main that must be removed in order to install the new water main or services is considered incidental to construction.
- 8. Length of dead end main shall be no greater than two times the outside diameter of the main.

 The work will be paid for at the Contract Lump Sum Price for each ABANDONMENT OF EXISTING WATER MAINS AND APPURTENANCES, at the locations and types noted.

1.27 FILLING OF EXISTING WATER MAIN PIPE FOR ABANDONMENT

A. Description:

1. The work of this Pay Item shall be completed in accordance with Section 31 23 79 and shall consist of filling existing water main pipe within Will County right-of-way with Low Density Cellular Grout, including concrete brick and mortar plugs; temporary plugging of pipes and subsequent removal of temporary plugs; providing fill, vent, and drain holes in the concrete brick and mortar plugs; and all labor, equipment, and materials.

B. Method of Measurement:

- 1. The work will be measured in cubic yards of Low Density Cellular Grout used to fill the water main pipes.
- 2. Measurement quantity for pipes will be the volume of pipe determined by the length of pipe from inside face of a structure or cut end to inside face of a structure or cut end, and the nominal inside diameter of the pipe.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per cubic yard for FILLING OF EXISTING WTER MAIN PIPE FOR ABANDONMENT.

1.28 SANITARY SEWER REMOVAL AND REPLACEMENT

A. Description:

- The work of this Pay Item consists of the removal and replacement of existing sanitary sewer with Class 52 ductile iron pipe, including sawcutting, and removal and disposal of existing pavement; protection, repair or replacement of utilities; excavation; removal and disposal of waste excavated material; trench dewatering, including erosion and sedimentation control; watertight connections to existing pipes or structures; installation of pipe; bracing, bedding and covering of pipe; granular trench backfill material at pipe crossing and below all pavement; and cleanup.
- 2. Where new water main is being installed parallel to storm sewer, the ductile Replacement shall extend on each side of the crossing until the perpendicular distance from the water main to the sewer is at least 10 feet.

B. Measurement:

 The work will be measured in lineal feet along the centerline of the sanitary sewer pipe removed and replaced.

 The work will be paid for at the Contract Unit Price per lineal foot for SANITARY SEWER REMOVAL AND REPLACEMENT, of the size and type indicated.

1.29 STORM SEWER REMOVAL AND REPLACEMENT

A. Description:

- 1. The work of this Pay Item consists of the removal and replacement of existing storm sewers with Class 52 ductile iron pipe, including sawcutting, and removal and disposal of existing pavement; protection, repair or replacement of utilities; excavation; removal and disposal of waste excavated material; trench dewatering, including erosion and sedimentation control; watertight connections to existing pipes or structures; installation of pipe; bracing, bedding and covering of pipe; granular trench backfill material at pipe crossing and below all pavement; and cleanup.
- 2. Replacement shall extend on each side of the crossing until the perpendicular distance from the water main to the sewer is at least 10 feet.

B. Measurement:

 The work will be measured in lineal feet along the centerline of the storm sewer removed and replaced.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per lineal foot for STORM SEWER REMOVAL AND REPLACEMENT, of the size and type indicated.

1.30 INLET PROTECTION FILTER

A. Description:

1. The work of this Pay Item consists of providing and installing inlet protection filters.

B. Measurement:

1. Inlet protection filters will be measured as per each item.

C. Basis of Payment:

 This work will be paid for at the Contract Unit Price per each INLET PROTECTION FILTER.

1.31 TEMPORARY HOT-MIX ASPHALT PAVEMENT

A. Description:

1. The work of this Pay Item consists of temporary hot-mix asphalt pavement complete in place at pipelines, including subgrade preparation, for use

- during the period between backfilling the trench and constructing the permanent pavement.
- 2. Install temporary hot-mix asphalt pavement only at high traffic areas such as intersections.

B. Measurement:

- The work will be measured in lineal feet along the centerline of the pipe and the quantity computed in square yards, with a maximum width of 6 feet used for payment calculations.
- 2. All areas disturbed shall be restored, but the maximum allowable trench widths used for pay quantity calculations will be 6 feet centered on the pipe. If the Contractor chooses to relocate the trench farther away from the curb and gutter, the pavement patch will extend from the sawcut in the existing pavement to the face of the gutter and the maximum allowable width of 6 feet will be used for pay quantity calculations. If the Owner or Engineer relocates the trench away from the curb and gutter, additional quantities will be calculated based on the actual trench width as approved by the Engineer.
- 3. Areas damaged due to Contractor's neglect or beyond the maximum pay width will not be measured for payment.

C. Basis of Payment:

This work will be paid for at the Contract Unit Price per square yard for TEMPORARY HOT-MIX ASPHALT PAVEMENT regardless of the depth of the pipeline.

1.32 HOT-MIX ASPHALT REMOVAL AND REPLACEMENT

A. Description:

- The work of this Pay Item consists of replacing street pavement that has been removed as part of the water main work. Restoration will consist of hot-mix asphalt binder course and hot-mix asphalt surface course of the thicknesses specified, installed within the trench limits to match the existing surface. The following items are associated with and shall be considered incidental to this Pay Item:
 - a. All quality control testing in accordance with Article 1030.05(d).
 - b. Prime coat applications.
 - c. Sawcutting to provide clean, straight edges is mandatory, regardless of the number of times required.
 - d. The thickness to be used are 2.5" binder and 1.5" surface course for all streets except Briggs St.
 - e. The thickness to be used at the intersection of Briggs St. and Elgin Ave is 10" binder and 2" surface course.

B. Measurement:

- Street restoration of trenches will be measured in place and the area computed in square yards.
 - a. The length will be measured down the centerline of the pipe.

b. All areas disturbed shall be restored, but the maximum allowable trench widths used for pay quantity calculations will be 6 feet centered on the pipe. If the Contractor chooses to relocate the trench farther away from the curb and gutter, the pavement patch will extend from the sawcut in the existing pavement to the face of the gutter and the maximum allowable width of 6 feet will be used for pay quantity calculations. If the Owner or Engineer relocates the trench away from the curb and gutter, additional quantities will be calculated based on the actual trench width as approved by the Engineer.

C. Basis of Payment:

 The street pavement restoration at water main and sewer trenches will be paid for at the Contract unit price per square yard of the thicknesses indicated as HOT-MIX ASPHALT REMOVAL AND REPLACEMENT.

1.33 HOT-MIX ASPHALT DRIVEWAY REMOVAL AND REPLACEMENT

A. Description:

- The work of this Pay Item consists of constructing hot-mix asphalt driveway aprons to match current City standards. The following work items are associated with and shall be considered incidental to, this Pay Item:
 - a. All quality control testing in accordance with Article 1030.05(d).
 - b. Excavate and dispose of existing subgrade material to proposed subgrade elevation.
 - c. Prepare Subgrade.
 - d. Replace steel edging if applicable.
 - e. Install minimum 3" aggregate Base Course.
 - f. Install 1.5-inch hot-mix asphalt base for driveways.
 - g. Install Bituminous Materials (Prime Coat).
 - h. Place hot-mix asphalt Surface Course, (1.5-inch compacted depth).
 - i. Utilize driveway mix for Surface Course if so directed by Engineer.

B. Measurement:

- 1. Driveway removal and replacement will be measured in place prior, and the area computed in square yards.
- 2. Additional quantities of driveway, subbase, or subgrade removed due to neglect by the Contractor, as determined by the Engineer, will not be measured for payment.

C. Basis of Payment:

 This work will be paid for at the Contract Unit Price per square yard for HOT-MIX ASPHALT DRIVEWAY REMOVAL AND REPLACEMENT of the thickness indicated in the Specifications.

1.34 PCC DRIVEWAY REMOVAL AND REPLACEMENT

A. Description:

- 1. The work of this Pay Item consists of constructing Portland Cement Concrete driveway and subgrade as directed by the Engineer. The following work items are associated with, and shall be considered incidental to, this Pay Item:
 - a. All quality control testing in accordance with Article 1020.09.
 - b. Excavate and dispose of existing subgrade material to proposed subgrade elevation.
 - c. Prepare Subgrade.
 - d. Replace steel edging if applicable.
 - e. Install minimum 3" aggregate Base Course.
 - f. Install 6 x 6-W4.0 x W4.0 welded wire fabric.
 - g. Install Portland Cement Concrete.
 - h. Apply Protective Coat.

B. Measurement:

- 1. Driveway removal and replacement will be measured in place, and the area computed in square yards.
- 2. Additional quantities of driveway, subbase, or subgrade removed due to neglect by the Contractor, as determined by the Engineer, will not be measured for payment.

C. Basis of Payment:

1. This work will be paid for at the Contract Unit Price per square yard for PCC DRIVEWAY REMOVAL AND REPLACEMENT.

1.35 GRAVEL SHOULDER RESTORATION

A. Description:

- 1. The work of this Pay Item consists of complete restoration of gravel shoulders removed as part of the trenching work, including removal and disposal of existing pavement materials, sawcutting, and subgrade preparation.
- 2. Sawcutting to provide clean, straight edges is mandatory, regardless of the number of times required, and is incidental to installation of the pavement.

B. Measurement:

1. Gravel shoulders will be 18 inches wide by 6-inch in depth, and will be measured in lineal feet parallel to the street.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per lineal foot for gravel shoulder as GRAVEL SHOULDER RESTORATION.

1.36 CONCRETE CURB AND GUTTER REMOVAL AND REPLACEMENT

A. Description:

- 1. The work of this Pay Item consists of constructing concrete curb and gutter of the type indicated. The following work items are associated with, and shall be considered incidental to, this Pay Item.
 - a. All quality control testing in accordance with Article 1020.09.
 - b. Sawcut existing curb and gutter, including steel reinforcement.
 - c. Break, remove, and dispose of existing curb and gutter.
 - d. Prepare aggregate Base Course below curb and gutter.
 - e. Install Combination Concrete Curb and Gutter.
 - f. Install all expansion joints, contraction joints, slip dowels, and tie bars.
 - g. Install Protective Coat.
 - h. Provide handicap ramps at sidewalks and depressed curb at driveways.

B. Measurement:

- Curb and Gutter Removal and Replacement will be measured in lineal feet along the flow line of the gutter from full-depth sawcut to full-depth sawcut, with a maximum pay quantity of 10 lineal feet per crossing unless otherwise noted on the Drawings.
- Curb and gutter damaged by the Contractor during construction operations beyond the limits described above, will be replaced but will not be measured for payment.
- 3. The repair of adjacent pavement, driveways, sidewalks, and landscaping is included in this item unless specifically identified for separate payment on the Drawings.

C. Basis of Payment:

 This work will be paid for at the Contract Unit Price per lineal foot for CONCRETE CURB AND GUTTER REMOVAL AND REPLACEMENT.

1.37 PCC SIDEWALK REMOVAL AND REPLACEMENT

A. Description:

- 1. The work of this Pay Item consists of constructing sidewalks to match existing. The following work items are associated with, and shall be considered incidental to this Pay Item:
 - a. All quality control testing in accordance with Article 1020.09.
 - b. Excavate and dispose of existing materials.
 - c. Install 3 inches of compacted Aggregate Base Course.
 - d. Form and pour Portland cement concrete sidewalk.
 - e. Remove forms and backfill areas adjacent to concrete sidewalk.
 - f. Apply protective coat.
- 2. Sidewalk damaged by the Contractor during construction operations beyond the limits of the trench will be replaced at no cost to the Owner.

B. Measurement:

- 1. Portland Cement Concrete Sidewalk will be measured in place, and the area computed in square feet, up to the limits indicated on the Drawings.
- 2. A maximum width of ten lineal feet per crossing will be utilized for pay quantity calculations.

C. Basis of Payment:

1. This work will be paid for at the Contract Unit Price per square foot for PCC SIDEWALK REMOVAL AND REPLACEMENT of the material thicknesses indicated in Section 32 10 00.

1.38 DETECTABLE WARNINGS

A. Description:

1. The work of this Pay Item includes constructing detectable warnings with cast-in-place 24" by 48" detectable warnings comprised of two 24" by 24" square cast iron plates. The panel shall comply with ADA requirements. The domes located on the plate shall be parallel to the pavement crosswalk with the closest corner at the back of curb. Installation shall be according to the manufacturer's recommendations.

B. Basis of Payment:

 The work will be paid for at the Contract Unit price per square foot for DETECTABLE WARNINGS.

1.39 PAVEMENT MARKING

A. Description:

- 1. The work of this Pay Item consists of complete replacement of pavement marking, lines, letters, or symbols of the patterns, sizes, and colors.
- 2. Replace all pavement markings removed or damaged within the limits and construction, as shown on the Drawings, as recorded on the "Job Set" plan prior to removal, or as directed by the Engineer. This work shall be done as specified in Sections 780, 781, and 783 of the IDOT "Standard Specifications", and Section 32 10 00.19 of these Specifications.

B. Measurement:

1. The work will be measured in lineal feet of line, or in square feet for letters and symbols.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per lineal feet for line, of the widths indicated, and per square foot for letters or symbols as PAVEMENT MARKING, of the type noted.

1.40 RESTORATION OF LAWNS AND PARKWAYS

A. Description:

 The work of this Pay Item consists of complete surface restoration of lawns, parkways, and other grassed areas disturbed as a result of the construction, including topsoil, lime, fertilizer, erosion control materials, seeding, and maintenance.

B. Measurement:

- 1. The work will be measured in lineal feet along the centerline of the pipe and the area computed in square yards, with a maximum allowable width of 6 feet for main line pipe and maximum width of 6 feet for service pipes.
- 2. Areas damaged by or disturbed beyond the work area will be restored at no additional cost to Owner.
- 3. Areas damaged or disturbed on private property is incidental and will be restored at no additional cost to the Owner.

C. Basis of Payment:

 This work will be paid for at the Contract Unit Price per square yard for RESTORATION OF LAWNS AND PARKWAYS.

1.41 REMOVE AND REPLACE DRAIN TILES

A. Description:

- 1. This work shall include all labor, equipment and materials required to remove and replace existing drain lines on private property that are damaged during installation of new water service lines. PVC and corrugated plastic drains shall be replaced to match existing pipe material. VCP drains shall be replaced with PVC.
- 2. This Pay Item covers pipe sizes as listed in the Bid Form, since the actual size and location of the tiles is unknown.

B. Measurement:

1. The work will be measured along the centerline of the drain tile replaced, for the length of the replacement.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per lineal foot as REMOVE AND REPLACE DRAIN TILES, of the size range indicated.

1.42 ADJUSTING SANITARY SEWER SERVICE LINE

A. Description:

The work of this Pay Item shall be completed in accordance with the latest edition of the "Standard Specifications for Water and Sewer Construction in Illinois", and shall consist of the removal and replacement, or adjustment and relocation, of sanitary sewer service lines in conflict with the water main at crossing locations complete in place, including connections to the

- existing service lines; couplings; excavation; bracing; bedding and covering of pipe; trench dewatering; finish grading; removal and disposal of waste excavated materials; protection, replacement, or repair of utilities; and backfilling with granular backfill materials.
- 2. The work shall include either the removal and replacement of at least 21' of service pipe at the existing location, slope, and elevation of the existing service pipe; or the vertical adjustment of the existing service pipe to cross below the water main.
 - a. Adjustment and relocation will require replacement of enough sanitary sewer service pipe to obtain an adequate elevation difference to cross below the water main with positive slope toward the main line sewer.

B. Basis of Payment:

The work will be paid for at the Contract Unit Price for each ADJUSTING SANITARY SEWER SERVICE LINE, regardless of the depth, length, size, or pipe material of the sanitary sewer service.

1.43 DUST CONTROL

A. Description:

- 1. The work of this Pay Item consists of dust control by the uniform application of sprinkled water and utilization of sweepers.
- 2. DUST CONTROL shall be performed on an as needed basis as operations and weather dictates, and when directed by the Engineer. Frequency of DUST CONTROL operations is anticipated to be ½ of all working days per month, as defined by IDOT.
- 3. All equipment used for this work shall be approved by the Engineer prior to beginning work.

B. Measurement:

 The work will be measured per each working day in which dust control operations are performed.

C. Basis of Payment:

1. The work will be paid for at the Contract Unit Price per day for DUST CONTROL.

1.44 REMOVAL AND REPLACEMENT OF UNSUITABLE MATERIAL

A. Description:

1. The work of this Pay Item consists of the removal of organic, soft, spongy or otherwise unsuitable soils found at the bottom of open trench pipe installations or at the bottom of tunnel shafts and the replacement with gradation No. CA 1 stone complete, including removal and disposal of unsuitable materials; protection, replacement, or repair of utilities; and dewatering, including erosion and sedimentation control methods and devices to provide protection to environment from all pumping operations.

2. Payment will be allowed for only the material below an elevation one foot below the bottom of the pipe barrel, and when removal and replacement is approved by the Engineer.

B. Measurement:

The work will be measured in the field and computed in cubic yards.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per cubic yard for REMOVAL AND REPLACEMENT OF UNSUITABLE MATERIAL, at the locations noted.

1.45 SOIL TESTS FOR CONTAMINANTS

A. Description:

1. The work of this Pay Item consists of providing the services of a geotechnical engineer to take samples of the soil at the site of the work and provide laboratory testing of the samples to determine if contaminants are present in the soil.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each sample, as SOIL TESTS FOR CONTAMINANTS.

1.46 CONTAMINATED WASTE DISPOSAL

A. Description:

1. The work of this Pay Item consists of testing soil samples and disposal of material from the work site that is determined to be contaminated, based upon criteria of Illinois Environmental Protection Act for CCDD materials. Soil testing should be coordinated with the City laboratory.

B. Measurement:

1. The work will be measured based on verified load tickets.

C. Basis of Payment:

 The work will be paid for at the Contract Unit Price per cubic yard as CONTAMINATED WASTE DISPOSAL.

1.47 EXPLORATORY EXCAVATION

A. Description:

 This item shall cover all costs associated with performing an exploratory excavation where requested by Engineer including equipment, labor, and materials. Trench backfill where needed.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for EXPLORATORY EXCAVATION for every occurrence of the depth specified.

1.48 TREE ROOT PRUNING

A. Description:

The work of this Pay Item consists of root pruning using an approved mechanical root pruning saw prior to excavating around the tree and where indicated on the Drawings or as directed by the Engineer; and in such a manner as to preserve the natural growth habit of each tree complete including an equivalent amount of the top vegetative growth of the plant material within one week following root damage, the application of fertilizer nutrients, and supplemental watering. Root pruning depth to be 12" minimum.

B. Basis of Payment:

1. The work will be paid for at the Contract Unit Price for each TREE ROOT PRUNING.

1.49 TREE TRIMMING

A. Description:

- 1. The work of this Pay Item consists of trimming of trees and designated by the Engineer.
- 2. The Contractor must walk the site with the Engineer prior to initiating trimming operations, and receive approval for limits of trimming.
- Only that amount of tree trimming necessary to complete the work will be allowed, and the Contractor must visit the site to determine the amount of clearing that will be necessary to complete the work and therefore base the Bid.

B. Basis of Payment:

1. The work will be paid for at the Contract Lump Sum Price for TREE TRIMMING.

1.50 WATER SERVICE LOCATE

A. Description:

 The work of this Pay Item consists of locating all newly installed water services with spray paint. This work must be completed after final paving and sweeping of roadways.

B. Basis of Payment:

 The work will be paid for at the Contract Lump Sum Price for WATER SERVICE LOCATE.

END OF SECTION

MEASUREMENT AND PAYMENT 01 22 29-26 (210039.40)

SECTION 01 26 13

REQUESTS FOR INTERPRETATION

PART 1 - GENERAL

1.1 SUMMARY: REQUESTS FOR INTERPRETATION (RFI)

A. The Contractor may submit Requests For Interpretation (RFI) to the Engineer to expedite the Contractor's performance on the Project. RFIs will be submitted following the requirements, all as described in this Section.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Sections in Division 01 of these Specifications.
- 2. Individual requirements for submittals will be described in pertinent Sections of these Specifications.

C. Work not included:

- 1. Incomplete submittals will not be reviewed by the Engineer.
- The Contractor may require his subcontractors to provide drawings, setting diagrams, and similar information to help coordinate the Work, but such data shall remain between the Contractor and his subcontractors and will not be reviewed by the Engineer unless specifically called for within the Contract Documents.
- D. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- E. Make submittals of RFIs in accordance with the provisions of this Section.
- F. Prior to submitting each RFI, the Contractor shall first carefully study and compare the Contract Documents, field conditions, other Owner provided information, Contractor prepared Coordination Drawings, and prior Project correspondence and documentation to determine that the information requested is not reasonably obtainable from such sources.

- G. The Contractor shall submit each RFI sufficiently in advance of the date by which such information is required to allow the Engineer sufficient time, in the Engineer's professional judgement, to permit adequate review and response and to permit Contractor compliance with the latest construction schedule.
- 1.3 QUALITY ASSURANCE (Reserved).
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 This Subsection intentionally left blank.

PART 3 - EXECUTION

3.1 IDENTIFICATION OF SUBMITTALS

- A. Each RFI shall be submitted to the Engineer, in writing, on such form and with such accompanying information as the Engineer may require for such purpose. Each RFI shall identify the specific sources which were reviewed by the Contractor in its efforts to determine the information requested, and a statement to the effect that the information being requested could not be determined from such sources.
- B. Consecutively number all submittals.
 - 1. When material is submitted for any reason, transmit under a new letter of transmittal and with a new transmittal number.
 - 2. On re-submittals, cite the original submittal number for reference.
- C. Accompany each submittal with a letter of transmittal showing all information required for identification and checking.
 - 1. Use Request for Interpretation (RFI) Form, Section 01 26 13.13.
- D. On at least the first page of each submittal, and elsewhere as required for positive identification, show the submittal number in which the item was included.
- E. Submittal log:
 - Maintain an accurate submittal log for the duration of the Work, showing current status of all submittals at all times, the date of the request, to whom the request was made, by whom the request was made, the nature of the request, and the Engineer's resolution thereof.

- 2. Make the submittal log available to the Engineer for the Engineer's review upon request.
- 3. Review this log at each Project Meeting and make the resolution of RFIs a part of the minutes of such meetings.

END OF SECTION

RFI NO._____

01 26 13.13

REQUEST FOR INTERPRETATION (RFI) FORM

By submission of this form the Contracto Contract Documents and coordinated th	be considered upon receipt of this completed RFI Form. or attests to the fact that having carefully reviewed the e Work with the appropriate trades and reviewed field I cannot be determined from such efforts as called for in
Date: Project: _	
Го:	
Specification References:	
Drawing References:	
Proposed method of resolving issue.	
Response Date: o maintain project schedule. (Allow suffic	List date by which response by Engineer is requested cient time for response).
	, Project Superintendent sibility for accuracy and completeness of information.
ENGINEER'S RESPONSE	
esolve issues or remarks in response t	neer's action on method proposed by the Contractor to to RFI when no Contractor recommendation has been and/or project time shall be processed using standard Sketches and/or Pages AttachedYesNo
Signed:	Date:

SECTION 01 31 19

PROJECT MEETINGS

PART 1 - GENERAL

1.1 SUMMARY

A. Work included: To enable orderly review during progress of the Work, and to provide for systematic discussion of problems, the Engineer will conduct project meetings throughout the construction period.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
- 2. The Contractor's relations with his subcontractors and materials suppliers, and discussions relative thereto, are the Contractor's responsibility and normally are not part of project meetings content.

C. References:

1. (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- E. Agenda items: To the maximum extent practicable, advise the Engineer at least 24 hours in advance of project meetings regarding items to be added to the agenda.

F. Minutes:

- The Engineer will compile minutes of each project meeting, and will furnish three copies to the Contractor and required copies to the Owner.
- 2. Recipients of copies may make and distribute such other copies as they wish.

1.3 QUALITY ASSURANCE

- A. For those persons designated by the Contractor to attend and participate in project meetings, provide required authority to commit the Contractor to solutions agreed upon in the project meetings.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 No products are required in this Section.

PART 3 - EXECUTION

3.1 MEETING SCHEDULE

- A. Project meetings will be scheduled at the Preconstruction Meeting.
- B. Coordinate as necessary to establish mutually acceptable schedule for meetings.

3.2 MEETING LOCATION

A. The Engineer will establish meeting location. To the maximum extent practicable, meetings will be held at the job site.

3.3 PRECONSTRUCTION MEETING

- A. Preconstruction Meeting will be scheduled to be held within 20 working days after the effective date of the Agreement.
 - 1. Provide attendance by authorized representatives of the Contractor and major subcontractors.
 - 2. The Engineer will advise other interested parties, including the Owner, and request their attendance.
- B. Minimum agenda: Data will be distributed and discussed on at least the following items:
 - 1. Organizational arrangement of Contractor's forces and personnel, and those of subcontractors, materials suppliers, and Engineer.
 - 2. Channels and procedures for communications.
 - 3. Construction schedule, including sequence of critical work.
 - 4. Contract Documents, including distribution of required copies of original Documents and revisions.

PROJECT MEETINGS 01 31 19-2 (210039.40)

- 5. Processing of Shop Drawings and other data submitted to the Engineer for review.
- 6. Processing of Bulletins, field decisions, and Change Orders.
- 7. Rules and regulations governing performance of the Work; and
- 8. Procedures for safety and first aid, security, quality control, housekeeping, and related matters.
- 9. American Iron and Steel requirements, including records that Contractor must keep and items that Contractor must submit.
 - See Section 01 62 13.

3.4 PROJECT MEETINGS

- A. Progress Meetings:
 - Progress meetings will be held throughout progress of the Work at intervals agreed to by Owner, Engineer, and Contractor. Interval will generally be weekly.
- B. Contractor's project manager, job superintendent, major subcontractors, and suppliers shall attend as appropriate to address agenda topics for each meeting. Contractor's representatives shall have authority to bind Contractor to decisions at the meetings.
- C. The project schedule shall be updated monthly and shall be reviewed at each progress meeting. Contractor shall provide the following information in written form at each meeting.
 - 1. Construction progress, including:
 - a. Activities completed this reporting period.
 - b. Activities in progress this reporting period.
 - c. Activities scheduled to commence this reporting period.
 - 2. Description of problem areas.
 - 3. Current and anticipated delays.
 - a. Cause of the delay.
 - b. Corrective action and schedule adjustments to correct the delay.
 - c. Impact of the delay on other activities, on milestones, and on completion dates.
 - 4. Changes in construction sequence.
- D. American Iron and Steel requirements shall be on the Agenda of each project meeting and at a minimum, the following shall be addressed:
 - 1. Status and updates to the accepted American Iron and Steel List from the American Iron and Steel Conference or previous project meeting.
 - 2. De Minimis List and items that need to be added to it.
 - 3. Minor Components Waiver: Identify any items that meet the criteria for being classified as "minor" and will be furnished under this waiver.
 - 4. Short Term National Product Waiver: Identify any items that will be furnished under this waiver.
 - 5. Pig Iron and Direct Reduced Iron Waiver: Identify any items that will be furnished under this waiver.

6. Project Specific Waiver: Identify any items for which Contractor intends to pursue a Project Specific Waiver and the status of that pursuit.

END OF SECTION

SECTION 01 32 16

CONSTRUCTION PROGRESS SCHEDULES

PART 1 - GENERAL

1.1 SUMMARY

A. Prepare and maintain the schedules and reports described in this Section to assure adequate planning and execution of the Work so that the Work is completed within the Contract Times, and to assist the Engineer in appraising the reasonableness of the proposed schedule and in evaluating progress of the Work.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
- 2. Requirements for progress schedule: General Conditions.
- 3. Construction period: Form of Agreement.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- E. Comply with pertinent provisions of Section 01 33 01.
- F. Construction schedule: Prior to submission of the first Application for Payment, but no later than 30 calendar days after Contract Times commence, submit to the Engineer one reproducible copy and four prints of a construction schedule prepared in accordance with Part 3 of this Section.
- G. Periodic reports: Update the construction progress monthly and submit it to the Engineer prior to submittal of each Application for Payment for completed work.
 - 1. Submit four prints of the construction schedule updated as described in Part 3 of this Section.

1.3 QUALITY ASSURANCE

- A. Perform data preparation, analysis, charting, and updating in accordance with standards approved by the Engineer.
- B. Reliance upon the approved schedule:
 - 1. The construction schedule as approved by the Engineer will be an integral part of the Contract and will establish interim completion dates for the various activities under the Contract.
 - 2. Processing of the first Application for Payment will not be completed by the Engineer until the construction schedule has been submitted in accordance with 1.2 F. above.
 - 3. Processing of the 50 percent and 80 percent progress payment applications will not be completed by the Engineer until the periodic reports have been submitted in accordance with 1.2 G. above.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 CONSTRUCTION ANALYSIS

- A. Graphically show by Critical-Path (CPM), Program Evaluation and Review Technique (PERT), Precedence Methods, bar-chart, or other means acceptable to the Engineer, the order and interdependence of all activities necessary to complete the Work, and the sequence in which each activity is to be accomplished, as planned by the Contractor and his project field superintendent in coordination with all subcontractors whose work is shown on the diagram.
- B. Include, but do not necessarily limit indicated activities to:
 - 1. Project mobilization.
 - 2. Work elements.
 - 3. Special material and equipment installation and testing.
 - 4. Final cleanup.
 - Final inspecting and testing.
 - 6. All activities by the Engineer that affect progress, required dates for completion, or both, for all and each part of the work.
 - 7. Contractor's anticipated working dates.

PART 3 - EXECUTION

3.1 CONSTRUCTION SCHEDULE

- A. As soon as practicable after receipt of Notice to Proceed, complete the construction schedule in preliminary form, meet with the Engineer, review contents of the proposed construction schedule, and make all revisions agreed upon.
- B. Submit in accordance with Paragraph 1.2 F. above.

3.2 PERIODIC REPORTS

- A. As required under Paragraph 1.2 G. above, update the approved construction schedule.
 - 1. Indicate "actual" progress in percent completion for each activity;
 - 2. Provide written narrative summary of revisions causing delay in the program, and an explanation of corrective actions taken or proposed.

3.3 REVISIONS

A. Make only those revisions to approved construction schedule as are approved in advance by the Engineer.

END OF SECTION

SECTION 01 32 36

PRECONSTRUCTION VIDEO RECORDING

PART 1 - GENERAL

1.1 SUMMARY

- A. Provide color audio-video documentation of construction areas prior to start of construction, as specified herein.
 - 1. Provide documentation on USB Flash drives.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Promptly upon award of the Contract, notify all pertinent personnel regarding requirements of this Section.
- B. Require that all personnel who will enter upon the Owner's property certify their awareness of and familiarity with the requirements of this Section.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 REQUIREMENTS

- A. Include coverage of at least 100 feet in each direction along the street and/or pipeline when documenting areas of work that are of minimal length or are isolated locations.
- B. Upon acceptance of the USB Flash drives, the USB Flash drives become the property of the Owner. The Contractor may duplicate the video for his own use at his expense.

1.8 SCHEDULING

A. Schedule video so that no area is recorded more than 6 months prior to the actual beginning of construction in that area.

PART 2 - EQUIPMENT

2.1 GENERAL

A. Furnish all equipment, accessories, and materials to perform this service, except the specific plans of the proposed area to be televised, which will be furnished by the Owner.

2.2 TECHNICAL REQUIREMENTS

- A. Provide the total audio-video recording system and procedures as required to produce a finished product that will fulfill the technical requirements of the project and the more subjective requirements of high quality video and audio production. Produce the video portion of the recording with bright, sharp, clear pictures with accurate colors and free from distortion, tearing, rolls, or other forms of picture imperfection. Produce the audio portion of the recording with proper volume, clarity, and free from distortion.
- B. Make recordings with a digital video recorder.
- Provide the documentation on USB Flash drives.

PART 3 - EXECUTION

3.1 GENERAL

A. Audio-video coverage will be required in areas not accessible by conventional wheeled vehicles. Obtain such coverage by walking or special conveyance approved by the Engineer.

3.2 TIME AND POSITION RECORDS

A. Audio:

1. Begin each recording with the current date, project name, and Owner; followed by the general location, e.g., station along the route, name of the street, viewing side, and direction of progress.

3.3 RECORDING IDENTIFICATION

A. Properly identify all USB Flash drives with number, location, and project name and municipality in a manner acceptable to the Owner.

3.4 CONSTRUCTION AREA

- A. Include coverage of all surface features located within the zone of influence of construction, supported by appropriate audio description. Make audio description simultaneously with video coverage. Include coverage, but not be limited to, all existing driveways, sidewalks, curbs, ditches, roadways, landscaping, trees, culverts, headwalls, bridges, and retaining walls or buildings located within such zone of influence.
- B. Identify houses and buildings visually by house or business number, when visible, in such a manner that structures of the proposed system, e.g., valve vaults, can be located by reference.
- C. Record at a rate of speed not exceeding 48 feet per minute in the general direction of travel. Control panning rates and zoom-in and zoom-out rates to provide clarity of the viewed object during playback.
- D. Perform all recording during times of good visibility. Perform no documentation during periods of visible precipitation, or when more than 10 percent of the ground area is covered with snow or fallen leaves, unless otherwise authorized by the Engineer.
- E. The Owner has the authority to designate what areas may be omitted or added for audio-video coverage.

3.5 CONTRACTOR'S RESPONSIBILITIES

- A. Use no electrical circuits within private property building structure at any time. Perform all documentation during regular business hours, unless otherwise specified by the private property owner or Engineer. Enter and leave private property in a professional and orderly, workmanship like manner.
- B. Provide for an audio-visual presentation for review.
 - 1. The Owner has the authority to reject all or any portion of the video recording not conforming to Specifications.

- 2. Re-record any coverage not acceptable to the Owner at no cost to the Owner.
- 3. Reschedule recording of unacceptable coverage as soon as possible after being notified of non-acceptance of previous recordings.
- 4. Submit completed audio-visual USB Flash drives to the Owner and Engineer before commencement of construction.
 - a. Provide at least two (2) copies of each USB Flash drive.
- C. All acceptable recordings are the property of the Owner.

END OF SECTION

SECTION 01 33 01

SUBMITTALS

PART 1 - GENERAL

1.1 SUMMARY

A. Provide submittals required by the Contract Documents, and revise and resubmit as necessary to establish compliance with the specified requirements, all as described in this Section.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- 2. Individual requirements for submittals will be described in pertinent Sections of these Specifications.
 - a. The process for securing approval of proposed substitutions is described in Section 01 62 01, "Product Options and Substitutions".

C. Work not included:

- 1. Submittals not required by the various Specification Sections of the Contract Documents will not be reviewed by the Engineer.
- 2. The Contractor may require his subcontractors to provide drawings, setting diagrams, and similar information to help coordinate the Work, but such data shall remain between the Contractor and his subcontractors and will not be reviewed by the Engineer unless specifically called for within the Contract Documents.
- D. References (Reserved).

1.2 SUBMITTALS

A. Provide submittals of Shop Drawings, Samples, Substitution Requests, progress schedules and other items required in the Contract Documents in accordance with the provisions of this Section.

1.3 QUALITY ASSURANCE

- A. Coordination of submittals:
 - 1. Review and coordinate all aspects of each item being submitted.
 - 2. Verify that each item and the submittal for it conform in all respects with the specified requirements.
 - 3. Certify that this coordination has been performed by affixing the Contractor's signature to each Contractor's Submittal Transmittal Form Attachment 01 33 01.

SUBMITTALS 01 33 01-1 (210039.40)

- B. Resubmittals and reimbursement of Engineer's costs.
 - 1. The Engineer will record all time used by the Engineer in the review of any third and subsequent submittals.
 - 2. The Owner will reimburse the Engineer at the Engineer's standard hourly rate for all time spent in such third and subsequent reviews and deduct such costs from payments due the Contractor.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 SHOP DRAWINGS

- A. Provide Shop Drawings accurately to a scale sufficiently large to show all pertinent aspects of the item and its method of connection to the Work.
 - 1. Shop Drawings are not required for manholes, valve vaults, catch basins, pipe, and appurtenances needed for infrastructure systems (storm sewers, sanitary sewers, and water distribution) so long as the items are the materials and manufacturers specified in the project manual.
- B. Submit Shop Drawings electronically to the Engineer as a single .pdf file set.
 - 1. Attach, as the first page of each Shop Drawing, a completely executed Contractor's Submittal Transmittal Form Attachment 01 33 01.
 - 2. Collate the electronic .pdf file to include all data pertaining to the Shop Drawing Submittal in one .pdf set.
 - a. Separate .pdf files submitted will be cause for rejection and the Shop Drawing will be returned to the Contractor.
 - 3. In cases where Electronic Shop Drawing files exceed a size that is practical for electronic transmission via electronic mail or through an FTP site, the Contractor may and will be required to submit up to five (5) .pdf file shop drawing submittals on separate compact discs or removable USB storage, if requested, plus the quantity of discs or removable USB storage that will be required to be returned to the Contractor.
- C. Submit all required shop drawings for a specification section at the same time under one Contractor's Submittal Transmittal Form Attachment 01 33 01.
- D. Do not submit partial submittals of an item within a specification section or use a separate Contractor's Submittal Transmittal Form for separate items within a particular section.
- E. Identify exceptions or items that do not comply with the specifications and provide explanation for exception or non-compliance.

SUBMITTALS 01 33 01-2 (210039.40)

- F. For Shop Drawings required to be resubmitted for review, include the following:
 - 1. A completely executed cover sheet Contractor's Submittal Transmittal Form Attachment 01 33 01.
 - A cover letter responding to each of the review comments returned to the Contractor by the Engineer with the previous review and specifically stating:
 - a. If the equipment and resubmitted data provided complies with the review comment(s) then provide:
 - (1) How the equipment complies.
 - (2) Specifically indicate where support documentation can be located in the shop drawing.
 - b. If the equipment and resubmitted data provided cannot or does not comply with the review comment(s) then provide:
 - (1) What is being provided to comply instead.
 - (2) Justify why the Contractor feels the Engineer should consider it is acceptable to allow the Contractor to not comply with the specification.
 - 3. Resubmission of a complete and fully-inclusive shop drawing with all data pertinent to the item(s) being submitted.
 - a. Partial submission of data that only addresses the Engineer's specific review comments, or a portion thereof, and does not include all data for a complete resubmittal, will be cause for immediate rejection.
- G. Upon completion of the Engineers review of the Shop Drawings, one electronic pdf file will be returned to the Contractor for their distribution.
 - 1. Upon the request of the Engineer or Owner at any time, the Contractor shall provide up to four (4) color, hard copy, shop drawing submittals.
 - 2. Upon the request of the Engineer or Owner at any time, the Contractor shall provide up to four (4) .pdf file shop drawing submittals on separate compact discs or removable USB storage.

2.2 MANUFACTURERS' LITERATURE

- A. Where contents of submitted literature from manufacturers includes data not pertinent to the submittal, clearly indicate which portion of the contents is being submitted for review by highlighting, circling, or other means, or by crossing out contents that do not pertain to the submittal and are not to be considered.
 - 1. This also applies to specifically indicating, when applicable, which optional items will or will not be provided with items specified.

2.3 SAMPLES

- A. Provide Sample or Samples identical to the precise article proposed to be provided.
 - 1. Identify as described under "Identification of Submittals" below.

- B. Number of Samples required:
 - 1. Unless otherwise specified, submit Samples in the quantity which is required to be returned, plus one which will be retained by the Engineer.
 - 2. By prearrangement in specific cases, a single Sample may be submitted for review and, when approved, be installed in the Work at a location agreed upon by the Engineer.
 - 3. Because submittals shall be submitted to the Engineer in an electronic format as described herein, the Contractor shall specifically indicate on the Contractor's Submittal Transmittal Form Attachment 01 33 01 included with each submittal (when samples are required) when and where the physical samples will or have been transmitted for physical observation.
 - 4. Include as part of the electronic submittal a .pdf copy of any and all transmittals, shipping information, signatures of receipt, etc. identifying the transmission and receipt of the said sample(s).

2.4 COLORS AND PATTERNS

A. Unless the precise color and pattern is specifically called out in the Contract Documents, and whenever a choice of color or pattern is available in the specified products, submit accurate color and pattern charts to the Engineer for selection.

2.5 MANUFACTURERS' RECOMMENDED INSTALLATION PROCEDURES

- A. Maintain in a safe place at the site one copy of manufacturers' recommended installation procedures for all equipment and materials.
 - 1. Make these installation procedures readily available to the Engineer for reference.
- B. When the manufacturers' recommended installation procedures are submitted as part of the shop drawings required by the Contract Documents, approval of such installation procedures by the Engineer will not be required.

PART 3 - EXECUTION

3.1 IDENTIFICATION OF SUBMITTALS

- A. Consecutively number all submittals, beginning with identifying number "001" for the first submittal delivered by the Contractor.
 - 1. When items are submitted for any reason, transmit under a new Contractor's Submittal Transmittal Form Attachment 01 33 01 and with a new transmittal number.
 - 2. When material is resubmitted for any reason, cite the original identifying submittal number followed by insertion of a letter "A" for the first resubmittal, "B" for the second resubmittal, and so on.

- B. Accompany each submittal with a letter of transmittal showing all information required for identification and checking.
 - 1. Use Contractor's Submittal Transmittal Form Attachment 01 33 01.
- C. On at least the first page of each submittal, and elsewhere as required for positive identification, show the submittal number in which the item was included.

3.2 GROUPING OF SUBMITTALS

- A. Shop Drawings may be submitted for different specification sections under one Contractor's Submittal Transmittal Form Attachment 01 33 01, provided the items are specifically and directly related to each other such that review of the items from different specification sections is pertinent for a complete review.
 - 1. Identify any and all items and their specific specification section(s) if included with and submitted under a differing main specification section submittal.
 - 2. Partial submittals may be rejected as not complying with the provisions of the Contract.
 - 3. The Contractor may be held liable for delays so occasioned.
 - 4. Do not submit unrelated items in group submittals.

3.3 TIMING OF SUBMITTALS

- A. Make submittals far enough in advance of scheduled dates for installation to provide time required for reviews, for securing necessary approvals, for possible revisions and resubmittals, and for placing orders and securing delivery.
- B. In scheduling, allow at least ten working days for review by the Engineer following the Engineer's receipt of the submittal.

END OF SECTION

ATTACHMENT 01 33 01

CONTRACTOR'S SUBMITTAL TRANSMITTAL FORM

TO: BAXTER & WOODMAN, INC. 8678 RIDGEFIELD ROAD CRYSTAL LAKE, IL 60012		DATE:		
ΑT	TN:	_		
PR	OJECT NAME:			
FR	OM:	SPEC NO.		
		ENGR. DWG. NO	S	
		TRANSMITTAL N	O	
1.	The following submittals are forwarded	I for your review:		
	<u> </u>	<u>Description</u>	Drawing <u>No.</u>	<u>Date</u>
2.	Have all field measurements, field corcriteria, materials, dimensions, numbers, and similar data been determined?	nstruction catalog		No
3.	Has work indicated in this submit coordinated with all trades?	tal been	Yes	No
4.	Is work by all trades being provinecessary to accommodate this submit		Yes	No
5.	Contractor's description and justifications from Contract Documents (Use additional sheet if necessary.)	ation for		
6.	Remarks:			
	Signature:			

1/15 - IL

SECTION 01 41 26

PERMITS

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes permit requirements for building, work in highway rights-of-way, work in railroad rights-of way and for stormwater discharges.
- B. Related Sections:
 - Documents affecting work of this Section include, but are not necessary limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
 - 2. Other permits requirements may also be described in other Sections of these Specifications.
- C. References (Reserved).
- 1.2 SUBMITTALS (Reserved).
- 1.3 QUALITY ASSURANCE
 - A. Promptly upon award of the Contract, notify all pertinent personnel regarding requirements of this Section.
 - B. Require that all personnel who will enter upon the Owner's property certify their awareness of and familiarity with the requirements of this Section.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).
- 1.7 WORK IN HIGHWAY RIGHTS-OF-WAY
 - A. Conform in all respects with the requirements of the controlling authority in performing construction work in the rights-of-way of Township, County, State, and Federal highways where required by the locations shown on the Drawings.
 - 1. Comply with IDOT "Permit Special Provision Governing Permit Work on State Right-of-Way".
 - B. Execute all necessary highway permit forms, provide and pay for any bond requirements, execute and comply with any required Traffic Control Authorization

PERMITS 01 41 26-1 (210039.40) Requests, and submit digital pictures (on a USB Flash drive) of the project showing all right-of-way that will be affected, and referenced to the project stationing.

- 1. The County of Will requires the Contractor to provide a Letter of Credit valued at 125% of the value of the work being performed within the County right-of-way; and a Letter of Credit "maintenance bond" valued at 10% of the value of the work for one year after the work is completed.
- C. Notify the highway authority involved not less than ten (10) days prior to starting construction in highway rights-of-way.

1.8 NPDES PHASE II STORMWATER PERMIT

- A. ILR10:
 - A General NPDES Permit for Stormwater Discharges from Construction Site Activity (General NPDES Permit No. ILR10) is not required because the project involves less than 1.0-acre of disturbed area.

END OF SECTION

SECTION 01 42 13

ABBREVIATIONS AND ACRONYMS

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes abbreviations referenced in the Contract Documents.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Promptly upon award of the Contract, notify all pertinent personnel regarding requirements of this Section.
- B. Require that all personnel who will enter upon the Owner's property certify their awareness of and familiarity with the requirements of this Section.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 ABBREVIATIONS

- A. Referenced Standards:
 - Where the Contract Documents reference any published specifications or standards of any organization or association, comply with the requirements of the specification or standards which are current on the date of

Advertisement for Bids. In case of a conflict between the referenced specifications or standards, the one having the more stringent requirements shall govern.

2. In case of conflict between the referenced specifications or standards and the Contract Documents, the Contract Documents shall govern.

B. Abbreviations:

1. The following are definitions of abbreviations that may be used within the Project Manual:

AA - Aluminum Association

AASHTO - American Association of State Highway and Transportation Officials

ACI - American Concrete Institute

AISC - American Institute of Steel Construction

ANSI - American National Standard Institute

ASTM - American Society for Testing and Materials

AWG - American Wire Gauge

AWS - American Welding Society

AWWA - American Water Works Association

CBM - Certified Ballast Manufacturers Association

CRSI - Concrete Reinforcing Steel Institute

HCF - Hundred Cubic Feet

ICEA - Insulated Cable Engineers Association

IEEE - Institute of Electrical and Electronics Engineers, Inc.

ISA - Instrument Society of America

FS - Federal Specifications

NAPF – National Association of Pipe Fabricators

NEC - National Electrical Code (NFPA 70)

NECA - National Electrical Contractors' Association

NEMA - National Electrical Manufacturer's Association

NFPA - National Fire Protection Association or National Forest Products Association

NSF - National Sanitation Foundation

OSHA - U.S. Department of Labor, Occupational Safety and Health Department

PS - United States Products Standards

IDOT "STANDARD SPECIFICATIONS" - Illinois Department of Transportation "Standard Specifications for Road and Bridge Construction" including Recurring Special Provisions and Interim Special Provisions.

SSPC – Society for Protective Coatings

UL - Underwriter's Laboratories, Inc.

SECTION 01 45 29

TESTING LABORATORY SERVICES

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes testing to be provided by an independent testing laboratory service.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
 - 2. Requirements for specific tests will be described in various Sections of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Provide the services of a testing laboratory approved by the Engineer.
- B. Upon completion of each test and/or inspection, promptly distribute copies of test or inspection reports to the Engineer, to governmental agencies requiring submission of such reports, and to such other persons as directed by the Engineer.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 TESTING AGENCY DUTIES AND LIMITS OF AUTHORITY

- A. Cooperate with the Engineer and the Contractor; provide qualified personnel and equipment to perform the scope of testing work outlined.
- B. Acquaint the Engineer and the Contractor with testing procedures for special conditions encountered at the site.
- C. Perform specified monitoring, sampling, and testing of the materials and construction.
 - 1. Comply with specified standards, ASTM, other authorities, and as specified.
 - 2. Ascertain compliance with the Contract Documents.
 - 3. Obtain written acknowledgment of sampling or testing.
- D. Give prompt written notice to the Engineer and the Contractor of irregularities or deficiencies of work which are observed during performance of service.
- E. The Laboratory is not authorized to release, revoke, alter or enlarge the Contract requirements, nor to approve or accept any portion of the work, nor to perform the duties of the Contractor.

PART 2 - PRODUCTS

2.1 PAYMENT FOR TESTING

- A. Include within the Contract Price an amount sufficient to cover all testing required of the Contractor under pertinent Sections of these Specifications, and to cover all testing and inspecting required by governmental agencies having jurisdiction.
- B. The Owner will pay for all testing and inspecting specifically requested by the Engineer over and above those described in Paragraph 2.1 A. above.
- C. When tests indicate noncompliance with the Contract Documents, all testing and subsequent retesting occasioned by the noncompliance shall be performed by the same testing laboratory and the costs thereof shall be paid by the Contractor.

PART 3 - EXECUTION

3.1 TAKING SPECIMENS

A. Except as may be specifically otherwise approved by the Engineer, have the testing laboratory secure and handle all samples and specimens for testing.

3.2 COOPERATION WITH TESTING LABORATORY

- A. Provide access to the Work at all times and at all locations where the Work is in progress. Provide facilities for such access to enable the laboratory to perform its functions properly.
- B. Furnish casual labor and facilities:
 - 1. To obtain and handle samples at the site or at the source of the product to be tested.
 - 2. To facilitate testing operations.
 - 3. For laboratory's exclusive use for storage and curing of test samples on site.
- C. Notify the testing agency sufficiently in advance of operations to allow for assignment of personnel and scheduling of its operations.
- D. Provide the testing laboratory with copies of approved relevant shop drawings.

SECTION 01 50 00

TEMPORARY FACILITIES AND CONTROLS

PART 1 - GENERAL

1.1 SUMMARY

A. This Section describes construction facilities and temporary controls required for the Work.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- 2. Comply with pertinent safety requirements and regulations for temporary facilities and controls.
- 3. Equipment normally furnished by the individual trades in execution of their own portions of the Work is not part of this Section.
- 4. Permanent installation and hookup of the various utility lines are described in other Sections.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Promptly upon award of the Contract, notify all pertinent personnel regarding requirements of this Section.
- B. Require that all personnel who will enter upon the Owner's property certify their awareness of and familiarity with the requirements of this Section.

1.4 DELIVERY, STORAGE, AND HANDLING

A. Maintain temporary facilities and controls in proper and safe condition throughout progress of the Work.

- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 REQUIREMENTS

- A. Provide construction facilities and temporary controls needed for the Work including, but not necessarily limited to:
 - 1. Temporary utilities and services such as heat, water, electricity, telephone, and internet access.
 - 2. Sanitary facilities.
 - 3. Enclosures such as fencing, tarpaulins, barricades, and canopies.
 - 4. Temporary fencing of the construction site.
 - 5. Fire extinguishers.
 - 6. Dust and mud control.
 - 7. Traffic control.
 - 8. Security.
 - 9. Right-of-way and property line control.

PART 2 - PRODUCTS

2.1 UTILITIES AND SERVICES DURING CONSTRUCTION

A. Water:

- 1. The Owner will provide for all water used for filling, flushing, and testing of the main and all other construction activities.
- 2. Provide necessary temporary piping and water supply and, upon completion of the work, remove such temporary facilities.
- 3. The Contractor must meter and record all water used for filling, flushing, and testing of the main and all other construction activities.
- 4. The City can provide the Contractor with a rental water meter, deposit required, to record all water taken from the City's water distribution system. Upon completion of the work, the Contractor shall return the water meter to the City with the total volume of water supplied based on the final water meter reading less the initial meter reading recorded.
 - a. Meter rental rate of \$0.123/day applies. A deposit and \$10.00 service fee are required at the time of pickup. The initial deposit will then be returned minus the daily meter rental rate and service fee.
 - b. The Contractor has the option to provide their own meter. The meter must be approved by the Public Utilities Superintendent prior to the start of construction. The Contractor must provide a backflow prevention device to be utilized between the hydrant and the Contractor's equipment. The Engineer will record monthly meter readings and provide to the City.
- 5. The Contractor shall not connect to any portion of the City's water distribution without prior approval by the City.

- B. Electricity:
 - 1. Provide necessary temporary wiring and, upon completion of the Work, remove such temporary facility.
 - 2. Provide and pay for electricity used in construction.
- C. Heating: Provide and maintain heat necessary for proper conduct of operations needed in the Work.

2.2 SANITARY FACILITIES

- A. Provide temporary sanitary facilities meeting federal, state, and local health department requirements.
 - 1. Maintain in a sanitary condition at all times.

2.3 ENCLOSURES

A. Provide and maintain for the duration of construction all scaffolds, tarpaulins, canopies, warning signs, steps, platforms, bridges, and other temporary construction necessary for proper completion of the Work in compliance with pertinent safety and other regulations.

2.4 TEMPORARY FENCING

- A. Provide and maintain for the duration of construction a temporary fence of design and type needed to prevent entry onto the Work by the public.
- B. Jacking and receiving pit fences.
 - 1. As outlined in Section 33 05 23 of these Specifications.

2.5 FIRE EXTINGUISHERS

A. Provide and maintain not less than two fire extinguishers, multi-purpose dry chemical type with UL rating of 4A-60 B:C, 10-pound capacity, Amerex Model ABC, or equal, enclosed in suitable protecting cabinets and conveniently located for proper protection.

PART 3 - EXECUTION

3.1 MAINTENANCE AND REMOVAL

- A. Maintain temporary facilities and controls as long as needed for safe and proper completion of the Work.
- B. Remove such temporary facilities and controls as rapidly as progress of the Work will permit, or as directed by the Engineer.

3.2 DUST AND MUD CONTROL

- A. Take necessary precautions to control dust and mud associated with the Work, subject to the approval of the Engineer.
 - 1. In dry weather, spray dusty areas daily with water in order to control dust.
 - 2. Apply calcium chloride having a minimum chemical content of 77 percent calcium chloride at an application rate of 3 pounds per square yard of surface covered at locations as directed by the Engineer.
- B. Take necessary steps to prevent the tracking of mud onto adjacent streets and highways.
 - 1. Wash mud resulting from the construction traffic off the adjacent streets and highways.
 - Clean all permanent roadways used for construction activities by using motorized street sweeper that utilizes vacuum and water to pick up debris, when directed by Engineer.

3.3 TRAFFIC CONTROL

- A. Protect and maintain traffic by the proper use of barricades, warning lights, flares, and necessary traffic control and safety devices, conforming to federal, state, and local regulations regarding their use.
- B. Use forms of traffic control on public roadways required by the construction operations in accordance with the "Manual on Uniform Traffic Control Devices for Streets and Highways", Articles 107.09 and 107.14 of the IDOT "Standard Specifications", and the "Highway Standards".
- C. Unless the Contractor has obtained written permission from the Engineer to temporarily close any street, alley, or other traveled way, keep such traveled way open to traffic on the existing pavement.
- D. Maintain alternating one-way traffic from opposing directions during working hours. At all other times, provide sufficient width within existing shoulders or curbs to permit one lane of traffic in each direction.
- Ensure that all barricades, warning signs, lights, and other devices are operational 24 hours each day, including Sundays and holidays, during the time the contract is in force.
 - 1. In the event of severe weather conditions, provide any additional personnel necessary to properly maintain all traffic control devices.
- F. At the preconstruction meeting, furnish the name of the individual in the Contractor's direct employ who is to be responsible for the installation and maintenance of the traffic control for this project.
 - 1. If the actual installation and maintenance are to be accomplished by a Subcontractor, obtain the Engineer's consent at the time of the preconstruction meeting.

- 2. The Owner will provide the Contractor with the name of its representative who will be responsible for the administration of the traffic control plan.
- G. Provide access to private driveways at all times except during pipeline installation across a driveway.
 - 1. Keep driveway closure times to an absolute minimum.
 - 2. Warn homeowners 24 hours in advance of a driveway closure, and again just prior to closing to allow homeowners to move their vehicles.
 - 3. Maintain access for emergency vehicles at all times.
- H. Provide access to commercial or industrial driveways at all times except during pipeline installation across a driveway.
 - 1. Keep driveway closure times to an absolute minimum.
 - 2. If two driveway entrances exist, keep one driveway open at all times.
 - 3. Establish schedule for driveway closures with commercial or industrial establishments that will not obstruct passage of employees or customers during heavy use period.
 - 4. Maintain access for emergency vehicles at all times.
- I. The costs for traffic control will be considered incidental to the Contract and no additional compensation will be allowed.

3.4 SECURITY

- A. Take whatever measures are necessary to protect the safety of the public, workmen, and materials.
 - 1. Provide inspection of work area daily.
 - 2. Provide the security of the site, both day and night.

3.5 RIGHT-OF-WAY AND PROPERTY LINE CONTROL

- A. Protect all right-of-way markers, property line iron pins, and easement iron pins during construction.
 - 1. Flag such control points prior to construction, and protect the points during the course of construction.
- B. Establish tie-down control for any right-of-way markers or iron pins that may be lost or damaged during the work.
- C. Re-establish any right-of-way markers or iron pins that are lost or damaged during construction, after completion of restoration work.
- D. Provide the services of a Registered Land Surveyor for replacement of lost markers and pins.
 - 1. The cost for this work will be considered incidental to the Contract, and no additional compensation will be allowed.

3.6 WORK BOUNDARY FENCE

- A. Provide fence along edges of all easement limits, shaft work zones, and access roads to delineate limit of Contractor's Work area, where indicated on the Drawings.
- B. Maintain fences until removal is approved by Engineer.
 - 1. Provide 6-foot tall chain link fence in place of standard work boundary fence where indicated on the Drawings.

SECTION 01 62 13

AMERICAN IRON AND STEEL DOCUMENTATION

PART 1 - GENERAL

1.1 SUMMARY

- A. Comply with the American Iron and Steel provisions of the Consolidated Appropriations Act of 2014.
- B. Provide documentation verifying compliance with the requirements of the American Iron and Steel provisions of the Consolidated Appropriations Act of 2014, including, but not limited to, requirements described in this Section.
- C. Related work:
 - A. Documents affecting work of this Section include, but are not limited to, General Conditions, Supplementary Conditions, and Division 01 General Requirements of these Specifications.
- D. References (Reserved).

1.2 SUBMITTALS

- A. Provide Attachment 01 62 13-01 Step Certification of American Iron and Steel Compliance for each product and/or material from each handler (supplier, fabricator, manufacturer, processor, etc.) of the iron and steel products subject to American Iron and Steel requirements certifying that their step in the process was domestically performed.
 - 1. Multiple certifications are required for products and/or materials which have more than one entity involved from iron and/or steel production through final product and/or material completion ready for delivery.
- B. Provide a list of project specific items subject to American Iron and Steel requirements.
 - 1. Submit list prior to submission of the first Application for Payment, but no later than 7 calendar days after the date of the Pre-Construction Meeting.
 - 2. Include in the list:
 - a. Item identification.
 - b. Related specification section.
 - c. Paragraph number.
 - d. Installation location.
 - e. Identification of which waivers will apply, if any:
 - (1) De Minimis Waiver.
 - (2) Minor Components Waiver.
 - (3) Short Term National Product Waiver.

- (4) Pig Iron and Direct Reduced Iron Waiver.
- (5) Project Specific Waiver.
- 3. Provide updates to list at each application for payment.
- C. Preliminary Documentation submittal:
 - 1. For items requiring shop drawings:
 - Provide a draft example Attachment 01 62 13-01 submittal at time of shop drawing submittal.
 - (1) Include a separate declaration on company letterhead from each entity in the chain of custody declaring ability and intent to meet the requirements of the American Iron and Steel provisions.
 - 2. For items not requiring shop drawings:
 - a. Provide a draft example Attachment 01 62 13-01 submittal no later than sixty days prior to shipment.
 - (1) Include a separate declaration on company letterhead from each entity in the chain of custody declaring ability and intent to meet the requirements of the American Iron and Steel provisions.
- D. Final Documentation Submittal:
 - 1. Provide submittal of final Attachment 01 62 13-01 compliance documentation no later than 7 calendar days prior to shipment.
- E. De Minimis Final Utilization and Certification Form:
 - 1. Update Attachment 01 62 13-02 monthly.
 - 2. Submit updated Attachment 01 62 13-02 with each pay request.
 - 3. Submit final Attachment 01 62 13-02 at substantial completion for the record of the Owner.
- F. Comply with pertinent provisions of Section 01 33 01.
- 1.3 QUALITY ASSURANCE (Reserved).
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS – (Reserved).

3.1 AMERICAN IRON AND STEEL CONFERENCE:

- A. Schedule conference with Owner and Engineer.
- B. The purpose of the conference is to review:
 - 1. The Contractor's list of items subject to American Iron and Steel requirements.
 - 2. The procedure for preparation, preliminary submittal, and final submittal of Step Certifications for:
 - a. Items requiring shop drawings.
 - b. Items not requiring shop drawings.
 - 3. Examples of step certifications.
 - 4. Engineer's procedure for review of Step Certifications.
 - 5. The dates the Contractor intends to submit Preliminary Documentation and Final Documentation, especially for those items that do not require shop drawings and are needed early in the construction process.
 - 6. The De Minimis Final Utilization and Certification Form (Attachment 01 62 13-02) and how Contractor is to use it to track De Minimis items.
 - 7. Project Specific Waivers, if any, that the Contractor intends to pursue.
- C. Complete American Iron and Steel Conference on or before the 75th day of the Contract Time.
- D. The accepted list will be used to track American Iron and Steel certification submittals by the Contractor.

ATTACHMENT 01 62 13-01 STEP CERTIFICATION OF AMERICAN IRON AND STEEL COMPLIANCE

[The following information is provided as a sample letter of step certification for American iron nd Steel compliance. Documentation must be provided on company letterhead. Provide a r r

separate certification from <u>each</u> entity in the chain of production of the product and/o material from source iron and/or steel production through completed product and/o material.]
Date
Company Name
Company
Address City,
State Zip
Subject: American Iron and Steel Step Certification for EJ Walsh Sub Water Main Improvements City of Joliet
I, (company representative), certify that the (list processes here: melting, bending, forming, rolling, drawing, finishing, coating, galvanizing, cutting, etc.) processes for (manufacturing of fabricating) the following products and/or materials shipped or provided for the subject project are in full compliance with the American Iron and Steel requirement as mandated in EPA's State Revolving Fund Programs.
Item, Products and/or Materials:
1. Xxxx 2. Xxxx 3. Xxxx
Such processes took place at the following location(s):

If any of the above compliance statements change while providing material to this project we will immediately notify the prime Contractor and the Engineer.

Signed by company representative

ATTACHMENT 01 62 13-02 DE MINIMIS FINAL UTILIZATION AND CERTIFICATION FORM

Public Law 114-113, the "Consolidated Appropriations Act 2016" and subsequent "Continuing Resolutions" passed by Congress require the use of American Iron & Steel in SRF-funded projects. Under the authority of the Consolidated Appropriations Act 2016, the EPA has issued a public interest waiver for De Minimis incidental components. The assistance recipient wishing to use this waiver should consult with their contractor(s) to maintain an itemized list of components covered under De Minimis. At the conclusion of the project, this form should be completed and retained by the loan applicant. A copy must be provided to the Agency if requested. It is strongly recommended that you maintain a list as the project progresses. Please print clearly or type.

Loan Recipient:					
Project Name:				Loan No.	
NOTE: The De Minimis waiver (labor, installation costs, etc. applicable tax.) in the "Total Cost o	of Materials". The	cost of a material i	=	to the site and any
Funds used for de minimis inci of the materials used in and inc	corporated into a pr	oject; the cost of a		may not exceed 1 per	
Total Cost of Materials:		5% Limit:		1% limit:	
Manufacturer & Component Description	Part/Model #	Quantity (if applicable)	Cost per Unit (if applicable)	Component's Total Cost	How is Cost Documented?*

Manufacturer & Component Description	Part/Model #	Quantity (if applicable)	Cost per Unit (if applicable)	Component's Total Cost	How is Cost Documented?*
Use additional sheets as necessary	Total D	e Minimis Cost	of Components:		
* Documentation must demonst Completed by:	rate confirmation of	the components	' actual costs (invoid	ce, etc.).	
Name:			Title:		-
Signature:			Date:		

SECTION 01 66 11

STORAGE AND PROTECTION OF MATERIAL AND EQUIPMENT

PART 1 - GENERAL

1.1 SUMMARY

- A. Protect products scheduled for use in the Work by means including, but not necessarily limited to, those described in this Section.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
 - 2. Additional procedures also may be prescribed in other Sections of these Specifications.
- C. References (Reserved).
- 1.2 SUBMITTALS (Reserved).
- 1.3 QUALITY ASSURANCE (Reserved).
- 1.4 DELIVERY, STORAGE, AND HANDLING
 - A. Comply with the requirements of this Section for off-site storage.
 - 1. The Engineer reserves the right to visit and observe the off-site storage areas.
 - B. Store equipment and materials in accordance with the manufacturer's instructions.
 - C. Provide temporary weathertight enclosures to protect products from damage by the elements.
 - D. Protect finished surfaces through which equipment and materials are handled.
 - E. Provide protection for finished floor surfaces in traffic areas prior to allowing equipment or materials to be moved over such surfaces.
 - F. Maintain finished surfaces clean, unmarred, and suitably protected until accepted by the Owner.
 - G. Do not store plant maintenance equipment, furniture, and laboratory equipment on site until they are needed by the Owner or for progress of work.

- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 MANUFACTURERS' RECOMMENDATIONS

A. Except as otherwise approved by the Engineer, determine and comply with manufacturers' recommendations on product handling, storage, and protection.

1.8 PACKAGING

- A. Deliver products to the job site in their manufacturer's original container, with labels intact and legible.
 - 1. Maintain packaged materials with seals unbroken and labels intact until time of use.
 - 2. Promptly remove damaged material and unsuitable items from the job site, and promptly replace with material meeting the specified requirements, at no additional cost to the Owner.
- B. The Engineer may reject as non-complying such material and products that do not bear identification satisfactory to the Engineer as to manufacturer, grade, quality, and other pertinent information.

1.9 REPAIRS AND REPLACEMENTS

- A. In event of damage, promptly make replacements and repairs to the approval of the Engineer and at no additional cost to the Owner.
- B. Additional time required to secure replacements and to make repairs will not be considered by the Engineer to justify an extension in the Contract Time of Completion.

SECTION 01 74 26

PIPELINE CONSTRUCTION SITE CLEANING

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes requirements for maintaining cleanliness on underground pipeline construction projects.
 - 1. Throughout the construction period, maintain the site at a standard of cleanliness as described in this Section.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- 2. In addition to standards described in this Section, comply with requirements for cleaning as described in other pertinent Sections of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Conduct daily inspections, and more often if necessary, to verify that requirements for cleanliness are being met.
- B. In addition to the standards described in this Section, comply with pertinent requirements of governmental agencies having jurisdiction.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 CLEANING MATERIALS AND EQUIPMENT

A. Provide required personnel, equipment, and materials needed to maintain the specified standard of cleanliness.

2.2 COMPATIBILITY

A. Use only the cleaning materials and equipment which are compatible with the surface being cleaned, and also capable of providing the cleanliness required.

PART 3 - EXECUTION

3.1 PROGRESS CLEANING

A. General:

- Retain stored items in an orderly arrangement allowing maximum access, not impeding traffic or drainage, and providing required protection of materials.
- 2. Do not allow accumulation of scrap, debris, waste material, and other items not required for construction of this Work.
- 3. At least twice each month, and more often if necessary, completely remove all scrap, debris, and waste material from the job site.
- 4. Provide adequate storage for all items awaiting removal from the job site, observing requirements for fire protection and protection of the environment.
- 5. Daily, and more often if necessary, inspect the site and pick up all scrap, debris, and waste material. Remove such items to the place designated for their storage.
- 6. Weekly, and more often if necessary, inspect all arrangements of materials stored on the site. Restack, tidy, or otherwise service arrangements to meet the requirements of Paragraph 3.1 A. 1. above.
- 7. Maintain the site in a neat and orderly condition at all times.

B. Daily Cleaning:

- 1. Take necessary steps to prevent tracking of mud onto adjacent streets, highways, driveways, sidewalks, and other paved areas.
 - a. Wash mud resulting from construction traffic off of adjacent pavements.
 - b. Provide a vacuum/washer/sweeper combination machine to clean pavements on a regular basis, and when requested by the Engineer.

3.2 FINAL CLEANING

- A. "Clean", for the purpose of this Article, and except as may be specifically provided otherwise, shall be interpreted as meaning the level of cleanliness generally provided by skilled cleaners using commercial quality maintenance equipment and materials.
- B. Prior to completion of the Work, remove from the job site all tools, surplus materials, equipment, scrap, debris, and waste. Conduct final progress cleaning as described in Paragraph 3.1 above.
- C. Unless otherwise specifically directed by the Engineer, broom and wash clean paved areas on the site and public paved areas adjacent to the site.
 - 1. Completely remove resultant debris.
 - 2. Provide vacuum/washer/sweeper combination machine if requested by the Engineer.
- D. Schedule final cleaning as approved by the Engineer to enable the Owner to accept a completely clean Work.

SECTION 01 77 01

CONTRACT CLOSEOUT

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes an orderly and efficient transfer of the completed Work to the Owner.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
 - 2. Activities relative to Substantial Completion and Contract closeout are described in the General Conditions.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Prior to requesting that the Engineer issue a certificate of Substantial Completion in accordance with Paragraph 14.04 or 14.05 of the General Conditions, use adequate means to assure that the Work is completed in accordance with the specified requirements and is ready for a joint inspection by Owner, Contractor, and Engineer.
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

1.7 PROCEDURES

- A. Substantial Completion:
 - 1. Prepare and submit the list required by the first sentence of Paragraph 14.04.A of the General Conditions and submit it along with a written request that Engineer issue a certificate of Substantial Completion.
 - 2. Within a reasonable time after receipt of the list, Owner, Contractor and Engineer will jointly inspect the Work to determine status of completion.
 - 3. Should the Engineer determine that the Work is not substantially complete:
 - a. The Engineer will so notify the Contractor, in writing, giving the reasons therefore.
 - b. Remedy the deficiencies and notify the Engineer when ready for reinspection.
 - c. Owner, Contractor and Engineer will reinspect the Work.
 - 4. When the Engineer concurs that the Work is substantially complete:
 - a. The Engineer will prepare a tentative "Certificate of Substantial Completion," accompanied by the Contractor's list of items to be completed or corrected, as verified by the Engineer.
 - b. The Engineer will submit the tentative Certificate to the Contractor for acceptance.
 - c. After Contractor signs and returns the tentative Certificate to Engineer, Engineer will submit the tentative Certificate to Owner accompanied by a tentative list of items to be completed or corrected before final payment.
 - d. Owner will have seven days after receipt of the tentative Certificate during which to make objection to Engineer as to any provisions of the Certificate on attached list.
 - (1) If Owner objects, Engineer will consider Owner's objections. If, after considering Owner's objections, Engineer concludes that the Work is not substantially complete, Engineer will, within fourteen days after submission of the tentative Certificate to Owner, notify Contractor in writing, stating reasons therefore. If, after considering Owner's objections, Engineer considers the Work substantially complete, Engineer will within said fourteen days execute and deliver to Owner and Contractor, a definitive Certificate of Substantial Completion (with a revised tentative list of items to be completed or corrected) reflecting such changes from the tentative Certificate as Engineer believes justified after consideration of any objections of Owner.
 - (2) If Owner has no objections, Engineer will within fourteen days after submission of the tentative Certificate to Owner and Contractor issue a definitive Certificate of Substantial Completion.
 - e. At the time of delivery of the tentative Certificate of Substantial Completion, Engineer will deliver to Owner and Contractor a written recommendation as to division of responsibilities pending final payment between Owner and Contractor with respect to security,

operation, safety, maintenance, heat, utilities, insurance, warranties, and guarantees. Unless Owner or Contractor advise the Engineer in writing of any objections within seven days after delivery of the tentative Certificate of Substantial Completion, the Engineer's aforesaid recommendation will be binding on Owner and Contractor until final payment.

B. Final Completion:

- 1. Prepare and submit the notice required by the first sentence of Paragraph 14.06A of the General Conditions.
- 2. Verify that the Work is complete including, but not necessarily limited to, the items mentioned in Paragraph 14.07.A of the General Conditions.
- 3. Certify that:
 - a. Contract Documents have been reviewed.
 - b. Work has been inspected for compliance with the Contract Documents.
 - c. Work has been completed in accordance with the Contract Documents.
 - d. Equipment and systems have been tested as required, and are operational.
 - e. Work is completed and ready for final inspection.
- 4. Owner, Contractor, and Engineer will make a joint inspection to verify status of completion.
- 5. Should the Engineer determine that the Work is incomplete or defective:
 - a. The Engineer will so notify the Contractor, in writing, listing the incomplete or defective work.
 - b. The Contractor will remedy the deficiencies promptly, and notify the Engineer when ready for reinspection.
- 6. When the Engineer determines that the Work is acceptable under the Contract Documents, he will request the Contractor to make closeout submittals.
- C. Closeout submittals include, but are not necessarily limited to:
 - 1. Project Record Documents described in Section 01 78 39.
 - 2. Warranties and bonds.
 - 3. Section 00 65 36 Warranty Form.
 - 4. Section 00 65 36.03 Contractor's Affidavit Letter.
 - 5. Keys and keying schedule.
 - 6. Spare parts and materials extra stock.
 - 7. Evidence of compliance with requirements of governmental agencies having jurisdiction including but not necessarily limited to:
 - a. Certificates of Inspection.
 - b. Certificates of Occupancy.
 - 8. Certificates of Insurance for products and completed operations.
 - 9. Evidence of payment and release of liens.

- List of subcontractors, service organizations, and principal vendors, including names, addresses, and telephone numbers where they can be reached for emergency service at all times including nights, weekends, and holidays.
- D. Final adjustment of accounts:
 - 1. Submit a final statement of accounting to the Engineer, showing all adjustments to the Contract Price.
 - 2. If so required, the Engineer will prepare a final Change Order showing adjustments to the Contract Price which have not been made by previous Change Orders.

SECTION 01 78 39

PROJECT RECORD DOCUMENTS

PART 1 - GENERAL

1.1 SUMMARY

A. Throughout progress of the Work, maintain an accurate record of changes in the Contract Documents, as described in Paragraph 3.1 below and, upon completion of the Work, submit the recorded changes as described in Paragraph 3.2 below.

B. Related work:

- Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- 2. Other requirements affecting Project Record Documents may appear in pertinent other Sections of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- E. Comply with pertinent provisions of Section 01 33 01.
- F. The Engineer's approval of the current status of Project Record Documents may be a prerequisite to the Engineer's approval of requests for progress payment and request for final payment under the Contract.
- G. Prior to submitting each request for progress payment, secure the Engineer's approval of the current status of the Project Record Documents.
- H. Prior to submitting request for final payment, submit the final Project Record Documents to the Engineer and secure his approval.

1.3 QUALITY ASSURANCE

- A. Delegate the responsibility for maintenance of Record Documents to one person on the Contractor's staff as approved by the Engineer.
- B. Accuracy of records:
 - 1. Thoroughly coordinate changes within the Record Documents, making adequate and proper entries on each page of Specifications and each sheet of Drawings and other Documents where such entry is required to show the change properly.
 - 2. Accuracy of records shall be such that future search for items shown in the Contract Documents may rely reasonably on information obtained from the approved Project Record Documents.
- C. Make entries within 24 hours after receipt of information that the change has occurred.
- D. Do not conceal any work until the required information is recorded.

1.4 DELIVERY, STORAGE, AND HANDLING

- A. Maintain the job set of Record Documents completely protected from deterioration and from loss and damage until completion of the Work and transfer of all recorded data to the final Project Record Documents.
- B. In the event of loss of recorded data, use means necessary to again secure the data to the Engineer's approval.
 - 1. Such means shall include, if necessary in the opinion of the Engineer, removal and replacement of concealing materials.
 - 2. In such case, provide replacements to the standards originally required by the Contract Documents.
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 RECORD DOCUMENTS

A. Job set: Promptly following receipt of the Owner's Notice to Proceed, secure from the Engineer at no charge to the Contractor one complete set of all Documents comprising the Contract.

3.1 MAINTENANCE OF JOB SET

A. Immediately upon receipt of the job set described in Paragraph 2.1 A. above, identify each of the Documents with the title, "RECORD DOCUMENTS - JOB SET".

B. Preservation:

- Considering the Contract completion time, the probable number of occasions upon which the job set must be taken out for new entries and for examination, and the conditions under which these activities will be performed, devise a suitable method for protecting the job set.
- 2. Do not use the job set for any purpose except entry of new data and for review by the Engineer.
- 3. Maintain the job set at the site of Work where designated by the Engineer.

C. Making entries on Drawings:

- 1. Using an erasable colored pencil (not ink or indelible pencil), clearly describe the change by graphic line and note as required.
- 2. Date all entries.
- 3. Call attention to the entry by a "cloud" drawn around the area or areas affected.
- 4. In the event of overlapping changes, use different colors for the overlapping changes.
- D. Make entries in the pertinent other Documents as approved by the Engineer.

E. Conversion of schematic layouts:

- In some cases on the Drawings, arrangements of conduits, circuits, piping, ducts, and similar items, is shown schematically and is not intended to portray precise physical layout.
 - a. Final physical arrangement is determined by the Contractor, subject to the Engineer's approval.
 - b. However, design of future modifications of the facility may require accurate information as to the final physical layout of items which are shown only schematically on the Drawings.
- 2. Show on the job set of Record Drawings, by dimension accurate to within one inch, the centerline of each run of items such as are described in Paragraph 3.1 E. 1. above.

3.2 REVIEW AND SUBMITTAL

- A. Submit the completed set of Project Record Documents to the Engineer as described in Paragraph 1.2.H. above.
- B. Participate in review meetings as required.

C. Make required changes and promptly deliver the final Project Record Documents to the Engineer.

3.3 CHANGES SUBSEQUENT TO ACCEPTANCE

A. The Contractor has no responsibility for recording changes in the Work subsequent to Final Completion, except for changes resulting from work performed under Warranty.

SECTION 31 23 79

TRENCHING, BACKFILLING, AND COMPACTING

PART 1 - GENERAL

1.1 SUMMARY

- A. Trench, backfill, and compact as specified herein and as needed for installation of underground pipelines and utilities associated with the Work.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

- A. Use adequate numbers of skilled workmen who are thoroughly trained and experienced in the necessary crafts and who are completely familiar with the specified requirements and the methods needed for proper performance of the work of this Section.
- B. Use equipment adequate in size, capacity, and numbers to accomplish the work in a timely manner.
- C. Comply with requirements of governmental agencies having jurisdiction.

1.4 DELIVERY, STORAGE, AND HANDLING

- A. Comply with pertinent provisions of Section 01 66 11.
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

2.1 GRANULAR PIPE BEDDING AND COVERING MATERIALS

A. Provide well graded, washed, mixture of gravel or crushed stone aggregate free of clay, loam, dirt, calcareous or other foreign matter conforming to the IDOT "Standard Specifications" gradation No. CA 7 or No. CA 11, or the Standard Specifications for Water and Sewer Construction in Illinois, with the following gradation:

Sieve Size	Percent Passing
1½-inch	100%
1-inch	90 -100%
1/2-inch	30 - 60%
No. 4	0 -10%

- For flexible thermoplastic pipes including sewer pipes, sewage force mains, and water mains: Comply with ASTM D2321, Class I or II as modified below.
 - Exclude sharp angular materials.
 - b. Do not use natural rounded or smooth stone or gravel (such as pea gravel or similarly rounded materials).
 - c. Limit maximum particle size to ½-inch (IDOT CA 15).
 - d. Do not use Class II materials in wet conditions.
- 2. For rigid pipes, comply with ASTM C12, Bedding Class B.

2.2 EXCAVATED BACKFILL MATERIALS

A. Provide soil materials free from organic matter, rubble, or frozen material, containing no rocks or lumps over 6 inches, and with not more than 15 percent of the rocks or lumps larger than 2 inches.

2.3 GRANULAR BACKFILL

- A. Provide granular material.
 - 1. Granular material: Use 100 percent crushed stone or gravel complying with the IDOT "Standard Specifications" gradation No. CA 6 or No. CA 7.

2.4 FLOWABLE FILL TRENCH BACKFILL MATERIALS

- A. Provide controlled low-strength material (CLSM) for excavations within pavement or within 2 feet of pavement located in the Will County Right-of-Way.Use CLSM for trenches under 50 LF or 500SF.
 - 1. Provide a portioned cement, fly ash, fine aggregate and water mix.
 - 2. Comply with IDOT Section 593 for Controlled Low-Strength Material, Backfill for material mix design criteria, and testing.

- a. Acceptable products:
 - (1) Geofill by MixOnSite.
 - (2) Or equal.

2.5 TEMPORARY AGGREGATE PAVEMENT MATERIAL

A. Provide well graded, 100 percent crushed gravel or crushed stone aggregate free of clay, loam, dirt, calcareous or other foreign matter conforming to the IDOT "Standard Specifications" gradation No. CA 6.

2.6 GEOTEXTILE FABRIC

- A. Provide geotextile fabric for separation of granular material and native soil in areas where trench is over-excavated to remove unsuitable materials.
 - 1. Acceptable manufacturers:
 - a. Mirafi: 160N.
 - b. Synthetic Industries: 601.
 - c. Or equal.

2.7 WATER MAIN REPAIR

- A. Repair water main or water services damaged during construction utilizing products of type and manufacturers as approved by the Owner.
- B. Pipe couplings for joining of sections of cut water main where a section of new pipe is used to replace a broken pipe.
 - 1. Acceptable manufacturers:
 - a. Dresser Style 38.
 - b. Smith-Blair CC-441.
 - c. Or equal.
- C. Repair clamps for broken or cracked pipe and sealing of existing corporation stop opening.
 - 1. Use full-circle single band all stainless steel clamps.
 - 2. Acceptable manufacturers:
 - a. Dresser Style 360.
 - b. Smith-Blair 200 Series.
 - c. Or equal.
 - 3. Replace damaged service corporation stops by installation of full-circle single band all stainless steel clamps, with service outlet, matching manufacturer's and styles used for repair of a cracked pipe.

2.8 DRAIN TILE REPLACEMENT

- A. Replacement pipe: New pipe of the same size. Match material or use new PVC SDR-26 pipe per ASTM D3034.
- B. Utilize flexible couplings with stainless steel bands for connecting new pipe to old pipe.

C. Provide CA-6, CA-7, CA-11 aggregate for backfill material.

2.9 PIPE INSULATION

- A. Rigid Pipe Insulation:
 - 1. Provide extruded polystyrene sheathing conforming with ASTM C578, Type IV.
 - 2. Thickness: 2 inches.
 - 3. R-Value: 10.
 - 4. Water absorption: No greater than 0.10% by volume per ASTM C272.
 - 5. Acceptable product:
 - a. Formular Rigid Foam Insulation, Owens-Corning.
 - b. Or equal.

PART 3 - EXECUTION

3.1 SURFACE CONDITIONS

A. Examine the areas and conditions under which work of this Section will be performed. Correct conditions detrimental to timely and proper completion of the Work. Do not proceed until unsatisfactory conditions are corrected.

3.2 GENERAL CONSTRUCTION REQUIREMENTS

- A. Protection of existing facilities:
 - Unless shown to be removed, protect existing structures, conduits, active
 utility lines and all other facilities shown on the Drawings or otherwise made
 known to the Contractor. If damaged, repair, replace, or restore to a
 condition equal to or better than the original condition at no additional cost
 to the Owner.
 - 2. Notify all persons, firms, corporations, or agencies owning or using any existing structures, conduits, or utilities which may be affected by the Work prior to the start of construction.
 - 3. Make arrangements to locate, maintain, protect, and/or relocate facilities in order to complete the Work.
 - 4. Make such exploration as is necessary to determine the exact location of underground utilities.
 - 5. Exercise care during the progress of work in the area to prevent damage to the utilities.
 - 6. Whenever it becomes necessary to relocate underground gas mains, telephone conduit, or electrical lines or support or relocate utility poles, the utility company involved will make such relocation or provide pole support. Notify the utility company promptly.
 - 7. Whenever it becomes necessary to relocate water or other pipes or conduits in direct conflict with the proposed pipe (exclusive of culverts) which are not shown on the Drawings, obtain the direction from the Engineer for the relocation. Compensation will be allowed only for such quantities as directed by the Engineer.

- 8. Do not block or obstruct sidewalks, streets, and pavements.
- 9. Whenever during construction operations any loose material is deposited in the flow line of gutters, drainage structures, or ditches such that the natural flow line of water is obstructed, remove this loose material at the close of each working day. At the conclusion of construction operations, keep all drainage structures and flow lines free from dirt and debris.
- 10. Do not obstruct accessibility of fire hydrants.
- 11. Maintain access to adjacent areas at all times.

B. Protection of Trees and Shrubs:

- Protect trees and shrubs from damage.
- 2. Do not remove trees or shrubs unless indicated on the Drawings or authorized in the field by the Engineer.
- 3. Where trees which are to remain interfere with normal excavation operations, use the following procedures:
 - Prior to excavation, carefully remove trees with trunk diameters of less than 4 inches, shrubs, and other plantings in the way of construction.
 - Do not machine excavate within a distance of three trunk diameters or 12 inches (whichever is greater) of any tree, and do not cut roots over 2-inch diameter unless approved by the Engineer.
 - c. Excavate by hand when closer than three tree trunk diameters or 12 inches (whichever is greater).
 - d. Tree tunneling where necessary to be determined by the Engineer.
 - e. Tie back shrubs and tree limbs to prevent loss or damage.
 - f. Prune damaged limbs and branches.
 - g. Provide plank wrappers wired in place to protect tree trunks from being damaged by trench machinery, tractors, or trucks; remove protective planking as soon as practical after work in vicinity has been completed.
 - h. Remove spoil banks from around trees by hand to prevent damage to trunks by construction machinery.
- 4. Replace trees and shrubs that cannot be protected or are damaged during construction:
 - a. Replant or replace with stock of like character, quality, variety, size, shape, color and condition upon completion of the construction.
 - b. Replace 4-inch diameter and larger trees with one 4-inch diameter size tree for each 6" of original tree diameter or fraction thereof.
 - c. Replace trees smaller than 4-inch diameter and shrubs with same kind and type.
 - d. As an option, replant trees smaller than 2-inch diameter or shrubs that are not damaged.
- 5. Remove and replace trees and shrubs which do not survive in good condition for a period of 18 months after time of planting.

C. Work on private property:

1. Construct work on private property within easements obtained by the Owner as shown on the Drawings.

- a. The Contractor will be permitted construction privileges within construction easement lines as shown on the Drawings.
- b. Perform the work in a manner such as to minimize damage to lawns, shrubs, trees and other plantings, driveways, sidewalks, fences, outbuildings, and any other miscellaneous improvements, using proper size and type of equipment.
- c. The Engineer has the authority to prohibit the use of any equipment that in his judgment is too large or otherwise unsuitable for the conditions of the work on private property.
- 2. Remove and replace fences, outbuildings and other miscellaneous improvements in the way of construction to the satisfaction of the property owner.
- 3. When working in cultivated fields or gardens, remove original topsoil to a depth of 12 inches prior to excavation, and replace the topsoil to its original depth and grade upon completion of trench backfill.
- 4. Restore the private property to its original condition or better, free of debris, stones and excess materials.

3.3 TRENCHING

- A. Do not advance trench excavation more than 50 feet ahead of completed pipe installation except as approved by the Engineer.
- B. Provide and maintain sheeting, shoring, and bracing necessary for protection of the Work, adjacent property, and for the safety of personnel.
 - 1. Remove temporary sheeting and bracing after backfilling to an elevation which will prohibit caving of exposed sidebanks.
 - 2. Fill voids left by the withdrawal of sheeting with compacted sand.
 - 3. The Engineer may direct that supports in trenches be cut off at any specific elevation to protect adjacent facilities or property. Compensation for support left in place will be negotiated.
 - 4. No extra payment will be made for the supports left in place without the direction of the Engineer.
 - 5. Do not leave supports within 4 feet of the ground or pavement surface in place without the permission of the Engineer.
- C. Provide pumping, bailing, wellpointing, and construct ditches and dikes required to dewater and drain ground water, sewage, or stormwater to keep the excavation and site dry for the completion of the Work.

D. Excavation:

- 1. Excavate by open cut unless otherwise indicated on the Drawings.
- 2. Excavate trenches to the depths and grades necessary for the pipelines with allowances for bedding material.
 - a. Comply with the following minimum depth of cover unless otherwise noted on the Drawings.
 - b. Water pipelines: 6 feet.
 - c. Sewage and sludge pressure piping: 5 feet.

- d. Air and gas piping: 3 feet.
- e. Electrical or wiring conduits and cables: 30 inches.
- 3. Over-excavate organic, soft, spongy, or otherwise unsuitable soils found at or below the bottom of the trench to meet firm subsoil or as directed by the Engineer.
- 4. Comply with the following maximum trench widths at the top of pipelines:

Nominal	
Pipe Sizes	Trench Widths
(inches)	<u>(inches)</u>
12 or smaller	30
14 - 18	36
20 - 24	42
27 - 30	48
33 and larger	1-1/3 times pipe OD

3.4 EXCAVATION FOR APPURTENANCES

- A. Excavate for manholes and similar structures to the depths as shown on the Drawings and to a distance sufficient to leave at least 12 inches clear between outer surfaces and the embankment or shoring that may be used to hold and protect the banks.
- B. Over-depth excavation beyond depths indicated on the Drawings that has not been directed will be considered unauthorized. Fill with sand, gravel, or lean concrete as directed by the Engineer, and at no additional cost to the Owner.

3.5 BEDDING AND COVERING OF PIPE

A. General:

- Bedding is defined as the shaped and tamped material which supports the pipes. Covering is defined as the compacted material which protects and covers the pipes.
- 2. Provide continuous bedding and covering for underground pipelines, except where concrete encasement, concrete cradles, boring or jacking are indicated.

B. Pipe bedding:

- 1. Provide compacted granular pipe bedding and covering material with a minimum thickness of 4 inches under pipe barrels and 2 inches under bells.
- 2. Wherever the trench is over-excavated due to the removal of unsuitable material, refill the excavated area to the bottom of the pipe bedding with material conforming to the IDOT "Standard Specifications" gradation No. CA 1 as indicated in the "Open Cut Pipeline Installation Unsuitable Material Removal/Replacement Detail".
 - a. Removal and replacement of material, or unsuitable material, to a depth of one foot below the bottom of the pipe barrel is considered incidental to installation of the pipe.

- 3. Wherever the trench is over-excavated to remove unsuitable material, install geotextile fabric between native soil and granular material:
 - Install fabric to cover bottom and sides of trench to heights as follows:
 - (1) For all flexible pipe and rigid pipe 24-inch and smaller: to envelop entire bedding and covering material and overlap 1-foot at the top.
 - (2) For rigid pipe 27-inch and larger: to cover bedding material and from sides of trench to edge of pipe.
 - (3) Where undercut is of a depth that requires more than one piece of fabric to provide envelope, provide sewn seams between sections of fabric.
- 4. Wherever two or more pipes or conduits are placed in the same trench or excavated area, backfill the trench with granular pipe bedding and covering material to support the uppermost pipe or conduit.
- 5. Provide sand bedding with a minimum thickness of 3 inches under electrical and wiring conduits and cables.

C. Pipe covering:

- 1. Following placement of pipe and inspection of joints, provide compacted granular pipe bedding and covering material for the full width of the trench to the following levels unless otherwise shown on the Drawings:
 - a. To 12 inches above the top of the pipe for all pipe sizes.
- 2. Place granular pipe bedding and covering material in uniform loose layers not exceeding 8 inches thick.
 - a. Compact each layer firmly by ramming or tamping with tools approved by the Engineer in such a manner as not to disturb or injure the pipe to yield a minimum density of 95 percent of maximum dry density as determined according to ASTM D1557 or AASHTO-T180.
- 3. Where trench is widened by installation of structures, jacking pits, or tunnel shafts, extend bedding and covering materials to total width of excavations and compact as noted in following section.
 - a. As a contractor's option, to eliminate compacting of the material, replace bedding and covering material with flowable fill at no additional cost to Owner.

3.6 TRENCH BACKFILLING AND COMPACTING

A. General:

- 1. Backfill trench from the top of pipe cover to topsoil, paving subgrade, or foundation level.
- 2. If trenches settle during the period of construction and within the guarantee period of the work, fill trench back to the surrounding grade, and restore the surfaces.
- B. For trench in lawns, parkways, and other improved areas not subject to vehicular traffic:

- 1. Backfill with excavated materials in uniform loose layer not exceeding 12 inches thick.
- Compact each layer of trench backfill materials to yield a minimum of 85 percent of maximum dry density as determined according to ASTM D1557 or AASHTO-T180.
- C. For trench in unimproved areas and cultivated fields:
 - 1. Backfill with excavated materials.
 - 2. Provide crowned surface to compensate for settlement.
- D. For trench in streets, parking areas, driveways, sidewalks, curb and gutter, or within 2 feet of any proposed curb and gutter, sidewalk, or other paved areas:
 - 1. Backfilling with granular backfill materials:
 - a. Place in uniform loose layer not exceeding 12 inches thick and compact with vibrating roller or equivalent.
 - b. Water jetting may not be be used in lieu of vibratory compaction.
 - c. Fill the top of trenches with temporary aggregate pavement material to the depth(s) required to provide aggregate base and pavement base, binder and surface courses of the depths(s) shown in the Details in the Drawings.
 - 2. Compacting requirements:
 - a. Compact each layer of trench backfill materials to yield a minimum density of 90 percent of maximum dry density as determined according to ASTM D1557 or AASHTO T-180.
 - b. Determine the density of compacted backfill at intervals of not more than 500 feet at locations selected by the Engineer.
 - c. Provide the services of an independent testing laboratory for the density tests complying with the pertinent provisions of Section 01 45 29.
 - 3. Maintain temporary aggregate pavement level with adjoining pavement surfaces until the permanent pavement is placed.
- E. For trench in pavement of within 2 feet of pavement located in Will County Rightof-Way, use CLSM. Use CLSM when trench is less than 50 LF or 500 SF:
 - 1. Backfill with Portland cement, fly ash, fine aggregate and water flowable fill mixture.
 - 2. Comply with IDOT Section 593 for Controlled Low-Strength Material (CLSM) for placement.

3.7 BACKFILL AND BEDDING FOR APPURTENANCES

- A. Provide 3 inches of sand or granular bedding material unless otherwise shown on the Drawings.
- B. Do not backfill until new concrete has properly cured, and any required tests have been accepted.
- C. Backfill in lawns and landscaped areas with excavated materials.

D. Backfill in pavement around manholes, catch basins, inlets, valve vaults, and other structures as directed by the Engineer with granular backfill materials.

3.8 FINISH GRADING

A. General:

- 1. Provide finish grading and filling to achieve the lines and grades.
- 2. Slope grades to drain away from structures.
- 3. Replace culverts damaged during the construction with new culverts of the same size and type unless instructed differently in other sections of these Specifications.

B. Finish grading:

- Except where mounding over trenches is specified, grade smooth areas of the Work including previously grassed areas that have been disturbed, and adjacent transition areas.
- 2. Fill and compact depressions from settlement and round tops of embankments and breaks in grade.
- 3. Protect newly graded areas from traffic and erosion. Repair settlement or washing away that may occur prior to surface restoration and re-establish grades to the required elevations at no additional cost to the Owner.

C. Disposal of waste excavated material:

- 1. Remove unsuitable and surplus excavated materials not used for backfilling from the project site.
- 2. Do not deposit on public or private property without written permission from property owner or authorized representative of appropriate public agency.

3.9 TEMPORARY HOT-MIX PAVEMENT SURFACE

- A. Provide a premixed hot-mix asphalt wearing surface for use during the period between backfilling the trench and constructing the permanent pavement surface at locations as shown on the Drawings or as directed by the Engineer.
 - 1. Comply with requirements of Section 32 10 00 of these Specifications.
- B. Remove the temporary pavement surface at the time of permanent pavement construction.

3.10 WATER MAIN REPAIR

- A. Whenever existing water mains and water service pipes are damaged during construction, stop the pipe installation work and immediately repair the damaged portion of the existing piping.
- B. Contact the Engineer and Owner immediately to report the location and extent of the damage.

- C. Repair the water main with methods complying with the "Standard Specifications for Water and Sewer Construction in Illinois", and any additional requirements required by the Owner.
- D. Utilize only materials of repair as noted in the products section of this specification or as dictated by the Owner.
- E. Where water services have been stripped or pulled from the water main, replace the corporation stop as instructed by the Engineer and Owner, and replace the water service pipe to a point as directed by the Owner.
- F. Comply with disinfection requirements as dictated by the Owner.
- G. Do not cover the repair until work is inspected and approved by Owner.

3.11 PIPE INSULATION

- A. Rigid Pipe Insulation:
 - Place rigid insulation board above the pipe bedding material to the width of the trench.
 - 2. Place rigid insulation board to the required thickness and in the locations as shown on the Drawings.

3.12 DRAIN TILE REPLACEMENT

- A. Replace all drain tile disturbed or damaged with new pipe of the same size, at same grade and slope, and utilizing flexible couplings for connection to existing pipe.
- B. Indicate location, depth, and size of drain tile on Job Set of plans.
- C. Provide aggregate backfill from bottom of trench to a minimum of 12 inches above new drain pipe.
- D. Inform Engineer immediately when drain tiles are encountered as a part of the trenching operation.
 - 1. Drain tile repair conducted without the observance of the Engineer will not be paid for.

END OF SECTION

SECTION 31 25 00

EROSION AND SEDIMENTATION CONTROLS

PART 1 - GENERAL

1.1 SUMMARY

A. Provide soil erosion and sediment controls as shown on the Drawings, as specified herein, and as needed for a complete and proper installation.

B. Related work:

 Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.

C. References:

 "Illinois Urban Manual", a technical manual designed for Urban Ecosystem Protection and Enhancement, prepared by the United States Department of Agriculture's (USDA) Natural Resources Conservation Service (NRCS) and the Illinois Environmental Protection Agency (IEPA), latest revision.

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).

1.3 QUALITY ASSURANCE

A. Provide adequate numbers of skilled workers who are thoroughly trained and experienced in the necessary crafts and who are completely familiar with the specified requirements and methods needed for proper performance of the work of this Section.

1.4 DELIVERY, STORAGE, AND HANDLING

- A. Comply with pertinent provisions of Section 01 66 11.
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 GENERAL

- A. Provide soil erosion and sediment controls in accordance with the "Illinois Urban Manual" and the IDOT "Standard Specifications".
- B. Provide manufacturer's certification that product meets the minimum specified value, if requested by the Engineer.

2.2 TEMPORARY EROSION AND SEDIMENT CONTROL SYSTEMS

A. Inlet Protection Filter:

- 1. Provide a drainage structure inlet filter assembly consisting of a steel frame, replaceable geotextile fabric bag, reinforced filter basket, stainless steel band suspended from the frame, and an overflow feature.
- 2. Provide a frame of rigid galvanized steel meeting the requirements of ASTM-A36.
- 3. Provide a filter bag constructed of a non-woven geotextile fabric with the following properties:
 - a. Minimum weight: 4 ounces per square yard.
 - b. Minimum flow rate: 145 gallons per minute per square foot.
 - c. Minimum silt and debris capacity: 2 cubic feet.
- 4. Provide reinforced filter basket of polyester or stainless steel mesh.
- 5. Acceptable products:
 - a. FlexStorm Inlet Filter by Inlet & Pipe Protection, Inc.
 - b. Catch-All by Marathon Materials, Inc.
 - c. Or equal.

B. Silt bag:

- 1. Provide filter or silt bags for dewatering pump discharge hoses that are utilized at well points or on dewatering pumps used to lower the water table and allow trench work or structure installation.
- 2. Utilize bags made of non-woven geotextile material.
- 3. Acceptable products:
 - a. Silt Bag by Layfield.
 - b. Dirtbag by Geo-Synthetics.
 - c. Or equal.

PART 3 - EXECUTION

3.1 INLET PROTECTION FILTER

A. General:

- 1. Verify the number and dimensions of the drainage structure frames for installation of the inlet filter assemblies.
- 2. Inspect and clean filters weekly and after every rainfall.

- 3. Dispose of debris removed at an approved location.
- 4. Remove filter assembly as directed by the Engineer.
- 5. The drainage structure inlet filter assembly will remain the property of the Contractor.

3.2 SILT BAG

- A. Provide protection from sedimentation to storm sewers, streams, ponds, or wetlands during well pointing or trench dewatering by the use of silt bags or silt socks at the hose or pipe discharge points.
- B. Dispose of collected silt or sediment offsite or at locations approved by the appropriate permitting authority.

END OF SECTION

SECTION 32 10 00.19

STREET, DRIVEWAY, AND SIDEWALK REPLACEMENT

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes replacement of streets, driveways, curbs and gutters, and sidewalks removed or damaged during pipeline construction.
- B. Provide base course and surface courses as specified herein, and as needed for a complete replacement of all streets, driveways, curbs and gutters, and sidewalks.
- C. Construct streets, driveways, concrete curbs and gutters, and sidewalks in accordance with IDOT "Standard Specifications for Road and Bridge Construction", hereby referred to as IDOT "Standard Specifications".
- D. Related work:
 - 1. Comply with only the pertinent provisions of this Section for the type of replacement required for the Work.
 - 2. Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Sections in Division 01 General Requirements of these Specifications.

E. References:

- (Reserved).
- 1.2 SUBMITTALS (Reserved).
- 1.3 QUALITY ASSURANCE (Reserved).
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 BASE COURSE

A. Aggregate:

1. Provide well graded, 100 percent crushed gravel or crushed stone aggregate free of clay, loam, dirt, calcareous or other foreign matter conforming to the IDOT "Standard Specifications", gradation No. CA 6.

- B. Hot-mix asphalt:
 - 1. Comply with applicable articles of Section 355 of the IDOT "Standard Specifications" and the most recent edition of IDOT District 1 Special Provisions, "HMA Mixture Design Requirements", "Ground Tire Rubber (GTR) Modified Asphalt Binder", "Friction Aggregate", the BDE Special Provisions for "Hot-Mix Asphalt Density Testing of Longitudinal Joints".
 - a. Hot-Mix Asphalt Binder Course, IL 19.0, N50.
 - b. 4% air voids @ 50 gyrations.
- C. Portland Cement concrete:
 - 1. Comply with applicable articles of Section 353 of the IDOT "Standard Specifications" for materials and material preparation.

2.2 SURFACE COURSE

- A. Aggregate surface course (Type A or Type B):
 - 1. Use crushed stone or crushed gravel complying with the IDOT "Standard Specifications" Articles 1004.01 and 1004.04 gradation No. CA 6.
- B. Hot-mix asphalt: Comply with applicable articles of Section 406 of the IDOT "Standard Specifications" and the most recent IDOT District 1 Special Provisions, "HMA Mixture Design Requirements", "Ground Tire Rubber (GTR) Modified Asphalt Binder", "Friction Agreement" and the BDE Special Provision for "Hot-Mix Asphalt Density Testing for Longitudinal Joints".
 - 1. Hot-mix asphalt surface course: Use Hot-mix asphalt Surface Course, Mix "D", N50 (0-10,000 ADT).
 - a. Mix type: IL 9.5 mm.
 - b. 4% air voids @ 50 gyrations.
 - 2. Hot-mix asphalt binder course:
 - a. Use Hot-mix asphalt Binder Course, IL-19, N50 (0-10,000 ADT).
 - (1) Mixture type: IL 19.0.
 - (2) 4% air voids @ 50 gyrations.
 - 3. Hot-mix asphalt leveling binder:
 - Use Hot-mix asphalt Leveling Binder (Machine Method), N50 (0-10,000 ADT).
 - (1) Mixture type: IL 9.5 mm.
 - (2) 4% air voids @ 50 gyrations.
 - 4. Hot-mix asphalt driveway:
 - a. Use Hot-mix asphalt Surface Course, Mix "D", N50.
 - (1) Mixture type: IL 9.5 mm.
 - (2) 4% air voids @ 50 gyrations.
- C. Portland Cement concrete pavement:
 - 1. Comply with applicable articles of Section 420 of the IDOT "Standard Specifications" for materials and material preparation.
- D. Portland Cement concrete driveway pavement:
 - 1. Comply with Section 423 of IDOT "Standard Specifications" for materials and material preparation.

- E. Portland Cement concrete walk:
 - 1. Comply with applicable requirements of Section 424 of the IDOT "Standard Specifications".
- F. Detectable warnings:
 - Comply with Section 424.09 of IDOT "Standard Specifications" for materials, installation and certification requirements.

2.3 PORTLAND CEMENT CONCRETE DRIVEWAY

A. Comply with applicable requirements of Section 423 of IDOT "Standard Specifications".

2.4 CURB AND GUTTER

- A. Portland Cement concrete:
 - Comply with applicable requirements of Section 606 of the IDOT "Standard Specifications".

2.5 PROTECTIVE COAT

- A. Use combination of 50 percent boiled linseed oil and 50 percent petroleum spirits by volume.
 - 1. Comply with Section 1023 of the IDOT "Standard Specifications".

2.6 TEMPORARY HOT-MIX ASPHALT PAVEMENT MATERIAL

- A. Provide hot-mix asphalt during summer months while plants are operating.
- B. Materials:
 - 1. Hot-mix asphalt binder or surface course, Mix "D", N50.
 - a. See paragraph 2.2, Hot-mix asphalt binder course and Hot-mix asphalt surface course.
- C. Provide asphalt (cold-patch) wearing surface during winter months when hot-mix is not available.
- D. Minimum compacted thickness: 2 inches.

2.7 PAVEMENT MARKINGS

- A. General:
 - Comply with IDOT "Standard Specifications" applicable articles of Sections 780, 781, and 783, for all preparation and materials to remove and replace all obstructed pavement markings.
- B. Materials Permanent:
 - 1. Thermoplastic: Article 1095.01.
 - 2. Preformed thermoplastic pavement marking: 1095.05.

- 3. Raised reflective pavement markers: Section 1096.01.
- C. Materials Temporary:
 - I. Tape: Comply with IDOT "Standard Specifications" for Road and Bridge Construction, Article 1095.06 for materials.
 - a. Do not use paint for temporary pavement markings.

PART 3 - EXECUTION

3.1 GENERAL CONSTRUCTION REQUIREMENTS

- A. Comply with the applicable articles of the following Sections of the IDOT "Standard Specifications" except as modified herein:
 - Base courses:
 - a. Aggregate Type A or Type B: Section 351.
 - b. Hot-mix asphalt: Section 355.
 - c. Portland Cement concrete: Section 353.
 - Surface courses:
 - a. Aggregate, Type A, or Type B: Section 402.
 - b. Hot-mix asphalt: Section 406.
 - c. Portland Cement concrete: Section 420.
 - Prime coats: Section 406.
 - 4. Curb and gutter: Section 606.
 - 5. Portland Cement concrete driveways: Section 423.
 - 6. Area Reflective Crack Control Treatment, System A: Section 443.
- B. Maintain the temporary pavement material installed during trench backfilling as temporary driving surface.
 - 1. Provide and maintain temporary culverts or other drainage devices.
 - 2. Shape the road bed to provide positive drainage.

3.2 REMOVAL OF EXISTING PAVEMENT SURFACES

- A. Remove and dispose of all existing pavement surface materials as necessary to complete the Work.
 - Remove all existing street and driveway pavements, including surfaces, bases and stabilized subbases, curbs and gutters, and sidewalks to a width of not less than one foot on undisturbed ground on each side of the trench excavation.
 - Saw cut to a full depth to provide a straight line joint between the existing pavement, curb and gutter or sidewalk which will remain and the portion to be removed and replaced.
 - a. Remake the saw cut joint as required at the time of resurfacing.
 - 3. Do not use any broken pavement, curb and gutter or sidewalk materials for backfilling the trench.

3.3 REPLACEMENT OF PAVEMENT SURFACES

- A. Reconstruct all street and driveway pavement including surfaces, bases, and stabilized sub-bases, curbs and gutters, and sidewalks that are removed or damaged as a result of the construction.
 - 1. Set all manholes, valve vaults, inlets, service boxes, and other structures and appurtenances to the established finish grade before pavement replacement.
 - 2. Construct in accordance with the details as shown on the Drawings.
 - 3. Replace all surface courses, and base courses to match the grade, cross-section, and thickness of existing, except:
 - a. Comply with the minimum thickness as specified herein.

B. Base courses:

- 1. Use existing temporary aggregate pavement material installed during trench backfilling as aggregate base.
- 2. Remove portion of the existing temporary aggregate pavement material necessary to install base, binder and surface courses.
 - a. Scarify, grade, and shape the aggregate base courses.
 - b. Place and compact additional material as directed by the Engineer.
- 3. Reuse the removed temporary pavement material for any of the following:
 - a. Trench backfill or bedding.
 - b. Temporary pavement at another location.
 - c. Fill material for subgrade removal and replacement.
 - d. Subbase granular material or aggregate base course if the Engineer determines that there is not significant segregation or contamination of the material and if it is constructed as specified in these Specifications.
- 4. Replace existing PCC base course with a minimum of 10 inches of Portland Cement concrete base course.
- 5. Driveway replacement shall include a minimum of 3 inches of aggregate base course.

3.4 PRIME COATS

- A. Bituminous prime coat:
 - 1. Apply prime coat to all aggregate base courses prior to installation of asphalt binder courses at a minimum rate of 0.40 gallons per square yard.
 - a. Apply prime coat to saw-cut edges of existing pavement or driveway that abuts trench limits, by hand swabbing, if necessary.
 - 2. Apply prime coat to all existing concrete or asphalt surfaces prior to installation of new hot-mix asphalt leveling binder, binder, or surface courses at a rate of 0.05 to 0.10 gallons per square yard.
 - a. Prime may be eliminated between new binder and surface courses only when surface course is applied immediately after binder course, when binder has cooled to the appropriate temperature.
 - b. Apply prime coat to new binder course if traffic has been allowed on binder course prior to installation of surface course at a rate of 0.10 gallons per square yard.

3. Do not apply when ambient temperature is less than as specified by IDOT Section 406, or when local conditions indicate that rain is imminent.

3.5 SURFACE COURSES

- A. Hot-mix asphalt binder courses and surface courses:
 - 1. Install hot-mix asphalt binder with a minimum thickness of 2½ inches and hot-mix asphalt surface courses with a minimum thickness of 1½ inches to attain a minimum total thickness of 4 inches.
 - a. For hot-mix asphalt driveways, install hot-mix asphalt binder with a minimum thickness of 1½ inches and hot-mix asphalt surface course with a minimum thickness of 1½ inches.
 - b. Install hot-mix asphalt binder with a minimum thickness of 10 inches and hot-mix asphalt surface courses with a minimum thickness of 2 inches to attain a minimum total thickness of 12 inches at the intersection of Elgin Ave and N Briggs St.
 - c. Install thick binder courses in lifts approved by Engineer and provide steel plate over trenches if necessary to allow cooling.
 - 2. Repair settled trenches, spalled asphalt, and other defective binder before placement of hot-mix asphalt surface course.
 - 3. Resaw-cut edges of trench limits, if necessary, to provide a clean-straight edge prior to installation of hot-mix asphalt surface course.
- B. Portland Cement concrete street or driveway pavement:
 - Replace concrete street pavement with a minimum of 12 inches of Portland Cement concrete.
 - 2. Replace concrete driveway pavement with a minimum of 6 inches of Portland Cement concrete.
 - 3. Provide steel reinforcing bars equal in size and spacing to reinforcement of original pavement.
 - 4. Provide 6 x 6-W4.0 x W4.0 welded wire fabric if steel reinforcing does not exist.
 - 5. Provide #4 grade 60 steel dowels around total perimeter of sawcut and adjacent curb and gutter where PCC pavement is replaced.
 - a. Provide 24-inch long dowels.
 - b. Drill existing concrete to a depth of 12 inches and set dowels with epoxy.
 - c. Space dowels at 24 inches center to center.
- C. Aggregate surface courses and driveways:
 - 1. Replace aggregate surface courses and driveways with a minimum of 8 inches of surface aggregate.

3.6 CURB AND GUTTER

- A. Replace curb and gutter with curb and gutter of the same size and cross-section.
- B. Provide protective coat if so directed by the Owner.

C. Provide two 1/2-inch steel reinforcing bars in full length of new curb and gutter and two 12-inch long slip dowels extended 6 inches into existing curb and gutter.

3.7 SIDEWALKS

- A. Construct to match existing walks.
- B. Portland Cement concrete walks:
 - 1. Erect temporary forms for placement of concrete as specified in Article 424.05 of the IDOT "Standard Specifications".
 - 2. Construct in accordance with applicable articles of Section 424 of the IDOT "Standard Specifications".
 - a. Provide a minimum thickness of 5 inches.
 - 3. Provide ramps accessible to the disabled in locations as shown on the Drawings and in accordance with current IDOT Highway Standard Details.
 - a. Refer to IDOT Standard Details 424001, 424006, 424011, 424016, 424021, and 424026, latest revision.
 - 4. Provide detectable warnings at ramps as specified in Section 424.09 of IDOT's "Standard Specifications".

3.8 TEMPORARY ASPHALT PAVEMENT SURFACE

- A. Provide a hot-mix asphalt wearing surface for use during the period between backfilling the trench and constructing the permanent pavement surface.
- B. Provide 2-inch (compacted) minimum wearing surface.
 - 1. Utilize mechanical tamper, vibrating tamper, or self-propelled roller to achieve compaction.
 - 2. To facilitate extra compaction by traffic, surface of patch may be finished up to 1/2-inch above existing pavement.
 - 3. Provide smooth, drivable surface, free of ruts or depressions.
 - 4. Remove all loose materials from edges of patch, and provide temporary sawcuts if directed by the Engineer.
- C. Utilize premix asphalt (cold patch) wearing surface only during winter months when hot-mix asphalt is not available.
 - 1. Comply with paragraph 3.8 B. when installing asphalt cold-patch.
- D. Remove and dispose of all temporary asphalt pavement surface materials at the time of permanent pavement construction.

3.9 STRUCTURE ADJUSTMENT

- A. Comply with applicable articles of Section 603 of IDOT "Standard Specifications" for materials and installation except as modified below:
 - 1. Do not use steel, ductile iron, or cast iron adjusting rings.
 - 2. Do not use common brick.

B. Provide watertight joints between concrete adjusting rings, frame, and structure when adjusting sanitary sewer manhole frames.

3.10 PAVEMENT MARKINGS

A. General:

- 1. Replace all existing pavement markings that are removed or damaged during pipe installation operations.
- 2. Mark locations (by dimensions) on Job Plan of all existing pavement lane markings, turn arrow, stop bars, and parking stall markings prior to excavation.
- 3. Replace all existing markings after placement of final surfaces.
- 4. Replace pavement markings with same kind and type unless otherwise noted on the Drawings or in Contract Documents.

B. Removal:

1. Remove pavement markings or reflectors per applicable articles of IDOT "Standard Specifications" Section 783.

C. Temporary - tape:

- Use temporary marking tape to delineate temporary lanes of traffic, stop bars, or parking stall lanes on temporary asphalt pavement, asphalt base courses, and surface courses if indicated on the Drawings or if so directed by the Engineer.
- 2. Remove temporary marking tape prior to installation of permanent markings.

END OF SECTION

SECTION 32 92 00.13

LAWNS AND GRASSES

PART 1 - GENERAL

1.1 SUMMARY

- A. Provide topsoil, seeding, sodding, and care of grass during establishment period for a complete surface restoration of lawns, parkways, and other areas disturbed as a result of the construction.
- B. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.
- C. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals (Reserved).
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- 1.3 QUALITY ASSURANCE (Reserved).
- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 TOPSOIL

A. Provide a mixture of black dirt having at least 90 percent passing a No. 10 sieve, free of large roots, brush, sticks, weeds, and stones larger than 1/4-inch in diameter, and any other debris.

2.2 AGRICULTURAL LIMESTONE

A. Provide agricultural-grade ground limestone, ground sufficiently fine so that at least 80 percent will pass through a No. 8 sieve, containing not less than 80 percent calcium carbonate equivalent. Moisture content at time of delivery not exceeding 8 percent.

2.3 FERTILIZER

A. Provide commercial grade fertilizer, having nutrient content of 16 percent nitrogen, 6 percent phosphorus, and 24 percent soluble potash.

2.4 EROSION CONTROL MATERIALS

A. Excelsior blanket:

- 1. Provide excelsior blanket consisting of a machine produced mat of wood excelsior of 80 percent 6 inches or longer fiber length.
- 2. Provide cover with a 60-day photodegradable extruded plastic (polypropylene) or biodegradable natural (jute fiber) mesh netting having an approximate minimum opening of 5/8" x 5/8" to maximum opening of 2" x 2".
- 3. Comply with the following:

Minimum width: 24 inches. Minimum weight: 0.9 lbs./sq.yd. Minimum length of roll: 150 feet.

B. Knitted straw mat:

- Provide knitted straw mat consisting of machine assembled blanket of clean, weed-free straw from agricultural crops, approximately 1/2-inch loose thickness.
- 2. Cover the top side of the blanket with a photodegradable plastic (polypropylene) or biodegradable natural (jute fiber) mesh netting of 3/8" x 3/8" square openings adhered to the straw by a knitting photodegradable or biodegradable thread.
- 3. Supply the blanket in a protected rolled mat form of 6'-6" width and the dry weight not less than 0.70 pounds per square yard.

C. Straw/coconut fiber mat:

- 1. Machine-assembled blanket of clean, weed-free 70 percent straw from agricultural crops and 30 percent coconut fiber, of approximately ¼-inch thickness.
- 2. Cover the top side of the mat with a photodegradable plastic (polypropylene) or biodegradable natural (jute fiber) mesh netting with 3/8-inch by 3/8-inch square openings.
 - a. Attach mesh to the mat with photodegradable or biodegradable knitting thread.
- 3. Supply blanket in protected rolled mat form of 6'-6" width, with a dry weight of 0.50 pounds per square yard minimum.

- 4. Acceptable products:
 - a. SC150BN by North American Green.
 - b. Or equal.

D. Coconut fiber blanket:

- 1. Machine-assembled mat of 100 percent coconut fiber, of approximately 1/4-inch thickness.
- 2. Cover the top side of the mat with a photodegradable plastic (polypropylene) or biodegradable natural (jute fiber) mesh netting with 3/8-inch by 3/8-inch square openings.
 - Attach mesh to the mat with photodegradable or biodegradable knitting thread.
- 3. Supply blanket in protected rolled mat form of 6'-6" width, with a dry weight of 0.50 pounds per square yard minimum.
- 4. Acceptable products:
 - a. C125BN by North American Green.
 - b. Or equal.
- E. Provide 6-inch long staples of 11 gauge wire to hold mat or blanket in place.

2.5 MULCH

- A. Vegetative mulch:
 - Provide vegetative mulch for seeded areas of a high-quality, air-dried straw of wheat, rye, oats, beans, or other approved straw, free from grass, broom sedge, noxious weeds, and weed seeds detrimental to growth of grass.
- B. Hydraulic mulch:
 - 1. Provide virgin wood cellulose fibers complying with the following properties (percent by weight):

, , ,	
Moisture content	15
Organic matter, minimum	95
Water holding capacity	400
pH	4.3-8.5

2.6 SEED

A. Provide new crop seed furnished in standard sealed containers bearing seed tags showing purity, germination, and weed seed content, free of all primary noxious weed seeds and seeds of other noxious weeds as stipulated in the IDOT Standard Specifications, complying with the following minimum requirements:

Seed Type	<u>Purity</u>	Pure Live Seed
Kentucky Bluegrass	97%	80%
Red Top	90%	78%
Creeping Red Fescue	97%	82%
Tall Fescue	98%	83%
Annual Ryegrass	97%	85%

Perennial Ryegrass	97%	85%
Oats	92%	88%
Winter Wheat	92%	89%

B. In level areas to be used for lawns, use the following seed mixture:

Seed Type	<u>Lbs/Acre</u>
Kentucky Bluegrass	100
Perennial Ryegrass	60
Creeping Red Fescue	40

C. In level and sloped areas where reduced mowing will occur, use the following seed mixture:

Seed Type	<u>Lbs/Acre</u>
Tall Fescue	100
Perennial Ryegrass	50
Creeping Red Fescue	40
Red Top	10
Annual Ryegrass	15

D. For temporary erosion control seeding, use 130 lbs/acre of oats between March 1 and July 31 or 130 lbs/acre of winter wheat between August 1 to November 15.

2.7 SOD

- A. Provide field or nursery grown sod that is native to the locality of the Project.
- B. Provide sod that will not break, crumble or tear during handling and placing, free of stones, crab grass, noxious weeds, and other objectionable plants or substances injurious to plant growth.
- C. Provide sod having at least 1-inch of soil adhering firmly to the roots and cut in rectangular pieces with the shortest side not less than 12 inches. At the time of cutting sod, mow the grass height not less than 2 inches nor more than 4 inches.
- D. Do not use sod cut for more than 48 hours.

PART 3 - EXECUTION

3.1 TOPSOIL PLACEMENT

- A. Scarify the compacted subgrade to a depth of 3 inches to receive the topsoil.
- B. Spread at least 4 inches of prepared topsoil in areas of new grading raked smooth and level.
- C. Grade flush with walks, curbs, and paving.

3.2 PREPARATION FOR SODDING OR SEEDING

- A. Do not start preparation until all other site and utility work and finished grading within the areas to be seeded have been completed.
- B. Till topsoil to a depth of at least 3 inches and smooth out all surface irregularities resulting therefrom. Leave area free of rocks or hard soil clods which will not pass through the tines of a standard garden rake.
- C. Take a test of the site soils to determine the need for application of agricultural limestone (soil pH less than 7.0). If agricultural limestone is needed, then at least 7 days before applying fertilizer, spread lime uniformly in sufficient quantity to produce in the soil a pH of 7.0. Work lime thoroughly into topsoil to a depth of 3 inches.
- D. Apply fertilizer uniformly at a rate of 7 lbs. per 1,000 sq. ft. Work fertilizer into soil prior to seeding or sodding.

3.3 SODDING

- A. Provide sod in developed areas that were grassed prior to construction and as indicated on the Drawings. Sodding shall also be used in ditches and drainage swales and on all embankment slopes steeper than 3 to 1 unless protection is provided against erosion of seeding. At the Contractor's option, sodding may be substituted for seeding, but at no additional cost.
- B. Place sod with the edges in close contact and alternate courses staggered. Lightly tamp or roll to eliminate air pockets. On slopes 2 to 1 or steeper, stake sod with not less than 4 stakes per square yard and with at least one stake for each piece of sod. Stakes shall be driven with the flat side parallel to the slope. Do not place sod when the ground surface is frozen or when air temperature may exceed 90 degrees F. Water the sod thoroughly within 8 hours after placement and as often as necessary to become well established.
- C. In ditches, the sod shall be placed with the longer dimension perpendicular to the flow of water in the ditch. On slopes, starting at the bottom of the slope, the sod shall be placed with the longer dimension parallel to the contours of the ground.
- D. All exposed edges of sod shall be buried flush with the adjacent turf.

3.4 SEEDING

- A. Seed all grassed areas disturbed by construction operations and not receiving sod, and as indicated on the Drawings.
- B. Sow seed between September 1 and November 1, or in spring from time ground can be worked until June 1.

- C. Apply seed during favorable climatic conditions. Do not seed in windy weather or when soil is very wet. Sow seed at the rate specified for each seed mixture.
- D. Broadcasting seeding method:
 - 1. Sow seed with mechanical seeder in two directions at right angles to each other to achieve an even distribution of seed.
 - 2. After seeding, rake seed lightly into ground and roll with a roller weighing between 100 and 200 pounds per foot of roller width.
- E. Hydraulic seeding method:
 - 1. When seed is applied with a hydraulic seeder, apply at a rate of not less than 1,000 gallons of slurry per acre containing the proper quantity of seed specified above.
 - 2. When using a hydraulic seeder, apply the fertilizer in a separate operation.

3.5 TEMPORARY EROSION CONTROL SEEDING

- A. Seed all erodible/bare areas with a temporary cover crop within 7 days of disturbance, unless the area is to have an alternative temporary or permanent soil erosion control measure implemented within 7 days of disturbance, or as directed by Engineer.
- B. Seed bed preparation will not be required if the soil is in a loose conditions. Light disking shall be done if the soil is hard packed or caked. Fertilizer will not be required.

3.6 EROSION CONTROL MATTING OR FABRIC

- A. Immediately after rolling seeded areas, place erosion control excelsior blanket or fiber mat on slopes steeper than 4 horizontal to 1 vertical. Unless otherwise indicated, also place erosion control material at sides and bottoms of ditches, swales, and all areas within 10 feet of catch basins in seeded areas.
- B. Apply erosion control in accordance with the IDOT "Standard Specifications".

3.7 MULCHING SEEDED AREAS

- A. Immediately after rolling seeded areas, apply mulch at the rate of 2 tons per acre within 24 hours after seeding. Use vegetative mulch on all seeded areas unless hydraulic seeding method is used.
- B. If the hydraulic mulch application method is to be used, apply the hydraulic mulch at a rate of 2,000 pounds per acre.
- C. Apply mulch in accordance with the IDOT Standard Specifications.

3.8 WATERING

A. Immediately after placing erosion control matting or mulch, water seeded areas thoroughly with a fine mist spray. Keep soil thoroughly moist until seeds have sprouted and achieved a growth of 1-inch. For sod, immediately begin watering and continually keep moist until the sod has firmly knit itself to the topsoil.

3.9 PROTECTION OF WORK

A. Protect newly seeded and sodded areas from all traffic by erecting temporary fences and signs. Protect slopes from erosion. Properly and promptly repair all damaged work when required.

3.10 APPLICATION OF FERTILIZER

A. Six weeks after completion of seeding or sodding apply granular fertilizer over all areas at the rate of 2 lbs. of nitrogen nutrients per 1,000 sq.ft. of area.

3.11 CLEAN-UP

A. At the time of final inspection of work, but before final acceptance, remove from seeded and sodded areas all debris, rubbish, excess materials, tools, and equipment.

3.12 MAINTENANCE

- A. Provide watering, mowing, and replanting and continue as necessary until a close healthy stand of specified grasses is established.
- B. Replace lawns not showing a close uniform stand of healthy specified grasses at the end of the guaranty period and maintain until acceptance.

END OF SECTION

SECTION 33 05 23.16

WATER SERVICES DRILLED UNDER PAVEMENTS

PART 1 - GENERAL

1.1 SUMMARY

- A. This Section describes water service pipes installed by the trenchless horizontal directional drilling (HDD) method as shown on the Drawings, as specified herein, and as needed for a complete installation.
- B. Provide labor, materials, tools, and equipment necessary to perform all work specified in this Section.
- C. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01
 General Requirements of these Specifications.

1.2 SUBMITTALS

- A. Comply with pertinent provisions of Section 01 33 01.
- B. Submit shop drawings, including:
 - 1. Manufacturer's data on installation procedures.
 - a. Obtain approval prior to beginning work.
- C. Submit detailed construction scheduling plan at preconstruction meeting.
 - Comply with all portions of Section 01 32 16 concerning schedules and submittals.
 - 2. Provide detailed plan of means and methods to maintain clean and safe conditions in the event drilling material escapes to surface or adjacent storm sewers, including list of material and equipment that will be on-site during drilling and pipe insertion.
- D. Prepare a written, comprehensive contingency plan to address the response to, and clean-up of, hydrofracture (frac-outs) and surface spill events (release events).
 - 1. Provide and submit the contingency plan at the Preconstruction Meeting in accordance with Project Meetings Section 01 31 19 and Submittals Section 01 33 01.

1.3 GENERAL CONSTRUCTION REQUIREMENTS

A. Comply with the requirements of the permit issued by the controlling agency.

- B. Provide the insurance and other information required for the above referenced permits.
- C. Notify the controlling agency of the construction schedule not less than 10 days prior to the start of the construction.
- D. Coordinate and pay for inspection services by controlling agency.
- E. Provide all excavation, pits, installation and removal of tight sheeting, leaving of sheeting in place if indicated on the Drawings, backfilling of pits, compaction of excavated materials, and providing and compacting granular backfill materials where indicated on the Drawings.

PART 2 - PRODUCTS

2.1 PIPE

- A. Service lines: Type K soft temper seamless copper water tubing complying with ASTM B-88.
- B. Water service corporation stops, curb stops, and service boxes:
 - 1. Comply with Section 33 11 43.

2.2 INSERTION AND RECEIVING PITS

- A. Sheeting: Comply with OSHA requirements.
- B. Granular backfill material:
 - Comply with Section 31 23 79.

2.3 DIRECTIONAL DRILLING SYSTEMS

- A. Compact directional drilling systems.
 - 1. Provide system capable of working in tight conditions.
 - 2. Acceptable systems:
 - a. GRUNDODRILL by TT Technologies, Inc.
 - b. Or equal.
- B. Pit launched mini-directional drill system.
 - 1. Provide package system capable of being installed in a pit small enough to be inserted in parkway tight conditions.
 - Acceptable systems:
 - a. GRUNDOPIT by TT Technologies, Inc.
 - b. GRUNDOPIT K Keyhole by TT Technologies, Inc.
 - c. Or equal.

3.1 GENERAL

- A. Provide all excavation, insertion and receiving pits, piping, and security.
- B. Install pipes.
 - 1. Location of pipes will be determined in the field at time of construction.
- C. Location, protection, repair or replacement of underground utilities will be complete responsibility of the Contractor.
 - 1. No additional compensation will be allowed for damage to underground utilities, or time lost due to utility repair or relocation.

3.2 INSERTION AND RECEIVING PITS

- A. Provide pits as required to insert drilling equipment and install pipe.
 - 1. Provide tight sheeting where required to provide protection to public, permitting agency and public property, and adjacent utilities.
 - 2. Comply with OSHA requirements for type, installation, and removal of sheeting.
 - 3. Leave sheeting in place where indicated on the Drawings.
 - 4. Provide fencing of pits to provide protection to the public and security.
- B. Provide pits of length and width as necessary to install pipes and sized to fit area available for Work.
- C. Provide dewatering as required to allow excavation of pits and installation of pipes.
 - 1. Provide protection to environment from erosion or sedimentation resulting from all pumping operations.

D. Backfilling of pits:

- 1. Backfill with compacted special granular backfill materials whenever pit is below or within 2 feet of paved surfaces or curb and gutter or where indicated on the Drawings.
- 2. Backfill with compacted excavated materials where special granular backfill material is not indicated.
- 3. Comply with compaction requirements outlined in Section 31 23 79.
- 4. Re-grade slopes and ditches to match preconstruction condition and adjacent area.
- 5. Remove all construction debris, materials, excess excavated material, and sheeting from construction site upon completion of the Work.
- 6. Restore areas disturbed as outlined in Section 32 10 00 for Pavements, and Section 32 92 00 for Lawns and Grasses.

3.3 SERVICE PIPE INSTALLATION

A. Install water service pipe below paved areas by use of directional drilling method.

- B. Install water service pipe outside of paved areas by use of open cut methods, in compliance with Section 33 11 43 of these Specifications.
- C. Make connections to new water main and existing services as outlined in Section 33 11 43 of these specifications.

END OF SECTION

SECTION 33 11 43

WATER DISTRIBUTION SYSTEM

PART 1 - GENERAL

1.1 SUMMARY

- A. Provide water distribution system as shown on the Drawings, as specified herein, and as needed for a complete and proper installation, and in accordance with the latest edition of the "Standard Specifications for Water and Sewer Construction in Illinois", except as revised herein.
- B. Provide labor, materials, tools, chemicals and equipment necessary to perform the pressure and leakage tests and disinfection.
- C. Related work:
 - Documents affecting work of this Section include, but are not necessarily limited to, General Conditions, Supplementary Conditions, and Division 01 - General Requirements of these Specifications.
- D. References (Reserved).

1.2 SUBMITTALS

- A. Shop Drawing Submittals
 - 1. Manufacturer, make, and model information for all fire hydrants, valves, corporation stops, and curb stops.
- B. Operation and Maintenance Manuals (Reserved).
- C. Certificates and Guarantees (Reserved).
- D. Spare Parts (Reserved).
- E. Furnish two (2) copies of bacteriological test reports.

1.3 QUALITY ASSURANCE

- A. Ensure all brass and bronze alloys supplied with the products contain less than 15 percent zinc, unless otherwise specified.
 - 1. Brass that will come in contact with potable water shall contain no more than 0.25% lead.
 - a. Brass fittings shall be marked with industry standard marking to indicate the amount of lead (no lead, low lead, etc.) in the brass.
 - b. Brass for service saddles may contain more than 0.25% lead to improve ductility of the saddles.

- 1.4 DELIVERY, STORAGE, AND HANDLING (Reserved).
- 1.5 SITE CONDITIONS (Reserved).
- 1.6 MAINTENANCE (Reserved).

PART 2 - PRODUCTS

2.1 PIPE AND FITTINGS

- A. General:
 - 1. Provide ductile iron pipe materials where indicated on the Drawings.
- B. Ductile Iron Pipe:
 - 1. Provide ductile iron pipe complying with ANSI A21.51, thickness Class 52, with joints complying with ANSI A21.11.
 - a. Use cement lining complying with ANSI/AWWA C104/A21.4, standard thickness.
 - b. Exterior Coating:
 - (1) The exterior of ductile iron pipe shall be coated with a layer of arc-sprayed zinc per ISO 8179.
 - (2) The mass of the zinc applied shall be 200 grams per square meter of pipe surface area.
 - (3) A finishing layer of asphaltic coating shall be applied over the zinc coating.
 - i. The mean dry film thickness of the finishing coat shall not be less than 3 mils, with a local minimum not less than 2 mils.
 - (4) The coating system shall conform in every respect to ISO 8179-1 "Ductile iron pipes - External zinc-based coating -Part 1: Metallic zinc with finishing layer. Second edition 2004-06-01".
 - c. In areas where petroleum contamination is present use Buna-Nitrile gaskets.
 - 2. Provide restrained joint pipe system, where indicated on the Drawings, that utilizes one of the following methods:
 - a. Lock rings welded into place around pipe barrel.
 - b. Bolted rings installed around pipe barrels that fit inside pipe bells.
 - c. Gaskets which include stainless steel locking segments vulcanized into the gasket.
 - d. Mechanical joint retainer gland systems that provide locking segments shaped to pipe barrel that do not create stress points on pipe barrel.
 - (1) Do not use set screw type retainer glands.
 - e. Acceptable products:
 - (1) Meg-A-Lug System.
 - i. Series 1100 Megalug for MJ to pipe.

- ii. Series 1700 Megalug Harness for push on joint.
- iii. As recommended by manufacturer for connection to existing pipes.

C. Fittings:

- Ductile iron fittings: Provide mechanical joints complying with ANSI A21.10 or A21.53.
 - a. Cement lining: Comply with ANSI A-21.4, standard thickness.
 - b. Bolts and nuts:
 - (1) Use A-304 stainless steel bolts with nuts and washers of series 300 stainless steel per ASTM A194.
- D. Polyethylene sheet: Comply with ANSI/AWWA C105/A21.5:
 - 1. Thickness: Three layers of co-extruded linear low-density polyethylene (LLDPE) fused into a single thickness of not less than 8 mils.
 - 2. The inside surface of the polyethylene wrap to be in contact with the pipe exterior shall be infused with a blend of anti-microbial biocide to mitigate microbiologically influenced corrosion and a volatile corrosion inhibitor to control galvanic corrosion.
 - 3. Markings: The following information will be clearly marked on the sheet at minimum increments of 2 feet along its length:
 - a. Manufacturers name or trademark.
 - Year of manufacture.
 - c. Minimum film thickness and material type (LLDPE).
 - d. Applicable range of nominal pipe diameter size(s).
 - e. Warning Corrosion Protection Repair Any Damage.
 - 4. Acceptable manufacturers:
 - a. V-BIO® Enhanced Polyethylene Encasement.
 - b. Or approved equal.

E. Conductivity appurtenances:

- 1. Provide wedges of serrated silicon bronze or #10-copper cable and tapping devices specifically designed for this purpose.
- 2. Use devices provided by the pipe manufacturer.
- 3. Standard mechanical joints, Field-Lok, Megalug joints or similar joints/restrained joint systems do not provide conductivity.

2.2 VALVES

- A. Provide valves with clockwise closing direction.
- B. Gate valves:
 - 1. Valves 3-inch through 14-inch:
 - Design in accordance with AWWA C515 (ductile iron body), bronze fitted, resilient wedge and seat type with non-rising stem and O-ring packing.
 - b. Provide gear operator for valves 14-inch and larger.

- 2. Provide mechanical joint ends for buried valves and ANSI Class 125 flange ends or mechanical joint ends for valve installed in vaults as indicated on the Drawings.
 - a. Provide restrained type joints for all mechanical joint end valves.
 - b. Provide and install nuts and bolts matching the nuts and bolts used for fittings.
- 3. Acceptable valve manufacturers:
 - a. American Flow Control.
 - b. Clow.
 - c. EJ.

2.3 VALVE BOXES

- A. Provide adjustable valve boxes on buried valves:
 - 1. Valve box compatible with size and type of valve protected.
 - 2. Extend box to finished grade.
 - 3. Mark valve box cover "WATER" for potable water piping valves. Other valve box covers unmarked.
 - 4. Stainless steel valve extension stems and 2-inch square operating nuts 2 inches below the cover.
 - 5. Two valve operating tee wrenches with 2-inch square socket, 24-inch long, 1½-inch pipe handle, and 48-inch long 1½-inch pipe stem.
- B. Provide valve box stabilizer for all valve boxes.
 - 1. Acceptable manufacturers:
 - a. Alberico.
 - b. American.
 - c. Adaptor, Inc.
 - d. EJ Series 8550.
 - e. Tyler 6850.
 - f. Or equal.

2.4 VALVE VAULTS

A. Precast:

- Provide precast reinforced concrete manhole sections, bottoms, and flat top slabs complying with ASTM C478 unless otherwise indicated on Drawings.
- 2. Provide concentric cone section for gate valves.
- Provide eccentric cone section for butterfly valves.
- 4. Provide precast reinforced concrete monolithic or separate base.
- 5. Design flat slab tops for AASHTO HS20-44 wheel loading.

B. Concrete:

1. Provide 4,000 psi concrete using Type I Portland Cement complying with ASTM C150.

C. Mortar:

1. Mix one part Portland Cement to three parts fine aggregate.

D. Joints for precast sections:

- Provide joints of either flexible watertight rubber gaskets or preformed bituminous plastic gaskets consisting of a homogeneous blend of refined hydro-carbon resins and plasticizing compound reinforced with inert mineral filler.
 - a. Acceptable manufacturers:
 - (1) Henry Company, RN103 RAM-NEK Preformed Flexible Plastic Gasket (Coils).
 - (2) ConSeal Concrete Sealants, Inc., Type CS-102.
 - (3) Or equal.

E. Steps:

- 1. Provide steps with a minimum width of 12 inches and a minimum projection of 5 inches.
- 2. Use steps consisting of copolymer polypropylene plastic with a continuous ½-inch steel reinforcement as manufactured by M.A. Industries, Inc.; cast iron steps, Neenah R-1980-I; or equal.

F. Frames and covers:

- Provide cast iron frames and covers with heavy duty, indented top, with solid self-sealing lids and machined bearing surfaces, stamped with the work "WATER" and "CITY OF JOLIET".
 - a. East Jordan 1050Z1.
 - b. Neenah R-1710.
 - c. Or equal.

G. Flexible pipe connectors:

- 1. Provide flexible rubber gasket collar for connecting pipe to the manhole.
 - a. Comply with ASTM C923.
 - b. For pipe 24-inch and smaller, use PSX gasket system by Press-Seal Gasket Corporation, or equal.
 - c. For pipe 27-inch and larger, use resilient gasket by A-LOK Products, Inc., or equal.

2.5 FIRE HYDRANTS

A. General:

- 1. Comply with AWWA C502.
- Paint fire hydrants in colors selected by the Owner.
 - a. Public water: red.
 - b. Raw water: green.
 - c. Private systems: yellow.
- 3. Match the fire hydrants generally installed in the Owner's water system.

- 4. Acceptable manufacturers:
 - a. Clow Medallion.
 - b. EJ 5BR 250.
 - c. Mueller Centurion A423.

B. Materials:

- 1. Provide compression type with a 5¼-inch minimum size main valve assembly, O-ring seals, two 2½-inch hose nozzles, and a 4½-inch pumper nozzle with National Standard threads, a National Standard operating nut, and an above ground break flange.
- 2. Provide a 6-inch auxiliary resilient seat type gate valve with restrained type joints or bituminous coated metal tie rods between the valves and the tee fittings.
 - a. Provide and install nuts, bolts, and tie rods matching the nuts and bolts used for fittings.
- 3. Provide valve boxes with cover marked with the word "WATER".
 - a. Stainless steel valve extension stems and 2-inch square operating nuts 2 inches below cover.
- 4. Provide valve box stabilizers on all fire hydrant auxiliary valves.
 - a. Acceptable manufacturers:
 - (1) Alberico.
 - (2) American.
 - (3) Adaptor, Inc.
 - (4) Or equal.

2.6 WATER SERVICES

A. General:

- 1. Provide service saddles, corporation stops, curb stops, service boxes, and water service tubing.
- 2. Comply with the most recent edition of AWWA C800 (or NSF/ANSI 372) for service lines and service line appurtenances.

B. Materials:

- 1. Service lines: Type K soft temper seamless copper water tubing complying with ASTM B88.
- Service saddles:
 - a. Provide service saddles on all water mains for services greater than 1-inch.
 - b. Ductile iron, epoxy coated bodies with stainless steel straps and double bolts (four bolts total) as manufactured by the specified corporation stop and curb stop manufacturer.
- 3. Corporation stops: A.Y. McDonald No. 74701BQ.
- 4. Curb stops: A.Y. McDonald No. 76104Q.
- 5. Services boxes: Tyler 6500 with BBAF valve box stabilizer.

2.7 TAPPING SLEEVES AND VALVES

- A. Tapping sleeves:
 - Use two-piece bolted sleeve ductile iron or stainless steel type with mechanical joints. Stainless steel sleeves only allowed with prior approval from City of Joliet Utilities Department.
 - 2. Provide joint accessories.
 - 3. Measure existing water main outside diameter to determine proper tapping sleeve size.
 - 4. Acceptable manufacturers:
 - a. Ductile iron: Clow F-5205, or equal.
- B. Tapping valves:
 - Use fully ported gate valves complying with AWWA C500.
 - 2. Use mechanical joints type, Clow F-5093, or equal.

2.8 RESTRAINED FLANGE ADAPTOR

- A. Provide a ductile iron flange adaptor dual ring system with bolt circles compatible with 125#/Class 150 bolt pattern.
 - 1. Provide adaptor with individual actuated gripping wedges that utilize torque limiting screws to insure proper initial set.
 - 2. Set screw "only" restraining adaptors are not acceptable.
 - 3. Provide system that allows joint deflection of up to 5°.
 - 4. Provide a fluoropolymer coating to the wedge and wedge assembly and powder coating to the restraint body.
- B. Acceptable manufacturers:
 - Series 2100 Megaflange by EBAA Iron:
 - a. Or approved equal.

2.9 PIPE INSULATION

- A. Rigid Pipe Insulation:
 - 1. Provide extruded polystyrene sheathing conforming with ASTM C578, Type IV where indicated in the drawings.
 - 2. Thickness: 2 inches.
 - 3. R-Value: 10.
 - 4. Water absorption: No greater than 0.10% by volume per ASTM C272.
 - 5. Acceptable product:
 - b. Formular Rigid Foam Insulation, Owens-Corning.
 - c. Or equal.

PART 3 - EXECUTION

3.1 PIPE INSPECTION, HANDLING, STORAGE, AND INSTALLATION

A. General:

- 1. Install in accordance with pipe manufacturer's recommendations.
- B. Ductile iron water mains and appurtenances:
 - Comply with AWWA C600.
 - 2. Protect pipe and fittings installed in corrosive soils containing cinders or having a high organic content by loose wrapping or tubing with polyethylene sheet.
 - Place polyethylene sheet or tube around the entire circumference of the pipe, tie or tape sheet securely to prevent displacement during backfilling.
 - b. Comply with ANSI/AWWA A21.5-99/C105 regarding installation of polyethylene protection.
 - 3. Install conductivity through joints by use of conductivity wedges or copper cable and taps.
 - a. Use two (2) wedges per joint for pipes 12 inches or smaller, and four (4) wedges per joint for pipe sizes larger than 12 inches.
 - b. Use number of copper cable connectors per joint as recommended by the pipe manufacturer.

3.2 DEPTH OF PIPE COVER

- Install water mains and water service lines with a minimum depth of cover of five
 (5) feet below finished grade ground level unless otherwise indicated on the Drawings.
- B. Where new water mains cross existing water mains, install new water main below existing main unless otherwise indicated on the Drawings.

3.3 CONNECTIONS TO EXISTING WATER MAINS

- A. Make connections to existing water mains.
- B. Use non-pressure connections except where pressure connections are shown on the Drawings, or required by the conditions encountered at the time of construction.
- C. Make one connection at a time except as approved by the Engineer.
- D. Do not operate the valves on the existing water mains. The employees of the Owner's Water Department will open and close the valves as required for the connections.

E. Coordinate shutdown of existing water mains with Owner's Water Department and Engineer with a minimum of 48 hours' notice prior to shutdown.

3.4 PIPE RESTRAINING SYSTEMS

A. General:

1. Provide protection from movement of water main piping, plugs, caps, tees, valves, hydrants, and bends of 11½ degrees or greater.

B. Restrained joint pipe, fittings and valves:

1. Provide restrained joint pipe to a distance indicated on the Drawings, or if not shown on the Drawings, to a distance not less than two pipe lengths on each side of a fitting or valve to be restrained.

C. Concrete thrust blocks:

- When conditions prevent the use of restrained joints, use concrete thrust blocks of an approved type. The use of concrete thrust blocks in lieu of restrained joint pipe, fittings and valves requires written approval of the Engineer.
- 2. Provide precast or cast-in-place concrete thrust blocking with a compressive strength of 3,000 psi in 28 days.
- 3. Locate thrust blocking between solid ground and the fitting to be anchored.
- 4. Unless otherwise shown or directed by the Engineer, place the base and thrust bearing sides of thrust blocking directly against undisturbed earth.
- 5. Sides of thrust blocking not subject to thrust may be placed against forms.
- 6. Place thrust blocking so the fitting joints will be accessible for repair.

3.5 SEWER CROSSING

A. General:

 Separate water mains and water service lines from sanitary sewer, storm sewers, combined sewers, house sewer service connections, and drains in accordance with the "Standard Specifications for Water and Sewer Construction in Illinois".

B. Water mains:

- 1. Wherever water mains cross storm sewers, sanitary sewers, or sewer service connections:
 - a. Install the water main so that its invert is at least 18 inches above the top of the sewer.
 - b. Maintain this vertical separation for that portion of the water main located within 10' horizontally of any sewer or drain crossed.
 - c. Center a length of water main pipe over the sewer to be crossed with joints equidistant from the sewer or drain.
- 2. When it is impossible to obtain the minimum 18 inches vertical separation, or when it is necessary for the water main to pass under a sewer or drain:
 - a. Construct the sewer or drain of pressure pipe, conforming to the specification for water main materials.

- b. Extend the sewer construction on each side of the crossing until the normal distance from the water main to the sewer or drain is at least ten feet.
- c. As an alternate, install either the water main or sewer inside a water main quality casing pipe for a distance of 10 feet measured perpendicular to the sewer on each side of the crossing.
- 3. Where a water main must cross under a sewer:
 - a. Maintain a vertical separation of 18 inches between the invert of the sewer and the crown of the water main.
 - b. Support the sewer or drain line to prevent settling and breaking the water main.

C. Water service lines:

1. Comply with the requirements of water main separation.

D. Sewer manholes:

1. Do not install water line through sewer manhole.

3.6 VALVE VAULTS

A. General:

- 1. Install pipe through valve vault as shown on the Detail.
- 2. Make vault watertight with use of flexible manhole connectors as per the Detail.

B. Steps:

- 1. Provide each valve vault over 3 feet deep with individual wall-mounted steps as shown on the valve vault detail.
- 2. Comply with the requirements of governmental agencies having jurisdiction.

C. Jointing:

- 1. Use flexible watertight gaskets for each joint.
- 2. Trim smooth and free from surplus gaskets.
- D. Frames and covers: Unless otherwise shown on the Drawings or as directed by the Engineer, set frames and covers:
 - 1. In paved areas: So that the top of the solid cover will be flush with the finished pavement.
 - 2. In unpaved areas: To drain away from the valve vault.
 - With flexible watertight gaskets.
 - 4. With grade rings not to exceed 8 inches.

3.7 FIRE HYDRANT INSTALLATION

A. Install fire hydrants plumb with the lowest hose connection at least 18 inches, but not more than 26 inches, above the finished grade ground level.

- B. Set fire hydrant and auxiliary valve on precast concrete blocks to provide firm support for the bases.
- C. Brace the fire hydrant base with solid concrete blocking between the base and undisturbed trench wall to counteract the reaction thrust of water pressure at the base.
 - 1. Provide mechanical joint anchoring fittings, or approved restrained joints.
- D. Brace the fire hydrant barrels during backfilling.
 - 1. Do not block the drain hole in fire hydrant.
- E. Place a minimum of 1/2 cubic yards of washed coarse stone at and around the base for proper drainage.
 - Cover stone with plastic before backfilling.
- F. Place and compact backfill materials in 6-inch layers around the fire hydrant and auxiliary gate valve.
- G. Cover new fire hydrant with plastic bag until new system is in service.

3.8 WATER SERVICE CONNECTION

A. General:

- 1. Make service connections at locations shown on the Drawings or determined by the Engineer at the time of construction.
- 2. Notify the City's Public Utilities Department immediately if existing lead water services are discovered.
- 3. Install water service pipe, corporation stop and curb stop as shown on the standard water service installation detail.
- 4. Set curb stop on a precast concrete block.
- 5. Do not splice the water service pipe.
- 6. Sequence of installation: See TESTING AND INSPECTING section of these Specifications for outline of when, in the sequence of construction, the corporation stops, curb stops, and service boxes are to be installed, leakage tested, and disinfected.

B. Service boxes:

- 1. Install over the curb stop in a truly vertical position.
- 2. Set the top of the box flush with the finished grade ground level.
- C. Direct tapping of polyethylene-encased D.I.P.:
 - 1. Wrap two or three layers of polyethylene adhesive tape completely around the pipe to cover the tapping machine and chain mounting area.
 - 2. Make the tap and install the corporation stop directly through the tap and polyethylene.
 - 3. After making the direct service connection, inspect the entire circumferential area for damage and make any necessary repairs.

4. Wrap the corporation stop and a minimum distance of 3 feet of the copper service pipe with polyethylene.

3.9 PRESSURE CONNECTION

- A. Support the exposed existing water main on concrete pedestals at sufficient intervals to properly carry its own weight, plus the weight of the tapping machinery and fitting.
 - 1. Provide a concrete thrust block behind tapping sleeve.
- B. The City of Joliet Water Department shall do the actual tapping of the main. The Contractor will be responsible for calling 815-724-4220 to schedule an appointment with City Water Department. Appointments are available on Tuesdays and Thursdays.
- C. Minimum tap hole diameter:
 - 1. For pipe 12-inch or smaller: 1/2-inch less than the nominal pipe diameter.
 - 2. For pipe 14-inch through 20-inch: 1½-inch less than the pipe diameter.
- D. Open valve to flush any foreign material after completion of the pressure tapping.

3.10 PIPE INSULATION

- A. Rigid Pipe Insulation:
 - 1. Place rigid insulation board above the pipe bedding material to the width of the trench.
 - 2. Place rigid insulation board to the required thickness at locations as directed by the Engineer.

3.11 POLYETHYLENE WRAPPING OF DUCTILE IRON PIPE AND APPURTENANCES

- A. Comply with requirements of ANSI/AWWA A21.5-99/C105.
 - 1. Place polyethylene sheet around the entire circumference of the pipe, tie or tape sheet securely to prevent displacement during backfilling.
 - 2. Wrap all water mains, fittings, valves, fire hydrant leaders, fire hydrants, and service lines.
 - a. Wrap copper service lines to a point 3 feet from center of water main.
 - b. Do not block fire hydrant weep hole.

3.12 TESTING AND INSPECTING

- A. Sequence of installation:
 - Install new water main but do not install corporation stops, service lines, curb stops, or service boxes; conduct pressure test, leakage test; and disinfection of new water main; flush main; after acceptance for use put main into service (while existing main continues to function): Install

corporation stops, curb stops, and new service boxes; and connect new service box to existing service lines.

B. Hydrostatic tests:

- 1. Where any section of a water line is provided with concrete thrust blocking for fittings, do not make hydrostatic tests until at least 5 days after installation of the concrete thrust blocking, unless otherwise directed by the Engineer.
- 2. Devise a method for disposal of waste water from hydrostatic tests, and for disinfection, as approved in advance by the Engineer.

C. Pressure tests:

- 1. Subject the new water mains and service lines, including valves and hydrants, to a hydrostatic pressure of 150 psi, meeting AWWA C600 requirements.
- 2. Test pressure shall not vary by more than ±5 psi for the duration of the test.
- 3. Carefully examine exposed pipe, joints, fittings, and valves.
- 4. Replace or remake joints showing visible leakage.
- 5. Remove cracked pipe, defective pipe, and cracked or defective joints, fittings, and valves. Replace with sound material and repeat the test until results are satisfactory.
- 6. Make repair and replacement without additional cost to the Owner.
- 7. Use only solid stainless full-body repair clamps as approved by the Engineer.

D. Leakage test:

- 1. Conduct a metered leakage test after the pressure test has been satisfactorily completed.
- 2. Duration of each leakage test: two (2) hours.
- 3. Subject the new water mains, including valves and hydrants, to a hydrostatic pressure of 150 psi.
- 4. Test pressure shall not vary by more than ±5 psi for the duration of the test.
- 5. Install water meter approved by the Engineer.
 - Provide double check valve assembly between water meter and existing water main.
- 6. Maximum allowable leakage: As recorded by a meter approved by the Engineer, with leakage to not exceed the number of gallons per hour (gph) as determined by the following formula:

gph = LD $(P^{1/2})/133,200$

in which: L = Length of pipe test, in feet

D = Diameter of water main, in inches

P = Average pressure, in pounds per square inch (gage)

7. Should any test of pipe disclose leakage greater than the maximum allowable amount, locate and repair the defective joint or joints and then repeat the 2-hour metered leakage test until the leakage is within the specified allowance, and at no additional cost to the Owner.

E. Time for making test:

- Except for joint material setting, or where concrete reaction backing necessitates a 5 day delay, pipelines jointed with rubber gaskets, mechanical, or push-on joints, or couplings may be subjected to hydrostatic pressure, inspected, and tested for leakage at any time after partial completion of backfill.
- 2. Perform the pressure and leakage tests satisfactorily prior to requesting the Engineer to witness the official tests.
- 3. Notify the Engineer at least 48 hours prior to the time of the requested official tests.
- 4. Depending on traffic conditions, public hazard, or other reasons, the Engineer may direct when to conduct the tests, and may order the tests to be made in relatively short sections of water mains.

3.13 PRELIMINARY FLUSHING

- A. Prior to disinfection, flush water main as thoroughly as possible.
 - 1. Flush water main until water runs clear.
 - 2. Provide a minimum flushing velocity of 2.5 feet per second in the water main.
 - 3. Where no fire hydrant exists on the end of a water main, the plug (or cap) on the end of the water main must be tapped with opening in the end for flushing purposes. After acceptance install threaded plug into tap.
 - a. 8", 10" and 12" mains: 2½-inch tap.
 - b. 16" mains: 3-inch tap.
 - c. Larger than 16" use temporary fire hydrants.
 - 4. Contractor has the option to use temporary fire hydrants in lieu of taps. Temporary fire hydrants must be removed after testing is complete.
- B. Coordinate time of flushing with Owner and Engineer, at least 72 hours in advance of flushing.
 - 1. Do not initiate flush without Owner's permission.

3.14 DISINFECTION

A. General:

- After the water main work has been satisfactorily completed and tested, disinfect the work in accordance with AWWA C651, and "Standard Specifications for Water and Sewer Construction in Illinois" as modified by these Specifications.
- B. Application of chlorine:
 - 1. Apply chlorine by gas feed or solution feed chlorinator as approved by the Owner.
 - a. Provide effective diffusion of the gas or solution into the water.
 - b. Provide means for preventing the backflow of water into the feeder.

2. Chlorine solution:

a. Apply solution of sodium hypochlorite into one end of the section of main to be disinfected while filling the main with water.

C. Requirement of chlorine:

- Initial chlorine solution in pipe: At least 50 mg/L, but not more than 100 mg/L.
- 2. Retain the disinfecting solutions in the work for at least 24 hours.
- 3. Chlorine residual after the retention period: At least 25 mg/L.

D. Flushing and testing:

- 1. Following chlorination, flush treated water thoroughly from the water mains until the chlorine concentration in the water flowing from the main is no higher than generally prevailing in the Owner's system, or less than 1 mg/L.
- 2. After flushing, the Owner will collect two samples on successive days at least 24 hours apart in sterile bottles treated with sodium thiosulfate. Notify the Engineer and Owner when leakage testing is complete and schedule the time for sample collection with the Owner. Provide materials and support to the Owner in collection of samples.
- 3. The Owner will perform bacteriological analysis.
- 4. Should the initial disinfection result in an unsatisfactory bacterial test, repeat the chlorination and sampling procedures until satisfactory results are obtained.

E. Swabbing:

- Flush and swab the piping, valves, and fittings that must be placed in service immediately and cannot be disinfected by the above specified methods, with 5 percent solution of calcium hypochlorite prior to assembly.
 - Secure the Engineer's approval before applying this method of disinfection.

3.15 DECHLORINATION

- A. Comply with AWWA C651-05 requirements to neutralize the residual chlorine in new water mains.
- B. After new water mains have passed disinfection requirements, utilize portable diffusing dechlorinators that utilize sulfur dioxide or other chemicals listed in Appendix C of AWWA C651 to lower chlorine residuals prior to discharge to the drainage system.
 - Lower concentration to 1 mg/L or less.

3.16 ABANDONMENT OF EXISTING WATER MAINS AND APPURTENANCES

A. Abandon water mains indicated on the Drawings as "to be abandoned" only after all requirements for testing and disinfection have been satisfied and all existing services have been connected to new water mains.

- B. Provide concrete plugs in all water main pipes to be abandoned at the limits of the trench excavations, or at other locations if so indicated by the Drawings.
- C. Provide ductile iron plugs, caps, or other necessary fittings, and thrust blocking, on ends of portions of existing water mains that are to remain in service. Tees should be removed and replaced with the appropriate diameter of water main, or tees shall have ductile iron plugs installed, as directed by the Engineer. Length of dead end mains shall be no greater than two times the outside diameter of the main.
- D. Close existing water valves only with the permission of the Engineer.
- E. Leave valve in place, but remove valve box to top of existing valve, and fill excavation with compacted granular material.
- F. Leave valve in place, remove frame and lid and top section of valve vault, and backfill excavation with compacted granular material.
- G. Remove fire hydrants in total, including auxiliary box, and backfill excavation with compacted granular material.
- H. Deliver valves, valve boxes, fire hydrants, and frames and lids to the Owner at 921 Washington Street, Joliet, Illinois.
- Where abandonment of existing water mains or appurtenances require work outside of the work zone, restore area of work as indicated on the Drawings or Section 32 92 00 of the Specifications.
- J. Removal of existing water mains that are being replaced by new water mains in the same location is considered incidental to the installation of the new water main and no additional compensation will be allowed.

END OF SECTION